

# Lincoln University Digital Thesis

# **Copyright Statement**

The digital copy of this thesis is protected by the Copyright Act 1994 (New Zealand).

This thesis may be consulted by you, provided you comply with the provisions of the Act and the following conditions of use:

- you will use the copy only for the purposes of research or private study
- you will recognise the author's right to be identified as the author of the thesis and due acknowledgement will be made to the author where appropriate
- you will obtain the author's permission before publishing any material from the thesis.

# **Campylobacteriosis in New Zealand: Causes and Control**

A thesis submitted in partial fulfilment of the requirements for the Degree of Doctor of Philosophy

at

Lincoln University

by

Ali Mohammed Y. Al-Sakkaf

Lincoln University

2020

# Abstract of a thesis submitted in partial fulfilment of the requirements for the Degree of Doctor of Philosophy

### Campylobacteriosis in New Zealand: Causes and Control

by

## Ali Mohammed Y. Al-Sakkaf

New Zealand (NZ) has a higher rate of reported campylobacteriosis cases than the rest of the developed world. The majority of human campylobacteriosis cases are attributable to the consumption of poultry products. The hypotheses explored in this study are: (i) NZ C. jejuni strains have a greater heat tolerance and hence are better able to survive cooking, (ii) NZ strains are more oxygen tolerant, (iii) secondary poultry processing practices increase the NZ chicken contamination with Campylobacter (iv) Food preparation home hygiene practices in NZ are poor compared with the developed world.

Quantitative microbial risk assessment (QMRA) was carried out by applying an advanced Bayesian approach to assess all the factors in the food chain, which impact the final campylobacteriosis risk estimation. Moreover, the current intervention practices applied at the poultry plant was investigated to find a better physical intervention method than the potential unsafe chemical intervention processes applied currently. Finally, an alternative approach to QMRA that could predict more accurately the annual campylobacteriosis risk in NZ was also investigated.

The kinetic parameters (D and z values) determined under the isothermal and dynamic condition in broths culture or on chicken skin for the most implicated strains in human campylobacteriosis cases in New Zealand were broadly in agreement with published international data. They did not indicate that NZ strains (ST 474, ST 190, ST 48, ST 45 from humans and poultry) were more heat resistant than the overseas strains. Similarly, the bacterial survival under oxidative stress showed that the oxygen tolerance of NZ strains was also similar to other internationally reported strains. A secondary practice (marination of chicken by needle injection) investigated at a poultry production plant did not significantly increase the chicken carcasses contamination level. The results confirm that the tested New Zealand C. jejuni strains do not have unusual characteristics, and the high rate of campylobacteriosis in New Zealand is not associated with the emergence of more environmental stress-resistant strains. This finding has a significant impact on the poultry industry, regulators, consumers and researchers as it confirms the scientific evidence to maintain the stipulated standards for the heat treatment practices at poultry plants, foodservices and at homes. Changes to heat treatment practices can cause unpredicted loss to the poultry industry. It is important for the regulatory bodies to convey to all consumers, the critical message of the cooking temperature given the reluctance of NZ consumers to use food thermometers to verify the home cooking temperatures.

The results of the consumers' food handling practices survey revealed that the mean score of food safety practices and food safety awareness of the New Zealanders who participated in the survey was below the average questionnaire score of 10.5 out of 21 (the total score ). The study confirmed that the awareness of basic food hygiene and safe food preparation practiced by people in New Zealander was lower than in other developed countries. Therefore, an innovative and more effective approach needs to be implemented for future consumer education strategy.

The QMRA conducted in this study indicated that hygiene has a significant impact on the total illness probability. The 'Bayesian hierarchal model', which provides a better insight into the food chain than the 'time series model' and is more informative as it incorporates all the factors that impact on the final risk estimation. The QMRA model was able to identify the consumer hygiene practices, the initial contamination prevalence at the farm and the practices at the processing plant, as the significant factors influencing the final risk estimate. The association between the reduction in birds/chicken carcases prevalence and improvement in home hygiene practice and the calculated reduction in risk of human disease was estimated to be 1:1, which agrees with international findings. Thus, QMRA model can easily determine the impact of any intervention in the food chain. Therefore, the effect of a new planned intervention such as consumer education on the final risk estimate should be clear for the policymakers, risk managers and health professionals. This can be achieved by altering the priors of the consumer hygiene practices in QMRA model. Similarly, other interventions at poultry plants and farms should also be assessed by altering the priors of farms and poultry plants. The Bayesian hierarchal model, which has been used in this study, was able to deliver the expected outcomes and to provide a fast response for policymakers and risk managers. The Auto-Regressive Integrated Moving Average (ARIMA) with an intervention model provided one of the best prediction of the annual campylobacteriosis risk in NZ from the ten models tested. ARIMA intervention model has the lowest forecast error with , only 9 % campylobacteriosis cases more than the actual notified cases. The Holt-Winters method, being local in nature, quickly adapts to the new post-intervention regime and also gives a good prediction which is comparable with the ARIMA intervention models. This method has the additional advantage that it is simple to calculate.

This study also highlighted the potential alternative physical interventions such as the immersion in hot water or hot water wash at poultry processing plants that could be employed with possible another physical intervention in case a ban on chemicals that are currently used to control campylobacteriosis is imposed in NZ.

iii

# Declaration

Some aspects of this thesis have been published/submitted for publication in international peerreview journals or as a commission report to the funding organisation (New Zealand Ministry of Primary industries). Some of the results have also been presented at conferences.

## Paper Published

- Al-Sakkaf, A. 2020. Comparison of three modelling approaches to predict the risk of campylobacteriosis in New Zealand. *Microbial Risk Analysis*. 14:100077 https://doi.org/10.1016/j.mran.2019.06.001.
- Al-Sakkaf, A. (2015). Domestic food preparation practices: a review of the reasons for poor home hygiene practices. *Health Promotion International.* 30, 3, pp 427–437 <u>http://dx.doi.org/10.1093/heapro/dat051</u>
- Al-Sakkaf, A. (2015). *Campylobacter* heat resistance past, current status and future prospect for New Zealand and beyond. *World Poultry Science Journal*, 71, pp 111-124. <u>http://dx.doi.org/10.1017/S0043933915000100</u>
- Al-Sakkaf A, Jones G. (2014). Comparison of time-series models for predicting campylobacteriosis risk in New Zealand. *Zoonoses and Public Health*. 61, 3, 167-174.<u>http://dx.doi.org/10.1111/zph.12046</u>
- Al-Sakkaf, A. (2013). Campylobacteriosis in New Zealand: A new twist to the tale? Part One (The pathogen and the poultry plant). *Food Control*, 33(2), 556-561. doi: <u>http://dx.doi.org/10.1016/j.foodcont.2013.03.051</u>
- Al-Sakkaf, A. (2013). Campylobacteriosis in New Zealand: A new twist to the tale? Part two (The consumer and the regulator). *Food Control*, 33(2), 562-566. doi: <u>http://dx.doi.org/10.1016/j.foodcont.2013.03.050</u>
- Al-Sakkaf, A. (2012). Evaluation of food handling practice among New Zealanders and other developed countries as a main risk factor for campylobacteriosis rate. *Food Control* 27(2), 330-337 doi: <u>http://dx.doi.org/10.1016/j.foodcont.2012.04.011</u>
- Al-Sakkaf, A. and Jones, G. (2012). Thermal inactivation of *Campylobacter jejuni* in broth. *Journal of Food Protection* 75, 6, 1029-1035 doi <u>http://dx.doi.org/10.4315/0362-028X.JFP-11-518.</u>

## **Commissioned Report**

 Al-Sakkaf, A., Jones, G., & Mawson, J. (2010). Survival of New Zealand relevant *Campylobacter* strains Final Report: N0174/06 Prepared for NZFSA available online <u>http://www.foodsafety.govt.nz/elibrary/industry/campylobacter-strains/index.htm</u>.

## Paper to be Submitted

Al-Sakkaf, A., Charles Brennan, Elizabeth Redmond and Ravi Gooneratne. (2020).
 Risk factors associated with consumer handing of raw poultry in the home: implications for campylobacteriosis in New Zealand.

## **Conference** papers

- July 2013 USA: "Quantitative Risk Assessment for campylobacteriosis in New Zealand by the Bayesian Belief Network Approach" at the Annual Meeting of IAFP. (July 28 – July 31, Charlotte, North Carolina) https://iafp.confex.com/iafp/2013/webprogram/Paper3707.html
- August 2010 USA Oral presentation at 99th Annual Meeting of International Association of Food Protection (IAFP). The title of my paper was: "Application of Kinetic Models to Describe Heat Inactivation of Selected New Zealand Isolates of *Campylobacter jejuni*". (August 1 August 4, Anaheim, California)
   <a href="https://www.hitpages.com/doc/5678762540662784/12">https://www.hitpages.com/doc/5678762540662784/12</a>
- September 2009 Japan: Oral presentations at the 15th International Workshop on *Campylobacter, Helicobacter* and Related Organisms (CHRO). Papers' titles: "Survival of some New Zealand *Campylobacter* strains under oxidative stress at different temperatures". (September 2-September 5, Niigata)
- September 2009 Japan Oral presentations at the 15th International Workshop on *Campylobacter, Helicobacter* and Related Organisms (CHRO). Papers' title "Thermal inactivation of *Campylobacter jejuni* on chicken skin". (September 2-September 5, Niigata).
- August 2008 USA: Poster at International Association for Food Protection (IAFP) "Thermal inactivation of *Campylobacter jejuni* in broth". (August 3-August 6, Columbus, Ohio).
- June 2008 New Zealand Poster at Annual Conference of the New Zealand Institute of Food Science and Technology (NZIFST). "Application of kinetic models to describe heat inactivation of selected New Zealand isolates of *C. jejuni*".(June 24-June 26, Rotorua)

## Acknowledgements

In the name of Allah, most benevolent, ever merciful

This PhD was a unique journey of tenacity and thanks to Almighty Allah, who has stood by my side all the way through. Thanks to many people for enabling me to complete the task of finishing this thesis. I appreciate the efforts of my supervisors Professors Ravi Gooneratne and Charles Brennan, as they always have been there when I needed them.

I would like to express my profound appreciation to Professor Geoff Jones from the School of Fundamental Science, Massey University, for his continued statistical support throughout this study.

I would like to thank my mother for her patient acceptance of absence from home for such a long period and her caring. I am grateful to my family in New Zealand, my wife and my children for their outstanding support, help and patient acceptance for my absence from home. My special thanks to my wife for her technical assistance in the laboratory for a few months.

It is hard to name and thank more than 30 individuals in New Zealand and overseas such as the experts and academics, laboratory managers, laboratory technicians, engineering and maintenance technicians, library staff and IT professionals who supported me throughout my PhD journey.

# Abbreviations

BBN	Bayesian Belief Networks.
BHI	Brain Heart Infusion broth
СА	Columbia Blood Agar
CEB	Campylobacter Enrichment Broth
cfu	Colony-forming unit
D	The decimal reduction time (s)
d	The actual ingested dose of the bacteria or the pathogen.
d <sub>bf</sub>	The probability of additional contamination on the chicken during slaughtering or transferring from farms to processing plants.
D <sub>ref</sub>	The <i>D</i> - value at a reference temperature $T_{ref}$ (s).
EC	European Commission
EU	The European Union
FAO	Food and Agriculture Organization of the United Nations
FDA	U.S.A Food and Drug Administration
	The number of chicken carcasses contaminated out of a number sampled in a survey or
gbs	routine sampling plan.
gfs	The number of chicken flocks contaminated out of a number sampled in a survey or routine sampling plan.
<b>g</b> its	The number of people suffering campylobacteriosis from whatever source in a year.
GlnaFit	An Excel add-in to fit inactivation curves to various models.
ICMSF	The International Commission on Microbiological Specifications for Foods
k <sub>T</sub>	A temperature-dependent death rate constant (per time unit).
m	The mean or expected value of the normal probability distribution describing on a logit
IIID	scale the probability of a chicken carcase being contaminated $(p_b)$
МС	Monte Carlo simulation which uses random numbers in the simulation algorithm
mCCDA	modified Cefoperazone Charcoal Deoxycholate Agar
МСМС	Markov chain Monte Carlo.
mc	The mean or expected value of the normal probability distribution describing on a logit
III	scale the probability of a bird in the chicken farm being contaminated $(p_f)$ .
MH	Muller Hinton Agar.
MLST	multi-locus sequence typing system
MSE	The mean squared error.
Ν	The number of cells per unit mass or volume.
N <sub>0</sub>	The initial count of bacteria.
n <sub>bs</sub>	The number of carcasses samples and tested after processing for Campylobacter prevalence

	in produced chicken.
2	The total number of households times 13, contributing to the total annual chicken
11 <sub>CS</sub>	consumption in NZ.
n <sub>csb</sub>	A total number of chickens purchased by NZ households during the year.
n <sub>fs</sub>	The number of birds samples and tested at a farm for <i>Campylobacter</i> prevalence in flocks.
n <sub>its</sub>	The average population of New Zealand during the year.
N <sub>res</sub>	The number of cells in a subpopulation of the total population N which are more resistant
	to the thermal treatment
Nt	The number of cells per unit mass or volume at time t (cfu ml-1).
NZFSA	New Zealand Food Safety Authority
р	An arbitrary (fitted) power parameter
рь	Probability of a chicken carcase being contaminated after processing in a poultry plant.
pbs	The probability that a processed chicken carcass being tested is found to be contaminated.
pe	The probability of an exposure event for a household over a four week period.
Pey	Probability of an individual being exposed during the one year period.
pf	Probability of a bird in a chicken farm being contaminated
pfs	The probability of contamination of a bird tested on a chicken farm.
ph	Probability of cross-contamination from a contaminated broiler in a consumer's kitchen.
P <sub>hc</sub>	Probability of bacterial transfer from one surface to another surface.
P <sub>hh</sub>	Probability of bad hygiene behaviour by a consumer during chicken preparation/handling
	or cooking.
D.:	The probability that an individual acquires campylobacteriosis due to broiler meat during
Pib	the year
Pie	The probability of illness given exposure to Campylobacter.
P <sub>in</sub>	The probability of illness given infection.
p <sub>iq</sub>	The attributable fraction of disease burden due to a known source.
D:	The probability that an individual acquires campylobacteriosis during the year, due to any
Pit	Campylobacter attributable source.
Dite	The probability that an individual acquires campylobacteriosis within one year, due to all
Pits	Campylobacter attributable sources after incorporating the data.
Pne	The probability that a single pathogenic organism causes infection.
QMRA	Quantitative microbial risk assessment
RA	Risk assessment
Sh	The standard deviation of the normal probability distribution describing on a logit scale
50	Probability of a chicken carcase being contaminated after processing in poultry plant ( $p_b$ ).
SE	The standard error.
Sf	The standard deviation of the normal probability distribution describing on a logit scale
~1	the probability of a bird in the chicken farm being contaminated (p <sub>f</sub> ).
Ss	The standard deviation of the possible error term between the $p_{it}$ and $p_{its}$ .
tL	The lag time

USDA	The U.S. Department of Agriculture
WHO	World Health Organisation
г.	The change of temperature (°C) required to reduce the <i>D</i> - value by tenfold.
η	A parameter describing the shape of the Weibull distribution function.
δ	A parameter describing the scale of the Weibull distribution function.
λ <sub>c</sub>	The rate of chicken consumption in a household during a four-week period.
$\lambda_{csb}$	The total annual rate of chicken consumption in NZ.
$\lambda_{e}$	The mean rate of exposure to Campylobacter per four week period.

# Table of contents

1	IN	ITROD	UCTION	2
	1.1	Prob	LEM DESCRIPTION	2
	1.2	Proje	CT AIM	2
	1.3	Нуро	THESIS	3
2	LI	TERAT	URE REVIEW	4
	2.1	Васк	GROUND	4
	2.2	THE P	ATHOGEN	6
	2.3	Surv	VAL AND GROWTH OF <i>C. JEJUNI</i> IN THE ENVIRONMENT	9
	2.	3.1	Atmosphere	10
	2.	3.2	Water Activity	11
	2.	3.3	Temperature	11
	2.4	Kinet	IC ANALYSIS OF BACTERIAL INACTIVATION	12
	2.	4.1	Linear models	12
	2.	4.2	Alternative models of thermal inactivation	15
	2.	4.3	Handling bacterial survival data	18
	2.	4.4	Non-isothermal models	19
	2.5	FACTO	DRS AFFECTING HEAT RESISTANCE OF MICROORGANISMS	20
	2.	5.1	Strain variation	20
	2.	5.2	Test or experimental conditions	21
	2.	5.3	Culturing media and recovery procedure	22
	2.	5.4	Effect of pH and presence of acids	22
	2.	5.5	Effect of age (growth phase) of microorganisms	23
	2.6	Heat	RESISTANCE STUDIES	23
	2.7	The r	OLE OF THE CONSUMER IN FOOD SAFETY	25
	2.8	The r	OLE OF THE FOOD REGULATORY BODY	26
	2.	8.1	Consumer education	26
	2.	8.2	Control of the poultry industry	27
	2.	8.3	Risk assessment (RA)	29
	2.9	QMR	A APPROACH DESCRIPTION	32
	2.	9.1	Monte Carlo simulation	33
	2.	9.2	Bayesian belief networks (BBN)	33
	2.10	A	NALTERNATIVE APPROACH FOR CAMPYLOBACTERIOSIS CASES PREDICTION	35
	2.	10.1	Time-series	35
	2.	10.2	Time-series modelling techniques	35
	2.11	Sc	OPE OF THIS WORK	38
	2.	11.1	The structure of this thesis	40

3	Μ	ICROE	BIOLOGICAL EXPERIMENTS MATERIALS & METHODS	41
	3.1	Самр	YLOBACTER ISOLATES	41
	3.2	Самр	YLOBACTER CULTURE GROWTH AND ENUMERATION	41
	3.3	THER	MAL INACTIVATION OF <i>C. JEJUNI</i> IN BROTH	42
	3.3	3.1	C. jejuni heat inactivation procedure	42
	3.4	THER	MAL INACTIVATION OF <i>C. JEJUNI</i> ON CHICKEN SKIN	42
	3.4	4.1	Chicken skin sample preparation and inoculation	43
	3.4	4.2	C. jejuni heat inactivation procedure.	43
	3.4	4.3	Enumeration of C. jejuni survivors	43
	3.5	THER	MAL INACTIVATION OF C. JEJUNI ON CHICKEN SKIN UNDER DYNAMIC CONDITIONS	44
	3.5	5.1	Meat sample preparation and inoculation	44
	3.5	5.2	Heat inactivation procedure	44
	3.5	5.3	Enumeration of C.jejuni survivors	44
	3.6	Survi	VAL OF C. JEJUNI UNDER OXIDATIVE STRESS AT DIFFERENT TEMPERATURES	45
	3.0	6.1	Exposure of C. jejuni to oxygen in broth	45
	3.0	6.2	Exposure of C. jejuni to oxygen on agar plates	45
	3.0	6.3	Exposure to H <sub>2</sub> O <sub>2</sub>	45
	3.7	Evali	JATION OF PROCESSING PRACTICES CONTRIBUTING TO CAMPYLOBACTER CONTAMINATION IN NEW ZEALAND CHICK	(EN
	AND C	HICKEN	I PRODUCTS	46
	3.2	7.1	Procedure	46
	3.2	7.2	Survival of C. jejuni in injected chicken products at the plant	46
	3.2	7.3	Recovery of C. jejuni from surfaces of the injected drumsticks	47
	3.2	7.4	Recovery of Campylobacter from the internal muscles of the injected drumsticks	47
	3.2	7.5	Survival of C. jejuni in injected chicken carcasses and chicken products in the retail market	48
	3.8	Evalu	JATION OF <i>CAMPYLOBACTER</i> MITIGATION STRATEGIES BY PHYSICAL METHODS IN A POULTRY PROCESSING PLANT	48
	3.8	8.1	Sample preparation and treatment	48
	3.8	8.2	Enumeration of C. jejuni survival.	49
4	TH	IERM/	AL INACTIVATION OF CAMPYLOBACTER JEJUNI IN BROTH	50
	4.1	INTRC	DUCTION	50
	4.2	Mate	RIALS AND METHODS	50
	4.2	2.1	Campylobacter culture growth and enumeration	50
	4.2	2.2	C. jejuni heat inactivation procedure	51
	4.3	Mod	ELLING AND SURVIVAL DATA FITTING	51
	4.3	3.1	Log-linear model	51
	4.3	3.2	The non–linear models	52
	4.4	STATI	STICAL ANALYSIS AND SOFTWARE	54
	4.5	Resui	TS AND DISCUSSION	54

	4.5.1	The log-linear model	54
	4.5.2	The non–linear models	59
	4.6 CON	CLUSIONS	62
5	THERM	AL INACTIVATION OF <i>C. JEJUNI</i> IN FOOD	63
	5.1 INTR	ODUCTION	63
	5.2 Mat	ERIALS AND METHODS	63
	5.2.1	Campylobacter culture growth and enumeration	63
	5.2.2	Meat sample preparation and inoculation	64
	5.2.3	C. jejuni heat inactivation procedure.	64
	5.2.4	Enumeration of C. jejuni survivors	64
	5.3 Mot	DELLING AND SURVIVAL DATA FITTING	65
	5.3.1	The log-linear model	65
	5.3.2	The non–linear models	65
	5.4 Stat	ISTICAL ANALYSIS AND SOFTWARE	65
	5.5 Resu	ILTS AND DISCUSSION	66
	5.5.1	The linear model	66
	5.5.2	The non–linear models	68
	5.6 Con	CLUSIONS	71
6	THERM	AL INACTIVATION OF C. JEJUNI IN FOOD UNDER DYNAMIC CONDITIONS	72
	6.1 INTR	ODUCTION	72
	6.2 MAT	ERIALS AND METHODS	73
	6.3 CAM	PYLOBACTER CULTURE GROWTH AND ENUMERATION	73
	6.3.1	Meat sample preparation and inoculation	73
	6.3.2	Heat inactivation procedure	73
	6.3.3	Enumeration of C.jejuni survivors	74
	6.4 Mot	DELLING AND SURVIVAL DATA FITTING	74
	6.5 Stat	ISTICAL ANALYSIS AND SOFTWARE	74
	6.6 Resu	ILTS AND DISCUSSION	75
	6.7 Con	CLUSIONS	78
7	SURVI	AL OF C. JEJUNI UNDER OXIDATIVE STRESS AT DIFFERENT TEMPERATURES	80
	7.1 INTR	ODUCTION	80
	7.2 MAT	ERIALS AND METHODS	82
	7.2.1	Campylobacter culture growth and enumeration	82
	7.2.2	Exposure of C. jejuni to oxygen in broth	82
	7.2.3	Exposure of C. jejuni to oxygen on agar plates	82
	7.2.4	Exposure to H <sub>2</sub> O <sub>2</sub>	83
	7.2.5	Statistical analysis	83

	7.3	RESUI	TS AND DISCUSSION	83
		7.3.1	Exposure to oxygen in broth.	83
		7.3.2	Exposure to oxygen on agar plates.	88
		7.3.3	Exposure to H <sub>2</sub> O <sub>2</sub> .	89
	7.4	Conc	LUSION	92
8		EVALUA	TION OF PROCESSING PRACTICES CONTRIBUTING TO CAMPYLOBACTER CONTAMINATION IN	
NE	W	ZEALANI	D CHICKEN AND CHICKEN PRODUCTS	93
:	8.1	. INTRC	DUCTION	93
:	8.2	2 Mate	RIALS AND METHODS	94
		8.2.1	Campylobacter culture growth and enumeration	94
		8.2.2	Procedure	94
		8.2.3	Survival of C. jejuni in injected chicken products at the plant	95
		8.2.4	Recovery of C. jejuni from surfaces of the injected drumsticks	95
		8.2.5	Recovery of Campylobacter from the internal muscles of the injected drumsticks	95
		8.2.6	Survival of C. jejuni in injected chicken carcasses and chicken products in the retail market	95
		8.2.7	Statistical analysis	96
:	8.3	RESUL	TS AND DISCUSSION	96
		8.3.1	Survival of C. jejuni in brine solutions.	96
		8.3.2	Campylobacter survival in injected chicken products at the processing plant.	97
		8.3.3	Survival of C. jejuni in injected chicken carcasses and chicken products in the retail market.	99
:	8.4	Conc	LUSIONS	103
9		EVALUA	TION OF CAMPYLOBACTER MITIGATION STRATEGIES BY PHYSICAL METHODS IN POULTRY	
PRO	C	ESSING I	PLANTS	104
9	9.1	. INTRC	DUCTION	104
9	9.2	2. Mate	RIALS AND METHODS	107
		9.2.1	Campylobacter culture growth and enumeration	107
		9.2.2	Sample preparation and treatment	107
		9.2.3	Enumeration of C. jejuni survival.	108
9	9.3	RESUL	TS AND DISCUSSION	108
9	9.4	Conc	LUSIONS	113
10		QUANTI	TATIVE MICROBIAL RISK ASSESSMENT FOR CAMPYLOBACTERIOSIS IN NEW ZEALAND BY THE	
ΒA	YES	SIAN APF	PROACH	115
	10.	.1 IN	TRODUCTION	115
	10.	.2 M	ETHODS	116
		10.2.1	The modelling approaches	116
		10.2.2	Data sources	118
	10.2.3		Computing tools	128

	10.	3	Resul	TS AND DISCUSSION	129
		10.3.	1	Sensitivity analysis	134
	10.	4	Conc	LUSIONS	138
11		COMPARISON OF TIME-SERIES MODELS FOR PREDICTING CAMPYLOBACTERIOSIS RISK IN NEW			
		139			
	11.	1	Intro	DUCTION	139
	11.	2	Метн	ODS	141
		11.2.	1	Data collection	141
		11.2.2	2	Time-series decomposition techniques	141
		11.2.	3	ARIMA modelling	143
		11.2.4	4	Model evaluation	145
	11.	3	Resul	TS AND DISCUSSION	145
	11.	4	Conc	LUSIONS	149
12		EVAL	UATIC	IN OF FOOD HANDLING PRACTICE AMONG NEW ZEALANDERS AND CITIZENS OF OTHER	
DE	VEL	OPE	D COU	NTRIES AS THE MAIN RISK FACTOR FOR CAMPYLOBACTERIOSIS RATE	150
	12.	1	Intro	DUCTION	150
	12.	2	Метн	ODS	151
		12.2.	1	Survey of consumers' knowledge	152
		12.2.2	2	Survey of consumers' food handling practice	152
		12.2.	3	Observational study	152
		12.2.4	4	Bacterial transfer	152
	12.	3	Resul	TS AND DISCUSSION	153
		12.3.	1	Survey of consumers' food safety knowledge	153
		12.3.	2	Survey of consumers' practice	154
		12.3.	3	Observational consumers practice	156
		12.3.4	4	Bacterial transfer	157
	12.	4	Conc	LUSIONS	160
13		CONS	SUME	R POULTRY HANDLING PRACTICE SURVEY	161
	13.	1	Intro	DUCTION	161
	13.	2	New	ZEALAND CONSUMER STUDIES	162
	13.	3	Метн	ODS	163
		13.3.	1	The quantitative design	163
		13.3.2	2	The study approaches	164
		13.3.	3	Questionnaire development	164
		13.3.4	4	The Questionnaire	165
		13.3.	5	Target participants	166
		13.3.0	6	Data collection	166

	13.3.	7	Data analysis	166
13	3.4	Resu	LTS AND DISCUSSION	167
	13.4.	1	Association between respondents' socio-demographic characteristics and mean score	167
	13.4.	2	Consumer food hygiene practices	176
	13.4.	3	Consumer cross-contamination prevention practices	177
	13.4.	4	Consumer food temperature control and proper storage practices	179
	13.4.	5	Consumer general food safety awareness	180
13	3.5	CONC	CLUSIONS	183
14	GEN	ERAL [	DISCUSSION	185
14	4.1	Васк	GROUND	185
14	4.2	The p	ATHOGEN	185
	14.2.	1	Heat resistance	185
	14.2.	2	Air tolerance	187
14	4.3	Poul	TRY PROCESSING PLANT	189
	14.3.	1	Marination or tenderisation by needle injection technology:	189
	14.3.	2	Evaluation of Campylobacter mitigation strategies	191
14	4.4	The C	ONSUMER	193
	14.4.	1	The consumer kitchen	193
	14.4.	2	The consumer's knowledge and practices	193
14	4.5	THE G	OVERNMENT	194
14	4.6	CONC	CLUSIONS AND OPPORTUNITIES FOR FUTURE RESEARCH	200
	14.6.	1	Conclusions	200
	14.6.	2	Future research	203
APPE	ENDIX	Α		205
APPE	ENDIX	В		215
APPE	ENDIX	с		220
APPE	ENDIX	D		226
APPE	ENDIX	E		229
APPE	ENDIX	F		233
APPE	ENDIX	G		235
15	BIBLI	OGRA	РНҮ	258

# List of Tables

Table 2-1 Different species of Campylobacter	8
Table 2-2 Growth conditions for Campylobacter jejuni	9
Table 2-3'Thermal inactivation parameter from Campylobacter heat resistance studies	14
Table 2-4 A list of some published quantitative risk assessments of foods	31
Table 4-1 Estimated D and z values for Campylobacter isolates by two-step regression	58
Table 4-2 Estimated D and z values for Campylobacter isolates by one-step regression	58
Table 4-3 Calculated $D$ and $z$ values for <i>Campylobacter</i> isolates	59
Table 4-4 Selected examples for illustrating the goodness of fit for nine models	61
Table 4-5 Comparison of heating time(s) required to achieve a 7 D reduction	62
Table 5-1 D and z values for C. jejuni isolates estimated by one step regression	68
Table 5-2 Calculated D and z values for C. jejuni isolates	68
Table 5-3 Estimated D and z values for C. jejuni isolates by two-step regression	68
Table 5-4 Comparison of heating times (s) required to achieve a 7 D reduction	71
Table 6-1 $D$ and $\chi$ values estimated from inactivation data at 56 °C	75
Table 7-1 Slope values for the linear model shown in Figure 7.1 and Figure 7.2	87
Table 7-2 Results of two-way analysis of variance for effects of storage atmosphere, isolates,	and
temperature on survival of <i>C.jejuni</i> in broth	88
Table 7-3 Effect of temperature and storage atmosphere conditions	88
Table 8-1 The Bayesian analysis output values for the interested parameters	99
Table 10-1 Description of most variates (vi), parameters (pa), covariates (cv)	119
Table 10-2 Prevalence of Campylobacter in New Zealand birds	121
Table 10-3 Prevalence of Campylobacter in New Zealand whole chickens	124
Table 10-4 Statistical summary for the most important variates of models	133
Table 11-1 Time-series statistical model comparisons for campylobacteriosis forecast in New	V
Zealand	147
Table 11-2 Time-series model comparisons for campylobacteriosis rate	147
Table 11-3 Forecast performance of the number of cases for the most successful models	148
Table 13-1 Respondents' details and demographic characteristics	168
Table 13-2 Score on reported practices of personal food hygiene	177
Table 13-3 Score on reported practices in cross contamination prevention	178
Table 13-4 Score on reported practices in food temperature control and food storage	180

Table 13-5 Score on general food safety awareness	182
Table 13-6 Correlation between questionnaire sections by Pearson test	183
Table 14-1 Comparison of models in predicting campylobacteriosis cases in New Zealand for	or the
year 2010	197

# List of Figures

Figure 2-1 2018 Campylobacteriosis rates by age group in New Zealand	4
Figure 2-2 World poultry consumption per capita in selected countries	6
Figure 2-3 Different survival curves observed during heat inactivation of microorganisms	s16
Figure 2-4 Most common survival curves which are applicable for the Weibull model	17
Figure 4-1 Typical inactivation curves for the Weibull model	53
Figure 4-2 Survival curve for C. jejuni (ST-190P) at 53.5 °C	55
Figure 4-3 z values for six isolates of C. jejuni	
Figure 4-4 D value as a function of temperature for C. jejuni inactivation	60
Figure 4-5 δ value as a function of temperature for C. <i>jejuni</i> inactivation	60
Figure 4-6 Variation observed in the Weibull shape factor $\eta$ fitted to <i>C. jejuni</i> survival cur	ves as a
function of temperature	60
Figure 5-1 Survival curve for <i>C. jejuni</i> strain ST-474 <sub>H</sub> inoculated on chicken skin	67
Figure 5-2 Variation observed in the Weibull Shape factor $\eta$ fitted to <i>C. jejuni</i> survival cur	ves as a
function of temperature	69
Figure 5-3 $\delta$ value as a function of temperature for <i>C. jejuni</i> inactivation	70
Figure 5-4 D value as a function of temperature for C. jejuni inactivation	70
Figure 6-1 Thermal inactivation of C. jejuni ST-474P under dynamic conditions	76
Figure 6-2 Thermal inactivation of C. jejuni ST-474 <sub>H</sub> under dynamic conditions	76
Figure 6-3 Thermal inactivation of C. jejuni ST-48 <sub>P</sub> under dynamic conditions	77
Figure 6-4 Thermal inactivation of <i>C. jejuni</i> ST-48 <sub>H</sub> under dynamic conditions	77
Figure 7-1 The linear model fitted to the observed data for survival of human Campylobad	ter jejuni
Figure 7-2 The linear model fit to the observed data for survival of poultry Campylobacter	<i>jejuni</i> 85
Figure 7-3 Survival of <i>C. jejuni</i> isolates at 4°C	
Figure 7-4 Survival of <i>C. jejuni</i> isolates at 10°C	
Figure 7-5 Survival of <i>C. jejuni</i> isolates at 20°C	
Figure 7-6 Susceptibility of C. jejuni isolates to different hydrogen peroxide concentration	is in
filter discs on MH agar	
Figure 8-1 Survival of ST-474 <sub>P</sub> in two replicate flasks containing solution A at 7 °C	97
Figure 8-2 Survival of ST-474 <sub>P</sub> in two replicate flasks containing solution B at 7 °C	97
Figure 8-3 Survival of <i>Campylobacter</i> in chicken drumsticks	
Figure 8-4 Survival of Campylobacter in chicken drumsticks (sampled at the processing line	e)99

Figure 8-5 Survival of Campylobacter on the surface of whole chickens (sampled at retail)	
Figure 9-1 Comparison of the effect of hot-water immersion on the survival of Campyloba	cter109
Figure 9-2 Temperature profile of water and chicken during heating to 60°C.	111
Figure 10-1 The complete model and the augmented model	
Figure 10-2 New Zealand chicken production data provided by the PIANZ	
Figure 10-3 Posterior marginal kernel densities	
Figure 10-4 Posterior marginal kernel densities	
Figure 10-5 Posterior marginal kernel densities	
Figure 11-1 Campylobacteriosis notifications in New Zealand from 1998 to 2010	146
Figure 14-1 Campylobacteriosis Number of cases in New Zealand (EpiSurv, 2019)	

# 1 Introduction

#### 1.1 **Problem description**

New Zealand (NZ) has a higher rate (~160 cases per 100,000) of reported campylobacteriosis cases than the rest of the developed world (French et al., 2019). This is 12 times higher than the rate for the United States, four times higher than the rate for Canada, Germany, Netherland and nearly six times higher than the rate in Norway (Kaakoush et al., 2015). The reasons for this high rate are not known. Due to the large economic and health consequences of campylobacteriosis, intervention programmes to reduce the disease rates are required urgently. For these to be effective, the causes for this high rate need to be investigated and addressed.

Several international and New Zealand epidemiological studies have established the link between the consumption of poultry meat and campylobacteriosis. The two main risk factors (consumption of undercooked meat or cross-contamination) identified in these studies need to be investigated in the NZ context.

There is also ambiguity and a lack of data concerning the heat-resistance and survival under aerobic conditions at low temperatures and an ambient temperature of NZ *Campylobacter* strains and international strains (Lake et al., 2003, Sagarzazu et al., 2010, Oh et al., 2019). Thus, there is a need to more fully determine the temperature-dependent growth of *Campylobacter* strains, to provide reliable data for decision-makers (such as government authorities), industry and consumers in NZ.

#### 1.2 **Project** aim

The aims of this PhD research were:

- To experimentally evaluate the thermal process lethality of NZ *C. jejuni* strains and to update the current knowledge of kinetic reaction parameters such as *D* and z values of *Campylobacter* strains in general.
- To study the survival of the NZ *Campylobacter* strains that are most frequently associated with poultry products during processing, storage and handling.
- To assess the processing practices and the interventions applied in NZ poultry processing plants such as needle injection and how *Campylobacter* infection possibly has been increased by any processing practices or could be minimised by alternative processing practices such as hot water immersion or hot water wash.

- To investigate the NZ context for cross-contamination during poultry preparation at home as a possible hypothesis to explain the high reported campylobacteriosis rate.
- To construct a model to accurately predict the number of campylobacteriosis cases in NZ.
- To present a structured approach for identifying, assessing, and summarising campylobacteriosis data. Such a structured methodology will provide the most relevant scientific research information to the food safety decision-makers.

Campylobacteriosis is an internationally recognised complex problem. It is important to investigate all the above aims to understand campylobacteriosis in NZ and thereby improve public health.

## 1.3 Hypothesis

The hypotheses for this research are:

- New Zealand *Campylobacter* strains from human and poultry (ST-474, ST-190, and ST-45) are heat resistant and survive normal cooking processes (chapter 4-6).
- New Zealand Campylobacter strains (ST474, ST190 and ST48) are oxygen tolerant (chapter 7).
- The secondary processing technique, marination by needle injection, may increase the *Campylobacter* contamination level in carcasses (chapter 8).
- The physical intervention by combining the hot water wash and hot water dipping would reduce *Campylobacter* level in carcasses, especially the heat-sensitive NZ strains (chapters 4 - 6). This is as an alternative to the EU banning of the use of chemicals in poultry processing in 2006 (chapter 9).
- Poultry farms practices, poultry processors practices, poultry cooking and handling practices in NZ homes are not effective in eradication *Campylobacter* (chapters 10, 12, 13).
- The use of time-series to forecast campylobacteriosis cases in NZ would be more accurate than the forecast provided by the risk assessment approach (chapter 11).

# 2 Literature Review

#### 2.1 Background

Worldwide, *Campylobacter* causes more cases of bacterial foodborne illness each year than any other bacterial pathogen (Jonsson et al., 2010, Rushton et al., 2019). It is the most commonly reported foodborne disease in industrial countries (Mylius et al., 2007, Facciolà et al., 2017).

The symptoms of campylobacteriosis include abdominal pain, watery or bloody diarrhoea, headache, fever, nausea, muscle pain (Gölz et al., 2018). It is sometimes accompanied by vomiting and late complications such as reactive arthritis, Reiter syndrome (RS) and Guillain-Barré syndrome (GBS). In NZ, the most affected population groups are adults between ages 20 and 29 and children < five years with 1057 and 884 respectively (EpiSurv, 2019) (see Figure 2.1 which is extracted from Episurve website)



#### Campylobacteriosis by Age group

Figure 2-1 2018 Campylobacteriosis rates by age group in New Zealand.

In NZ, the estimated annual cost for campylobacteriosis is more than NZ\$77 million per year (NZFSA, 2007). Several studies have estimated the burden of campylobacteriosis, expressed as disability-adjusted life-years (DALYs) (Babo Martins et al., 2017, Lackner et al., 2019). This refers to the number of years lost due to disability caused by sequelae of the infections that occur in GBS

and RS. Campylobacteriosis infection is considered the most expensive disease in Europe and NZ. The cost due to campylobacteriosis sequelae of the infections is estimated as 1568 DALYs (Lake et al., 2010).

In 1980, NZ legislated campylobacteriosis as a notifiable enteric disease. Since then, the incidence of reported campylobacteriosis has risen steadily, and it is believed that NZ has the highest rate of reported cases in the world. In 2006, the notification rate was 384 per 100,000, an increase of 56 % from 2001 and 14% since 2005. This is 35 times higher than the rate in the United States, four times higher than the rate in Australia and five times higher than the rate in the UK. (Baker et al., 2006a). The actual figures are estimated to be at least seven to eight times higher than the notified cases (Baker et al., 2006a). It is estimated (based on 7.6 multiplier factor commonly used in epidemiological studies) that the total number of actual campylobacteriosis cases as 120,000 per year (Baker et al., 2007). Another study reported that actual campylobacteriosis could be 30 times higher than the reported cases (Newman et al., 2015).

Baker (2006b) concluded that campylobacteriosis is a real public health burden, and it is not a surveillance artefact. The reason for this increase in the incidence is still not established, and many aspects of the epidemiology are not clear (Baker et al., 2006a). However, several international epidemiological studies have established the link between the consumption of poultry meat and campylobacteriosis (Wingstrand et al., 2006, Skarp et al., 2016). NZ epidemiological studies also indicate a similar link (French, 2008a, Mullner et al., 2009b, Baker et al., 2011, Nohra et al., 2018, French et al., 2019). In addition, the high level of prevalence and contamination of poultry meat (80 -100 %) with *Campylobacter* (French, 2008b) provide further evidence about this link. Moreover, the increase in the consumption of poultry is concomitant with the increase in the rate of reported campylobacteriosis. A dramatic drop in disease notification rate accompanied the withdrawal of poultry from the markets in Belgium and the Netherlands (Vellinga and Van Loock, 2002) also confirm such a link. The similar remarkable decline in the notification rate following the freezing of fresh poultry in Iceland indicated fresh chicken as the main risk factor for campylobacteriosis (Lake, 2006).

New Zealanders consume a slightly higher quantity of poultry per capita than Western European countries but less than the amounts consumed by North American and Australians as per the Food and Agriculture Organization of the United Nations (FAOSTAT) database for the year 2018 (Figure 2.2).



Figure 2-2 World poultry consumption per capita in selected countries

Poultry meat products are increasingly available to NZ consumers in a wide variety of formats: fresh or raw products (e.g. whole carcass, selected pieces, pre-prepared cuts), partially-processed raw products (e.g. frozen, marinated, coated, 'tenderised'), raw reconstituted meats (e.g. sausages, nuggets, patties, luncheon, bacon analogues) and many ready-to-eat (cooked) forms. Raw products are cooked by the home consumer or foodservice provider by a variety of methods including roasting, braising, grilling, frying and deep-frying.

It is important for processors, caterers and consumers to apply adequate heat treatment to destroy all food pathogens, including *Salmonella* and *Campylobacter* (Humphrey et al., 2007, NACMCF, 2007). Moreover, there is a possibility that undercooking may occur and/or a proportion of bacterial cells in some protected areas may survive the normal heating process. Thus, the consumption of undercooked chicken or poultry products is considered an important risk factor for campylobacteriosis (Butzler, 2004).

It has been reported that NZ *Campylobacter* strains differ genetically from international strains. The subtypes isolated in NZ have not been identified previously in other international databases, and some NZ strains are antimicrobial drug resistance (McKenzie, 2006, French et al., 2019). It is possible that local strains have a greater heat tolerance and thus are better able to survive cooking, an exposure route often discounted in risk assessments studies.

## 2.2 The pathogen

Most Campylobacter species are motile and possess a single flagellum at one or both poles. A majority

of *Campylobacter* are microaerophilic. The genus *Campylobacter* has changed dramatically since it was identified in 1963 (Sebald and V'Eron, 1963). At that time, the genus contained just two taxa, *C. fetus* and *C. bubulus. Campylobacter* contains 16 species and six subspecies (Boxall, 2005) (Table 2.1). However, there are currently 34 to 35 *Campylobacter* species recognised, but with the improved diagnostic techniques and genomic analysis, this number is expected to increase over time (Anonymous, 2017, Klein, 2016). Most human illnesses are caused by *C. jejuni. Campylobacter jejuni* and *C. coli* are considered the major species that threaten public health in NZ (Lake et al., 2003). A minority (1-5 %) of human campylobacteriosis cases are caused by other species (*C. upsaliensis* and *C. lari*).

Table 2-1 Different species of Campylobacter

Species	Subspecies	Biovar	Source
consisus			Human oral cavity
showae			Human oral cavity
curvus			Human oral cavity
rectus			Human oral cavity
gracilis			Human oral cavity, abscesses, lesions
			Human oral cavity, abscesses, lesions,
		sputorum	faeces, cows, sheep, pig's genital tract,
sputorum			aborted mater
		faecalis	Cow, sheep, faeces
		paraureolyticus	Cattle, human, faeces
hominis			Human oral cavity
			Pig intestine, oral cavity
			Aborted fetus Placenta, stomach
fetus	fetus		contents
	venerealis		Cows vaginal mucus, semen, prepuce
hyointestinalis	hyointestinalis		Pig intestine
	lawsonii		Pig stomach
lanienae			Human (abattoir workers)
	jejuni		Poultry, cows, sheep, human,
jejuni			Environment, faeces
	doylei		Human gastric tissue, diarrhoea, blood
coli			Pig faeces
lari			Birds, animals, water, shellfish, faeces
upsaliensis			Cats, dogs, faeces
helveticus			Cats, dogs, faeces
mucosalis			Pig intestine, oral cavity

Note: Adapted from (Boxall, 2005).

*Campylobacter jejuni* is a Gram-negative, slender, curved, and motile rod. It contains two subspecies that differ substantially in their ubiquity and to some extent ecology (*C. jejuni* subsp. *jejuni* and *C. jejuni* subsp. *doylei*). *C. jejuni* subsp. *jejuni* is often referred to as *C. jejuni* and represents the taxon first reported by Jones (Jones et al., 1931) as *Vibrio jejuni* isolated from bovine intestinal contents. Since the 1970s, this organism has been widely considered as the most commonly isolated bacterium that has caused human gastroenteritis (Facciolà et al., 2017). Its prevalence in food animals makes it one of the most important foodborne pathogens. Many studies (Dingle et al., 2001, Parkhill et al., 2000, Thépault et al., 2018) have reported its genetic diversity and genomic plasticity.

## 2.3 Survival and growth of *C. jejuni* in the environment

*Campylobacter* species exist in many wild and domestic warm-blooded animals (Adams and Moss, 2000). It does not survive well outside the gut, nor does it replicate readily. Most literature reports the best range for growth as 37- 43 °C with 42 °C as the optimum, similar to the chicken body temperature (Adams and Moss, 2000, CFSAN, 2008, FDA/CFSAN, 2000, Forsythe, 2000, Lake et al., 2003). Over the years, various studies targeted this pathogen to determine the environmental conditions necessary for growth and survival. The published values of temperature, pH, water activity and the optimal atmospheric conditions are summarised in Table 2.2.

Table 2-2 Growth conditions for Can	pylobacter	jejuni
-------------------------------------	------------	--------

Parameter	Optimal Conditions	Growth Range
Temperature	42 °C	30.5 to 45 °C.
Generation time	$\sim 1$ hour	
рН	6.5	4.9 to 9
Water activity	0.997	≥0.987
Gas composition	5% O <sub>2</sub> , 10% CO <sub>2</sub> , 85 N <sub>2</sub> %	

Note; Adapted from Microbiological Specifications of Food Pathogens (ICMSF, 1996)

Laboratory studies suggest that *Campylobacter* species are sensitive to adverse conditions. They are unable to grow in the presence of elevated levels of oxygen (above 5 % O<sub>2</sub>), difficult to multiply outside the animal host and are highly sensitive to environmental conditions such as drying, acidity, salting, freezing, chlorination, osmotic stress and pressure (Park, 2005, Burgess et al., 2016). Consequently, the organism can survive only for up to four to six hours on hands and moist surfaces (De Cesare et al., 2003, Humphrey et al., 2001). Moreover, *Campylobacter* is a poor competitor against other organisms in the environment (ICMSF, 1996). However, the pathogen can survive and remain in a viable but in a non-culturable state at 4 °C for up to four months and sometimes as high as seven months (Haddad et al., 2009). Others (Dasti et al., 2010) have reported the catalase activity, ATP generation, oxygen consumption, chemotaxis, and protein synthesis for the pathogen at the same temperature. Its viability at -20 and -70 °C, for 14 days and 56 days, respectively, has also been confirmed (Lee et al., 1998). Another study (Gunther Iv et al., 2015) reported that *C. jejuni* survived for 33 days at -20 °C. The survival ability of *Campylobacter* strains survival at low temperature remains poorly understood from the inconsistency of the results reported by the above studies and further research is required.

#### 2.3.1 Atmosphere

*Campylobacter* strains survive well in modified atmosphere and vacuum packaging, but poorly at the atmospheric oxygen concentration (21%) (Kelly, 2008). The optimum atmosphere or gas composition for the growth is 5 %  $O_2$ , 10 %  $CO_2$  and 85 %  $N_2$  (ICMSF, 1996). Studies that have investigated the growth of *Campylobacter* under different gas compositions with different techniques and substrates are in general agreement with this proposed optimum composition (Gharst et al., 2013, Meredith et al., 2014).

Several investigators have studied the effect of temperature on oxidative stress reviewed by (Park, 2005). This review reported that the survival of *Campylobacter* strains in poultry and meat was not influenced by the storage atmosphere or the treatment condition at 4 °C, which is in agreement with other studies. Another study indicated that the exposure to atmospheric oxygen in broth culture for five and 15 hours had no effect on the *C. jejuni* growth and the count was similar to the *C. jejuni* incubated under micro-aerobic conditions (Mihaljevic et al., 2007). This is consistent with other reports, also regarding the growth and adaptation of *C. jejuni* to aerobic metabolism (Oh et al., 2015). There are studies which report that the presence of oxygen increased the death rate at 4 °C in broth and milk (Butzler, 2014). Yamasaki (2004) reported that initial *C. jejuni* inoculum (9.5 log) concentration declined by half log after the first six hours under aerobic conditions at 37 °C followed by another one log reduction in the following six hours. Yamasaki observed that a rapid

reduction by about 5 log occur between 12 - 15 hours of incubation. Others reported that *C. jejuni* could survive oxidative stress at 4 °C for several weeks (Chan et al., 2001, Garenaux et al., 2008, Oh et al., 2017). A NZ study revealed that *C. jejuni* could adapt to aerobic growth after repeated subculture. In this study, about 81 % of the human and 65% of poultry *C. jejuni* isolates that can grow under aerobic conditions in the nutrient agar, a cocktail of isolates (human, poultry, veterinary) survived aerobically for more than four weeks in poultry meat at 5 °C and for more than a week at 25 °C (Chynoweth et al., 1998). From all of the above studies, it is concluded that there is ambiguity about the behaviour of *C. jejuni* under oxidative stress. Evidently, *Campylobacter* causes more difficulties in standardization and handling than any other pathogen (Bergsma et al., 2007, Jasson et al., 2007).

#### 2.3.2 Water Activity

Campylobacter is sensitive to drying, and its survival is dependent on the suspending substrate, the temperature and the extent of drying (ICMSF, 1996). The pathogen does not survive long in less moist environments compared to Salmonella and E.coli. For example, Campylobacter survives poorly on dishes with organic matter while Salmonella can survive for 24 h and E. coli even longer (Mattick et al., 2003). Coates et al. (1987) study reported a 3 - 7 log reduction in the pathogen numbers within two mins when the cells suspension were spread on fingertips. Handling of raw meat with moist hands enhances Campylobacter survival on fingertips which facilitates the cross-contamination or direct contamination during food preparation in the kitchen (Humphrey et al., 2001). Generally, moist hands transfer microorganisms more easily than do dry hands to skin, food and preparation surfaces (Mutters and Warnes, 2019). Others have reported that surface drying during air chilling in meat or poultry processing reduces water activity, retard bacterial growth and cause enough injury to pathogenic bacteria when pathogen levels on carcases are compared before and after airchilling red meat and at poultry processing abattoirs. It has been suggested that drying is responsible for the reduction of the pathogen numbers on the surfaces (Sánchez et al., 2002, Lu et al., 2019). Poultry is traditionally chilled by immersion in cold water, called the spin chiller, and are packed and stored in a high moisture environment which enables Campylobacter to survive longer. Many studies (Huezo et al., 2007, Sánchez et al., 2002, Lu et al., 2019) have considered that dry air chilling at the poultry processing plant is the best technology to reduce Campylobacter contamination level and to prevent the cross-contamination which is acquired in the wet chilling approach. Therefore, dry air chilling is implemented in many countries.

#### 2.3.3 Temperature

Campylobacter cannot grow below 30°C (Jacobs-Reitsma et al., 2008), which means that it should

not multiply during food processing or storage at a typical ambient temperature. However, it survives better in foods under refrigeration than at ambient temperature (Bhaduri and Cottrell, 2004). It grows only slowly at the optimum temperature (42 °C) in the laboratory. *Campylobacter* typically takes 48 h for good colony formation of on agar plates. Plates may be incubated at 37 °C or 42 °C, but it is a common practice to incubate at 42 °C to minimise the growth of contaminants and to select *C. jejuni* and *C. coli* for optimal growth (Anonymous, 2017). However, the International Organization for Standardization recommends incubating plates at 41.5 °C (Jacobs-Reitsma et al., 2019). *Campylobacter* failed to grow above 47 °C (Baserisalchi et al., 2006) as a result of the irreversible denaturation of the proteins, the thermal breakdown of the plasma membrane and the most sensitive parts of ribosomes. The kinetics of microbial inactivation and its importance in food processing are explained in the next section (2.4). Information on the biological aspects and mechanisms of inactivation at a molecular level to understand what takes place from a physiological and genetic perspective are not yet completely understood, but the current knowledge has been summarised in various studies (Park, 2000, Park, 2002).

The methods for measuring the heat resistance of food pathogens can be classified by the heating method (direct or indirect) or by the method of sampling (successive sampling systems or multiple replicate unit systems) (Pflug, 1990, Downes and Ito, 2001).

#### 2.4 Kinetic analysis of bacterial inactivation

#### 2.4.1 Linear models

Bacterial inactivation is a function of residence time and temperature in the environment to which they are exposed to. It is commonly recognized that the number of organisms decreases exponentially with time. This implies that the rate of inactivation is proportional to the number of viable cells and that death follows first-order kinetics (Chick, 1910, Davis et al., Holdsworth, 1998, Prescot et al., 1996). This is mathematically described at constant temperature by:

$$\frac{dN}{dt} = -k_T N$$
<sup>2-1</sup>

where N represents the number of cells per unit mass or volume (colony-forming unit; cfu/ml or cfu/g), t the total heating time at constant temperature (in s), and  $k_T$  a temperature-dependent death rate constant (per unit time)

On integration, this yields the following expressions:

$$\ln \frac{N_0}{N_t} = k_T t$$
<sup>2-2</sup>

$$\log_{10} \frac{N_0}{N_t} = \frac{t}{D}$$
 2-3

Where  $N_0$  is the initial count of bacteria, and *D* is the decimal reduction time. This is the time required to reduce the bacterial count by a factor of 10 and is equal to the number of seconds for the survival curve to traverse one log cycle.

The temperature dependence of the  $k_T$  or *D*- value can be expressed using the Arrhenius relationship or the *z* value model Equation 2.4.

$$\log_{10}(D_T) = \log_{10}(D_{ref}) + \frac{T_{ref} - T}{z}$$
 2-4

where:

 $D_{ref}$  is the *D*-value at a reference temperature  $T_{ref}$  (s).

T is the temperature (°C).

 $\chi$  is the change of temperature (°C) required to reduce the *D*-value by 10-fold.

A summary of D and z-values for various pathogens and products can be found in a report from the FDA/CFSAN (FDA/CFSAN, 2000). Table 2.3 summarises D and z-values reported for *Campylobacter*.

Substrate	Temp	D value (min)	z value ⁰C	References (as a group for each
	°C			section)
Skim milk	48	7.2 - 12.8		(Doyle and Roman, 1981)
	50	1 – 7.3		(Christopher et al., 1982)
	52	5.2	2.8 - 5.3	(Skirrow, 1982)
	53	1.56 - 1.95		
	55	0.74 – 1.1		
Peptone	49	14.9 - 15.2		(Blankenship and Craven, 1982)
water/saline	50	0.88 - 1.63	6-6.4	(Forsythe, 2000)
Buffer	51	4.90 - 7.02	4.94	(Humphrey and Lanning, 1987)
Scald water	51.5	2.45-3.8	4-5.2	(Sorqvist, 1989)
	52	6.4 - 8.7		(Gill and Harris, 1982)
	52	1.96 -10.82		(Doyle and Roman, 1981)
	53	1.71 - 2.70		(Nguyen et al., 2006)
	53.5	0.72 -2.14		(Oosterom et al., 1983)
	55	0.64 - 1.09		(Sorqvist, 2003)
	55	4.6-5.3		(Al-Sakkaf and Jones, 2012)
	56	0.78		(Lahou et al., 2015)
	57	0.25		
	58	0.24		
	60	0.13		
	60	0.26 - 0.95		
	60	0.02-0.07		
	65	0.22		
	70	0.18		

Table 2-3Thermal inactivation parameter from Campylobacter heat resistance studies

Chicken	49	20.5	4-8	(Blankenship and Craven, 1982)
Beef	50	5.9 - 6.3,	8.7 - 10.2	(Koidis and Doyle, 1983)
Lamb	50	5.9 - 13.3	7.6 - 11.3	(Bergsma et al., 2007)
Smoked	51	8.77 – 9.27	12.3	(de Jong et al., 2012)
Salmon	53	4.85 - 4.89	2.8-5.8	
	55	0.96–1.26		(Al-Sakkaf et al., 2010)
	55	2.12-2.25		(van Asselt and Zwietering, 2006)
	56	0.62 - 0.96		(Li et al., 2002)
	56.5	22.5- 41.6		(Jackowska et al., 2008)
	57	0.79 -0.98		(Lahou et al., 2015)
	58	0.25		
	60	0.21 - 0.26		
	60	11.2-18.0		
	100	1.90		
	109-	1.95		
	127			

## 2.4.2 Alternative models of thermal inactivation

Many studies in the literature have reported that bacterial survival curves (i.e. log N or ln N vs t) are not always strictly linear (Figure 2.3). They could be sigmoid curves, curves with shoulders, tailings, and concavity upwards and downwards (Campanella and Peleg, 2001, Cerf, 1977, Peleg et al., 2000, van Boekel, 2002). Models have been proposed to fit these different survival curves. For example, a simple two-phase model was suggested by Whiting (1993), used for *Listeria monocytogenes* (Breand et al., 1998)



Figure 2-3 Different survival curves observed during heat inactivation of microorganisms (Geeraerd et al., 2005).

when  $0 \le t \le t_L$ 

$$\log_{10}N_{t} = \log_{10}N_{0}$$
 2-5

when  $t \ge t_L$ 

$$\log_{10} N_{t} = \log_{10} N_{0} - \frac{1}{D} (t - t_{L})$$
<sup>2-6</sup>

Where t<sub>L</sub> is the lag time.

For the other forms of survival curves, a power-law modification of Equation (2-3) may be applied:

$$\log_{10} \frac{N_t}{N_0} = -\frac{t^p}{D}$$

Where p is an arbitrary (fitted) power; p = 1 gives the first-order model (Figure 2.4a), and when p>1 gives a convex shape (Figure 2.4b) results of p < 1 gives a concave curve (Figure 2.4c) (Huang and Juneja, 2001, Mafart et al., 2002).

Probabilistic models have also been developed in the last few years to describe the survival curve as a cumulative form of the temporal distribution of a lethality event (Peleg and Cole, 1998). For example, the Weibull frequency distribution has commonly been used to express the bacterial death time or bacterial inactivation (Cunha et al., 1998, Fernandez et al., 1999, Mafart et al., 2002, Peleg, 2002, Peleg and Cole, 1998).

$$\log_{10} N_t = \log_{10} N_0 - \left(\frac{t}{\delta}\right)^{\eta}$$
<sup>2-8</sup>

 $\eta$  and  $\delta$  are the two parameters relating to the scale and shape of the distribution function. The above model is another form of the power-law model (Equation 2.7) in which "D" value of Equation 2.7 is replaced by " $\delta$ "in the Weibull model. Most of the authors of these studies reported that the Weibull model as being successfully tested by using a variety of published heat resistance data and the authors have concluded that the Weibull model is considered a more suitable and simpler model to describe non-log-linear survival curve.



Figure 2-4 Most common survival curves which are applicable for the Weibull model (a: linear model when  $\eta = 1$ , b: convex curve when  $\eta > 1$ , c: concave curve when  $\eta < 1$ )

The non-linear models which were fitted for data in this study and covered most of the survival curves observed during the heat treatment shown below:

Shoulder and linear (Figure 2.3 brown dotted line), linear and tail (black straight line), shoulder linear and tail (straight brown line) (Geeraerd et al., 2000).

$$\log N = \log \left[ (10^{\log(N_0)} - 10^{\log(N_{res})} \right] \cdot e^{-k_{max}t} \cdot \left( \frac{e^{k_{max}s_l}}{1 + (e^{k_{max}s_l} - 1) \cdot e^{-k_{max}t}} \right) + 10^{\log(N_{res})}$$
2-9

Where:  $N_{res}$  is the number of cells in a subpopulation of the total population N, which are more resistant to the thermal treatment (Cerf, 1977).
Weibull with a tail (Albert and Mafart, 2005)

$$\log N = \log[(N_0 - N_{res}) \cdot 10^{(-(\frac{t}{\delta})^{\eta})} + N_{res}$$
<sup>2-10</sup>

Biphasic (Cerf, 1977) (Figure 2.4 dotted red)

$$\log N = \log N_0 + \log (f \cdot e^{-k_{\max 1} t} + (1 - f) \cdot e^{-k_{\max 2} t})$$
 2-11

f is the fraction of the initial population in a major subpopulation, (1-f) is the fraction of the initial population in a minor subpopulation ),  $f=N_{01}/N_0$  and (f-1)  $=N_{02}/N_0$ .

Biphasic with shoulder (Figure 2.4 dotted light blue) (Geeraerd et al., 2005)

$$\log N = \log N_0 + \log \left( f \cdot e^{-k_{\max 1} t} \cdot \frac{e^{k_{\max 1} s_l}}{1 + (e^{k_{\max 1} s_l} - 1) \cdot e^{k_{\max 1} t}} + (1 - f) \cdot e^{-k_{\max 2} t} \cdot \left( \frac{e^{k_{\max 1} s_l}}{1 + (e^{k_{\max 1} s_l} - 1) \cdot e^{-k_{\max 1} t}} \right)^{\frac{k_{\max 2}}{k_{\max 1}}} \right)$$
2-12

Generally, the non-linear models are based on experimental data without understanding the physiological or physical behaviour mechanisms of the microorganisms, and they are often over parameterised. In addition, the complexity of some of these models hampers their application in the calculation of the heat treatment as food industry practitioners and regulatory institutions are unwilling to abandon the z and D-value based method which through experience has demonstrated to be generally safe and reliable (Heldman and Newsome, 2003).

#### 2.4.3 Handling bacterial survival data

Two major methods are employed for determining bacterial survival curves. The most common method follows from heating a number of replicate samples for different time intervals at a fixed temperature. The samples are plated, and after proper incubation, the number of colonies (colony forming units; cfu) grown on plates is counted. The logarithm of the colony number is then plotted against the heating time to derive the survival curve.

The alternative negative fraction method is based on recording the presence or absence of bacterial growth (Pflug, 1990). Appropriate statistical and mathematical methods (e.g. fitting the observed data to pre-existing model or develop a new model) must then be applied to obtain the appropriate kinetic values (D and x). The final step is to validate the selected or developed model by comparing predictions from the models to observed growth, survival, death responses of bacteria in relevant foods. Detailed explanations of interpretation of survival data and modelling have been reported (Pflug, 1990, Pinheiro, 2000, Stumbo, 1973).

#### 2.4.4 Non-isothermal models

Heat treatments employed in the food industry invariably results in temperature fluctuations during the heating or the cooling stages. In practice, the thermal effect during the time required to achieve a constant temperature (the come-up time) must be accounted for, and in many cases, it has a much larger effect than the holding time at a constant temperature. Moreover, kinetic parameters used under isothermal conditions should not be applied to variable processing conditions without validation under dynamic conditions (Welt et al., 1997). Therefore, many researchers have tried to apply non-isothermal methods as an alternative to the study of microbial inactivation kinetics. Dynamic models describing microbial inactivation are expressed as differential equations which incorporate the imposed or observed time-temperature profile as an input to the dynamic models selected which describe the thermal death of micro-organisms. The resulting equations can be solved numerically to obtain the number of surviving organisms at a given position in the time-temperature profile. If the rate of temperature increase can be described by a simple function, the dynamic model equation can be solved analytically (Miles and Mackey, 1994). Several procedures proposed to accomplish this integration are described in the literature (Patashnik, 1953, Stumbo, 1973).

The determination of kinetic parameters under dynamic conditions can better reflect a realistic processing scenario and the 'come-up time' is explicitly a part of the experimental design. That is, by specifying the heating rate and time-temperature profile there will not be a limited temperature range from which to choose sufficient points for the experiments, particularly for heat-sensitive organisms (Welt and Tong, 1993, Welt et al., 1997).

The models employed in predictive microbiology have most commonly been obtained by differentiating a modified Gompertz equation with respect to time, in combination with an Arrhenius-type equation, to describe the microbial load as a function of both time and temperature

(Geeraerd et al., 2000). From this, a class of models have been developed used to describe the inactivation of micro-organism kinetics at time-varying temperatures (Conesa et al., 2003, Fernandez et al., 1999, Geeraerd et al., 2000, Schellekens et al., 1995, Van Impe et al., 1992). The results obtained for the kinetic parameters by non-isothermal methods have differed from those obtained by isothermal methods (Conesa et al., 2009, Decordt et al., 1992). However, Valdramidis (2008) succeeded in obtaining similar results from both methods and concluded that non-isothermal kinetic parameters are more accurate and can be obtained with less experimental effort.

#### 2.5 Factors affecting heat resistance of microorganisms

Many factors influence the heat resistance of a pathogen including between strain variation, presence of salt or acid, the growth phase of the cells, experimental conditions (e.g. heating media directly or indirectly), cooling after heating, and products or laboratory media used.

#### 2.5.1 Strain variation

Strains within an individual Campylobacter sp. genotype or phenotype vary considerably in their ability to survive in the environment and along the food chain and can exhibit large differences in their stress response. The diversity within Campylobacter jejuni strains is well known and has been detected at both the phenotypic and genotypic level (Gölz et al., 2018). The number of typed strains is rapidly increasing within data banks as different typing methods such as microarrays, multi-locus sequence typing system (MLST), pulsed-field gel electrophoresis (PFGE), Penner & Lior and other DNA-based methods and porA typing based on the sequencing of the major outer membrane protein (MOMP) encoding gene (Garaizar et al., 2006, Petersen et al., 2001, Wieczorek and Osek, 2019). Different Campylobacter strains show different capacities to resist environmental stress (Habib et al., 2010, Alter and Scherer, 2006, Chan et al., 2001, Duqué et al., 2019). A few studies have addressed differences in strain behaviour under environmental stress. Chan (2001) reported the variation in the survival of 19 different C. jejuni isolates subjected to temperature stress. The focus of the study was on the survival of the pathogen at low temperatures (4 °C and -20 °C). The results of the study have been used in the epidemiological investigation as human isolates showed greater cold-tolerance than poultry strains, although the freezing tolerance was strain-specific and independent of the strain source.

Similarly, Cools (2003) tested the survival of some *C. jejuni* strains in drinking water and reported their recovery after 64 days at 4 °C. Poultry strains survived for longer periods than human, water and bovine strains. Murphy et al. (2003) suggested that the pathogen acid tolerance may be strain-dependent. The carrying of *Campylobacter* strains by poultry carcass during processing is also strain-

dependent with the more resistant strains remaining on the carcass at the end of processing (Newell et al., 2001).

A study of NZ strains by Gill and Harris (1982) investigated six *Campylobacter* strains from human cases, six from calf and sheep and they reported a *D* value of less than a minute at 60 °C in broth medium. No poultry strains were included, and the study was conducted with the available methods, instrumentation and techniques at that time with regard to strain typing, isolation, recovery, enumeration, inactivation and incubation.

#### 2.5.2 Test or experimental conditions

The experimental conditions have a significant effect on the reported data of heat resistance. A variety of methods, used for determination of heat-resistance, have attempted to reduce the errors introduced by calculation of heat transfer (sample 'come up time' calculation), sample dilution and accuracy of time and temperature measurement. Sorqvist (1989, Sorqvist) compared the test tube types for the D- value determination and found marked differences between the capillary tube and other tube types. The *D*-value obtained by a test tube was up to 29 times higher than the *D*-value obtained by the capillary tube. Similarly, Miller et al. (2000) reported a lower D- value for Listeria monocytogenes by using a complete submerged system. A study by Line (2006) compared the type of enrichment vessels used for recovery of Campylobacter. Moore and Madden (Moore and Madden, 2000) compared the heat transfer prosperities of three flasks to select the best one which can be used later for their Campylobacter heat sensitivity experiment, and they concluded that stoppered glass diluent bottles were better than sealed glass universal bottles or universal plastic bottles in terms of their conductivity. Heat-resistance studies have used a range of equipment for heating including water bath, heating block, incubators, oil bath, thermoresistometer (Condon et al., 1993, Condon et al., 1989), submerged-coil heating unit (Cole and Jones, 1990), temperature gradient incubator (Elsgaard and Jørgensen, 2002), differential scanning calorimeter (DSC) heating unit (Brown, 1988), and Peltier units. The Petlier unit operates like a thermocouple in reverse and the junctions heat or cool as the voltage changes across them. A specially designed apparatus for heating food sample was developed for the Bug Death project (Foster et al., 2006). This apparatus had the ability to heat the surface of food from 8 to 120 °C in 14 s, cooling from 120 to 40 °C in 28 s and then to 8 °C in 300 s. A 1 ml volume aluminium cell was developed for investigating the kinetics parameters (Chung et al., 2008) and a modified thermos-resistometer with the ability for use during non-isothermal heat treatment (Conesa et al., 2009), fast-responding thermal-deathtime tubes (Büchner et al., 2012) and the heating block system (HBS) (Kou et al., 2016). Detailed descriptions of all the methods and equipment used for the study of microorganisms' heat

resistance are in the literature cited above and in major texts books (Pflug, 1990).

#### 2.5.3 Culturing media and recovery procedure

The isolation of Campylobacter and the methods used for its detection are well documented (Donnison, 2002, Line, 2001, Hsieh et al., 2018, Jacobs-Reitsma et al., 2019). Corry (1995) listed 40 liquid and solid selective types of media used for the isolation of Campylobacter. However, the detection and the recovery of heat-injured cells have not yet received much attention from any. Murphy (2005) concluded that there was a direct effect of the medium used on stress and survival of C. jejuni, and suggested that studies using different media may not be comparable. In that study, Murphy (2005) compared the growth of C. jejuni after acid stress in seven selected media and observed a remarkable difference in the growth of C. jejuni - up to a 5 log difference. Similarly, Jasson (2007) tried to describe a procedure for sub-lethal injury for Campylobacter stressed by different environmental factors (heat, cold, freeze, acid and oxidation) by comparing the number of the survival counts in selective (modified Cefoperazone Charcoal Deoxycholate Agar) and nonselective media (Columbia Blood Agar). The results showed that Campylobacter contributed to the high variation in test reproducibility and repeatability and that there were difficulties in standardization during handling the organism compared to other pathogens. Habib (2008) compared three selective media for *Campylobacter* (mCCDA, Karmali agar and CampyFood ID agar) and concluded that there were minor differences among the three selective media in terms of the reproducibility and repeatability. The mCCDA, which the International Organization for Standardization (ISO) has recommended has no major problem with the accuracy of the results obtained by using it. Hsieh et al. (2018) concluded that 37 °C is the optimal temperature for incubation and not 41.5 °C or 42 °C as reported in most other international studies (Jacobs-Reitsma et al., 2019). Others (Efimochkina et al., 2017) developed new media for detection and enumeration. Recently (Andritsos et al., 2020) compared the performance of four culture media used for enumeration and detection of Campylobacter species in chicken meat and concluded that the newly selected Campylobacter agar (CASA) developed by bioMériéux outperformed another recently developed agar by Oxiod BrillianceTM CampyCount agar (BCCA) and the ISO recommended mCCDA and Karmali agars.

#### 2.5.4 Effect of pH and presence of acids

The effect of pH is dependent on the type of *Campylobacter* strain used, suspension medium, water activity and the test temperature. Generally, the heat resistance is at its maximum at pH 7 and decreases as the pH shifts from neutrality (Varsaki et al., 2015). Humphrey and Lanning (1987) reported that the *D*-value was 11.5 min at 52 °C in a hot water tank at pH 7, and this was reduced

to 0.4 min and 1 min when the pH was adjusted to 4 and 10, respectively. They also reported that adjusting the scalding water tank pH to 9 significantly reduced the bacterial numbers in the scalding water but not the bacterial count on the carcass. Another study (Berrang et al., 2011) indicated that adjusting the scalding water tank pH to 9.89 was more effective to lessen the prevalence and the number of *Campylobacter* on broiler carcasses. Recently, another study (McCarthy et al., 2018) developed and validated a scalding model that accounted for both pH and temperature.

Another study (Coşansu and Ayhan, 2010) dipped chicken leg and breast in 1-3% solution of lactic acid and acetic acid (pH 2.2 - 2.3) and achieved a 0.3 to 2 log reduction in bacterial numbers. The combination of different lactic acid concentration with hot steam treatment on chicken breast samples was the objective of another study (Cil et al., 2019). This study concluded that the treatment with 4% lactic acid (pH 2.85) achieved more than 4 log reduction. However, 2% lactic acid with 15 seconds of treatment time by steam at 97 °C also achieved more than 4 log reduction.

#### 2.5.5 Effect of age (growth phase) of microorganisms

Usually, microorganisms in the stationary phase of growth are assumed to be the most resistant to heat stress (Kaur et al., 1998). However, a study by (Martinez-Rodriguez et al., 2004) reported that a strain isolated from the late stationary phase showed only a small increase in heat resistance over its parent. Kelly et al. (2001) showed that *Campylobacter* did not possess a greater heat resistance in the stationary phase and instead was more heat resistant in the exponential phase. However, these authors did not explain adequately why some cells were able to survive after a prolonged period of heat application of the cells in the stationary phase of growth although they reported that this phenomenon was due to the acquisition of resistance during prolonged periods of stress application.

#### 2.6 Heat resistance studies

Many reports have addressed the response of *Campylobacter* under thermal stress conditions by using broth or liquid media (Table 2.3). Some of these studies investigated changes in heat resistance combined with changes in pH or gas composition. A review study (Juffs and Deeth, 2007) has summarised different milk pasteurisation time and temperature combinations to determine D and z values in milk and to investigate whether pasteurisation could adequately destroy *Campylobacter* present in milk. They concluded that the organism was inactivated by both batch and HTST pasteurisation, with a wide margin of safety. Peptone water or saline has been used widely as the heating medium to determine D and z values. Poultry meat slurry was used in modelling the combined effect of heat and high pressure (Lori et al., 2007). *Campylobacter* survives better under

aerobic and ambient temperature in biofilms (Tang and Schraft, 2000, Joshua et al., 2006). However, Trachoo and Brooks (2005) claimed that Campylobacter in biofilms is not heat resistant. Bergsma et al. (2007) reported the highest D- value for C. jejuni in the fried chicken fillet, 1.95 min at a surface temperature between 109 to 127 ° C. de Jong et al. (2012) confirmed this and reported a D-value of 1.9 min at 100 °C in a boiled chicken fillet. Such extremely high D-values have not been confirmed by other studies. Gunsen et al. (2008) could not detect any Campylobacter colonies after cooking an artificially inoculated chicken drumstick for three to five min to reach a core temperature of 70° - 80 °C respectively in an oven at 200 °C. However, Sampers et al. (2010) found the 4.5 log cfu/g of C. jejuni inoculated to chicken burgers were reduced to below the detectable level (<10 cfu/g) after four min frying to an internal temperature of 57.5 °C and also 2.5 log cfu/g of naturally contaminated chicken burger reduced to below detectable level after four min frying to an internal temperature 52.1 °C. Moreover, C. jejuni was not detected after simulated home panfrying of artificially inoculated steaks or fillets, hamburgers and meat strips (Lahou et al., 2015). All meat samples of this study (Lahou et al., 2015) were from various animal species (pork, beef, chicken, lamb and some turkey, horse, and kangaroo), inoculated with 4 log cfu/g of Campylobacter jejuni. Only after a further enrichment step, C. jejuni was detected in few samples of pork hamburger, horse steak and crocodile steak. This is probably due to the matrix effect of the meat in these animals.

*C. jejuni* could not be detected when meatballs were artificially inoculated with 5 log cfu/g and cooked in a deep fryer at 180°C for one, two and three mins, grilled for four, eight and 12 min to reach a core temperature of at least 72.1°C, or cooked in an oven for five, 10 and 15 min at 200°C to reach a core temperature of 98.4 °C (Bostan, 2001). In some poultry processing plant studies, Yang (2001) reported a reduction up to 2 log while Guerin et al. (2010) reported a reduction of 1.3 to 2.9 cfu/g in a commercial scalding operation at 49 °C and 55.4 °C respectively.

Several studies examined the treatment of poultry carcasses by immersion or spraying with hot water or steam to reduce the contamination level. The reduction in the carcass contamination level was to a maximum of 3.8 log (Corry et al., 2007, Cox et al., 2005, Purnell et al., 2004, Zhang et al., 2013). However, these treatments cause a change in the appearance of carcasses, mostly a likelihood of the skin to shrink and become more fragile and also cause a slight colour change in any exposed muscle (Zhang et al., 2013).

The survival of *Campylobacter* following poultry scalding and chilling (Yang et al., 2001) was modelled with the Weibull distribution (Yang et al., 2002). Results of another study (Purnell et al., 2004) showed that immersing carcasses in the water at 65–70 °C did not reduce *Campylobacter* contamination level in naturally contaminated carcasses but by increasing the temperature to

>75°C reduced contamination but caused skin damage, discolouration and thus changed the external appearance.

A predictive model was developed for *Campylobacter* inactivation by heat and high pressure (Lori et al., 2007). Survival of *Campylobacter* was also studied in butter (Zhao et al., 2000). Based on the peak campylobacteriosis notification rate in early 2006, a NZ study investigated the cooking time and temperature required to eliminate naturally contaminated livers (Whyte et al., 2006). The results showed that the total *Campylobacter* elimination of 4 log cfu/liver could be achieved by the pan-frying method if the internal temperature was > 70 °C and held at this temperature for two to three min. If the effect of heat inactivation has to be estimated for a risk model, it is difficult to choose the appropriate *D*-value from the list in Table 2.3.

The development of a precise and accurate mathematical model(s) able to describe the inactivation behaviour of micro-organisms on the food subjected to stress factors (e.g. high temperature, pH, water activity) is crucial if the food industries are to design efficient and reliable preservation systems, to predict the shelf-life of the product or to use the model for quantitative microbial risk assessment (QMRA). QMRA estimates the magnitude of microbial exposure at different steps in the food production chain to assess the risk of foodborne illness (Cahill, 2005).

#### 2.7 The role of the consumer in food safety

Consumers play an important role in making certain the foods they prepare, handle and serve safe. Consumers can take steps to prevent cross-contamination, which is the transfer of bacteria directly or indirectly from a contaminated source to a non-contaminated product during food preparation which can potentially cause foodborne illness. Cross-contamination is also considered as the most critical food handling practice in the transmission of campylobacteriosis (Kapperud et al., 2003, Nauta et al., 2008, Kosa et al., 2015). *Campylobacter* is considered heat-sensitive, unable to grow in the presence of air, unable to multiply or grow < 30 °C (Jacobs-Reitsma et al., 2008).

Quantitative risk assessment studies, which will be described later, consider the development of campylobacteriosis to be due to direct exposure to *Campylobacter* spp. in animal feed and water. Many of these risk assessment studies consider cross-contamination during food preparation as the major pathway that causes campylobacteriosis (Brynestad et al., 2008, Hartnett et al., 2001, Lindqvist and Lindblad, 2008, Mylius et al., 2007, Nauta et al., 2007).

Many consumer reports associate the foodborne illnesses to the food consumed outside of the home, such as at retailers, restaurants, fast food outlets, bars and cafes (Redmond and Griffith, 2003). However, many studies have confirmed that up to 87% of the foodborne illnesses are related to food usually eaten at home due to malpractices in the domestic kitchens (van Asselt et al., 2009,

Humphrey et al., 2001, Redmond and Griffith, 2003, Hillers et al., 2003, Lake et al., 2007). It is now believed that improper food handling practices contribute to at least 40 – 60% of foodborne illnesses (Cogan et al., 2002, Humphrey et al., 2001). A NZ analysis of gastroenteritis disease outbreaks for the period of 1998 to 2000 revealed that 39.3% of the cases were acquired at home (Gilbert et al., 2007). Another New Zealand study (UMR Research, 2007) revealed that between 2003 and 2005, about 20% of the participants experienced foodborne illness. However, only 6% of those participants indicated that their illness was caused by food cooked at home. In 2007, 30% of NZ study participants reported foodborne illnesses.

There are review papers that discuss home food safety practices throughout the world (Patil et al., 2005, Redmond and Griffith, 2003). Some of these studies (Langiano et al., 2012, Frewer et al., 2016) addressed consumer beliefs about hazards, their knowledge about food safety and their self-reported practices. There are observational studies of consumer behaviour in a laboratory kitchen and in real environmental conditions. A few of the observational studies linked specific consumer behaviour to the microbial contamination of select artificially or naturally contaminated meals (Redmond et al., 2004). Models for cross-contamination have been designed (Kusumaningrum et al., 2004, Perez-Rodriguez et al., 2006, Rosenquest et al., 2003, Schaffner, 2004b, Mylius et al., 2007, van Asselt et al., 2008, Fravalo et al., 2009) and validated by quantitative observational studies (van Asselt et al., 2009).

It appears that cross-contamination contributes to the high notification rate in NZ. Understanding of the relationship between consumer knowledge, attitudes and food handling practices, and discussions of the discrepancy between the predicted consumer behaviour and the observed behaviour are key issues that need to be addressed in the campylobacteriosis epidemic in NZ. Moreover, investigation of the best hygiene measures that can be used is vital to provide public health, risk managers and regulatory authorities with the scientific support for health promotion plans that will discourage consumers' risky, unsafe food preparation practices.

#### 2.8 The role of the food regulatory body

Generally, the main role of any food regulatory body is to identify the most successful and realistic interventions and measures to eradicate or reduce the pathogen level at significant points in the food chain and ensure their removal.

#### 2.8.1 Consumer education

The aim of health professionals and risk managers is to encourage consumers to practice desirable and safe practices and discourage improper or unsafe ones. A study of consumer knowledge, attitude and behaviour can help in designing an effective health promotion initiative. Therefore, research and consumer education regarding the risk of food safety malpractices is an important element of precluding food-borne disease. Traditionally, the communication process is a transfer of information or education process with one way flow of objective scientific information from experts (government, academia, authorised organisation or industry) to the public considering the public as lacking knowledge and skills necessary to assess and judge the food risk appropriately (Hansen et al., 2003). Moreover, it is vital to investigate the best hygiene measures that can be used to provide public health professionals, risk managers and regulatory authorities with the scientific support for future health promotion plans that will discourage consumers from risky, unsafe food preparation practices. Laws and regulations cannot enforce controls and measures to prevent unsafe practices at home, as is the case for commercial food processors and retailers. The most logical way to change risky consumer practices is by educating consumers.

#### 2.8.2 Control of the poultry industry

A regulatory body is responsible for developing and setting targeted risk-based control measures throughout the food chain. The regulatory body has to audit and monitor in general the food safety system in place (food code for hygienic practice) for compliance, effectiveness and success. Moreover, the regulatory body has to identify and provide guidance about the most successful and practical interventions to reduce the risk of campylobacteriosis in any step in the food chain. A simplified poultry processing steps at a plant is shown in the appendix E2.

Generally, good hygienic practices alone during slaughtering and processing of broilers cannot totally eradicate the contamination level of *Campylobacter* from poultry carcases, and poultry meat as the good hygiene practices can only reduce the *Campylobacter* contamination level (Boysen and Rosenquist, 2009, Cui et al., 2005). There are many review papers detailing poultry processing interventions (Hendricks et al., 2000, Keener et al., 2004, Oyarzabal, 2005, Loretz et al., 2010, Lu et al., 2019). The aim of all interventions such as the chemical (organic acids, chlorine-based and phosphate-based treatments), physical (hot water, freezing, irradiation, and steam) and biological (bacteriophages and bacteriocins) is to achieve a further reduction of raw chicken contamination level.

Chemical intervention with chlorine- and phosphate-based treatment of broiler carcasses has been used in the USA for several years (Oyarzabal, 2005) and in European countries and other countries for decontamination of poultry but it is not as successful as required (Boysen and Rosenquist, 2009). The new regulation regarding the strict use of chemicals in EU (EC) No. 853/2004 implemented in 2006 permits the use of a chemical, only after the European Food Safety Authority

(EFSA) has provided a risk assessment about the possible impact of the intended chemical (Hugas and Tsigarida, 2008, EFSA, 2011b). Thus far, no such chemicals have actually been authorised by the EU with the exception of EU regulation EU2016/672 (EU, 2016) which allows the application of peroxyacetic acid to be used in poultry processing but subject to compliance with certain specifications and conditions. Therefore, only few poultry processors in Europe have used peroxyacetic acid due to the strict conditions and most European consumers have no appetite for chicken washed in chemicals (Siekkinen et al., 2012, Korzen et al., 2011, Boysen and Rosenquist, 2009). EU experts have debated against the use of chemicals because of the adverse health effects for consumers due to formation of toxic compounds during the processing, and they emphasised that the use of chemicals should not replace the good hygienic practice required to be implemented in all poultry processors and that there was no indication that antimicrobial resistance would develop (Hugas and Tsigarida, 2008, Koolman et al., 2014).

The physical methods to decontaminate carcasses during poultry processing have acquired new importance as a result of 1998 EU regulations and the ban of imports of all US poultry carcases which were treated by chemicals. Irradiation techniques have succeeded in achieving 3 log reduction of bacterial contamination level from the chicken carcass, but consumers do not readily accept the use of such technology and the European Commission discouraged the use of this technology in 1994 and enforced strict regulation about the labelling of irradiated food in 1999. European Commission (EC) allowed initially only a limited range of products such as aromatic herbal, seasonings and a few other products with a legal absorbed irradiation dose to be marketed in EC (Kume et al., 2009, Wilcock et al., 2004). However, 2011 EC legislation (EFSA, 2011a) stipulated that any irradiated food or food ingredient, even if it is present in trace amounts in a non-irradiated compound food, has to be labelled as "irradiated" or "treated with ionizing radiation" in order to allow consumers an informed choice. European consumers are reluctant to take any risk and generally have a negative public attitude toward food irradiation.

Ultrasound with a steam treatment (Boysen and Rosenquist, 2009, Hansen and Larsen, 2007) achieved more than 2.5 log reduction, but the adverse quality effect on the treated carcases such as skin splits between the breast and thigh, pitted appearance of the outside of the thighs and drums and skin shrinkage can hinder the commercialisation of this combination treatment (Biggs, 2009). Freezing to -20 °C for five weeks has been used in Norway to reduce *Campylobacter* contamination level (Sandberg et al., 2005). Generally, freezing can achieve 1.3 -2.2 log reduction in naturally contaminated chicken carcasses and it is also used in Ireland and Denmark (Boysen and Rosenquist, 2009, Georgsson et al., 2006, Hofshagen and Kruse, 2005). However, it is not a feasible option due to the consumers' satisfaction, especially for those with a higher income and their

demand for chilled fresh chicken and poultry products in the market. Steam under atmospheric pressure can reduce up to 3.8 log in artificially contaminated carcasses or chicken parts, but the boiled appearance of the treated products has prevented its use in the industry (James et al., 2007, Loretz et al., 2010, Whyte et al., 2003).

Many studies reported a 0.9 - 2.1 log reduction mainly in artificially contaminated carcasses or chicken parts by using hot water immersion or spray as a decontamination method (Corry et al., 2007, De Ledesma et al., 1996, Göksoy et al., 2001, Loretz et al., 2010, Purnell et al., 2004). James (2007) combined immersion in hot water and crust freezing and achieved 3.2 log reduction in artificially contaminated chicken carcasses. There are new physical promising interventions such as electrolyzed water, ozone water and cold plasma treatment which are summarised in the literature (Lu et al., 2019).

In NZ, there is no data published regarding any poultry physical intervention method except those related to freezing (-12°C and - 20°C) which can achieve a 3.3 - 3.5 log reduction in artificially contaminated chicken portions after six weeks (McIntyre, 2008, McIntyre, 2009).

#### 2.8.3 Risk assessment (RA)

Risk assessment is a scientifically based process of identifying and characterising the hazards and risk factors associated with a given food system. The risk posed by a particular hazard or process is calculated. Thus, risk assessment is the scientific evaluation of known or potential adverse health effects resulting from human exposure to foodborne hazards. This approach employs mathematical models that are used to explain: (i) the introduction of pathogens into food, (ii) the replication of pathogens in food over time, (iii) the destruction of pathogens by heat treatment or other techniques, (iv) consumption of pathogens in food, and (v) subsequent probability of illness. The core model is the risk pathway which starts at the source of the hazard and ends at the adverse health consequence. This pathway is called 'farm to fork' pathway where the hazard source is usually the farm "biological hazards reservoir" and end with the dose consumed by an individual or a number of the individuals becoming ill from the consumption of the biological hazard.

Quantitative microbial risk assessment (QMRA) uses the best measurements about microbe behaviour to identify where they can become a danger and estimate the risk. QMRA helps in assessing an intervention applied to the food chain steps by providing insight into microbial hazard dynamics which allows identification of targeted control strategies in order to reduce the risk of foodborne illness (Havelaar et al., 2007, Membré and Boué, 2018). QMRA describes the food chain step by a mathematical model where the change in model inputs will cause a change in the model output (Membré and Boué, 2018). Thus, if a new intervention is applied or a desired hypothetical

intervention is introduced to any step in the food chain, the expected outcome of that intervention can be examined and evaluated (Membré and Boué, 2018). QMRA is thus the scientific aspect of the risk analysis framework, which also comprises risk communication and risk management as defined by FAO/WHO (2007).

Risk management involves the development of policies and executing them to reduce the risk explained by the risk assessment study. The risk manager considers the social, economical and political factors during the evaluation of the risk and develops the policies to tackle the health impact on the community. Risk communication is all about the communication between the risk assessors, risk managers, and targeted stakeholders or individuals. The risk managers' (government officials) task is to use the risk assessment findings to establish a practical limit for the hazard if total elimination was impossible when developing a new policy or a standard according to the findings of the risk assessment (Tuominen, 2009).

Many countries have developed QMRA to comply with the sanitary and photo-sanitary agreement (SPS) of the World Trade Organisation. The agreement in 1994 allowed member countries to give priorities to food safety over international trade only if a sound scientific risk assessment has been conducted to justify their regulatory measures. The goal was to protect consumers and facilitate international trade in a transparent and reliable approach in spite of the limited resources of national governments as QMRA is a labour-intensive process with a data gap in many areas and it requires extra time and financial cost (Membré and Boué, 2018).

The Codex Alimentarius Commissions set standards for food in international trade and provide a framework for conducting risk assessment (Codex Alimentarius Commission, 1999) to harmonise and promote the methodology internationally. The first published QMRA was in Canada for *E.coli* 0157 in-ground beef hamburger (Cassin et al., 1998) and the first QMRA for *Campylobacter* was also in Canada (Fazil et al., 1999). QMRA methodology is used internationally by the World Health Organisation (WHO), Food and Agriculture Organisation (FAO), and The Food and Drug Administration (FDA), US Department of Agriculture, EFSA, Food Safety and Inspection Service (USDA-FSIS). Furthermore, the QMRA methodology is used nationally and regionally or on a smaller industrial scale for particular pathogens in a particular product or process. The use of QMRA in investigations of human health risks from poultry meat has increased in recent years and especially for *Campylobacter* as it causes more cases of foodborne illness each year than any other bacterial pathogen. Table 2.4 summarises several published QMRA.

Pathogen	Food	Country	Year	Source
E. coli 0157	Ground hamburger	Canada	1998	(Cassin et al., 1998)
Salmonella enteritidis	Shell and egg products	USA	1998	USDA/FSIS
Campylobacter jejuni	Chicken	Canada	1998	(Fazil et al., 1999)
Salmonella	Shell egg	USA	2000	(Whiting et al., 2000)
Campylobacter jejuni	Chicken	U.K	2001	(Hartnett et al., 2001)
Listeria	Ready to eat	International	2001	FAO/WHO
Salmonella	Pasteurised liquid eggs	USA	2001	FDA/USDA
Campylobacter spp (fluoroquinolone-	Poultry	USA	2001	FDA/Centre for Veterinary
resistant)	-			Medicine
Vibrio parahaemolyticus-us	Row molluscan shellfish	USA	2001	USDA/Centre for Food Safety
				Applied Nutrition
Vibrio spp	Seafood	USA	2002	FAO/WHO
Salmonella	Sheep meat	NZ	2002	(Sabirovic, 2002)
Salmonella spp.	Eggs and chickens	International	2002	FAO/WHO
Thermophilic Campylobacter species	Chicken	Denmark	2003	(Rosenquest et al., 2003)
Listeria	Ready to eat	USA	2003	FDA/USDA
Campylobacter jejuni	Chicken	France	2003	(Mégraud and Bultel, 2003)
Salmonella	Pork products	Italy	2004	(Giovannini et al., 2004)
Campylobacter	Broiler meat, other routes	Netherlands	2005	(Nauta et al., 2005)
<i>Campylobacter</i> spp	Poultry meat	Belgium	2006	(Uyttendaele et al., 2006)
<i>Campylobacter</i> spp	Poultry food chain	New Zealand	2007	(Lake et al., 2007)
Campylobacter	Poultry food chain	France	2008	(Albert et al., 2008)
Thermophilic Campylobacter spp	Chickens	Sweden	2008	(Lindqvist and Lindblad, 2008)
<i>Campylobacter</i> spp	Chicken	Germany	2008	(Brynestad et al., 2008)
Campylobacter jejuni	Chicken	Italy	2008	(Calistri and Giovannini, 2008)
<i>Campylobacter</i> spp	Broiler chickens	International	2009	FAO/WHO
<i>Campylobacter</i> spp	Broiler chickens	EU	2011	(EFSA, 2011b)

#### 2.9 QMRA approach description

The concentration of the pathogen is the variable the QMRA quantifies in order to follow the dynamics of the pathogen through the food chain. The final pathogen concentration is usually dependent on the variable describing the concentration at a previous stage in the food chain and other process parameters such as temperature, time, pH, and water activity.

Generally, QMRA separates the food chain into a specific number of modules which is subdivided into sub-modules. Each module or sub-module can describe a stage, process and part of the realworld problem. Those modules are expressed by functional relationships containing the process parameters and the bacterial concentration at a previous stage as an argument of the functions (Smid et al., 2010). The final model is calculated by linking the modules, transferring information from one module to the subsequent ones (Smid et al., 2010). In QMRA, the consumer phase model (CPM) is an essential link between previous phases in the food chain and the dose-response model. Several CPMs have been developed (Signorini et al., 2013, Lopez et al., 2015, Pouillot et al., 2012, Evers and Chardon, 2010) and reviewed widely in the literature (Chapman et al., 2016, Nauta and Christensen, 2011).

Generally, a QMRA model is defined as a schematic description of a system, theory or phenomenon that considers its known or deduced properties and may be used for further study of its characteristics (Tuominen, 2009). The main objective of the modelling process is to organise the data and relevant knowledge in a coherent framework. This can lead to a better ability to estimate the human risk and to use the model as a tool to assess the input variable changes and consequently assess the intervention which may lead to modify the output of the model and eventually reduce the risk (Lake et al., 2007). The model input variables or parameters can be changed during the execution, and the model equations can quantify the change in the observed variables in terms of other variables. The variables can be represented by one numerical value or point estimates. If all variables are expressed by one value, then the model is called a deterministic model (Vose, 2008). Usually, differential equations are used to express the rate of decay which depends on the initial count of bacteria and other variables and parameters. This can sometimes be solved to give an explicit algebraic formula for the number of surviving bacteria. However, the accuracy of the deterministic approach is questionable as it does not provide any information about uncertainty (the lack of knowledge of the system) or variability (heterogeneity of population).

Usually, uncertainty about the error in the parameters used in the model due to measurement, sampling, simplification or assumptions of realistic processes should be identified clearly to derive scientific results (Lindqvist et al., 2002, Neves et al., 2018). Uncertainty can be reduced by

increasing knowledge or improving the measurement, but variability cannot be reduced further as it is a natural variation of the system.

Stochastic models account for the uncertainty and variability by expressing the results or the parameters used in the models with probabilistic statistical distributions. Thus, stochastic models are preferred and widely used in QMRAs (Cahill, 2005, Vose, 2008, Neves et al., 2018). These distributions contain all possible values of the estimated parameter or the possible values of the expected results by using simulation techniques (for example Monte Carlo) which calculates model output multiple times with different initial values of input parameters. Each time a model calculates with a particular initial value called iteration. A single simulation consists of one to any number of iterations, to obtain a series of possible distributions considering the random variation in the input variable.

#### 2.9.1 Monte Carlo simulation

Monte Carlo (MC) simulation is the random sampling of each input of the probability distribution of a variable or a parameter in the model for each iteration of a simulation model (Vose, 2008). The main idea of MC method is to represent the solution of a mathematical problem by a parameter of a true or hypothetical distribution and to estimate the value of this parameter by sampling from this distribution reflecting the uncertainty and/or the variability of the parameter (Lewerenz, 2002, Hadjicharalambous et al., 2019). All probability distributions of variables, which are independent variables, are sampled per iteration to form the shape of the future distribution (Mataragas et al., 2010). The model outcome reflects the range of possible model inputs and evaluates each possible outcome by the probability of its occurrence based on the frequentist theory, which depends on the availability of the data. To obtain reliable results by the MC method may require many iterations to get accurate results which may be a time-consuming process. However, MC simulation of these models is relatively easy to implement using commercial software to automate the tasks, and they produce acceptable results (Robert and Casella, 2004, Huang et al., 2018). Another drawback of MC is that this technique performs poorly when modelling rare events (Vose, 2008). Other methods of calculating outcome distributions for QMRA exist and Bayesian belief networks have gained increased interest by researchers in recent years.

#### 2.9.2 Bayesian belief networks (BBN)

Bayesian Networks or Bayesian Nets technique is based on Bayes' theorem as described by Thomas Bayes in 1763 (Moore et al., 2009) as:

Prior Information + Evidence → Posterior Information (refined and updated information)

BBN is a method to express the relationship between causes and consequences represented by nodes and arcs in a directed acyclic graph. Each node describes an uncertain variable (cause/sequence) expressed by a probability distribution. Each arc or line describes the direction of the relationships (influential or casual) between nodes. Usually, the model is built from prior distributions consisting of the joint probability distributions of all the nodes derived from the prior distributions. Bayesian analysis, which is revising the assigned probabilities in the light of actual measured or observed data, is solved analytically by one of several software packages described by Uusitalo (2007). Sometimes, the complicated integrals of arbitrary probability distributions can be solved by combining with another Bayesian-based analysis method called Markov Chain Monte Carlo (sampling) which can cope with a larger class of posteriors than the Gaussian or any other fixed form distribution (Uusitalo, 2007).

BBN is a popular method for modelling in engineering, environmental management, image processing, decision analysis, gene sequencing, financial predictions, neural networks, robotics, geology, geography, forensic science, ecology, complex epidemiological model, text analysis, evaluation of scientific evidence and medical diagnoses (Smid et al., 2010, Uusitalo et al., 2005, Bromley et al., 2005, Yuen, 2010). More than 600 published papers on Bayesian method are listed on the Agenarisk web site (https://www.agenarisk.com). However, Bayesian approach has rarely been used in food science or in QMRA, and only a few studies have been published (Albert et al., 2008, Barker et al., 2002, Maijala et al., 2005, Ranta et al., 2005, Tuominen et al., 2006, Delignette-Muller et al., 2006, Mikkelä et al., 2016, Evers and Bouwknegt, 2016).

The Bayesian approach has many advantages such as being an appropriate method in handling missing data, accounting for the uncertainty in all estimated parameters, enabling data combination with the expert prior belief or knowledge. They allow back-propagation of prior variables with the observed data, assist understanding of underlying relationships between variables to provide a method to avoid overfitting of data and offer satisfactory prediction accuracy (Smid et al., 2010). Bayesian networks are not yet a preferred methodology in MRA research field due to the complexity involved in constructing and assigning probabilities from expert knowledge or prior information, linking the nodes, discretising the data, complex software and solving the resultant joint probability distributions. However, they are currently slowly gaining interest in food science and are likely to increase their relevance as one of the standard methods of QMRA. Therefore, the application of the Bayesian approach will be valuable in NZ as a NZ QMRA model (Lake et al., 2007) to use the MC simulation techniques which fail to predict the actual notified campylobacteriosis cases correctly. The NZ model prediction is about ten times more than the

actual notification value. The main drawback of the NZ adapted model is that it relies on the international literature data which are not reflected in the actual scenario in NZ poultry processing and does not consider the implemented interventions by the industry and the government. Moreover, the challenges faced in general in the probabilistic QMRA approach used in the NZ model have, to some extent, contributed to the current model inaccuracy (Havelaar et al., 2008). However, there are limitations to QMRA in general, since inadequate and inaccurate data can result in false predictions in any type of modelling. Moreover, QMRA model development can be resource-intensive and time-consuming and require an additional cost to fill the data gaps and update the model accurately with correct and appropriate data (Chung, 2004, Membré and Boué, 2018).

#### 2.10 An alternative approach for campylobacteriosis cases prediction

Predicting campylobacteriosis have become a matter of considerable concern in NZ. Moreover, the intervention strategies selected by policymakers to combat the high rate of the disease in the country can be evaluated by the cost and benefit in terms of campylobacteriosis cases prevented. Therefore, there is a need to develop a model to predict the number of campylobacteriosis cases in NZ accurately. Time-series forecasting methods have been successfully applied in the fields of engineering, science, sociology and economics (Brockwell and Davis, 2002). Other studies have shown that a time-series analysis is an appropriate methodology to clarify trends of campylobacteria, salmonella and listeria outbreaks to forecast future cases and to test the impact of interventions on the burden of food-borne diseases (Maertens de Noordhout et al., 2017, Weisent et al., 2010).

#### 2.10.1 Time-series

Finding the most appropriate time-series model for NZ data has many practical considerations. It is a complex task given the seasonality and the change of pattern in response to any implemented interventions. The classical time-series modelling approach is considered as an appropriate technique to predict campylobacteriosis in NZ. A reliable method for making time-series predictions would provide a great benefit to epidemiologists and public health officials (Altizer et al., 2006, De Greeff et al., 2009, Nobre et al., 2001).

#### 2.10.2 Time-series modelling techniques

#### **Decomposition methods**

These decompose the time-series into several components as any time-series may contain four

components; i) trend 'T': increase or decrease in a time-series over a long period of time, ii) cycle 'C': smooth fluctuations around the trend line, iii) season 'S': yearly repeated fluctuation (not necessarily an identically repeated fluctuation), iv) Error 'E': random variations which remain after the removal of all of the above components from the time-series. Estimates of those components can be used for forecasting (Makridakis et al., 1998).

The decomposition method assumes that the time-series can be expressed as a sum (additive model which is mainly used where the time-series exhibits constant seasonal variation) or product (a multiplicative model which is mainly used where the time-series exhibits increasing or decreasing seasonal variation) of up to three or 45 components only for a more practical decomposition by ignoring the cycle component. The decomposition method has been used successfully in predicting campylobacteriosis risk in the USA (Weisent et al., 2010).

#### **Regression-based methods**

These methods use a simple model for the trend, which represents the average level that changes over time. This is a linear trend model (straight line growth or decrease), but other possibilities are quadratic trend (curvilinear) and the pth-order polynomials model trend (with one or more reversals in curvature). Least squares point estimates of the parameters in these models are obtained by using standard regression techniques. These assume that the error term E represents random fluctuations that are independent and identically distributed. Variables are retained if they improve predictive value ( $R^2$ ), or indicate significance (p>0.05), i.e. with significant regression coefficients. This is a "global" model in the sense that it assumes the trend model and seasonal factors are the same throughout time, and all data points contribute equally to estimating the fixed parameters.

#### Holt's techniques

The exponential smoothing method without a trend or seasonal components which rely on simple updating equations to predict or forecast can be generalised to handle a time-series with a trend and season components (Chatfield, 1978). These techniques give a "local" model in which the components are constantly changing over time, and data points are given more weight in the estimation when they are close to the time of estimation. The Holt Winter method is a simple method to implement and is widely used in practice (Chatfield, 2004).

#### Moving average (MA)

In a time-series with a MA process, the model describes how each observation is a function of the previous errors. In the MA, errors are the average of this period's and the last period's random errors. Thus, the observed series is a weighted moving average of the unobserved series (Armstrong, 2001).

Autoregressive(AR): A simple way to model dependence on past observations is to use ideas from

regression. AR model expresses the current time-series observation as a linear function with its past observed values. It is also called a Markov process (Chatfield, 2004).

#### Box and Jenkins model

This involves identifying an appropriate model to fit the data and then using the fitted model for forecasting. ARMA models, combining MA and AR components described by Box and Jenkens (1976) have been specifically developed for the analysis of time-series, and the advantages of these models are well documented (Benschop et al., 2008, Reichert et al., 2004). However, the drawback of the above ARMA model is that it is valid only for stationary time-series, i.e. when there is no systematic change in mean (a constant mean and constant variance in time with no trend). Thus, to use the ARMA model, it is necessary to remove the non-stationary source of variation such as trend and seasonal variation to model the residuals of the time-series as those are more suitable for modelling as a stochastic process. In some cases, the trend and the seasonality of the time-series may have more interest than the residuals of the time-series. To transform into a stationary-series, the observations usually first need to be filtered (removing of trend and seasonality) usually by subtracting consecutive observations from each other to achieve stationarity. The resulting model is capable of relating certain types of non-stationary series and is known as an integrated model or ARIMA as the stationary model that is fitted to the differenced data summed or integrated to provide a model for the original non-stationary data (Chatfield, 2004).

Thus, each ARIMA process has three parts: AR, the integrated (I), and MA parts. The models are written as ARIMA(p, d, q) where p describes the AR part, d the integrated part and q the MA part. Series with seasonal components also need equivalent terms at seasonal lags (S), leading to the full seasonal ARIMA model written as ARIMA(p, d, q)× (P, D, Q)S. A multiplicative form of the ARIMA can be obtained by making a logarithmic transformation of the data, fitting an ARIMA to the transformed data, and then back-transforming the predictions.

#### Intervention models

These models are mainly used when exceptional, sudden, external events affect the variable to be predicted. Possible examples of those events are worker strikes, disasters, and policy changes, to name a few (Coshall, 2005). However, the ARIMA with intervention is considered an advanced method which requires further knowledge of statistics and the use of additional complex software. New Zealand has applied interventions in 2007 and in 2008 to reduce the rate of campylobacteriosis. Thus, those models were considered to accommodate interventions by the use of an intervention variable which is used to build the intervention model. Such variables are called step functions, impulse functions or response functions. There are different types of response functions. Historically, an improvement in forecasting has been noticed by many authors after

applying the ARIMA intervention model (Goh and Law, 2002, Min et al., 2010, Rashed et al., 2017). These studies have shown that ARIMA intervention models had the lowest forecast error.

#### 2.11 Scope of this work

Due to the increase in notified cases of campylobacteriosis in NZ, especially when compared with other developed countries, it is vital to investigate and to understand the reasons and the risk factors for this increase in order to develop intervention strategies to protect consumers and lower the rate of human illness associated with *Campylobacter*. The strategies are intended to identify the preventive steps within the food production chain. There are many hypotheses for this increase, and more than one have contributed to this staggering historical worldwide notification rate record, as outlined below:

- > NZ has more heat resistant strains to survive normal cooking.
- ➢ NZ has more oxygen tolerant strains.
- NZ poultry processing interventions are less effective than those used elsewhere, or the secondary processing practices may increase the contamination level.
- New Zealanders have the worst home hygiene practices than the citizens of other developed countries.

Campylobacteriosis is an internationally recognised complex problem, and it is important to investigate all the above hypotheses to solve New Zealand's public health, economical and historical crisis. The thermal treatment destroys *Campylobacter* and is one of the primary interventions used to ensure the safety of foods. Therefore, the survival and heat-resistance of *Campylobacter* under different conditions need to be investigated.

D values are not always determined under the same conditions as are encountered in the actual situation or scenario. Moreover, the physiological state of the bacteria and the specific contaminating strain present is often unknown. It is then unclear which D- value to use from the large datasets available and whether to include factors that may influence heat resistance. When there is an effect reported for a factor (e.g. food product), this has usually been tested in only one laboratory for a few variable conditions and a few strains. It is unclear whether such effects are relevant overall in comparison with other conditions. Therefore, the determination of the D and z-values for NZ strains is a high priority research task. Under-cooking in food manufacturing facilities is not currently causing widespread food safety problems (Lambertz et al., 2012). The multistate European survey has confirmed that manufacturing cooking is not a food safety concern (EFSA, 2016). Most of the surveys of cooked poultry products showed only negligible prevalence and concentration or no contamination (Egan, 2006). However, continued development of new

products and processes, the possible emergence of new heat resistant strains and the significantly higher reported *D*- value values for *Campylobacter* by Bergsma (2007) and de Jong (2012) require investigation. Moreover, the on-going regulatory changes regarding food safety codes and NZ *Campylobacter* strains which genetically differ from international strains (McKenzie, 2006) necessitate a practical stance in ensuring appropriate evaluation of thermal process lethality and in updating the current knowledge of kinetic parameters of the different strains. It is important to focus future research on the most prevalent poultry, human, and animal strains to identify specific risk factors that may be linked to greater heat resistance strains with the usage of all the best methods, techniques, and instrumentation. This will lead to a better understanding of the survival of *Campylobacter* in the food chain and provide improved targeted information on health and food risks.

Further research is needed to develop growth, survival and death models specifically for *Campylobacter* either on or in food under dynamic temperatures. Most studies conducted for *Campylobacter* until now were not designed to generate sufficient data to develop such models (Grigoriadis et al., 1997, Svedhem et al., 1981, Bergsma et al., 2007, de Jong et al., 2012, Sampers et al., 2010). Rather, most of the models developed were static models, which are applicable only for static or invariant environmental conditions. In addition, most of these models were developed on broth and not on the solid foods with which *Campylobacter* is mostly associated with.

To properly assess the risks and risk minimisation strategies require: i) accurate knowledge of the survival and heat resistance during processing, storage and handling of the NZ strains that are most frequently associated with poultry products and humans, ii) evaluation of the processing practices and the interventions applied in NZ poultry processing plants, iii) investigation of the cross-contamination route in the NZ context as a major risk factor for the high rate of campylobacteriosis, iv) provision of insight into consumers' knowledge, attitude and behaviour which can help in designing an effective health promotion initiative, v) advising decision-makers on the current state of knowledge for designing an effective health promotion strategy by a structured approach to identify, assess, review and summarize data on the most relevant scientific research, as consumer information intervention may lead to prevention of campylobacteriosis by increasing consumer protection.

Cumulatively, this PhD research was planned to solve the mystery of a very complex problem, the high rate of NZ campylobacteriosis and to provide the means for control and prevention by finding the optimal industry and consumer interventions to combat the disease. Moreover, this study is intended to develop a model to predict accurately the number of campylobacteriosis cases in NZ as the control strategies selected can be evaluated by the cost and benefit in terms of

campylobacteriosis cases prevented. Thus, this study was tailored to serve as a scientific response to address a major public health problem in NZ and to provide a fast response and a guide in analysis of the causes of the problem and recommend relevant solutions with scientific evidence in an acceptable time frame.

#### 2.11.1 The structure of this thesis

Chapter one includes the introduction. Chapter two contains the literature review. Chapter three describes the materials and methods used in the microbiological experimental chapters (a total of six chapters of the thesis). Chapters four, five and six report on the investigations of heat resistance characteristics of most relevant NZ C. jejuni isolates from human cases and in poultry. Chapter four shows data on the heat resistance of those isolates in broth medium. Chapter five reports the heat resistance studies on the chicken skin. Chapter six tests the heat resistance under dynamic conditions, simulating processing scenarios such as cooking. Chapter seven evaluates the survival of C. jejuni isolates under oxidative stress conducted in various ways and under different temperatures simulating storage and handling temperatures. Chapter eight identifies the effect of secondary poultry processing practices at the plant level, such as needle injection, on final contamination level and prevalence in poultry carcasses and poultry products sold in NZ. Chapter nine explores a safe alternative physical decontamination intervention that could be applied in poultry plants to replace the unsafe chemical decontamination intervention currently applied in NZ, but it is banned in the EU. Chapter 10 reports on the quantitative risk assessment for campylobacteriosis in NZ by the Bayesian approach. Chapter 11 assesses the time-series models for predicting campylobacteriosis risk in NZ. Chapter 12 addresses the cross-contamination route in the transmission of *Campylobacter* spp and highlights the most important unsafe consumer practices. Chapter 13 shows the data on the consumers survey about their poultry handling practice at home and their basic food safety knowledge. Chapter 14 discusses the thesis findings and presents conclusions and future research.

## 3 Microbiological experiments materials & methods

In this chapter, materials and methods used in all microbiological experiments (in chapters four to nine) are described.

### 3.1 Campylobacter isolates

All the *Campylobacter* multilocus sequence typed (MLST) isolates used in this study were provided by the Hopkirk Molecular Epidemiology Laboratory in Palmerston North. The MLST method provides a discriminatory molecular profile, is reproducible and easy to interpret (Maiden, 2006). Use of this method has resulted in the recognition of major genetic lineages or clonal complexes in *C. jejuni* populations from human infections and animal and environmental sources (Dingle et al., 2002).

Human faeces submitted to MedLab Central, Palmerston North that were positive for *Campylobacter* by ELISA (ProSpecT R, Remel, USA) were sent to the Hopkirk Molecular Epidemiology Laboratory over a three year period from 1st March 2005 to 29th February 2008 (Mullner, 2009). Human faecal swabs were cultured on modified cefoperazone charcoal deoxycholate agar (mCCDA) plates (Fort Richard, Auckland) and in Bolton broth (Lab M, Bury, England) and incubated at 42 °C in a microaerophilic atmosphere (85% N<sub>2</sub>, 10% CO<sub>2</sub>, 5% O<sub>2</sub>) for two days. A single colony resembling *Campylobacter* was subcultured in Blood Agar (BA) (Fort Richard, Auckland) and incubated micro-aerophilically at 42 °C for two days before DNA preparation for MLST. Cultures were frozen at -80 °C in glycerol broth (Difco, USA).

Over the same period (1st March 2005 to 29th February 2008), 12-18 fresh whole poultry carcasses from different poultry suppliers in the region were sampled each month from retail outlets in Palmerston North, with the number of samples collected per supplier reflecting market share. Chickens samples were analysed for campylobacter. Single colonies of *Campylobacter* species were subcultured to BA and incubated micro-aerophilically at 42 °C for two days before DNA preparation for MLST. Bacterial cultures were frozen at -80 °C (Mullner, 2009). A total of 62 different MLSTs were detected in human and poultry samples. The major isolates frequently detected in human and poultry samples according to Mullner et al. (2009), namely ST - 474, ST -48, ST - 45 and ST 190 were used in this study.

### 3.2 Campylobacter culture growth and enumeration

*Campylobacter* strain cultures were stored at -80 °C on beads (Microbank, Pro-Lab Canada). They were sub-cultured onto CA and incubated at 41.5 °C for 48 h in a microaerobic atmosphere within

a specialized workstation (Forma System model 1024, USA) supplied with continuous optimal atmospheric conditions conducive for *Campylobacter* growth, 5 % O<sub>2</sub>, 10 % CO<sub>2</sub> and 85 % N<sub>2</sub>. This atmosphere was used in all subsequent incubations in this study. Colonies grown on CA were subcultured on to another CA plate and incubated for 48 h in a micro-aerobic atmosphere at 41.5 °C to obtain a plate with prolific growth for inoculum preparation. Each inoculum for testing heat resistance was prepared by transferring all colonies from a CA plate into Brain Heart Infusion broth (BHI). The final culture concentrations used for inoculation were in the range of 10<sup>7</sup> to 10<sup>9</sup> cfu ml<sup>-1</sup> or per portion/drumstick. The determination of inoculum concentration was conducted by a serial dilution of the inoculum in BHI followed by plating in triplicate on mCCDA by the Surface Plate Method (Downes and Ito, 2001). Plates were incubated micro-aerobically at 41.5 °C for 48 h as described above.

### 3.3 Thermal inactivation of *C. jejuni* in broth

The typed isolates of *C. jejuni* examined were ST-474, ST-190 and ST-45 isolated from both poultry and humans. These three strains were implicated in more than 42 % of campylobacteriosis cases in a Manawatu study (French, 2008a). An isolate from a wild bird for strain 45 was also tested at two different temperatures (53.5 and 56 °C) in this study.

#### 3.3.1 C. jejuni heat inactivation procedure

The bacterial suspensions obtained from above were heated to a predetermined temperature using a submerged coil heating apparatus (Protrol instruments Ltd., U.K). This comprised of a stainless-steel coil fully submerged in a thermostatically-controlled water bath which allows microbial suspensions to be heated within a short time to achieve a rapid temperature equilibration. Samples were taken at predetermined intervals (from 2 s -200 s with a total heating time range from 12 s – 1050 s, depending on the investigated temperature and the isolate heat resistance) automatically and collected in a series of vials in an automated carousel (Cole and Jones, 1990). The inoculum population and the number of surviving organisms in suspension after heat treatment were determined by dilution in BHI followed by plating on mCCDA using an automated spiral plater (Don Whitley, Yorkshire, United Kingdom) and triplicate plates used at each dilution. The inoculated plates were incubated micro-aerobically at 41.5 °C for 48 h.

### 3.4 Thermal inactivation of *C. jejuni* on chicken skin

The typed isolates of *C. jejuni* examined were ST-474 and ST-48 isolated from both poultry and humans.

#### 3.4.1 Chicken skin sample preparation and inoculation

The chicken skin was purchased from a local butchery. Discs of chicken skin (18 mm diameter and 4 mm thick) were prepared using a stainless steel circular cutting tool. The prepared chicken skin discs were sent for gamma irradiation (25 -32 kGy) at the irradiation plant (Schering-Plough Animal Health, Wellington) to ensure that *Campylobacter* no longer exists in the tested sample before the artificial inoculation. Skin samples were stored at 4 °C until use. Each irradiated chicken disc was equilibrated to room temperature and placed aseptically in an aluminium test cell (Appendix B, Figure B1) specially designed for kinetic studies (Chung et al., 2008). An inoculum of 20  $\mu$ l was pipetted to the chicken skin centre in the lower part of the aluminium cell. After inoculation, the cell was left for 60 min to allow *C. jejuni* attachment to chicken skin to mimic natural contamination.

#### 3.4.2 *C. jejuni* heat inactivation procedure.

Eight aluminium cells were placed in a rack submerged in a temperature-controlled water bath (Grant, UK) maintained at a 56.5 to 65 °C temperature range for a predetermined time period (Appendix B, Figure B2). One aluminium cell was removed at each prescribed time interval and then transferred immediately to an ice bath for C. jejuni survivor analysis. All aluminium cells were equipped with a wire holder for ease and speed of handling during the heating period in the water bath. The time it took for the chicken skin sample to reach the equilibration temperature ('come up time') was measured by a very fast response thermocouple sensor with one second response time (Labfacility, UK). The temperature was recorded with a Measurement Advantage USB-TEMP logger (Measurement Computing, USA). A graph of the temperature profile is shown in Appendix B (Figures B3-B6). The 'come-up time was 55 s from a number of experiments for both the investigated temperatures 56.5 °C and 60 °C. There was a significant amount of variation caused by i) the chicken skin moisture and fat content both of which affect the thermal properties of the chicken skin, and ii) shrinkage of the chicken skin during heating inside the cell which may have created an air gap and slowed down the heat transfer from the aluminium cell walls to the chicken skin (because the air can form an insulation layer). All data points used in the calculation of the kinetic parameters were considered to estimate the come-up time.

#### 3.4.3 Enumeration of *C. jejuni* survivors

Each chicken sample was transferred from the aluminium cell to a vial containing 4 ml BHI and the aluminium cell was flushed out with a further 5 ml BHI to remove any survivors from the internal surface of the cell. The sample was mixed using a vortex mixer and serially diluted in BHI. All the samples remained in BHI for at least two hours to allow recovery of injured cells, followed by plating in triplicate on mCCDA by the Surface Plate Method, a 200 µl spread by L shaped sterilised spreader (Downes and Ito, 2001). Plates were incubated microaerobically at 41.5 °C for 48 hours as described above. The average of the logarithm (base 10) of three plate counts at each sampling time was used for modelling and analysis.

# 3.5 Thermal inactivation of *C. jejuni* on chicken skin under dynamic conditions

The typed isolates of *C. jejuni* examined were ST-474 and ST-48 isolated from both poultry and humans.

#### 3.5.1 Meat sample preparation and inoculation

Same as detailed above in section 3.4.1

#### 3.5.2 Heat inactivation procedure

For each run, 16 aluminium cells containing inoculated chicken skin samples were placed in a programmable water bath (developed locally at the university, described in Appendix C, (Figure C1). The water temperature was programmed to increase linearly to reach 65 °C for a period of either 8 or 18 min simulating the scenarios that can occur in retail food service, industry and in domestic cooking (Persson et al., 2002, van Asselt et al., 2009, NACMCF, 2007). Duplicate samples (two aluminium cells) were withdrawn at different predetermined intervals and transferred immediately to an ice bath until analysed for the counts of surviving culturable C. jejuni. All aluminium cells were equipped with a wire holder for ease and speed of handling during the heating inside the water bath. The come-up time for the chicken skin sample within the cell to equilibrate with the external medium was short for both heating regimes (8 and 18 min). The difference at any given time between the temperature of the skin and the bath temperature did not exceed 0.25 °C for 18 min heating regime or 1 °C for the eight min heating regime. For most of the heating interval of the eight min heating regime, the difference was between 0.3 and 0.7 °C. These values were measured using a very fast response thermocouple sensor (1s response time; Labfacility, UK) and the temperature was recorded by a Measurement Advantage USB-TEMP logger (Measurement Computing, USA). The temperature profiles for both regimes were embedded in the individual survival curves and are shown in the results section.

#### 3.5.3 Enumeration of *C.jejuni* survivors

As described in 3.3.2, the average of the log<sub>10</sub> counts for four plates at each sampling time was used

for modelling in contrast to the three plates used in section 3.3.2. This was to improve the accuracy of the results of this experiment because less number of experiments are required to conduct under the dynamic conditions.

#### 3.6 Survival of C. jejuni under oxidative stress at different temperatures

The multilocus sequence typed strains ST-474, ST-48, ST-190 were used in this study.

#### 3.6.1 Exposure of *C. jejuni* to oxygen in broth

Inoculums of 3 ml were added to 27 ml BHI which was then dispensed to form a shallow layer of 30 ml in a 125 ml conical flask equipped with a special bung (Bug stopper, Whatman) that allows air exchange with the surrounding atmosphere but prevents microbial contamination. This flask was allowed to equilibrate at the appropriate temperature and atmosphere (aerobically or microaerobically) by gentle shaking using an electrical shaker (Lab-line Junior Orbit Shaker, U.K). The number of surviving organisms in samples withdrawn from each flask at predetermined intervals after exposure to the controlled atmosphere conditions at each temperature (4, 10, and 20 °C) was determined by dilution in BHI followed by plating on mCCDA. Duplicate plates were made at each dilution. The plates were incubated micro-aerobically within a specialised workstation (Thermo scientific, USA) continuously supplied with 5 % O<sub>2</sub>, 10 % CO<sub>2</sub> and 85 % N<sub>2</sub> at 41.5 °C for 48 h.

#### 3.6.2 Exposure of *C. jejuni* to oxygen on agar plates

50 µl of the culture suspension in BHI was spread on CA plates (blood-free 16 to 40 plates for each isolates at the specified incubation temperature based on a previous primary trial data collected before the actual experiment) using a spiroplater (Don Whitley, Yorkshire, U.K) and incubated aerobically at 4 °C and 25 °C for one to five weeks. The CA control plates containing the culture spread were simultaneously incubated under microaerobic conditions at 4 °C and 25 °C using anaerobic plastic jars with a microaerobic atmosphere generating system (Pack MicroAero, Mitsubishi Gas Chemical Co. Inc). Two plates from each aerobic condition were sampled every day and incubated at 41.5 °C for 48 h in a microaerobic atmosphere within a specialised workstation (MACS VA 500, Don Whitley Scientific U.K).

#### 3.6.3 Exposure to $H_2O_2$

Each culture suspension prepared in BHI was spread by a swab on Muller Hinton Agar plates and allowed to grow under microaerobic conditions for 48 h. Subsequently, filter discs (6 mm) inoculated with 10 µl of 1 mM, 10 mM, 100 mM, or 1 M hydrogen peroxide were placed onto the

plates (Disc Diffusion Method) (Fields and Thompson, 2008). These plates were then incubated at 41.5 °C under microaerobic conditions for 48h.

# 3.7 Evaluation of processing practices contributing to *Campylobacter* contamination in New Zealand chicken and chicken products

One MLST isolate was used in this study, namely ST-474, which was isolated from poultry.

#### 3.7.1 Procedure

Two brine solutions were tested. One, solution A, is usually used for frozen products and contained 5% w/v marination powder mix (salt (0.5-1%), mineral salt (451), hydrolysed vegetable protein (soy), thickeners (415, 412), vegetable oil (canola, antioxidant 319), anti-caking agent (551) and maltodextrin). The other brine solution (B; 'Tenderpaste' marinade) is usually used for fresh products and contain 15% powder mix (salt (6-7.5%), mineral salts (450, 451, 452), thickeners (412, 415) and sugar. Two 50 ml portions of each brine solution were prepared according to the manufacturer's instructions and placed in sterile 125 ml flasks and inoculated with *C. jejuni* at 10<sup>6</sup> cfu ml<sup>-1</sup>and the flasks held at 7 °C. A 1 ml sample of each brine solution was withdrawn for enumeration of *C. jejuni* at the start of incubation, and another 1 ml was withdrawn at predetermined intervals from Solution A (0.5 -1 % salt) and for Solution B (6-7.5% salt), and again after 24 h for both brines.

Serial 10-fold dilutions of each sample were also prepared in BHI by transferring a one ml sample to nine ml of sterile BHI. The number of surviving organisms in the sample suspensions was determined by plating on mCCDA as described above. Duplicate plates were made at each dilution. The plates were incubated micro-aerobically within a specialized work station as described above. The number of *Campylobacter* is expressed as log count (mean + SE) for the two duplicates counts for all experiments.

#### 3.7.2 Survival of *C. jejuni* in injected chicken products at the plant

Fresh chicken drumsticks were purchased from the local supermarket and were sealed in a vacuum packaging bag and irradiated with a 25 kGy dose at Schering –Plough Animal Health plant located in Wellington (NZ) and stored at 4 °C until use. The irradiated chicken drumsticks were injected with brine during normal processing operations at a commercial plant processing. The needle injection machine conveyor carries the drumsticks to the injection chamber, and the brine is injected through needles. The brine was prepared in a mixing tank and was pumped through a permanent line to another feeding tank and from there circulated continuously through a

refrigeration unit and filtration unit to remove chicken meat particles. The brine was maintained at 6-8°C. Ten irradiated and injected chicken drumsticks were processed during the production run (at four hours) and another ten irradiated and injected drumsticks were processed at the end of the production run (eight hours) for the two brine types commonly used in the processing plant (solutions A and B).

#### 3.7.3 Recovery of *C. jejuni* from surfaces of the injected drumsticks

Five irradiated injected chicken drumsticks from each brine concentration at the end of the run and another five drumsticks from the middle of the run from each brine solution (giving a total of 20 drumsticks) were examined by removing 10 g of skin aseptically, using scalpels with sterilised disposable stainless-steel blades. The skin sample was placed in a stomacher bag with a filter containing 90 ml *Campylobacter* Enrichment Broth (CEB) without blood (Lab m135) and pummelled for two min by Colworth Stomacher model 400 (A.J. Seward & Co. Ltd, London), then 1 ml of stomacher fluid was pipetted from inside the filter sleeve for plating. 200  $\mu$ l of the same liquid was separately plated on duplicate mCCDA plates. Serial dilutions were performed for the stomacher fluid by transferring one ml fluid to nine ml CEB without blood and plated as described above. The final results were also expressed as cfu per portion (~10g) as the whole skin surface was used in each sample. Confirmation that the colonies were *C. jejuni* was achieved by subculturing to CA and incubating at 37 °C aerobically. Additionally, the hippurate test was performed later.

#### 3.7.4 Recovery of *Campylobacter* from the internal muscles of the injected drumsticks

Five irradiated injected chicken drumsticks from each brine concentration collected at the middle of the run and another five at the end of the run for each brine solution (a total of 20 drumsticks) were dipped in boiling water for 15 s. The initial studies confirmed a reliable recovery of internal injected inoculum after the 15 s treatment with boiling water and those studies also confirmed that the irradiated chicken drumsticks were free from *Campylobacter* from the surface and from the internal tissues. Using a special tool, the injected chicken drumsticks were removed in batches of three from the water and placed individually in a new sterilised bag to cool to room temperature. The skin and about a 3-4 mm layer from the surface were aseptically removed with a scalpel with sterilised disposable stainless steel blades and sterilized forceps. Strips of 2 mm thick were cut from the deep internal muscles using the scalpel and the sterilized forceps and transferred to a weighed stomacher bag with a filter, which contained 90 ml of CEB without blood, to obtain a 10 g sample and the cells were enumerated as described above. The final results were expressed as cfu per portion for consistency with the surface samples described above in spite of the fact the 10 g sample did not represent the whole internal tissue of the drumstick. We assumed this to represent

the whole internal tissue of that drumstick layer.

# 3.7.5 Survival of *C. jejuni* in injected chicken carcasses and chicken products in the retail market

A total of 20 injected drumsticks (of the same drumstick type as those injected in the plant) were randomly selected between the periods of mid-June to mid-July 2009 from different local retail shops. All products were analysed for internal and external contamination as described above. In addition, ten fresh whole chickens injected with solution B were analysed for *Campylobacter* presence by rinsing for two min with 400 ml CEB without blood and proceed by plating as described above in section 3.7.3.

# 3.8 Evaluation of *Campylobacter* mitigation strategies by physical methods in a poultry processing plant

One MLST isolate was used in this study, namely poultry isolate ST474,

#### 3.8.1 Sample preparation and treatment

The rough surface of chicken skin facilitates attachment and survival of C. jejuni during multiple stages of processing. Therefore, we used chicken drumstick with skin as a model to represent the whole carcass for testing the efficacy of the hot water dip that is used against C. jejuni. Fresh chicken drumsticks were purchased from the local supermarket and were stored at 4 °C until use. Each drumstick was allowed to condition at room temperature for at least ten h before a 200 ul of inoculum was introduced on the skin surface and was left for 60 min to allow C. jejuni to allow for chicken skin attachment to mimic the natural contamination scenario. Chicken drumsticks were held by tongs and submerged in a temperature-controlled water bath (GD 100; Grant, UK) maintained at various temperatures in the range 56.5 to 70 °C for a predetermined time interval. The chicken drumsticks were removed at each prescribed time interval and then transferred immediately to an ice bath until analysis for C. jejuni survivors. The following treatment temperatures, based on the D and z value obtained in previous sections and the literature, 70, 65, 63, 60, 56.5 °C were applied. A combination of 11 different times and temperatures were used to investigate the best time and temperature combination to achieve a high Campylobacter log reduction with the minimum adverse effects on the visual appearance of the treated chicken skin or chicken drumstick.

The time it took for the chicken skin sample to reach the equilibration temperature ('come up time') was measured by a very fast response thermocouple sensor with one-second response time

(Labfacility, UK). The temperature profile was recorded with a Measurement Advantage USB-TEMP logger (Measurement Computing, USA). A graph of the temperature profile is shown in Figure 3.1. Ten drumsticks were used for each treatment, and five for the control group, which was only immersed in a different water bath GD 100 (Grant, UK) maintained at room temperature (20°C) for a similar time interval to the heated drumsticks in order to account for the wash effect (Smith, 1992). Two repeated trials were carried out for only one time-temperature combination.



Figure 3.1 Temperature profile of water and chicken during heating the water bath to 60°C.

#### 3.8.2 Enumeration of *C. jejuni* survival.

The treated drumsticks and control drumsticks were examined by removing the whole of the skin aseptically using scalpels with sterilised disposable stainless steel blades. The skin sample was placed in a stomacher bag with a filter containing 90 ml *Campylobacter* Enrichment Broth (CEB) without blood (Lab m135) and pummelled for two minutes using the Colworth Stomacher model 400 (A.J. Seward & Co. Ltd., London). Then 1 ml of stomacher fluid was pipetted from inside the filter sleeve for plating, and 200  $\mu$ l of the same liquid was separately plated on triplicate mCCDA plates. Serial dilutions were performed for the stomacher fluid by transferring one ml fluid to nine ml CEB without blood and plated as described above. Confirmation that the colonies were *C. jejuni* was achieved by subculturing to CA and incubating at 37 °C aerobically. Additionally, the Hippurate test for *Campylobacter* identification was also performed.

# 4 Thermal inactivation of Campylobacter jejuni in broth

#### 4.1 Introduction

*Campylobacter* causes more cases of food-borne gastroenteritis each year than any other bacterial pathogen worldwide (Jonsson et al., 2010, Samuel et al., 2004a, Schönberg-Norio et al., 2004, Rushton et al., 2019). The reason for this high incidence is still not established, and many aspects of the epidemiology of the infection are not clear. However, several international epidemiological studies have established the link between the consumption of poultry meat and campylobacteriosis (Eberhart-Phillips et al., 1997, Ikram et al., 1994, Kapperud et al., 2003, Michaud et al., 2004, Skarp et al., 2016, Neimann et al., 2003) and the high level of contamination of poultry meat in NZ (~80 %) with *Campylobacter* supports this assumption (French, 2008a).

It is important that processors, caterers and consumers apply heat treatments that are effective at destroying all food pathogens. In general, the heat treatments required to eliminate *Salmonella* will also destroy *Campylobacter* (NACMCF, 2007). However, there remains a possibility that undercooking may occur or a proportion of cells in some protected areas may survive the normal heating process leading to ingestion of an infectious dose. Moreover, the remarkable high D value for *C. jejuni* reported (D<sub>100°C</sub> value of about two min, (de Jong et al., 2012)) may indicate the emergence of heat resistant strains. Therefore, it is appropriate that the current knowledge of thermal inactivation of NZ *Campylobacter* strains is updated and expanded.

The objective of this study was to determine the thermal inactivation kinetic parameters (D and z) for NZ-relevant strains of C. *jejuni* isolated from both humans and poultry. The effects of different time-temperature combinations on survival and variations in survival within the same typed strain (isolated from different sources) and between different typed strains were evaluated.

#### 4.2 Materials and methods

#### 4.2.1 Campylobacter culture growth and enumeration

*Campylobacter* strain cultures of ST-474, ST-190 and ST-45 were prepared as it was described in section 3.2 in details to obtain a plate with prolific growth for inoculum preparation. Each inoculum for testing heat resistance was prepared by transferring all colonies from a CA plate into Brain Heart Infusion broth (BHI). The final culture concentrations used for inoculation were in the range of  $10^7$  to  $10^9$  cfu ml<sup>-</sup>. The determination of inoculum concentration was conducted by a serial

dilution of the inoculum in BHI followed by plating in triplicate on mCCDA by the Surface Plate Method (Downes and Ito, 2001).

#### 4.2.2 C. jejuni heat inactivation procedure

The bacterial suspensions obtained from above were heated to a predetermined temperature using a submerged coil heating apparatus (Protrol instruments Ltd., U.K). As described in detail in section 3.3 that samples were taken at predetermined intervals (from 2 s -200 s with a total heating time range from 12 s - 1050 s, depending on the investigated temperature and the isolate heat resistance) automatically and collected in a series of vials in an automated carousel (Cole and Jones, 1990). The inoculum population and the number of surviving organisms in suspension after heat treatment were determined by dilution in BHI followed by plating on mCCDA using an automated spiral plater (Don Whitley, Yorkshire, United Kingdom)

#### 4.3 Modelling and survival data fitting

Survival curves were constructed by plotting the log of surviving bacterial counts against heating time for each temperature. These were then fitted using a range of kinetic models. The log-linear kinetic model (Equation 4.1) has been used extensively in the calculation of the thermal inactivation parameters, *D* and z. However, non-linear survival curves have also been reported (Cerf, 1977, Peleg and Cole, 1998) and several other models have been proposed to describe the patterns observed (Cerf, 1977, McKellar and Lu, 2004).

#### 4.3.1 Log-linear model

D values were estimated by two regression methodologies: (i) the classical two-step regression, and (ii) one step regression or global regression. With classical regression, the D value was calculated from the negative reciprocal of the slope of the linear regression line of the plot of log counts versus heating time (Equation 4.1). The z-value of each organism was subsequently calculated from the negative reciprocal of the slope of the plot of log D versus temperature (Equation 4.2).

$$\log_{10}(N_t) = \log_{10}(N_0) - \frac{t}{D}$$
 4-1

#### Where:

 $N_t$  represents the number of cells per unit mass or volume at time t (cfu ml<sup>-1</sup>).

 $N_0$  is the initial count of bacteria (cfu ml<sup>-1</sup>).

*D* is the decimal reduction time (s).

*t* is the total heating time at constant temperature (s).

$$\log_{10}(D) = \log_{10}(D_{ref}) + \frac{T_{ref} - T}{z}$$
 4-2

Where:

 $D_{\text{ref}}$  is the D value at a reference temperature  $T_{\text{ref}}$  (s).

T is the temperature (°C).

z is the temperature (°C) required for a one log D reduction.

In the one-step regression method, all the data are considered as a whole, and the parameters D (at a reference temperature, usually taken as the middle of the range of the temperatures investigated) and z are estimated by incorporating Equation 4.2 into Equation. 4.1 to express the temperature dependency of the D value (Equation4.3). By using numerical optimisation methods, the D and z values were estimated simultaneously using all the data (Valdramidis et al., 2004). The advantage of this method is that the error in estimating the D value is incorporated appropriately into the estimation of the z value. By contrast, in the classical method, the error in the estimation of D from Equation 4.1 is not included in Equation 4.2 for the calculation of the z value. Therefore, one-step regression is better in preventing the accumulation of fitting errors.

$$\log_{10} N_t = \log_{10} N_0 - \frac{t}{D_{ref} 10^{(T_{ref} - T)/z}}$$
 4-3

#### 4.3.2 The non–linear models

There are many reports in the literature which indicate that bacterial survival curves (e.g. log N vs. t) are not always strictly log-linear as described by Equation 4-1. Sigmoidal curves, curves with shoulders, tailing, and concavity upwards and downwards have all been reported (Campanella and Peleg, 2001, Cerf, 1977, Peleg et al., 2000, van Boekel, 2002). Various models have been proposed to fit these different survival curves.

The non-linear models which were tested for this study data are described in the literature review chapter two (section 2.4.2):

- 1. Shoulder before log-linear decrease.
- 2. Tail after a log-linear decrease.
- 3. Shoulder and tailing behaviour.
- 4. Concave or convex curves.

- 5. Concave or convex carves followed by tailing.
- 6. Biphasic inactivation kinetics.
- 7. Biphasic inactivation kinetics preceded by a shoulder.

The Weibull model has been the most widely studied and for this model bacterial inactivation is described by Equation 4-4 (Cunha et al., 1998, Fernandez et al., 1999, Mafart et al., 2002, Peleg, 2002).

$$\log_{10} N_t = \log_{10} N_0 - \left(\frac{t}{\delta}\right)^{\eta}$$
 4-4

In Equation 4.4,  $\eta$  and  $\delta$  are the parameters describing the shape and scale of the Weibull distribution function, respectively. The shape parameter gives a convex curve when  $\eta > 1$ , or a concave curve when  $\eta < 1$ . When  $\eta$  is equal to 1, a straight line corresponding to the log-linear model is obtained (Figure 4.1). This model has been successfully tested using a variety of published heat resistance studies in which it was concluded that the Weibull model was both more suitable and simpler for describing non-log-linear survival curves than other models (van Boekel, 2002).



#### Time

Figure 4-1 Typical inactivation curves for the Weibull model.

(a) a straight line when the shape parameter  $\eta = 1$ , (b) a convex curve when  $\eta > 1$ , (c) a concave curve when  $\eta < 1$ . (Adapted from (Cunha et al., 1998)).

The secondary models describe the dependency of one or more of the parameters of the primary models on a change in environmental conditions. In this study, the classical simple  $\chi$  value model was also used to describe the evolution of the Weibull model's scale parameter's ( $\delta$ ) dependency on temperature (van Boekel, 2002).

The temperature dependency of the scale parameter  $\delta$  for the popular Weibull model is expressed
by the z value model (Eq4-2) in the following form (by substitution of log  $\delta$  by a,  $T - T_{ref} = T$  and z = 1/b):

 $\log_{10}\delta = a - bT \qquad 4-5$ 

# 4.4 Statistical analysis and software

The software tool which was used to fit the data is *GlnaFit*, v 1.5 (Geeraerd et al., 2005). An example of the *GlnaFit* output is shown in Appendix A (Figure A1). The statistical criterion of the mean squared error (MSE) which was automatically reported for each model and the Akaike criterion were compared for the fitting of each survival curve using a simple ranking analysis (1 = best fit, 2 =  $2^{nd}$  best fit, etc.). The model with the minimum rank-sum value after fitting all the data sets was considered the best fitting model. The software has some built-in logical constraints on the parameter values to be estimated and will deliver an error message when selecting a model which seems not to be substantiated with the tested data. The R software (version 2.9.1) was used to calculate the *D* and z value by one step regression which was applied only for the log-linear model. The codes used for both parameters calculation are detailed in appendix A. For comparison purposes, the linear regression analyses for all data sets were performed using the Statistical Analysis System v.9.1 (SAS Institute Inc, USA) to estimate the *D* value from the negative reciprocal of the slope along with the *D* values obtained by GlnaFit software.

# 4.5 Results and discussion

#### 4.5.1 The log-linear model

The survival curves at different temperatures for the six isolates were obtained. A typical survival curve is shown in Figure 4.2. Most of the survival curves were fitted by the log-linear model, and the  $r^2$  values were > 0.90 for all regressions with only a few showing a small lag (shoulder) during heating at a temperature ranges from 51.5 to 60 °C. However, due to the heat sensitivity of ST 474 to 60 °C temperature no sufficient data points were collected for both isolates of this strain. Thus, no measured *D* value at 60 °C was reported for these strains but its *D* value was estimated from the z value and by one step regression technique. The non-linear model fitting is discussed later in this section.



Figure 4-2 Survival curve for C. jejuni (ST-190P) at 53.5 °C.

The linear model was fitted to all the experimental data and the *D* and z values obtained were 1.5 - 228.3 s, 4.1 - 4.7 °C respectively, by the two-step approach (Table 4.1, Figures 4.2, 4.3), and 1.3 - 187.8 s and 4.0 - 5.2 °C respectively by the one-step approach (Table 4.2, 4.3, appendix A for the model code in R software and the model output). These results are in agreement with international published data (Li et al., 2002, Ray, 2014, Juffs and Deeth, 2007). In fact, the *D* values for all NZ isolates at 60 °C (1.3 - 4.2 s)(Table 4.1) is slightly lower than the minimum *D* value of the published international data which range from seven to 23 s (ICMSF, 1996). Similarly, the *z* values calculated for all isolates from Table 4.1 or Table 4.2, ranged from 4.0 -5.2 °C were within the range of the published *z* values (2.8 - 5.8 °C) (Sorqvist, 2003). It is clear from Tables 4.1- 4.3 and Figure 4.4 that the *z* values did not vary significantly for each isolate.



Figure 4-3 z values for six isolates of C. jejuni.

The fact that the minimal *D* values estimated here are lower than published values may be due to many factors such as the strains used in the study, experimental conditions (e.g. heating media directly or indirectly), cooling after heating, equipment or laboratory media used, presence of salt or acid (pH), the growth phase of the cells and other factors (Pflug, 1990). Moreover, most published data for *Campylobacter* were obtained during the early 1980s with the available methods, instrumentation and techniques available at that time with regard to strain typing, isolation, recovery, enumeration, inactivation and incubation. The only *Campylobacter* D value reported by (Scanlon et al., 2015) in an experiment which used similar methodology and heating equipment to this study, reported higher D values at 60°C in the range of 42.4 s to 74.7 s. However, the study by Scanlon et al. (2015) investigated four different *Campylobacter* species have not been found in NZ campylobacteriosis patient clinical samples in New Zealand, according to a NZ epidemiological study (Nohra et al., 2016). This epidemiological study screened 1,601 of *Campylobacter* clinical isolates and found only 47 *C. coli* (2.9 %). Moreover, the difference between this study D values at 60°C and Scanlon et al. study (2015) is not remarkable.

Our data showed only a minor variation as the standard errors for the estimated D and z and values were from 0.3 - 1.1 s and 0.04 - 0.13 °C respectively. This was mainly achieved by using the one-

step regression approach due to the increased number of degrees of freedom (van Boekel, 1996). Strains ST-474 and ST-45 had similar D values, and they are slightly more heat-sensitive than ST-190. In general, the source of the isolate, either poultry or human, do not significantly affect the D value of the strain. Only the human isolate of ST-190 was (to a limited extent) more heat resistant than the poultry isolate of the same strain at all of the measured temperatures. Human strains have previously been reported to be more tolerant of chilling temperature than poultry strains (Chan et al., 2001). In addition, the isolate of ST-45 from a wild bird was more heat resistant than the human and poultry isolates of the same strain and its D value was similar to the D value of the poultry strain ST-190 (data not shown). The poultry and human isolates of ST-474 and ST-45 showed source-independent variations in their observed D values due to random biological variation. This random biological variation was observed to be more significant at the lowest heating temperature of 51.5 °C. This phenomenon has also been observed by others. For example, Doyle (1981) reported that as the temperature increased the range of rates of inactivation among the different strains narrowed. This was also observed in our study, particularly for strain ST-190 and ST-45, as shown in Tables 4.1, 4.2. and 4.3.

Table 4-1 Estimated $D$ and $z$ values for <i>Campylobacter</i> isolates by two-step regression at different
temperatures (SE in square brackets, $P = poultry$ isolate, and $H = human \text{ or clinical isolate, } n.r =$
not reported ).

	Γ	z value (°C)			
Strain	51.5	53.5	56.5	60	
ST-45P	178.9 [2.95]	46.8 [2.4]	8.5 [0.18]	1.5 [0.06]	4.1 [0.18]
ST-45H	228.3 [6.91]	45.3 [1.69]	8.0 [0.25]	1.8 [0.08]	4.1 [0.39]
ST-190P	n.r	85.2 [2.47]	17.0 [0.32]	2.8 [0.15]	4.4 [0.17]
ST-190H	n.r	128.2 [2.28]	24.1 [0.28]	4.2 [0.13]	4.7 [0.41]
ST-474P	147.5 [3.25]	47.5 [0.76]	11.0 [0.32]	n.r	4.4 [0.05]
ST-474H	156.5 [3.96]	43.3 [2.83]	11.1 [0.20]	n.r	4.4 [0.14]

Table 4-2 Estimated D and z values for *Campylobacter* isolates by one-step regression at different temperatures (P = poultry isolate and H = human or clinical isolate).

Strain	Temperature (°C)	D value (s)	SE	₹ value (°C)	SE
ST-45 <sub>P</sub>	55.8	14.7	0.31	4.0	0.05
ST-45 <sub>H</sub>	55.8	15.8	0.45	4.0	0.06
ST-190 <sub>P</sub>	56.8	17.1	0.47	4.5	0.11
ST-190 <sub>H</sub>	56.8	22.5	0.34	4.4	0.05
ST-474 <sub>P</sub>	54.0	46.1	0.79	5.2	0.13
ST-474 <sub>H</sub>	54.0	40.8	1.06	4.4	0.09

Table 4-3 Calculated $D$ and $z$ values for <i>Campylobacter</i> isolates at different temperatures based on
the D and $z$ values estimated by one-step regression in Table 4.2 for a comparison with the values
indicated in Table 4.1.

					z value
	D v	alue (s) at Tem	perature (°C	)	(°C)
Strain	51.5	53.5	56.5	60.0	•
ST-45 <sub>P</sub>	174.7	55.2	9.8	1.3	4.0
ST-45 <sub>H</sub>	187.8	59.4	10.6	1.4	4.0
ST-190 <sub>P</sub>	n.r	92.5	19.9	3.3	4.5
ST-190 <sub>H</sub>	n.r	126.5	26.3	4.2	4.4
ST-474 <sub>P</sub>	139.5	57.5	15.2	n.r	5.2
ST-474 <sub>H</sub>	151.0	53.0	11.0	n.r	4.4

#### 4.5.2 The non–linear models

In general, for any given run, one or more of the non-linear models produced better fits to the inactivation data than the log-linear model based on the goodness of fit criteria employed. This is in agreement with other studies (Campanella and Peleg, 2001, van Boekel, 2002, Keklik et al., 2011, Cebrián et al., 2019) which reported that non-linear models fit the survival data sets better. An example of a few data sets is shown in Table 4.4. Overall, the Weibull model (Equation 4.4) was the only non-linear model which could be successfully fitted to all data sets. The *GlnaFit* program identified unlikely or unreliable parameter estimates for the other non-linear models for at least some data sets. Therefore, the Weibull model was further used to determine the effect of temperature on the scale and shape parameters ( $\delta$ ,  $\eta$ ). The scale factor ( $\delta$ ) showed a similar log-linear dependency on temperature as the *D* value (Figure 4.4 and Figure 4.5). This temperature dependency of the scale factor has also been reported by others (Mishra and Puri, 2013, Farakos et al., 2014, Kaur and Rao, 2017, Li et al., 2014) However, the shape factor ( $\eta$ ) did not depend on the temperature systematically but varied significantly across the data sets (Figure 3.6). This is in agreement with most of the published data (van Boekel, 2002, Farakos et al., 2014).



Figure 4-4 D value as a function of temperature for C. *jejuni* inactivation

Figure 4-5  $\delta$  value as a function of temperature for C. *jejuni* inactivation



Figure 4-6 Variation observed in the Weibull shape factor  $\eta$  fitted to *C. jejuni* survival curves as a function of temperature.

Thus, it seems worthwhile to fix the  $\eta$  value. The mean value of  $\eta$  was selected as a characteristic of *Campylobacter* for the overall data sets (Mafart et al., 2002). The new generic model, therefore, can be represented as:

$$\log_{10} N_t = \log_{10} N_0 - \left(\frac{t}{\delta}\right)^{1.1}$$
 4-6

When the mean shape factor ( $\eta = 1.1$ , SE= 0.053) was employed, the goodness of fit to the survival data was poorer than for the Weibull model when fitting  $\eta$  and  $\delta$  individually (Table 4.4) and was also poorer than the linear model for many data sets, providing a better fit for only 50 % of the data sets. The highlighted mean squared error values and Akaike criteria values in Table 4.4 indicate the best goodness of fit for the Weibull model followed by the linear model which has equal goodness of fit to the 'Weibull with fixed parameter' model.

Table 4-4 Selected examples for illustrating the goodness of fit for nine models fitted to the survival data of selected strains of *C. jejuni*. (P =poultry isolate and H = human or clinical isolate).

		The goodness of fit (MSE/AIC)				
		ST-474 <sub>P</sub>	ST-45 <sub>H</sub>	ST-190 <sub>H</sub>	ST- 474 <sub>P</sub>	
	Model	53.5 °C	51.5 °C	53.5 °C	56.5 °C	
1	Linear	0.095	0.048	0.023	0.093	
		-65.731	-95.124	-85.376	-73.561	
2	Linear + shoulder	0.032	n/a	n/a	0.045	
u						
3	Linear + tail	n/a	0.020	0.018	n/a	
4	Linear + shoulder + tail	n/a	0.020	n/a	n/a	
5	Weibull	0.042	0.035	0.018	0.037	
		-91.789	-107.071	-92.390	-105.503	
6	Weibull + fix	0.062	0.064	0.038	0.071	
		-78.517	-85.949	-73.119	-82.4129	
7	Weibull + tail	n/a	0.020	n/a	n/a	
8	Biphasic	n/a	n/a	0.025	n/a	
9	Biphasic + shoulder	n/a	0.018	0.016	n/a	

The recalculated scale factor ( $\delta$ ') with fixed  $\eta$  showed a similar dependency on temperature as the D value in the log-linear model, and the  $\chi$  value 4.7 °C was unchanged ( $\chi$ ' = 4.7 °C) (Figures 4.4 and 4.5). The time required to achieve a given process extent (e.g. a 7 D reduction) was similar for the Weibull with fixed parameter' model and the log-linear model, as shown in Table 4.5.

Table 4-5 Comparison of heating time(s) required to achieve a 7 D reduction in the number of bacterial cells using the Log-linear (Equation 4.1) and the Weibull model (Equation 4.4) with pooled data D and  $\delta$  values at selected temperatures.

Temperature	7 D time based on Weibull	7 D time based on log-linear
(°C)		
51.5	1114.99	1270.86
53.5	416.17	474.35
56.5	94.90	108.17
60	16.91	19.28
65	1.44	1.64
70	0.12	0.14

# 4.6 Conclusions

The  $D_{60}$  and z values in broth medium for the most important strains of *C. jejuni* in human cases of campylobacteriosis in NZ were within the range from 2 - 5 s and 4 - 5.2 °C, respectively. These values are broadly in agreement with the published international data and do not indicate that the tested NZ *Campylobacter* strains are more heat resistant than other strains. The Weibull model fitted the survival data better than the log-linear model if the shape factor was allowed to vary between runs. When the mean shape factor was employed and the scale factor obtained by re-fitting the model, the goodness of fit was poorer. This 'Weibull with fixed parameter' model was only as good a fit as the log-linear model. That is not so strange if the value is 1.1, i.e. very close to 1. The scale factor of the Weibull model showed a similar dependency on temperature as the *D* value in the log-linear model and the time required to achieve a given process extent (e.g. a 7 *D* reduction) was similar for both models. The likelihood of a systematic error in using the log-linear model in quantitative risk models, therefore, appears to be low. It is important to verify that the *D* and *z* values obtained in broth are similar to those obtained in food samples to confirm that these NZ strains are not heat resistant and the next chapter addresses this point.

# 5 Thermal inactivation of *C. jejuni* in food

I

#### 5.1 Introduction

Different values of D and z for many pathogens have been reported in the literature (as summarised in Table 2.3). One major factor influencing the observed kinetics is the composition of the matrix with which the bacterium is associated with during the heat treatment (Stumbo, 1973). *Campylobacter* heat resistance has not been studied as extensively by researchers as other food pathogens probably because it is generally regarded as heat-sensitive. Nevertheless, due to the variation in reported kinetic parameters, it is difficult to choose a value for risk assessment and HACCP plans for any specific cooking process from the reported values in the literature (Bergsma et al., 2007, de Jong et al., 2012, Forsythe, 2000, Nguyen et al., 2006).

It is also known that *Campylobacter* attachment to chicken skin increases its heat resistance (Yang et al., 2001). The chemical environment around microorganisms during the heating can increase or decrease heat resistance (Doyle et al., 2001). Many studies have reported the importance of determining the kinetic parameters D and z in food matrices rather than in liquid media. WHO and the local regulatory bodies always emphasise that the evaluation of process lethality should be based on food data or real substrates contaminated with the target organism (Gaze et al., 1989). Therefore, the objective of this study was to determine the kinetic parameters (D and z values) of

selected NZ *C. jejuni* strains on chicken skin which is often significantly contaminated (Scherer et al., 2006). The finding of this study will confirm whether NZ *Campylobacter* strains are more heat resistant than overseas strains, which has been proposed as a reason for the high rate of campylobacteriosis in NZ.

# 5.2 Materials and methods

#### 5.2.1 Campylobacter culture growth and enumeration

As described in detail in section 3.2 *Campylobacter* strain cultures of ST-474 and ST-48 were grown on CA and sub-cultured on to another CA plate and incubated for 48 h in a micro-aerobic atmosphere at 41.5 °C to obtain a plate with prolific growth for inoculum preparation. Each inoculum for testing heat resistance was prepared by transferring all colonies from a CA plate into Brain Heart Infusion broth (BHI). The final culture concentrations used for inoculation were in the range of 10<sup>7</sup> to 10<sup>9</sup> cfu ml<sup>-</sup> The determination of inoculum concentration was conducted by a serial dilution of the inoculum in BHI followed by plating in triplicate on mCCDA by the Surface Plate Method (Downes and Ito, 2001). Plates were incubated micro-aerobically at 41.5 °C for 48 h.

#### 5.2.2 Meat sample preparation and inoculation

The chicken skin was prepared as described in details in section 3.4.1 and was equilibrated to room temperature and placed aseptically in an aluminium test cell (Appendix B, Figure B1) specially designed for kinetic studies (Chung et al., 2008). An inoculum of 20  $\mu$ l was pipetted to the chicken skin centre in the lower part of the aluminium cell. After inoculation, the cell was left for 60 min to allow *C. jejuni* attachment to chicken skin to mimic natural contamination.

#### 5.2.3 *C. jejuni* heat inactivation procedure.

AS described in detail in section 3.4.2, the eight aluminium cells were placed in a rack submerged in a temperature-controlled water bath (Grant, UK) maintained at a 56.5 to 65 °C temperature range for a predetermined time period (Appendix B, Figure B2). One aluminium cell was removed at each prescribed time interval and then transferred immediately to an ice bath for *C. jejuni* survivor analysis. The time it took for the chicken skin sample to reach the equilibration temperature ('come up time') was measured by a very fast response thermocouple sensor with one second response time (Labfacility, UK). The temperature was recorded with a Measurement Advantage USB-TEMP logger (Measurement Computing, USA). A graph of the temperature profile is shown in Appendix B(Figures B3-B6). The 'come-up time was 55 s from a number of experiments for both the investigated temperatures 56.5 °C and 60 °C. All data points used in the calculation of the kinetic parameters were considered to estimate the come up time.

#### 5.2.4 Enumeration of *C. jejuni* survivors

Each chicken sample was transferred from the aluminium cell to a vial containing 4 ml BHI and the aluminium cell was flushed out with a further 5 ml BHI to remove any survivors from the internal surface of the cell. The sample was mixed using a vortex mixer and serially diluted in BHI. All the samples remained in BHI for at least two hours to allow recovery of injured cells, followed by plating in triplicate on mCCDA by the Surface Plate Method, a 200  $\mu$ l spread by L shaped sterilised spreader (Downes and Ito, 2001). Plates were incubated microaerobically at 41.5 °C for 48 h as described above. The average of the logarithm (base 10) of three plate counts at each sampling time was used for modelling and analysis.

# 5.3 Modelling and survival data fitting

All the modelling and survival data were carried out as for the broth data in the previous chapter, section 4.3.

Survival curves were constructed by plotting the log of surviving bacterial counts against heating time for each temperature. These were then fitted using a range of kinetic models.

#### 5.3.1 The log-linear model

D values were estimated by two regression methodologies: (i) the classical two-step regression, and (ii) one step regression or global regression. With classical regression, the D value was calculated from the negative reciprocal of the slope of the linear regression line of the plot of log counts versus heating time (Equation 4.1). The z-value of each organism was subsequently calculated from the negative reciprocal of the slope of the plot of log D versus temperature (Equation 4.2). In the one-step regression method, all the data are considered as a whole, and the parameters D (at a reference temperature, usually taken as the middle of the range of the temperatures investigated) and z are estimated by incorporating Equation 4.2 into Equation. 4.1 to express the temperature

dependency of the *D* value (Equation 4.3). By using numerical optimisation methods, the *D* and z values were estimated simultaneously using all the data (Valdramidis et al., 2004).

### 5.3.2 The non–linear models

Various models have been proposed to fit these different survival curves.

The non-linear models which were tested for this study data are described in the literature review chapter two (section 2.4.2):

The Weibull model has been the most widely studied and for this model bacterial inactivation is described by Equation 4-4

The secondary models describe the dependency of one or more of the parameters of the primary models on a change in environmental conditions. In this study, the classical simple z value model was also used to describe the evolution of the Weibull model's scale parameter's ( $\delta$ ) dependency on temperature (van Boekel, 2002).

The temperature dependency of the scale parameter  $\delta$  for the popular Weibull model is expressed by the z value model (Eq4-2)

# 5.4 Statistical analysis and software

The software tool which was used to fit the data is *GlnaFit*, v 1.5 (Geeraerd et al., 2005). The R software (version 2.9.1) was used to calculate the D and z value by one step regression which was

applied only for the log-linear model. The codes used for both parameters calculation are detailed in appendix A. For comparison purposes, the linear regression analyses for all data sets were performed using the Statistical Analysis System v.9.1 (SAS Institute Inc, USA) to estimate the Dvalue from the negative reciprocal of the slope along with the D values obtained by GlnaFit software.

# 5.5 Results and discussion

#### 5.5.1 The linear model

A typical survival curve is shown in Figure 5.1. Most of the survival curves were linear, and the  $r^2$  values were > 0.92 for all regressions with only a few showing a small lag (shoulder,  $r^2 > 0.85$ ) during heating at the investigated temperature range (56.5 to 65 °C). Therefore, the linear model was chosen to fit all the experimental data. All isolates were rapidly inactivated at 65 °C and hence no data for that temperature. Sufficient data were obtained at 56.5 °C and 60 °C. The extrapolated *D* values at 65° for all the isolates were obtained from the measured *D* values at 56.5 °C and 60 °C. The extrapolated of °C. The calculated D<sub>65</sub> values were comparable with the *D* values obtained for at 65 °C by the dynamic method in the next chapter.

The *D* values at the temperature range 56.5 - 60 °C were 42 - 12 s when estimated by one step regression (see Table 5.1) and 42 - 11 s by the two-step regression (see Table 5.2). These results are lower than the published *D* values for *C. jejuni* on chicken skin, 90-130 s at 55 °C and 30 s at 60 °C (Yang et al., 2001).

However, the z values of all studied isolates (ranging from 7.6 - 11.3 °C, Table 5.1) were higher than some (extrapolated) published data, viz. 6 - 6.4 °C (Forsythe, 2000). However, they are in agreement with the extrapolated z value of 12.3 °C published by another study (van Asselt and Zwietering, 2006). Discrepancies between z values are well documented in the literature. For example, one study reported a z value for Listeria as 29 °C (Fang and Lin, 1994) when most textbooks reported a range of 5 - 12 °C (Ryser and Marth, 2007). The *D* and z values we report here are 2- to 3-fold higher than the values obtained for the same isolates in broth in the previous chapter, confirming the importance of making measurements in food.

The lower *D* values found in this study compared to published data for chicken may relate to the strains used in the study, experimental conditions (e.g. mechanism of heating of media), method of cooling after heating, equipment or laboratory media used, presence of salt or acid (pH), or the growth phase of the cells (Pflug 1990). Most published data for *Campylobacter* were acquired during the early 1980s with the then available methods, instrumentation and techniques used for strain typing, isolation, recovery, enumeration, inactivation and incubation. *Campylobacter* has been

described as having a 'capricious' nature because of the high variation detected between tests, which cause more difficulties in standardisation and handling this pathogen than with any other pathogen (Bergsma et al., 2007, Jasson et al., 2007). This study data also shows variations as the standard errors for the estimated D values were from 0.8 to 4.8 and 0.6 to 4.7 for the z values. The diversity within *C. jejuni* strains is well known and has been detected at both the phenotypic and genotypic level (Gölz et al., 2018).

There was a small but statistically significant difference in the D values between the poultry and human isolates of strain 474. However, there was no significant difference between the poultry and human isolates of strain 48 or between the human isolates of both strains with only a minor but statistically significant difference between the D values of both poultry strains. This statistical difference does not have any practical importance as the more heat resistant strain 48 is less attributed to campylobacteriosis in NZ than strain 474 which accounts for ~30 % of the cases (Müllner et al., 2010). Generally, the low D values obtained for both strains confirmed that NZ strains tested in this study are more heat-sensitive than the international strains and should be easily eliminated by normal cooking or processing treatment. Cross-contamination has been stated as the most important factor for the high rate of campylobacteriosis internationally (Luber, 2009), and it seems likely that the same situation applies in NZ.



Figure 5-1 Survival curve for *C. jejuni* strain ST-474<sub>H</sub> inoculated on chicken skin and heated at 60  $^{\circ}$ C.

	Strain	D value (s)		₹ value (°C)	Se
		at 58.25 °C	Se		
1	ST-474 <sub>P</sub>	19.0	0.8	7.6	0.6
2	ST-474 <sub>H</sub>	23.2	1.3	11.1	1.7
3	ST-48 <sub>P</sub>	27.3	4.8	9.6	4.0
4	ST-48 <sub>H</sub>	22.7	3.4	11.3	4.7

Table 5-1 D and z values for C. *jejuni* isolates estimated by one step regression at different temperatures.

Table 5-2 Calculated D and z values for C. *jejuni* isolates at different temperatures based on the D and z values estimated by one-step regression in Table 5.1 for a comparison purpose with the values indicated in Table 5.3.

Strain	D value (s)	₹ value (°C)		
	Temperature (			
	56.5	60.0	65.0	
ST-474 <sub>P</sub>	34.7	12.1	2.7	7.6
ST-474 <sub>H</sub>	33.3	16.1	5.7	11.1
ST-48 <sub>P</sub>	41.6	18.0	5.4	9.6
ST-48 <sub>H</sub>	32.5	15.9	5.7	11.3

Table 5-3 Estimated D and z values for C. *jejuni* isolates by two-step regression at different temperatures

	D value (s) at Temperature (°C)		₹ value (°C)
Strain	56.5	60	
ST-474 <sub>P</sub>	22.5 (0.31)	11.2 (0.43)	8.8 (3.77)
ST-474 <sub>H</sub>	33.3 (2.45)	16.1 (1.39)	11.1 (2.15)
ST-48 <sub>p</sub>	41.5 (11.07)	14.6 (0.12)	10.0 (5.14)
ST-48 <sub>H</sub>	29.5 (1.27)	15.9 (1.12)	11.1 (6.93)

# 5.5.2 The non–linear models

In general, for any given run, one or more of the non-linear models produced a better fit for the inactivation data than the linear model based on the goodness of fit criteria employed. Overall, the Weibull model (Equation 4.4) was the only non-linear model which could be successfully fitted to all data sets. The *GlnaFit* program identified unlikely or unreliable parameter estimates for the other non-linear models for at least some data sets. Therefore, the Weibull model was further analysed to determine the effect of temperature on the scale and shape parameters ( $\delta$ ,  $\eta$ ). The scale factor ( $\delta$ ) showed a similar log-linear dependency on temperature as the *D* value (Figures 5.3 and 5.4). However, the shape factor ( $\eta$ ) did not depend on the temperature systematically and varied significantly across the data sets (Figure 5.2). This is in accordance with most published data (van Boekel, 2002).



Figure 5-2 Variation observed in the Weibull Shape factor  $\eta$  fitted to *C. jejuni* survival curves as a function of temperature.



1.8 1.6 1.4 1.2 (s) Q Bol 0.8 -0.101x + 7.269 0.6  $R^2 = 0.740$ 0.4 0.2 56.5 59.5 56 57 57.5 58 58.5 59 60 60.5 Temperature (°C)

Figure 5-3  $\delta$  value as a function of temperature for *C. jejuni* inactivation

Figure 5-4 *D* value as a function of temperature for *C. jejuni* inactivation

Thus, it seems worthwhile to fix the  $\eta$  value. The mean value (1.11, SE=0.18) of  $\eta$  was selected as characteristic of *C. jejuni* for the overall data sets (Mafart et al. 2002). Hence, a new generic model is described below.

$$\log_{10} N_t = \log_{10} N_0 - \left(\frac{t}{\delta}\right)^{1.11}$$
 5-1

When the mean shape factor ( $\eta = 1.11$ ) was employed, and the Weibull model was refitted to the survival data, the goodness of fit was poorer than for the Weibull model when  $\eta$  and  $\delta$  were fitted individually. However, it was still better than the linear model for >60 % of the data sets. The recalculated scale factor ( $\delta$ ') with fixed  $\eta$  showed a similar dependency on temperature as the *D* value in the log-linear model (z' = 9.7 °C, Figure 5.3;  $\chi = 9.9$  °C, Figure 5.4).

The time required to achieve a given process extent (e.g. a 7 D reduction) was quite similar at the highest temperature for the Weibull fixed model and the log-linear model (Table 5.4). This finding suggests that both models give similar results. However, the Weibull fixed model should be investigated further as the pooled data approach may not represent the individual thermal characteristics of each *C. jejuni* isolate or strain. The primary comparison of the time for 7 D at 65 °C by the linear model, for example for ST-474<sub>P</sub> was 19 s from Table 5.2, whereas the pooled data value in Table 5.3 was 39 s. Therefore, Table 5.4 illustrates how important it is to select an appropriate model to fit inactivation survival data.

Temperature (°C)	7 D time based on Weibull	7 D time based on log-linear
51.5	736.67	956.31
53.5	462.67	595.11
56.5	230.29	292.14
60	102.04	127.38
65	31.90	38.91
70	9.97	11.89

Table 5-4 Comparison of heating times (s) required to achieve a 7 *D* reduction in the number of bacterial cells using the Log-linear and the Weibull model.

The maximum contamination level with *C. jejuni* in NZ is below  $10^6$  cfu per carcass (Chrystal et al., 2008) and inactivation at this level would be easily achieved by cooking to the temperature recommended by The Food Safety and Inspection Service for poultry products (75 – 80 °C). In fact, this level of contamination would be inactivated by far lower cooking temperatures than the recommended cooking temperature.

Overall, the objective of this experiment was to determine if the food matrix and the cell attachment can alter the heat resistance of *Campylobacter* significantly and it seems that the food matrix significantly increases the heat resistance of *Campylobacter* but not to the same extent as reported by a Netherlands study (de Jong et al., 2012)). The two *Campylobacter* strains tested here, which are the most commonly implicated strains of campylobacteriosis in NZ, should still be considered as heat-sensitive microorganisms. Moreover, another study confirmed that *C. jejuni* is heat sensitive as it was not detected after simulated home pan-frying of artificially inoculated with 4 log cfu/g of *Campylobacter jejuni* steaks or filets, hamburgers and meat strips (Lahou et al., 2015).

# 5.6 Conclusions

The presence of a food matrix altered the kinetic parameters for *Campylobacter* heat resistance. Nevertheless, the  $D_{60}$  and z values calculated for the most important strains of *C. jejuni* on chicken skin in NZ were within the range from 11 - 18 s and 8 - 11 °C, respectively. These values are broadly in agreement with the published international data and do not indicate that NZ *Campylobacter* strains are more heat resistant than other strains. However, it is important to verify that the *D* and z values obtained by isothermal conditions and the *D* and z values obtained under non- isothermal conditions reflect the real scenario in practice as the temperature is not always constant in a processing environment.

# 6 Thermal inactivation of *C. jejuni* in food under dynamic conditions

I

# 6.1 Introduction

Kinetic parameters describing thermal inactivation of micro-organisms are most commonly determined under nominally isothermal conditions for the apparent theoretical and practical simplicity. These kinetic parameters obtained under isothermal conditions should not be applied to variable processing conditions without validation under dynamic conditions. In practice, the thermal effect is incurred during the time required to achieve a constant temperature. This must be accounted for to avoid errors in parameters calculation. This time is called the 'come up time' which considers the heat transfer phenomena. As a result of excluding the come up time period from the heat treatment time, there may be a limited temperature range from which to choose sufficient points for the experiments, particularly for heat-sensitive organisms (Welt and Tong, 1993).

Dynamic models describing microbial inactivation are expressed as differential equations which incorporate the imposed or observed time-temperature profile as an input to the dynamic model selected which describes the thermal death of micro-organisms. The resulting equations can be solved numerically to obtain the number of surviving organisms at each time and temperature regime. If the rate of temperature increase can be described by a simple function, the dynamic model equation can be solved analytically (Miles and Mackey, 1994). Several procedures have been proposed to accomplish this integration (Patashnik, 1953, Stumbo, 1973).

The determination of kinetic parameters under dynamic conditions can better reflect realistic processing scenarios and the 'come-up time' is explicitly a part of the experimental design, for example by specifying the heating rate and time-temperature profile (Welt et al., 1997). In general, estimating the kinetic parameters under dynamic conditions has been hampered by the mathematical complexity of the calculations involved and the instruments required to conduct the experiments. However, modern software and rapid development in technology and instrumentation have helped to overcome these two drawbacks. Therefore, interest in mathematical modelling and analysis of dynamic processes have increased in the last two decades, and dynamic models have been developed for microbial growth and inactivation, heat transfer and changes in texture and sensory properties (Peleg, 2003, Conesa et al., 2003, Dolan et al., 2007, Valdramidis et al., 2008, Pesciaroli et al., 2019, Garre et al., 2018, Tarlak et al., 2020).

As *Campylobacter* is considered a heat-sensitive microorganism, the data for non-isothermal kinetic parameters are not available. Therefore, the objective of this study was to determine the microbial kinetic parameters (D and z) under dynamic conditions and to compare the data generated with the D and z values obtained under isothermal conditions for the most commonly implicated strains in campylobacteriosis cases in NZ.

# 6.2 Materials and methods

### 6.3 *Campylobacter* culture growth and enumeration

*Campylobacter* strain cultures of ST 474 and ST 48 were prepared as described in detail in section 3.2 grown on CA were sub-cultured on to another CA plate and incubated for 48 h in a micro-aerobic atmosphere at 41.5 °C to obtain a plate with prolific growth for inoculum preparation. Each inoculum for testing heat resistance was prepared by transferring all colonies from a CA plate into Brain Heart Infusion broth (BHI). The final culture concentrations used for inoculation were in the range of  $10^7$  to  $10^9$  cfu ml<sup>-1</sup>. The determination of inoculum concentration was conducted by a serial dilution of the inoculum in BHI followed by plating in triplicate on mCCDA by the Surface Plate Method (Downes and Ito, 2001).

#### 6.3.1 Meat sample preparation and inoculation

The chicken skin was prepared as described in details in section 3.4.1 and was equilibrated to room temperature and placed aseptically in an aluminium test cell (Appendix B, Figure B1) specially designed for kinetic studies (Chung et al., 2008). An inoculum of 20  $\mu$ l was pipetted to the chicken skin centre in the lower part of the aluminium cell. After inoculation, the cell was left for 60 min to allow *C. jejuni* attachment to chicken skin to mimic natural contamination.

#### 6.3.2 Heat inactivation procedure

As described in detail in section 3.5.2 For each run, 16 aluminium cells containing inoculated chicken skin samples were placed in a programmable water bath (developed locally at the university, described in Appendix C, Figure C1). The water temperature was programmed to increase linearly to reach 65 °C for a period of either 8 or 18 min simulating the scenarios that can occur in retail food service, industry and in domestic cooking (Persson et al., 2002, van Asselt et al., 2009, NACMCF, 2007). Duplicate samples (two aluminium cells) were withdrawn at different predetermined intervals and transferred immediately to an ice bath until analysed for the counts of surviving culturable *C. jejuni*.. The temperature profiles for both regimes were embedded in the individual survival curves and are shown in the results section.

#### 6.3.3 Enumeration of *C.jejuni* survivors

As described in 3.3.2, the average of the  $log_{10}$  counts for four plates at each sampling time was used for modelling in contrast to the three plates used in section 3.3.2. This was to improve the accuracy of the results of this experiment because less number of experiments are required to conduct under the dynamic conditions.

# 6.4 Modelling and survival data fitting

The dynamic modelling technique requires the application of a one-step regression approach (Valdramidis et al., 2008). The first-order kinetic model (Equation 6.1) was selected to model the survival data as it has been shown to provide an adequate fit and for its simplicity and global applicability. It is a common approach (Pflug, 1990)to incorporate the Bigelow model for temperature dependence by defining the parameter D value as a measure of the rate of microbial inactivation instead of the kinetic rate coefficient K. By incorporation of Bigelow model (Equation 6..2) and with K expressed in the following format:

$$\frac{dN_t}{dt} = -kN$$

$$k = \frac{\ln 10}{D_{ref}} e^{\ln 10/z(T-T_{ref})}$$

$$6-2$$

The final model can be expressed as follows:

$$\frac{dN_t}{dt} = -\frac{\ln 10}{D_{ref}} e^{\ln 10/z(T-T_{ref})} N$$
 6-3

The measured time and temperature profile were incorporated as an input to the above model to obtain the D and z values at a reference temperature. Following the common convention, this was selected to be the average temperature of 56°C of the experimental temperature range. This is midway between the initial inactivation temperature of the pathogen 47 °C for *C. jejuni* (Hazeleger et al., 1998) and the final temperature selected for the heat treatment 65 °C. This was to minimise the uncertainty in the estimation of D value (Poschet et al., 2004)).

#### 6.5 Statistical analysis and software

Parameter optimisation was accomplished using the MATLAB Optimisation Toolbox (MathWorks Inc. Natick, MA, USA). The Trust-Region-Reflective algorithm was used to

estimate the kinetic parameters. This is an iterative technique that locates the minimum of a multivariate function expressed as the sum of squares of non-linear real-valued functions. Initial guesses are made for the true value of the parameters. (Please see the appendix D for the MATLAB code used in this study). The standard errors of the estimated parameters were calculated by the Jacobian matrix (Valdramidis et al., 2008).

# 6.6 Results and discussion

The *D* and z values calculated at the reference temperature of 56 °C (the mid-point selected from the experimental values of 47 - 65 °C) were in the range of 30 - 40s and 8.7 - 10.2 °C, respectively. The results are summarised in Table 6.1 and Figures 6.1 - 6.4.

Table 6-1 D and z values estimated from inactivation data at 56 °C with two different heating time points (8 & 18 min).

					$N_0$	
Strain	$D_{56^{\circ}\mathrm{C}}(\mathrm{s})$	SE	₹(°C)	SE	(log cfu g <sup>-1</sup> )	SE
ST-474 <sub>P</sub>	37.6	3.9	10.2	1.3	8.6	0.3
ST-474 <sub>H</sub>	30.8	1.8	9.2	1.1	9.3	0.2
ST-48 <sub>P</sub>	39.8	2.2	8.7	0.7	9.5	0.2
ST-48 <sub>H</sub>	38.3	2.3	9.6	0.8	8.9	0.2

These results are in general agreement with the results obtained for the same isolates by the isothermal approach (Tables 5.1, 5.2) and do not indicate that the isolates are heat resistant. Therefore, the heat resistance of key NZ *C. jejuni* strains appears broadly to agree with most internationally reported values (Adams and Moss, 2000, Forsythe, 2000, Ray, 2014, Juffs and Deeth, 2007).

However, the z values listed in Table 6.1 (ranging from 8.7 - 10.2 °C) are higher than the isothermal published data (5 – 8 °C) (Forsythe, 2000, Juffs and Deeth, 2007) but are lower than the isothermal extrapolated z value (12.3 °C) published in the study by (van Asselt and Zwietering, 2006). The variation of z values reported internationally is well recognised, and in one study the z value of Listeria was reported as 29 °C (Fang and Lin, 1994) which is three to four times higher than the values published in the literature (Ryser and Marth, 2007).



Figure 6-1 Thermal inactivation of *C. jejuni*  $ST-474_P$  under dynamic conditions with two different heating profiles, 8 and 18 min (two straight lines).



Figure 6-2 Thermal inactivation of *C. jejuni* ST-474 $_{\rm H}$  under dynamic conditions with two different heating profiles, at 8 min and 18 min.



Figure 6-3 Thermal inactivation of *C. jejuni* ST-48<sub>P</sub> under dynamic conditions with two different heating profiles, 8 and 18 min.



Figure 6-4 Thermal inactivation of *C. jejuni*  $ST-48_H$  under dynamic conditions with two different heating profiles, 8 and 18 min.

This partial disagreement of the results of this study with the international isothermal data in terms of the estimated z values is due to many factors such as the strains used in the study, experimental

conditions (e.g. heating media directly or indirectly), cooling after heating, and the equipment or laboratory media used (particularly if in the presence of salt or acid), the growth phase of the cells, and other factors (Pflug, 1990). Most published data for *Campylobacter* were reported during the 1980s with the then available methods, instrumentation and techniques with regard to strain typing, isolation, recovery, enumeration, inactivation and incubation. The standard errors observed in this study for *D* and *z* values are 1.7 - 3.8 s and 0.7 - 1.2 °C respectively. Few recent studies have addressed the heat resistance of *Campylobacter jejuni* but none of these studies have reported the D values neither at constant temperature nor at dynamic temperature.

Sampers et al. (2010) found the 4.5 log cfu/g of *C. jejuni* inoculated to chicken burgers were reduced to below the detectable level (<10 cfu/g) after four min frying to an internal temperature of 57.5 °C and 2.5 log cfu/g of naturally contaminated chicken burger reduced to below detectable level after four min frying to an internal temperature 52.1 °C. Moreover, *C. jejuni* was not detected after simulated home pan-frying of artificially inoculated steaks or filets, hamburgers and meat strips (Lahou et al., 2015). It is noteworthy to indicate that this study (Lahou et al., 2015) has reported D values for the tested *C. jejuni* strains in broth medium only before inoculating the tested strains to the different meat types. All meat samples were from various animal species such as the pig, cattle, chicken, lamb and some turkey, horse, crocodile and kangaroo were inoculated with 4 log cfu/g of *C. jejuni*. Only after a further enrichment step *C. jejuni* was detected in few samples of pork hamburger, horse steak and crocodile steak. This is probably due to the different nutritional composition (e.g. fat content), texture (e.g. fibre structure) of the different meat types.

The D and z values estimated under dynamic condition are considered to be more precise and straightforward as they are i) derived from a complete temperature range, ii) estimated by one step regression approach with less uncertainty (smaller SE values), and iii) obtained with less experimental effort (Dolan et al., 2007). However, more validation is required before the estimated parameters are incorporated in subsequent thermal treatment calculation since the industry has a very good record on the use of the thermal parameters derived from the isothermal approach.

The estimation of the parameters under dynamic conditions overcomes the problem of calculating the 'come-up time' or the temperature transient (conduction) as the sample temperature is raised gradually from ambient temperature to minimises the delay that occurs in heating the sample.

# 6.7 Conclusions

The obtained D and z values for NZ strains under dynamic conditions are broadly in agreement with the published international isothermal data. Hence the tested NZ *Campylobacter* strains do not indicate that they are more heat resistant than other strains. The parameters obtained under dynamic conditions are more robust than the isothermal parameters and such information can be derived with less experimental effort. A question that needs to be addressed is, are the NZ strains similar to the international strains in terms of their heat resistance, and then what their ability is to survive under other environmental conditions such as atmospheric gas atmosphere. It is known that *Campylobacter* is strictly microaerophilic, and this question is addressed in the next chapter.

# 7 Survival of *C. jejuni* under oxidative stress at different temperatures

# 7.1 Introduction

Campylobacter is considered as the most common bacterial causative agent of gastrointestinal disease in humans (Allos and Acheson, 2001). Campylobacter strains survive well in modified atmosphere and vacuum packaging but poorly at atmospheric oxygen concentration (~21%) (Kelly, 2008). Campylobacter spp. are harboured in many wild and domestic warm-blooded animals. These pathogens do not survive well in the environment when compared to other bacterial foodborne pathogens, such as Listeria and Salmonella. These organisms are unable to grow in the presence of air, and traditionally, a microaerobic mix (5 % O2, 10 % CO2 and 85 % N2) and high temperature (42°C) have been used for the isolation of the thermotolerant Campylobacter spp. (C. jejuni, C. coli, C. lari and C. upsaliensis) (Garenaux et al., 2008, ICMSF, 1996). Many studies have investigated the pathogen's survival under oxidative stress, which is defined as cell damage leading to cell death caused by excessive levels of oxidant and free radicals in the environment including both free oxygen and compounds such as H2O2, hydroxyl radicals and superoxide anions. These are summarized in the literature, for example, the publication of the International Commission on the Microbiological Specifications for Foods (ICMSF, 1996). Studies that have investigated the growth of Campylobacter under different gas compositions with different techniques and substrates are in general agreement with the proposed optimum composition (5 % O<sub>2</sub>, 10 % CO<sub>2</sub> and 85 % N<sub>2</sub>) (Grigoriadis et al., 1997, Koidis and Doyle, 1983). However, it remains unclear exactly how this pathogen survives the high oxygen level in the atmosphere and the low ambient temperatures during transmission to its human host.

There are other studies that have reported the survival of *Campylobacter* strains on poultry and meat is not influenced by the storage atmosphere or the treatment condition (vacuum packing) at 4°C (Hanninen et al., 1984, Wesley and Stadelman, 1985). Few studies reported that the presence of O<sub>2</sub> increased the death rate at 4°C in broth and milk (ICMSF, 1996). A New Zealand study revealed that *C. jejuni* could adapt to aerobic growth after a repeated subculture. In this study, about 81% of human, 75% of water, and 65% of poultry *C. jejuni* isolates were capable of growth under aerobic conditions in Nutrient Agar, and a cocktail of isolates (human, poultry, veterinary) survived aerobically for more than four weeks in poultry meat at 5°C and for more than a week at 25°C (Chynoweth et al., 1998). It was reported that *C. jejuni* isolates (10 human and nine poultry) could survive up to four weeks at 4°C aerobically in Mueller Hinton Broth and sterilised chicken rinse with a strain variation from one week to four weeks (Chan et al., 2001).

Yamasaki (2004) reported that one log of the initial *C. jejuni* inoculums (9.5 log) decreased in concentration after the first six hours under aerobic conditions at  $37C^{\circ}$  followed by another one log reduction at the end of next six hours. They also observed that a rapid reduction by about 5 log occurred between 12 and 15 h of incubation. Another study reported the survival in broth and agar plates for poultry and human isolates at three different temperatures (4, 25, and 42 °C) but the data were collected only for a one week period (Garenaux et al., 2008). Another study also indicated that the exposure to atmospheric oxygen in broth culture for five and 15 hours did not affect the *C. jejuni* growth and the count was similar to that of *C. jejuni* incubated under microaerobic conditions (Mihaljevic et al., 2007). This is consistent with other reports regarding the growth and adaptation of *C. jejuni* in aerobic metabolism (Harvey and Leach, 1998, Jones et al., 1993). One study investigated the survival of *C. jejuni* under oxidative stress in broth and agar plates but only at  $37^{\circ}C$  (Kaakoush et al., 2009).

From all of the above studies, it is concluded that there is ambiguity about the behaviour of *C. jejuni* under oxidative stress. This may confirm the capricious nature of such bacterial organisms due to the high variation found between tests and discrepancies between the reported results of the different studies. Evidently, *Campylobacter* causes more difficulties in standardization and handling of this pathogen, than are seen with any other pathogen (Bergsma et al., 2007, Jasson et al., 2007). It is believed that NZ has the highest rate of reported campylobacteriosis cases in the world. In 2006, the notification rate was 422 per 100,000, thus an increase of 56% from 2001, and 14% since 2005. This is 35 times higher than the rate for the United States, four times higher than the rate for Australia, and five times higher than the rate for the U.K. One possible reason for the high rate of campylobacteriosis in NZ could be that local strains have a greater ability to survive under processing, storage and handling conditions in the country (Hansen et al., 2003).

The objective of this study was to investigate the behaviour of selected NZ isolates of *C. jejuni* under varying conditions of oxidative stress and temperature. The strains chosen were those commonly found in human cases of campylobacteriosis and contaminated food samples in NZ. Thus, the study will focus on those relevant isolates, in order to provide a fast response for the scientific community and risk managers in NZ, with regards to the isolates' aero-tolerance and resistance to oxidative stress. The provision of a fast response by this study will confirm the scientific, academic capacity of the University in NZ, as the response of a non-academic scientific institution in NZ can be usually obtained in a timely manner. A comprehensive study using more NZ isolates and international isolates is planned if the chosen isolates showed aero-tolerance or

resistance to oxidative stress. Tough NZ biosecurity measures regarding the importation of biological hazards and the resulting possible delays in the administrative process have prevented the inclusion of any international isolates in this study.

# 7.2 Materials and methods

#### 7.2.1 Campylobacter culture growth and enumeration

*Campylobacter* strain cultures of ST- ST-474, ST-48, ST-190 were prepared as it was described in section 3.2 in details to obtain a plate with prolific growth for inoculum preparation. Each inoculum for testing heat resistance was prepared by transferring all colonies from a CA plate into Brain Heart Infusion broth (BHI). The final culture concentrations used for inoculation were in the range of 10<sup>7</sup> to 10<sup>9</sup> cfu ml<sup>-</sup>. The determination of inoculum concentration was conducted by a serial dilution of the inoculum in BHI followed by plating in triplicate on mCCDA by the Surface Plate Method (Downes and Ito, 2001).

#### 7.2.2 Exposure of *C. jejuni* to oxygen in broth

As it was described in detail in section 3.6.1 an inoculums of 3 ml were added to 27 ml BHI which was then dispensed to form a shallow layer of 30 ml in a 125 ml conical flask equipped with a special bung (Bug stopper, Whatman) that allows air exchange with the surrounding atmosphere but prevents microbial contamination. This flask was allowed to equilibrate at the appropriate temperature (4, 10, and 20 °C) atmosphere (aerobically or microaerobically) by gentle shaking using an electrical shaker (Lab-line Junior Orbit Shaker, U.K). The number of surviving organisms in samples withdrawn from each flask at predetermined intervals after exposure to the controlled atmosphere conditions at each temperature was determined by dilution in BHI followed by plating on mCCDA.

#### 7.2.3 Exposure of *C. jejuni* to oxygen on agar plates

As it was described section 3.6.2 in details, a 50 µl of the culture suspension in BHI was spread on CA plates) and incubated aerobically at 4 °C and 25 °C for one to five weeks. The CA control plates containing the culture spread were simultaneously incubated under microaerobic conditions at 4 °C and 25 °C using anaerobic plastic jars with a microaerobic atmosphere generating system (Pack MicroAero, Mitsubishi Gas Chemical Co. Inc). Two plates from each aerobic condition were sampled every day and incubated at 41.5 °C for 48 h in a microaerobic atmosphere.

#### 7.2.4 Exposure to $H_2O_2$

As described in detail in section 3.6.3, each culture suspension prepared in BHI was spread with a swab on Muller Hinton Agar plates and allowed to grow under microaerobic conditions for 48 h. Subsequently, filter discs (6 mm) inoculated with 10 µl of 1 mM, 10 mM, 100 mM, or 1 M hydrogen peroxide were placed onto the plates (Disc Diffusion Method).

#### 7.2.5 Statistical analysis

All bacterial counts were log10 transformed prior to statistical analysis. The dataset was analysed using R software (version 2.9.1). Statistical differences between isolates, incubation atmosphere, time and temperature were undertaken by ANOVA, with P = 0.05 used as the statistical threshold for significance for only the survival experiment in BHI due to large amount of data generated from the experiment. The appropriate analysis of the data set as provided was to determine how the treatments were allotted and therefore linked to the observed values. The different treatment combinations were allotted to different number of plates and replicates. Measurements were therefore taken on plates and/or replicates in an unbalanced manner for the model that incorporated both fixed and random effects. The linear model was expressed graphically, and the slope of these lines (for each temperature and the individual isolates) were identified.

# 7.3 Results and discussion

#### 7.3.1 Exposure to oxygen in broth.

The results from exposure to oxygen in BHI from the duplicate plates at all temperatures (4°C, 10°C and 20°C) are shown in Figure 7.1 - 7.5, where Figure 7.1 is for the human isolates and Figure 7.2 for the poultry isolates. All constructed linear model slopes for each data set are shown in Table 7.1. The results from the ANOVA analysis using R software is shown in Table 7.2. At every temperature, the poultry isolate survived longer than the human isolates. The human isolates did not show differential survival between the atmospheres, suggesting they were less sensitive to oxygen than the poultry isolate.



Figure 7-1 The linear model fitted to the observed data for survival of human *Campylobacter jejuni* isolate (474) in broth stored at three different temperatures (green colour for 4°C, black for 10°C and red for 20°C) and two different atmospheres (the upper line of the two lines for microaerobic and the lower line for aerobic).



Figure 7-2 The linear model fit to the observed data for survival of poultry *Campylobacter jejuni* isolate (474) in broth stored at three different temperatures and two different atmospheres.



Figure 7-3 Survival of *C. jejuni* isolates at 4°C (H474 refers to human isolates 474, P474 refers to poultry isolates 474, A aerobic incubation, M refers to microaerobic AR/MR to aerobic/ microaerobic replicate data).



Figure 7-4 Survival of C. jejuni isolates at 10°C



Figure 7-5 Survival of *C. jejuni* isolates at 20°C

Temperature (°C)	Human	Poultry
4	-0.20336	-0.15033
10	-0.46046	-0.40743
20	-0.7338	-0.68077

Table 7-1 Slope values for the linear model shown in Figure 7.1 and Figure 7.2

Table 7-2 Results of two-way analysis of variance for effects of storage atmosphere, isolates, and temperature on survival of *C.jejuni* in broth.

				p-
	numDF	denDF	F-value	value
(Intercept)	1	228	20015	<.0001
Isolates	1	54	160	<.0001
Atmosphere	1	54	2	0.1638
Temperature	2	54	40	<.0001
Time	1	228	1411	<.0001
Isolates:Atmosphere	1	54	3	0.1145
Temperature:Time	2	228	219	<.0001
Isolates:Time	1	228	14	0.0002

# 7.3.2 Exposure to oxygen on agar plates.

The results from exposure to air in agar plates confirmed that survival was significantly longer at low temperatures compared to high. However, on agar, both poultry and human isolates survived longer microaerobically than aerobically. The data are summarised in Table 7.3. There was no significant difference between the survival of the poultry and human isolates when differences in the initial concentrations are accounted for.

Table 7-3 Effect of temperature and storage atmosphere conditions on survival (in days) of selected New Zealand *C. jejuni* strains on agar plates

Isolates/strain	At 4 °C		At 25 °C		
	Aerobic	Microaerobic	Aerobic	Microaerobic	
ST-474 <sub>H</sub>	21	>32*	6	7	
ST-474 <sub>P</sub>	18	23	4	7	
ST-190 <sub>H</sub>	22	>32	5	6	

\* The experiment was ended at 32 days. (ST-474<sub>H</sub> refers to human isolate and ST474<sub>P</sub> refers to poultry isolate)

#### 7.3.3 Exposure to $H_2O_2$ .

The exposure to  $H_2O_2$  revealed that all the selected NZ *C. jejuni* strains (ST-474, ST-48, and ST-190) tested were sensitive to all the concentrations tested, except for the lowest concentration of 1 mM (Figure 7.6).



Figure 7-6 Susceptibility of C. *jejuni* isolates to different hydrogen peroxide concentrations in filter discs on MH agar.

Most of the data reported and the *Campylobacter* survival studies in the ComBase database for predictive microbiology were either for a short period exposure to a specified temperature or investigations of more than two factors (temperature, atmosphere) targeted by this study, such as the addition of NaCl, CO<sub>2</sub> and other stress factors. Therefore, the comparison of our results to the international data which used similar experimental conditions used in my study is limited. The results obtained in this study are in agreement with the findings of (Garenaux et al., 2008) especially at 25°C and at 4°C for the first-week data and also with those of (Chan et al., 2001, González et al., 2009) at 4°C. The results obtained in this study at 4°C are somewhat in agreement with the results of Yoon (2004) in broth at the same temperature aerobically in terms of the first reduction which occurred after nine days of incubation compared to eleven days in Yoon (2004) study. This marginal difference of two days is possibly due to the agitation employed in this study which was not practised in Yoon's study. It is well known that the agitation creates strictly aerobic conditions
which may enhance the pathogen death rate (Meredith et al., 2014, Butzler, 2014). Yoon's study was planned to simulate semi aerobic conditions or mixed aerobic and anaerobic conditions. Moreover, this study results are in agreement with a NZ study for chicken (Chynoweth et al., 1998) although the chicken meat used by Chynoweth et al. (1998) as a substrate or a medium may have influenced the results to some extent of that study. In fact, NZ strains survive less than the international strains, especially in within the 20 - 25°C range. Blankenship and Craven (1982) reported that survival of sterilised ground chicken at 23°C was for > three weeks whereas NZ strains investigated in my study survived aerobically at 25°C for only four to five days in the agar plates and up to 7 days at the same temperature in microaerobic environments. This may be due to the different strain and different matrices used in the two studies.

In general, there are not many differences in  $H_2O_2$  sensitivity between the poultry and human isolates of each strain. These results are in agreement with a study published internationally with regards to the diameter of the inhibition zone (Fields and Thompson, 2008) Despite a report that *C. jejuni* has the good adaptive ability to  $H_2O_2$  stress (van Vliet et al., 2002), this study results confirmed that the tested NZ *C. jejuni* strains were sensitive to  $H_2O_2$ .

The investigation of the NZ strains tested in this study by three methods revealed that the air tolerance of the NZ strains was not different to that of the international strains (Chan et al., 2001, Garenaux et al., 2005, ICMSF, 1996) and the previously tested NZ strains (Chynoweth et al., 1998). The results of this study did not support the hypothesis that the *C. jejuni* survival was not influenced by the storage atmosphere aerobically or anaerobically, as the survival of *C. jejuni* was shorter in the agar and the broth experiments under aerobic conditions than when subjected to microaerobic conditions at all temperatures tested, except in the broth experiment at 10°C. After the fifth day, the survival rate was more or less similar at aerobic or anaerobic conditions. This is consistent with other reports regarding the growth and adaptation of *C. jejuni* to aerobic metabolism (Oh et al., 2015). However, the best known and fundamental hypothesis is that *Campylobacter* is a microaerophilic microorganism that requires a specific gas composition for its growth and survival (Butzler, 2014, Kelly, 2008, Gharst et al., 2013, ICMSF, 1996).

It is not yet completely understood the biological aspects and mechanisms responsible for microaerobic growth or better survival at a molecular level and the processes that occur at the physiological and genetic level. It is assumed that due to the inhibition of enzymes by a higher concentration of  $O_2$  (more than the maximum level 10%), the vulnerability to reactive oxygen species (ROS) and /or metabolic generation of free radicals are all capable of reacting with molecular components of cells to disturb their function in addition to producing more toxic components (Krieg and Hoffman, 1986, Kaakoush et al., 2009). The poultry isolates survived

longer than the human isolates especially at the incubation temperatures of 10° and 20°C, but there was no significant difference between the survival of the poultry and human isolates when differences in the initial concentrations were accounted for at the incubation temperature 4°C.

It is revealed from this study that the survival of C. jejuni from oxidative stress is more influenced by the temperature rather than the incubation atmosphere. At a low temperature of 4°C, there was less effect on C.jejuni survival and was able to survive for more than four weeks in both broth and agar experiments. This was assumed by Gareneux (2008) and Hazelegar (1998) to be due to a less active metabolism and a decrease in catalytic activity, or oxygen being less toxic for cells at 4°C. Despite the fact that this study was conducted with chicken meat, the combination of three methods to address the oxidative stress on C. jejuni increased the validity of the results reported by this study and saved time, resources and cost. Practically, international data on the survival of C. jejuni have been reported mostly for short periods and for above freezing temperatures for two days (Solow et al., 2003, Zhou et al., 2011) or for one to two weeks (Bhaduri and Cottrell, 2004, Garenaux et al., 2008, Davis and Conner, 2007, Oyarzabal et al., 2010). Moreover, the results obtained by this study are in agreement with the results reported in NZ with chicken mince and chicken nuggets (Chynoweth et al., 1998). A few studies (Byrd et al., 2011, Meredith et al., 2014) evaluated the effects of modified storage atmosphere and chilling on the survival of naturally occurring Campylobacter in raw poultry. For Byrd et al. study (Byrd et al., 2011) the modified packaging atmosphere applied for the stored chicken (n = 16) at 2 °C were 100% O<sub>2</sub>, normal air, 85% N<sub>2</sub>+10% CO<sub>2</sub>+ 5% O<sub>2</sub> and 100% CO<sub>2</sub>. Campylobacter was not detected after 14 days of storage in 13 out of 16 chicken samples stored with 100% O2(Byrd et al., 2011). Similarly, Campylobacter was detected in the 12 out of 16 chicken exposed to normal air and in 9 out of 16 chicken samples treated with 85% N<sub>2</sub>+10% CO<sub>2</sub>+ 5% O<sub>2</sub>.(Byrd et al., 2011). But, only in 7 out of 16 chicken samples, the *Campylobacter* was detected in chicken treated with 100% CO<sub>2</sub> (Byrd et al., 2011). Therefore, the comparison and discussion of this study are limited because there is only a limited amount of international data which had used similar experimental conditions to this study. Similarly, the results obtained by Garenaux et al. (Garenaux et al., 2008) using the Colombia agar plates were similar to the results of (Bhaduri and Cottrell, 2004) at 4°C using chicken mince or chicken skin. Both studies reported that at 4°C Campylobacter J. survived for one week. In fact, the implications of the spoilage of chicken meat samples or chicken skin samples during storage of the samples above 4° C hinders the use of a food matrix in long survival studies. Moreover, the use of irradiated skin or chicken meat samples eradicates the microflora (pseudomonads, micrococci, Staphylococci), which are naturally found in poultry and inhibits the growth of Campylobacter (Mai, 2003). The conclusion from such studies is that irradiated chicken meat must be treated with

caution due to the absence of microflora. Thus, the results of this study rejected the hypothesis investigated in this study, that the high rates of campylobacteriosis in NZ may be due to the emergence of unusual new strains with more oxygen tolerance, or that they have a unique survival ability at varied storage or handling temperatures (4, 10, 20, and 25 °C). Given the infectious dose of illness is about 500 cells (Robinson, 1981) and the length of time that *Campylobacter* can survive as revealed by this study for NZ strains which is more than four weeks This is a significant finding from a health perspective as the shelf life of a fresh chicken is eight days at 4°C. Thus, the contamination level of fresh poultry products will persist without any significant reduction until it reaches the kitchen of a consumer, and this poses a risk to the consumers. Poultry processing plants should apply the necessary intervention to ensure that chicken carcasses and poultry products are released to the retail market with as low a contamination as practically possible, and theoretically, the counts should not exceed 500 cells per serving of chicken portion.

The linear model constructed by this study is acceptable given the empirical nature of the Weibull model with its two parameters (shape  $\eta$  and scale  $\delta$ ), and the Weibull model, with a fixed parameter shape model, is only as good a fit as the log-linear model and only partially succeeded in the prediction of the tested *Campylobacter* data (Oyarzabal et al., 2010, Al-Sakkaf and Jones, 2012).

# 7.4 Conclusion

The results obtained from all the oxidative stress conditions are similar and do not indicate that the NZ strains differ in oxygen tolerance when compared with most of the other internationally reported strains at the investigated temperatures. At 20 or 25°C, NZ selected strains survived in media only up to one week aerobically, but at 4°C, they survived aerobically for more than three weeks. New Zealand higher rate of reported campylobacteriosis compared to other developed countries is possibly due to other factors, and these are explored in the next chapter(s).

Chapters 8 and 9 Embargoed until 31 December 2022

# 

# 10 Quantitative microbial risk assessment for campylobacteriosis in New Zealand by the Bayesian approach

# **10.1 Introduction**

For more than a decade, QMRA has been an important tool for food safety strategic planning and control. It has been promoted by international organisations and national organisations worldwide such as WHO, FAO, EFSA, USDA, FDA, WTO, and Codex Alimentarius. QMRA provides a unified approach to a public health problem. QMRA not only predicts the consequences of the disease acquired by the consumption of contaminated food but also provides a prior assessment of any possible intervention in the food chain. However, QMRA predictions of human health cases have always been higher than the number of actual reported cases and the estimates of epidemiological studies. This is due to the uncertainties of estimating QMRA parameters. For example, the lack of a reliable dose-response model, failure to account for the variability of immunity within the population, and the estimation of prevalence/concentration of pathogens through all the steps or pathways of the food chain, all of which influence the accuracy of the QMRA outcomes (2008). Many QMRAs were conducted using the Monte Carlo (MC) approach. Many iterations are required to obtain reliable and accurate results from the MC method. This can be a time-consuming process. However, simulation of these models by the Monte Carlo approach is relatively easy to implement using commercial software to automate the tasks and can produce acceptable results (Robert and Casella, 2004). Another drawback of the MC approach is that this technique performs poorly when modelling rare events (Vose, 2008).

Bayesian belief networks, approaches and inference have gained increased interest from researchers in recent years to calculate the outcomes of QMRA. Bayesian inference has many advantages such as being able to accommodate missing data, accounting for the uncertainty in all estimated parameters, enabling the combination of data with the experts' prior belief or knowledge. The Bayesian approaches allow for the backpropagation of prior variables with the observed data, assist in understanding underlying relationships between variables, and provide a method for avoiding overfitting of data (Smid et al., 2010). The Bayesian approaches offer satisfactory prediction accuracy (Smid et al., 2010). The Bayesian inference prediction is consistent with the surveillance data, where MC techniques fail to predict the actual notified campylobacteriosis cases correctly. Several authors have reviewed the literature regarding the published QMRA models for *Campylobacter* (Nauta et al., 2009, Nauta and Christensen, 2011, Chapman et al., 2016). The NZ

QMRA model prediction was about ten times more than the actual notification value. Therefore, the application of the Bayesian approach and inference is valuable for NZ, since the NZ QMRA model (Lake et al., 2007) used MC simulation which prevents the inclusion of surveillance data necessary to evaluate the benefit of any future intervention in the food chain in terms of the change in the number of illnesses. The main drawback of the NZ adapted model is that it relies heavily on the data of international literature, which does not reflect the actual NZ scenario in poultry processing and also does not take into consideration the implemented interventions by the industry and the government.

Moreover, the challenges faced in general by the traditional QMRA approach used in the NZ model have probably contributed to the current model's inaccuracy (Havelaar et al., 2008, Smid et al., 2010). The objective of this study was to conduct QMRA by using the Bayesian Belief Network and inference approach in spite of the complexity involved in constructing and assigning probabilities from expert knowledge or prior information, linking the nodes, discretising the data, complex software and solving the resultant joint probability distributions. Hence, the priority will be to assess, illustrate and present the Bayesian approach for tackling the problem of campylobacteriosis in NZ. A simplified model representing the entire food chain from the 'farm to fork' was used.

# 10.2 Methods

#### 10.2.1 The modelling approaches

The model developed by (Albert et al., 2008) in France for *Campylobacter* was adopted. The construction of the model and risk estimates are based on the classical Bayes' theorem, which can be presented in the case of two events as:

$$P(A | B) = \frac{P(B|A)P(A)}{P(B)}$$
 10-1

Where P(A|B) is the posterior probability (updated knowledge) of event A (unknown parameter), given that event B happens (observed data).

P(B|A) is the conditional probability of B given A, and it is the likelihood function for event A for a realised event B.

P(A) is the prior probability (prior knowledge) of event A indicating uncertainty about A before knowledge about B (data at hand).

P(B) is the normalizing factor ensuring the posterior probabilities sum to 1, and it is also the marginal probability of B. The posterior distribution P(A|B) is largely important as it expresses a

belief about A after obtaining data. It provides a solution to the problem of how our prior knowledge or estimate about an unknown parameter is improved using data.

Equation (10-1) can also be written as:

Posterior distribution  $= \frac{\text{likelihood x prior distribution}}{\sum \text{likelihood x prior distribution}}$  10-2

The objective of the Bayesian inference is to assess the total uncertainty based on all the available data and prior knowledge if it exists. The interpretation achieved by Bayesian inference can balance the model by completing gaps in the data. The Bayesian Belief Network model, as developed by Albert et al. (2008), consists of six sub-models, as shown in Figure 10.1. These are the chicken farm model, broiler production model, hygiene model, consumption model, exposure model and the illness model. Each sub-model is characterised by its variates, parent variates, complementary co-variates and parameters which are listed in Table 10.1. It is important to specify the joint distribution of the model using conditional independencies between some model parameters in constructing the typical Bayesian Network acyclic graph. Then the marginal distributions (priors) of the variates, complementary covariates and parameters must also be specified. Finally, the variates of interest are estimated from their posterior distributions after introducing the data to the model.



Figure 10-1 The complete model and the augmented model

adapted from Albert et al. 2008 with the variates (circled), parameters (triangles), variables and complementary variables expressed conditionally by their parent variates and linked by arrows (see Table 10.1 for the details)

#### 10.2.2 Data sources

All of the data incorporated in this study were extracted from the scientific literature published in peer-reviewed journals for the priors rather than the exclusive expert opinions which include subjectivity and bias that may impact the validity of the QMRA and consequently affect the decisions made by the risk managers or policymakers based on the QMRA results. Reliance purely on expert opinion was avoided where possible or reduced as much as possible. Mostly NZ based studies were included for the priors, and international studies were included when the NZ data was either lacking or was of poor quality. However, NZ surveillance data or data from the daily microbiological monitoring programme, such as the National Microbiological Database (NMD) were used to improve the priors used in the core model. The parameters of interest were then conditionally estimated from their posterior distributions by the newly incorporated observed data (NMD, or surveillance data). The incorporation of the new data may require additional modelling to link them with the core model.

Class	Variate	Parent (s)	Distribution/Relationship
Vi	p <sub>f</sub>	m <sub>f</sub> ,s <sub>f</sub>	$logit (p_f) \sim N (m_f, s f)$
Vi	p <sub>b</sub>	$m_b, s_b$	logit ( $p_b$ ) ~ N ( $m_b$ , s $_b$ )
Vi	p <sub>h</sub>	$p_{hc}$ , $p_{hh}$	$= p_{hc} * p_{hh}$
Vi	$\lambda_{c}$	-	Gamma(8, 2)
Vi	pey	pe	$=1 - (1 - p_e)^{13}$
Vi	p <sub>ib</sub>	pey	$= p_{ey} * p_{ie}$
Vi	p <sub>it</sub>	$p_{ey}$ , $p_{ib}$ , $p_{iq}$	$= p_{ib} / (1 - (1 - p_{iq})*(1 - p_{ey}))$
Cv	m <sub>b</sub>	m <sub>f</sub> , d <sub>bf</sub>	$= m_{\rm f} + d_{ m bf}$
Cv	$\lambda_{\rm e}$	$p_b, \lambda_c, p_h$	$= p_b^* \lambda_c^* p_h$
Cv	pe	$\lambda_{e}$	$= 1 - \exp(-\lambda_{e})$
Cv	$p_{ie}$	d,p <sub>ne</sub> ,P <sub>in</sub>	$= 1 - (1 - p_e)^d) * p_{in}$
Ра	m <sub>f</sub>	_	$\sim N(0,0.22), (0.511,20.66)$
Ра	Sf	_	~ Uniform (0,0.2)
Ра	$d_{\rm bf}$	_	~ N (0.1,0.0696)>0, (0.15,206.6)I(0,)
Ра	Sb	_	~ Uniform (0,0.2)
Ра	Phc	_	~ Beta (8,8)
Ра	$p_{\rm hh}$	_	~ Beta (22,20) & ~ Beta (166,345)
Ра	d	_	$\sim$ (1,2,10,100,300)with p =(0.5, 0.163, 0.222, 0.097, 0.018)
Ра	p <sub>ne</sub>	_	~ Beta (0.024, 0.011)
Ра	P <sub>in</sub>	_	=0.33
Ра	p <sub>iq</sub>	_	Beta (39,12)
Da	g <sub>fs</sub>	p <sub>fs</sub>	~ Binomial $(n_{fs}, p_{fs})$
Da	g <sub>bs</sub>	P <sub>bs</sub>	~ Binomial $(n_{bs}, p_{bs})$
Da	n <sub>csb</sub>	$\lambda_{csb}$	Poisson ( $\lambda_{csb}$ )
Da	gits	Pits	~ Binomial $(n_{its}, p_{its})$
Cv	p <sub>fs</sub>	m <sub>f</sub> , s <sub>f</sub>	logit ( $p_{fs}$ ) ~ N ( $m_f$ , $s_f$ )
Cv	p <sub>bs</sub>	m <sub>b,</sub> s <sub>b</sub>	$logit (p_{bs}) \sim N (m_{b}, s_{b})$
cv	λ <sub>csb</sub>	$\lambda_{cs}$	$= n_{cs} * \lambda_c$
cv	Pits	p <sub>it</sub> , s <sub>s</sub>	logit ( $p_{its}$ ) ~ N (logit ( $p_{it}$ ), $s_s$ )
ра	Ss	-	=1
со	n <sub>fs</sub>	-	188,150,148,195,
со	n <sub>bs</sub>	-	75,60,60,72,60,
со	n <sub>cs</sub>	-	21037900
со	n <sub>its</sub>	-	4292350

Table 10-1 Description of most variates (vi), parameters (pa), covariates (cv), data (da) and constant(co) used in the models depicted on Figure10.1

The new model with better data is called the augmented model, and its variates are described in Table 10.1 above the core model variates. Most of the model and sub-model details are summarised below. The priors or the scientific literature data are explained first before the surveillance, or the more reliable data are given in each model. More details about the model code and the calculations are also provided in Appendix E or by the cited literature.

# Farm model

The prior for the probability of a bird in a chicken farm being contaminated ( $p_f$ ) is expressed as a random variable with the logit normal distribution function centred at  $m_f$  with a value of 0.5 as an average for the values reported by Boxall (2005) where it was indicated that farm prevalence was 25%. Once any flock is discovered positive for being colonised with *C. jejuni*, the flock prevalence increases to 77%. The 0.5 value is also in agreement with the international value used by Albert et al. (2008). The data used to update the above prior come from the actual *Campylobacter* prevalence. This was obtained from the caecal sampling programme set up by the NZFSA under the *Campylobacter* in Poultry Risk Management strategy 2006-2009 for NZ broiler flocks at each batch in order to determine the prevalence of *Campylobacter*. A few examples of the data set are summarised in Table 10.2. The Poultry Industry Association of NZ (PIANZ) kindly provided the data.

	Plant A			Plant B			
Month	No.samples	detected	%	No.		%	
	collected		positive	sample	detected	positive	
				collected			
Mar-08	188	133	71	181	66	36	
Apr-08	150	108	72	160	140	88	
May-08	148	102	69	172	106	62	
Jun-08	195	128	66	229	133	58	
Jul-08	170	119	70	141	107	76	
Aug-08	309	131	42	168	114	68	
Sep-08	211	142	67	181	121	67	
Oct-08	144	107	74	176	119	68	
Nov-08	145	102	70	1596	142	89	
Dec-08	158	117	74	211	154	73	
Jan-09	107	95	89	163	126	77	
Feb-09	153	117	76	157	137	87	
Mar-09	150	125	83	140	132	94	
Apr-09	181	148	82	176	153	87	
May-09	160	122	76	146	98	67	
Jun-09	204	137	67	151	83	55	
Jul-09	57	20	35	67	17	25	
Aug-09	52	12	23	61	15	25	
Sep-09	62	17	27	61	20	33	
Oct-09	84	34	40	70	19	27	
Nov-09	87	24	28	64	16	25	
Dec-09	40	31	78	65	27	42	
Jan-10	83	50	60	60	23	38	
Feb-10	54	50	93	66	34	52	
Mar-10	84	63	75	75	55	73	
Ave.			64			60	

Table 10-2 Prevalence of *Campylobacter* in New Zealand birds.

#### Poultry plant model

Similarly, the probability of a chicken carcase being contaminated (pb) is expressed as a logit distribution function ( $p_b$ ) centred at  $m_b$  with a value of 0.5 with the limit  $m_f < m_b$ . It is possible that the carcasses become more contaminated during transport and slaughtering. This additional contamination was represented as a portion  $(d_{bf})$  in the model structure (Figure 10.1) and expressed by a truncated normal distribution  $(0.1, 0.0696)I(0, \infty)$  from expert opinion used by the Albert et al. (2008) Chrystal et al. (2008) reported that 45% of 163 whole chicken samples collected from the retail market in NZ were positive. However, in a relatively old but large survey conducted in 2003-2004, 89% of 230 chicken meat samples were found to be positive for *Campylobacter* (McIntyre et al., 2010). The survey by Chrystal et al. (2008) was probably conducted after the intervention applied by the poultry industry, and this is the reason for it being included in this study. Mullner et al. (2009a) reported that 80% of the samples were positive (454 out of total 652 chicken samples) over the three years from 2005 to 2008, but that study did not report the prevalence each year to better visualise the change in the prevalence over a 3-year period of the chicken survey. Therefore, the values reported by Mullner et al. (2009a) and McIntyre et al. (2010) could be tested for sensitivity analysis to test their effect on the global model output. The actual data of the prevalence of Campylobacter in chicken carcases were obtained from the whole chicken carcases sampling programme for NZ broiler. This programme aimed to survey the prevalence of *Campylobacter* and the contamination level (cfu) and to monitor compliance with the Campylobacter Performance Target (CPT) which was set up by NZFSA under the National Microbiological Database programme in February 2007. Therefore, all the microbial data for two major plants in NZ were included in this study, and a few examples of the data set are summarised in Table 10.3. The PIANZ kindly provided the data.

#### Hygiene model

International studies have shown that undercooking plays a negligible role in campylobacteriosis, and cross-contamination is the main reason for the disease. The consumer studies (van Asselt et al., 2008, van Asselt et al., 2009) confirmed that mishandling practices by the consumers lead to cross-contamination during food preparation in the kitchen.

Nauta and Christensen (2011) compared eight different consumer phase models in QMRAs. Most of them expressed the consumer mishandling practices and the bacterial transfer rate from contaminated raw chicken to equipment, hands and ready-to-eat or cooked food separately. International quantitative data for mishandling practices are scarce and are usually expressed as distributions derived from limited consumer observational studies or consumer self-reported surveys, where actual consumer mishandling practice may be far worse than the reported practice.

A NZ study (2007) has indicated that 28-41% of New Zealanders allow cross-contamination to occur in their kitchen. However, the same study reported that 61% of consumers do not wash their hands before food preparation, 78% do not wash their hands between food preparation steps, and 95% do not wash their hands after food preparation. Moreover, 73.6% do not wash their hands correctly (using hot water, soap, and drying them). Thus, different priors for poor home hygiene (P<sub>hh</sub>) could be tested in order for the model to reflect the available NZ data with beta distributions, as the parameters of the beta distribution are chosen to reflect the plausible range of parameter values; (166,345 for 28-41%) (mean=0.77, CI 95%=[0.64,0.87]), (22,20 for 28-74%). In spite of the possibility of measuring the bacterial transfer rate to food or utensils and vice-versa, there is a significant discrepancy in the values of transfer rates reported in the literature. These values have been discussed briefly previously in this thesis and by others (Gilbert et al., 2006). The studies by Luber et al. (2006), Kusumaningrum et al. (2004) and van Asselt et al. (2008) are considered to be the main studies that have addressed *Campylobacter* solely. However, different transfer rates are reported in each study. Thus, similar to poor home hygiene practice, different priors could be tested for the model to reflect the available data for the cross transfer rate (Phc) representing the least reported transfer rate value of 5% and the possible highest value of 100%. However, the reported transfer rates (3-63%; Beta 8,8) by Luber et al. (2006) have been used extensively (Albert et al., 2008), as the naturally contaminated chicken was used in that study and therefore reflects the actual consumer preparation scenario. Use of artificially inoculated bacteria or high inoculums in other studies disqualified them from inclusion in this study (Montville and Schaffner, 2003, Purnell et al., 2004, Whyte et al., 2003). Therefore, Luber et al. (2006) transfer rate value was selected to be used in this model. A NZ primary transfer study (Gilbert et al., 2006) reported a low transfer rate values from 0 - 6% only, but the actual study was not later conducted, and so the findings of the primary study were not confirmed. Thus, NZ transfer data has been excluded for use in this study. The total probability of cross-contamination from a contaminated chicken in the home  $(P_h)$  is expressed as the product of Phc and Phh, as has been expressed in other international models (Albert et al., 2008, Lindqvist and Lindblad, 2008).

		Plant A			Plant B		
	Month	No.samples	No.	% positive	No.samples	detected	%
		collected	detected		collected		positive
1	Mar-08	75	34	55	n.a	n.a	n.a
2	Apr-08	60	31	48	n.a	n.a	n.a
3	May-08	60	37	38	n.a	n.a	n.a
4	Jun-08	72	31	57	n.a	n.a	n.a
5	Jul-08	60	32	47	n.a	n.a	n.a
6	Aug-08	60	26	57	n.a	n.a	n.a
7	Sep-08	60	35	42	60	30	50
8	Oct-08	57	31	46	54	29	46
9	Nov-08	60	32	47	60	23	62
10	Dec-08	72	45	38	69	29	58
11	Jan-09	60	34	43	42	15	.64
12	Feb-09	57	26	54	57	27	.53
13	Mar-09	57	34	40	60	24	60
14	Apr-09	72	52	28	72	32	56
15	May-09	60	31	48	60	36	40
16	Jun-09	72	41	43	72	39	46
17	Jul-09	60	37	38	60	34	43
18	Aug-09	60	36	40	60 46		23
19	Sep-09	60	37	38	57	39	32
20	Oct-09	60	46	23	72	47	35
21	Nov-09	72	60	17	60	40	33
22	Dec-09	57	36	37	75 55		27
23	Jan-10	72	43	40	60 36		40
24	Feb-10	60	27	.55	55	39	29
25	Mar-10	72	21	71	75	32	57
	Average			44			45

Table 10-3 Prevalence of Campylobacter in New Zealand whole chickens.

#### Consumption model

Consumption data play an important role in any QMRA. However, poultry consumption surveys are insufficient in NZ. Moreover, the available data were not designed to be used or incorporated in a QMRA. The main parameters, such as, how frequently New Zealanders purchase chicken or how many chickens they purchase each time over a defined period of time, were scarce. Therefore, an alternative procedure was followed in order to overcome this data gap by expressing the purchase of chicken by New Zealanders as a Poisson random variable with intensity  $\lambda_c$ . In order to consider the variability of  $\lambda_c$  over the NZ population, it was modelled by a gamma probability distribution (8,2) (CI 95%=1.7,7.2) which has its mean at four chicken per month and a standard deviation of sqrt (2) = 1.414214. Consumption data were collected from the PIANZ as the total number of annual birds slaughtered in NZ (Figure 10. 2). The average value of the total annual birds slaughtered (79, 821 000 birds) for the two years 2008 and 2009 was used since the only microbial data available for this study also covered two years, 2008 and 2009, and the average value for both years was 1,609,500 households. The number of chickens consumed per year was divided by 13 to express the consumption over a four-week period.



Figure 10-2 New Zealand chicken production data provided by the PIANZ. (Poultry Industry Association of NZ).

#### Exposure model

For assessing the risk to human health from *Campylobacter* at purchase and preparation of chicken in the home, the potential of exposure to the pathogen during a specific period of the year or the number of times per year that chicken was brought home needs to be estimated. It was suggested to express the number of times the household was exposed to chicken by a random binomial variable (X,p) as the binomial process describing a repeated trial (X) with a constant number or probability of success p (Albert et al., 2008). The probability of success p was modelled as a product of the probability that the chicken brought home is contaminated ( $p_b$ ) times the total probability of cross-contamination occurring from a contaminated chicken in the home ( $P_h$ ).

This can be mathematically expressed as follows:

 $\lambda_e = \lambda_c * P_b * P_h$  10-3

Where  $\lambda_e$  represents the mean value of an exposure event.

The exponential distribution was used in the literature to express the probability of exposure to a hazard in order to represent the uncertainty and variability of the exposure event (Vose, 2008, Albert et al., 2008). Thus, the probability of exposure for the whole exposure event for New Zealanders over four weeks during a year is as follows:

$$P_e = 1 - \exp(-\lambda_e)$$
 10-4

Then, the probability of exposure for a home in one year is as:

$$P_{ey} = 1 - (1 - P_e)^{13}$$
 10-5

### Illness model

In the last sub-model, it is necessary to predict the probability of illness as a consequence of poultry consumption or purchase in a year. Conventionally, this model is built using the combination of the exposure model and a dose-response model. However, the exposure model and all the preceding models through the food chain were without the inclusion of bacterial counts due to the restrictions of the available data in NZ. All the farm samples were categorised into only positive/negative results, and poultry plant results were reported unfortunately in grouping methodology ((number of samples less than 1200/ 6000/10000/100000/750000 cfu per carcasses with missed values on a few occasions). Thus, the predicted ingested dose, which is the most uncertain variable in any QMRA, was represented in this study by a distribution derived from Section four of a CARMA report (Table 4.2.) (Nauta et al., 2005). Although there are questions about Table 4.2, the CARMA report was published in a peer-reviewed journal and satisfied the minimum criteria for the inclusion of the data in this study model. The values published in the CARMA report have also been used by Albert et al. (2008) in their model. It was considered by Albert et al. (2008) as the best information available to them as well. Due to the restriction of time and the lack of data for our knowledge, the CARMA report data was the only available option for use in this study as well. However, it is important to point out our reservations and concerns about the values reported in the CARMA report, and that it is considered as an area of investigation and a priority for improving the model in the near future.

An estimate of the probability of illness, given the exposure to *Campylobacter* (P<sub>ic</sub>), was derived from the simple binomial equation for the probability of infection as indicated in WHO guidelines (FAO/WHO, 2003) as follows:

$$P_{ie} = 1 - (1 - p_{ne})^d$$
 10-6

Where  $p_{ne}$  represents the probability that a single pathogenic organism can cause the infection. A Beta (0.024,0.011) distribution has been used by Albert et al. (2008) to express this probability with  $\alpha$  and  $\beta$  values estimated by Teunis et al. (2005). d is the dose or the number of pathogenic organisms in the exposure event and is expressed by the distribution based on the CARMA report (Table 4.2)

The multiplication of P(inf/d) by the probability of illness given the infection  $P_{in}$  (0.33) which has been derived from the only human dose-response study in the literature, (Black et al., 1988) estimates the final probability of illness given the exposure (Albert et al., 2008) It was reported by (Black et al., 1988) that only a third of the volunteers of his study became ill after they had ingested the pathogenic Campylobacter cells. As the surveillance data indicates the total reported cases of campylobacteriosis, but with no confirmation about the number of cases attributed to poultry consumption, the model has to consider how to incorporate the total reported cases in the model while only tracking the poultry source. The equation (11.8) which has been used to account for the flaw in surveillance data regarding the number of notified cases due to poultry was the same equation used by (Albert et al., 2008). That equation needs the relative contribution of poultry and non-food sources for acquiring campylobacteriosis in NZ. The determination of what proportion of human campylobacteriosis are acquired from poultry  $(P_{ib})$  has been obtained from a NZ study (Mullner et al., 2009a) which estimated it as 80%. The other sources which can contribute to campylobacteriosis were calculated in this study in order to provide decision-makers with valuable information about the total probability of illness from campylobacteriosis within one year, whatever the source of infection (Pit), to estimate the total possible campylobacteriosis cases in one year and to prioritise targeted interventions. Usually, source attribution is extrapolated from surveillance, outbreaks and epidemiological studies. New molecular biology tools such as multilocus sequence typing (MLST), polymerase chain reaction (PCR) assays and the DNA microarray have been used to provide the relative importance of each source in acquiring human diseases (Bumgarner, 2013). Mullner et al. (2009a) in a molecular-based study, reported that 80% (379 out of 474 cases) of human cases in NZ were attributed to poultry, which is translated into

Beta (39,12) (mean=0.77, CI 95%=[0.64,0.87]). Taking into account all of the literature, this is a high population attributable fraction, and it is only close to the value reported in Scotland (Sheppard et al., 2009). An expression used by Greenland and Drescher (1993) was used to define the attributable fraction ( $p_{iq}$ );

$$P_{iq} = \frac{\Pr(\text{disease}) - \Pr(\text{disease}|\text{no exposure})}{\Pr(\text{disease})}$$
10-7

After the substitution and rearrangement, the final total probability of illness was calculated as follows (Albert et al., 2008):

$$P_{it} = \frac{p.ey * p.ie}{(1 - (1 - p.iq)*(1 - p.ey))}$$
 10-8

The actual reported number of campylobacteriosis cases for the years 2008 and 2009 were obtained from annual surveillance reports prepared by the Institute of Environmental Science and Research Limited (ESR) (NZPHO, 2010). There was no significant difference between the reported cases for the years 2008 and 2009, 6693 and 7160, respectively. Thus, the average value (6926) was included in this study.

#### 10.2.3 Computing tools

The software, WinBUGS version 1.4.3 (Lunn et al., 2000) was used for numerical computation. It was developed from the original code by Albert et al. (2008) and ran for 700,000 iterations after a burn-in period of 240,000 iterations using Markov chain Monte Carlo (MCMC) sampling algorithms. The convergence of the algorithms was monitored by using the general methods for assessing the convergence of Markov Chain Monte Carlo iterative simulations (Toft et al., 2007). Three chains have been simulated in parallel, each with different starting values which are overdispersed relative to the posterior distribution. Convergence is assumed when the output from the three chains is similar, and the BGR convergence diagnostic scale reduction factor stays close to 1. The resulting burn-in of 240,000 iterations was discarded, and the remaining iterations summarised to provide the posterior distributions. The error of each posterior distribution standard deviation has to be less than 5% (Tuominen et al., 2006) all the variates or parameters have to behave reasonably, and the results have to be convincing to experts. The coda results produced by the WinBUGS simulations were processed using the statistical R software package to better represent the results of the WinBUGS software.

# 10.3 Results and discussion

The statistical output of the prior distribution and posterior distribution are summarised in Table 10.4. The Monte Carlo error (MC error) for each parameter, which assesses the accuracy of the posterior estimates, represents the difference between the mean of the sampled values (which are used as the estimate of the posterior mean for each parameter) and the true posterior mean, (Huang and McBean, 2007) are shown in Table 10.4. The statistics table also reports the sample standard deviation (SD) with the confidence level at 2.5% and 97.5%. The primary results from running the core model without incorporating the most relevant data, such as the surveillance data or robust data from the daily microbiological monitoring programme conducted by the poultry industry and monitored by the NZFSA were similar to that obtained by the Monte Carlo Simulation method based solely on the prior information available in the core model using WinBUGS. These prior distributions, which are depicted in Figure 10.3 (by dashed lines) can provide valuable information when relevant data is lacking. The incorporation of the augmented model has produced posterior distributions of the variates of interest which have a different shape to the prior distributions, as the data forced the change of the shape of the prior distributions, except for hygiene, farm and plant models, where the differences between both the distribution's shapes were marginal. The spread of the posterior distribution has been narrowed mostly as it was depicted in Figure 10. 3. However, it is important to indicate that this narrowing cannot yet establish a comprehensive response to the question of risk, as the distributions still have a large spread to reflect all the variability and uncertainty of all parameters, variates, co-variates and data used in the developed model. The resultant probability of illness derived from the priors' distributions without incorporating the observed data, was significantly higher than the surveillance data value by 100 times as was the case with the previous model developed in NZ by (Lake et al., 2007). The prediction of campylobacteriosis cases was calculated crudely by multiplying the mean probability of illness from all sources (p.it value shown in Table 10.4 of the results obtained from WinBUGS for each model with the total NZ population to estimate the possible campylobacteriosis cases given the available information from the priors and the most relevant data, respectively in the model. Evidently, the prediction for campylobacteriosis cases for the year 2010, by the Bayesian inference with the data, indicated 9,250 cases from all the sources. This prediction was close to the value reported by the surveillance in NZ in 2010, which was 7333 cases.



Figure 10-3 Posterior marginal kernel densities

(solid lines) and its prior densities (dashed lines) of the six variates defined in Table 10.4. for the main model.



Figure 10-4 Posterior marginal kernel densities

(solid lines) for the main model and posterior marginal kernel densities for the hypothetical model with a 50 % increase in hygiene value priors (dashed lines) of the six variates defined in Table 10.4.



Figure 10-5 Posterior marginal kernel densities

(solid lines) for the main model and posterior densities for the hypothetical model with 25 % increase in the values of the farm and the poultry plant priors (broken lines) of the six variates defined in Table 10.4.

The main model								
node	mean	SD	MC error	2.50%	median	97.50%	start	sample
p.b	0.6139	0.04616	0.00009293	0.5208	0.6149	0.7012	240000	1380000
p.ey	0.9825	0.02462	0.00004153	0.9137	0.991	0.9993	240000	1380000
p.f	0.6109	0.04743	0.00004136	0.5152	0.6119	0.7006	240000	1380000
p.h	0.1593	0.03998	0.00004215	0.08375	0.1587	0.2386	240000	1380000
p.ib	0.002126	0.002101	0.00004517	0.0002515	0.001419	0.007876	240000	1380000
p.it	0.002155	0.002128	0.00004578	0.0002552	0.001438	0.007982	240000	1380000
The model	with a 50 % incr	rease in hygiene	prior			•	•	
p.b	0.6138	0.04598	0.00009261	0.5212	0.6149	0.7009	240000	1380000
p.ey	0.997	0.008878	0.00002981	0.9771	0.9995	1.000	240000	1380000
p.f	0.6108	0.04744	0.00004116	0.5153	0.6119	0.7007	240000	1380000
p.h	0.2619	0.07457	0.00008285	0.129	0.2577	0.4186	240000	1380000
p.ib	0.002899	0.003173	0.00006833	0.0002396	0.001859	0.01119	240000	1380000
p.it	0.002901	0.003175	0.00006838	0.0002398	0.00186	0.01119	240000	1380000
The main n	nodel with priors	s only (without t	he data )	•				
p.b	0.5272	0.06267	0.00005304	0.4035	0.5277	0.6489	240000	1380003
p.ey	0.9568	0.06395	0.00005397	0.7688	0.9822	0.9999	240000	1380003
p.f	0.5001	0.06114	0.00005223	0.3803	0.5001	0.6198	240000	1380003
p.h	0.1594	0.04007	0.00003404	0.08368	0.1587	0.2387	240000	1380003
p.ib	0.1953	0.142	0.0001227	0.0000115	0.284	0.3298	240000	1380003
p.it	0.1972	0.143	0.0001235	0.00001161	0.291	0.3299	240000	1380003
The priors	only for the 50 %	/o increase in hyg	ziene	•				
p.b	0.5272	0.06265	0.00005328	0.4034	0.5277	0.6486	240000	1380003
p.ey	0.9877	0.03112	0.00002609	0.9004	0.9986	1.000	240000	1380003
p.f	0.5	0.06111	0.00005176	0.3803	0.5001	0.6197	240000	1380003
p.h	0.262	0.07467	0.00006367	0.129	0.2577	0.4189	240000	1380003
p.ib	0.2018	0.1457	0.000127	0.00001192	0.3072	0.33	240000	1380003
p.it	0.2024	0.146	0.0001272	0.00001195	0.3099	0.33	240000	1380003
With 25% i	ncrease in pf and	d dbf	•					
p.b	0.7647	0.03586	0.00007092	0.6893	0.7667	0.8295	240000	1380003
p.ey	0.9924	0.01499	0.0000666	0.9538	0.9972	0.9999	240000	1380003
p.f	0.7627	0.0364	0.00003248	0.6862	0.7646	0.8285	240000	1380003
p.h	0.1593	0.04006	0.00006607	0.08336	0.1587	0.2387	240000	1380003
p.ib	0.002416	0.002816	0.00006079	0.0002332	0.001475	0.0101	240000	1380003
p.it	0.00242	0.002821	0.0000609	0.0002337	0.001478	0.01011	240000	1380003
Priors only	of for 25% increa	se in pf						
p.b	0.6578	0.05704	0.00004848	0.5399	0.6601	0.763	240000	1380003
p.ey	0.976	0.0449	0.00003799	0.8434	0.9935	1.000	240000	1380003
p.f	0.6234	0.05745	0.00004864	0.5058	0.6251	0.731	240000	1380003
p.h	0.1593	0.04002	0.00003431	0.08357	0.1587	0.2386	240000	1380003
p.ib	0.1992	0.1443	0.0001223	0.00001173	0.2976	0.33	240000	1380003
p.it	0.2003	0.1449	0.0001228	0.0000118	0.3022	0.33	240000	1380003

Table 10-4 Statistical summary for the most important variates of models used representing the model with and without the use of the data along with possible two plan interventions (+50% in hygiene prior,+25% in farm & plant priors). The main model

Therefore, the prediction produced by Bayesian inference was of acceptable accuracy, given the uncertainty of the surveillance data due to under-ascertainment in the notifiable disease in general. Under-ascertainment can occur as a result of factors associated with a patient who may decide to seek treatment, the general practitioner who may request a sample, and/or the laboratory which analyses the sample. The under-ascertainment of disease and its implication have been discussed in the literature (Doyle et al., 2002) suggesting that the completeness of notification varies between areas, for different diseases, and during outbreaks. Thus, the rate of under-reporting of campylobacteriosis cases in NZ or internationally becomes a complex task. A NZ study (Lake et al., 2009) revealed that only 0.4 % of NZ community cases of acute gastrointestinal illness are notified to national surveillance. However, that study also indicated that their findings could not apply to the notification rate of campylobacteriosis, giardiasis or the norovirus infection, and presented a factor of 7.6 used internationally as an alternative to estimate the possible campylobacteriosis incidences. There were also some reservations concerning this factor.

#### 10.3.1 Sensitivity analysis

Testing the model with the two consumers' hygiene priors is fundamental to determine the constancy of the model outputs as part of the sensitivity analysis to change the different model priors, parameters and variates to evaluate their marginal significance and also to validate the model (Huang and McBean, 2007, Albert et al., 2008). However, only the main factors, such as consumer hygiene practice at home, and the farm, and processer intervention, were investigated in this study. This was due to the significant practical and possible benefits for regulators, policymakers, industry, public health professionals and researchers which may be gained from the outcomes of the indicated model priors by altering their input values. As discussed previously, cross-contamination plays a major role in the risk of campylobacteriosis (Kapperud et al., 2003, Mylius et al., 2007) in NZ Previous studies and next chapter indicate that most New Zealanders scored worse than those in other developed countries (Gilbert et al., 2007, Redmond and Griffith, 2003, Al-Sakkaf, 2012). Therefore, the first model was tested with the lowest value reported by Gilbert et al. (2007) and supported by the findings of the next chapters as a prior for poor hygiene, where 28 - 41% of New Zealanders allowed cross-contamination to occur in their kitchen, which resulted in an unpleasant prediction for the year 2010 as mentioned above. The other higher percentage of New Zealanders, who allow cross-contamination to occur in their kitchen was derived as an aggregate of the 78%of New Zealanders who did not wash their hand before food preparation, and the 74% who do not wash their hands correctly and the 95% who do not wash their hands after food preparation. This is represented by a beta distribution reflecting the 74%, as this value is considered more

reasonable and acceptable. Moreover, the 74% value was approximately 50% higher than the lower reported value. This can serve as a planned target for intervention by policymakers and risk managers and should be considered in their long term strategy to combat the disease in NZ. It is noticed from Table 10.4 that hygiene has a significant impact on the total probability of illness, as the increase of the hygiene percentage by approximately 50 % reflected an increase of approximately 50% in the probability of illness (from  $\sim 0.002$  to  $\sim 0.003$ ). The prediction by the model used in this study, with the use of a higher percentage for the hygiene prior (28-74 %) increased the total campylobacteriosis cases for the year 2010 to 12,452 cases, which is approximately 70% more than the notifiable incidences or surveillance value (7,333 cases). This may include the number of unreported cases also. This finding with the other evidence discussed previously confirms the hypothesis that poor hygiene can be the cause of the high rate of campylobacteriosis in NZ. While the other hypothesis regarding NZ Campylobacter strains may have extreme characteristics or NZ poultry plant malpractices increases the contamination level of poultry were eliminated as explained in the previous chapters. It is also important to indicate that there is a data gap in NZ regarding consumers' hygiene practices since new relevant data can improve the current model output as the current model relies only on the information of the priors or the limited data available regarding New Zealanders' hygiene practice. However, the finding of Bayesian inference model argues for the regulatory, policymakers, risk managers, health professionals and educators to review and to assess their current consumer education strategy plans. They should motivate all parties involved and responsible for protecting consumers health and for developing a more effective education strategy to improve consumer hygiene practices to avoid the unsafe practices which can lead to the contamination of their food by Campylobacter. This would reduce the campylobacteriosis cases and possibly other foodborne diseases in NZ. Unfortunately, *Campylobacter* has the remarkable capability of disseminating easily from raw poultry, directly onto cooked chicken meat or some other component of the meal or indirectly onto all utensils (knives, cutting boards, taps, kitchen cloths, etc.) or hands that have been in direct contact with raw meat or contaminated packages if consumers do not follow safe handling practices. Despite the fact that the importance of the microorganism cross-contamination route in preventing the risk of food poisoning and most of the international published QMRAs have considered it as the major or the only pathway that enables the development of campylobacteriosis (Christensen et al., 2005, Hartnett et al., 2001, Mylius et al., 2007, Nauta et al., 2007, Rosenquest et al., 2003), several of QMRA studies (Dogan et al., 2019, Akil and Ahmad, 2019, Jeong et al., 2019, Premarathne et al., 2017) have not conducted sensitivity analysis regarding the uncertainty of the hygiene model parameters. Especially the frequency of improper hygiene behaviour by consumers,

and its impact on the final risk estimate, as has been performed in this study and other studies (Calistri and Giovannini, 2008, Pang et al., 2017) which performed a sensitivity analysis as recommended in the literature (Zwietering and van Gerwen, 2000). Therefore, the effect of a new intervention, for example, an education plan, was clear for the policymakers, risk managers and health professionals before the implementation of the new intervention. This can be achieved by altering the priors of consumers' hygiene practice.

The results obtained from increasing the prior of the poultry farm model (the probability that a bird in a poultry farm has been contaminated or positive, pf) and the prior presenting the possible increase in the  $d_{bf}$  (the contamination prevalence due to the transportation of flock and slaughtering process) by 25% in the model (N~ 0.511, 20.66 and (0.15, 206.6) revealed a similar increase of approximately 25% in the final probability of illness (p.it, p.ib). The 25% increase in the actual microbiological data for birds prevalence (g.fs) and chicken prevalence (g.bs) incorporated in the model was performed to avoid any impact of the actual data on the shape of the resultant posterior distribution of the final probability of illness estimated by the model computation. The finding of the 25% increase in (pf,  $d_{bf}$ ) priors and data (g.bs, g.fs) has confirmed the significant impact of the two priors in altering the final risk estimation. This is in agreement with another international QMRA model (Rosenquest et al., 2007, Huang et al., 2018, Vigre et al., 2016).

The selection of the value of 25% was justified by the practical future pragmatism as a planned intervention. It was not practical to further reduce the contamination in poultry to 50% given the status and circumstances of the poultry industry in NZ in terms of the prevalence and load of the counts on birds and on slaughtered chicken in 2006. The contamination level/count loads reported in 2006 was from 2.6 up to 6.8 log (Chrystal et al., 2008) on slaughtered chicken and the prevalence was 80 -100 % (French, 2008a) where the prevalence in birds varied from 27 up to 100% (NZFSA, 2009, Boxall, 2005). However, to our knowledge, there is no reported data regarding the counts in bird caecal samples in NZ, and all the data available in NZ were based only on positive/negative analysis.

Moreover, the poultry industry reported that they implemented interventions across the industry that have seen *Campylobacter* detection rates drop by 25% between 2007 and 2010 in response to a NZFSA strategy set by (PIANZ, 2011). This supports the reasonable selection of the 25% value for use in the model to test the change in the final risk estimate output of the model in contrast to the 50% value selected for the consumer's hygiene variate prior, which has been discussed above. Therefore, the 25% level of contamination was a practical target for the poultry industry to achieve. The association between the reduction in flock/ poultry prevalence and reduction of poor hygiene practice and the calculated reduction in risk of human disease was estimated to be 1:1 which is in

agreement with international findings (Rosenquest et al., 2003, Vigre et al., 2016). The overall performance of the simplified proposed model was acceptable and fulfilled the primary proposed objectives such as the use of the Bayesian technique in QMRA for its strength, as was recommended (Smid et al., 2010, Nauta et al., 2009). The model proved to be useful in characterising particular risk mitigation strategies once they have been planned and expressed (Smid et al., 2010, Neves et al., 2018). A source attribution study in NZ noted that 74% of the decline in notification rate in 2007 and 2008 was attributed to poultry (Sears et al., 2011). This may support the finding of this risk assessment study by the Bayesian Belief Network and inference. The hypothetical interventions were selected in order to reduce consumer malpractice by 50% and improve the level of poultry contamination at farms and plants by 25%, which adds up to 75%. The model also highlighted areas of data gaps which need to be improved in the future. The model used in this study needs further improvements, such as the incorporation of the contamination level in terms of the bacterial count, since the current model was based only on the analysis of presence or absence due to the lack of count data in NZ, especially for live birds. The availability of data for other model variates parameters, variables and co-variables will improve the model globally since it is known that QMRA is a data-hungry process, requires enormous resources and takes a long time to complete. However, it is important to clarify that a QMRA model simplifies a complex reality system by the use of mathematics and making assumptions and hypotheses. The simplified model used in this study was able to identify consumer hygiene practice, the initial prevalence at the farm level, and the processing practice at the plant level, as the major factors influencing the final risk estimate. A QMRA model can provide additional insight into the system and identify relevant data gaps, but it cannot provide additional knowledge itself except for the provision of an evaluation of control measures all through the food chain in terms of human health risks (Nauta et al., 2009, Membré and Boué, 2018). Therefore, before making any decision regarding the allocation of resources to gather the data required for a more comprehensive QMRA model in NZ, it is important to ask: is it worth collecting more data? What is the cost-benefit? And what analysis could be done with the currently available data, and possibly with the data available shortly, with limited time and resources? (Vose, 2005). In spite of the short time frame and the limited resources, the outcomes of this study are better than the QMRA model developed earlier in NZ (Lake et al., 2007) which was guided from abroad (Denmark and U.K) (Nauta et al., 2009) and whose main function was to guide future research.

Moreover, the MC modelling technique was used for the old NZ model, ignored the farm step due to the lack of data, and tested only two scenarios out of the three scenarios listed for the flock contamination (Nauta et al., 2009). The Bayesian technique deserves further investigation, and the convergence of the model needs to be treated with caution and be tested and validated with other software. There are new software packages which may perform better than WinBUGS and which conduct a complete sensitivity analysis as an automated routine (Smid et al., 2010). The Bayesian inference technique has more advantages than the MC technique, (Albert et al., 2008, Smid et al., 2010, Tuominen, 2009) although it is always the purpose of the model to impact the decision of the appropriate technique to be used for the model. The Bayesian inference technique is more complex than the MC technique can fit the purpose for use as well (Smid et al., 2010).

# **10.4 Conclusions**

The Bayesian proposed technique can provide better information for QMRA than MC technique for its ability to incorporate the data. The Bayesian approach can provide better insight into the food chain, is more informative and can easily understand the impact of any intervention in the food chain. However, Bayesian methodology is a complex task and its use in QMRA it still in its infancy. Bayesian approach deserves further investigation, and the convergence of the model needs to be treated with caution and tested and validated with other software or packages. A simplified model used was able to identify the consumer hygiene practice, the initial prevalence at farm level and the processing practice at the processing plant level as the major factors influencing the final risk estimate. This study provided a fast response to regulatory, policymakers, risk managers regarding the main risk drivers in NZ campylobacteriosis epidemic and provided additional supportive evidence for the conclusions stated in the cross-contamination chapter regarding the possibility of the poor hygiene practice as the reason for campylobacteriosis in NZ. Future research needs to be conducted to improve the model, data and the understanding of the BBN technique and its feasibility for NZ QMRA studies and to use it as a tool to enhance food safety in NZ. However, predicting campylobacteriosis cases for the year 2010 by the Bayesian approach failed to provide an accurate estimation of the number of actual notified campylobacteriosis cases in 2010. Therefore, there is a need to explore other methods to predict campylobacteriosis more accurately. The next chapter has addressed this point.

# 11 Comparison of time-series models for predicting campylobacteriosis risk in New Zealand

# 11.1 Introduction

*Campylobacter* causes more cases of bacterial foodborne illness each year than any other bacterial pathogen worldwide (Schönberg-Norio et al., 2006, Schönberg-Norio et al., 2004, Facciolà et al., 2017, Samuel et al., 2004b). It is the most commonly reported foodborne disease in industrialised and European countries (Hilmarsson et al., 2006, Johnson et al., 2006, Jore et al., 2010, EFSA, 2018). The symptoms of campylobacteriosis include muscle pain, headache, fever, watery or bloody diarrhoea, abdominal pain, and nausea, sometimes accompanied by vomiting with late complications such as reactive arthritis, RS and GBS.

The estimated annual cost for campylobacteriosis in NZ is more than \$77 million per year, not including its effect on New Zealand's reputation in the world and the financial loss to the tourism industry (Mataragas et al., 2010). In 1980, NZ legislated campylobacteriosis as a notifiable enteric disease. Since then, the incidence of reported campylobacteriosis has risen steadily, and it is now believed that NZ has the highest rate of reported cases in the world. In 2006, the notification rate was 380 per 100,000, an increase of 56 % from 2001 and 14% since 2005. This rate is 35 times higher than the rate for the United States, four times higher than the rate for Australia and five times higher than the rate for the U.K. (Baker et al., 2006a). Because of under-reporting, the actual figures are estimated to be at least seven to eight times higher than the notified cases (Lake et al., 2007) or possibly up to 30 times higher (Newman et al., 2015).

Based on above, campylobacteriosis in NZ is considered as a national epidemic, raising a public demand for urgent action and in particular control of chicken contamination to protect New Zealanders and to reduce the economic impact of the disease in the country. The NZ Food Safety Authority (NZFSA) has developed a comprehensive risk management strategy aimed at achieving a significant reduction in *Campylobacter* levels in chicken meat by employing interventions at appropriate points in the food chain. In April 2007 NZFSA established a microbiological monitoring programme (NMD) to estimate on a national basis the prevalence of *Campylobacter* in flocks (sheds) of poultry at slaughter and the prevalence and numbers of *Campylobacter* on poultry carcasses after processing. All poultry processors have to collect three samples every processing day for monitoring of the prevalence and the contamination level in flocks before slaughtering and on carcasses before distribution to the market for each day. Thus, the producers have been able to

reduce the contamination level as the result of the daily analysis as monitored by NZFSA.

In April 2008, NZFSA imposed the Poultry *Campylobacter* Performance Target (CPT) of the maximum contamination level in chicken carcasses as 6000 cfu/carcase when released to the market. This probably forced the poultry processor in NZ to increase their efforts to implement more interventions throughout the processing lines to further reduce the contamination level in chicken carcasses. It should be noted that the European Union performance target is only 1000 cfu.

The intervention strategies implemented in New Zealand and reported by Sears et al. (2011) appear to have led to a noticeable reduction in incidence. However, a careful statistical analysis is necessary to evaluate the evidence that these interventions have resulted in a change-point in disease incidence and to evaluate their cost and benefit in terms of campylobacteriosis cases prevented.

Predicting campylobacteriosis cases has become a matter of considerable concern in NZ- The Dutch Microbial Risk Assessment Model used to predict the number of campylobacteriosis cases in NZ (Lake et al., 2007) failed to predict the accurate number of actual cases, giving a prediction about threefold higher than the actual notified cases for 2006 and about nine-fold higher than the actual notified cases in 2008. The main drawback of this NZ adapted model was that it relied on the international literature data which did not reflect the actual scenario in NZ poultry processing and also did not consider the interventions implemented by the industry and the government. Moreover, the challenges faced in general by the probabilistic QMRA approach and used in the NZ model have probably contributed to the current model's accuracy (Havelaar et al., 2008).

Here in this chapter, it was explored the classical time-series modelling approaches for predicting campylobacteriosis in NZ. Such time-series have been successfully applied in fields as diverse as engineering, science, sociology and economics (Brockwell and Davis, 2002). A reliable method for making time-series predictions would be of great benefit to epidemiologists and public health officials (Altizer et al., 2006, De Greeff et al., 2009). A study in Belgium, showed that a time-series analysis is an appropriate methodology to clarify trends of campylobacteriosis, salmonellosis and listeriosis, forecast future cases and test the impact of interventions on the burden of food-borne diseases (Maertens de Noordhout et al., 2017). A US study (Weisent et al., 2010) examined time-series forecasting of *Campylobacter* rates, comparing a variety of time-series models. They found that a seasonal decomposition model provided the fastest, most accurate, user-friendly method to capture the temporal patterns in disease risk based on their main evaluation criteria of evaluating mean absolute percentage error, mean squared error and coefficient of determination (R<sup>2</sup>). Finding the most appropriate time-series model for NZ data has additional practical considerations given a possible structural change, i.e. a specific and sudden change in the underlying nature of the time-

series, in response to the implemented interventions.

In this study, it has been considered the decomposition, smoothing and ARIMA methods reported in a US study (Weisent et al., 2010) and added an ARIMA with interventions to explicitly model the effect of an intervention. The developed models used the monthly disease incidence of campylobacteriosis calculated from NZ surveillance data from 1997 to 2008. Predictive accuracy was evaluated and compared using the data for 2009.

# 11.2 Methods

# 11.2.1 Data collection

Campylobacteriosis monthly notification reports were obtained from surveillance data published on the official Public Health surveillance website (Episurv) on behalf of the NZ Ministry of Health, from January 1998 to December 2009. Rates of notification were calculated using annual end-ofyear population estimates obtained from the governmental Statistics NZ web site, with spline interpolation used to estimate the monthly population figures. These risk estimates were presented as the number of cases/100 000 persons. The data from 1988 to 1996 were disregarded as the monthly notification figures were not available, with only the yearly notification totals obtainable (NZPHO, 2010). The data from the years 1998 to 2008 were used to model the time-series, with the year 2009 held out of the dataset for model validation. The best two models were then fitted to the full 1998-2009 data and used to predict for each month of 2010. The best two models were then fitted to the full 1998-2009 data and used to predict for each month of 2010 (Figure 11.1). A total of 144 points were used in this study, which is generally considered to be sufficient for timeseries methods (DeLurgio, 1998). The statistical analyses were performed in Minitab ® version 15 statistical software (Minitab Inc, USA) and R software (version 2.9.1).

#### 11.2.2 Time-series decomposition techniques

These methods decompose a time-series into several components regarded as operating at different time scales. Generally, any time-series is considered to contain up to four components: i) trend 'T ' or the increase or decrease in a time-series over a long period of time; ii) cycle ' C ' or the smooth fluctuations around the trend line; iii) season ' S ' or yearly repeated fluctuation, not necessarily an identically repeated fluctuation; iv) error ' E ' or random variations which have a transitory effect on the time-series. Estimates of the first three components can be obtained by decomposition of the series and used for forecasting (Makridakis et al., 1998). The decomposition method assumes that the time-series can be expressed as a sum (additive model) or product (multiplicative model) of the above components, as shown in Equation 11-1 (see below). A multiplicative model denotes

proportional variation when the time-series exhibits increasing or decreasing seasonal variation.

$$Y_t = T + C + S + E \text{ or } Y_t = T \cdot C \cdot S \cdot E$$
 11-1

It is often practical to ignore the cycle component or to subsume it into the trend component. There are two main approaches to decomposition, which are discussed below:

Regression-based methods

These methods assume a simple regression model for the trend, which represent the average level that changes over time, as follows:

$$T = \beta 0 + \beta 1 t \qquad 11-2$$

This is a linear trend model (straight line growth or decrease), but other possibilities are a quadratic trend (curvilinear) and a *p*th-order polynomial trend (with one or more reversals in curvature). The seasonal component S is modelled by using a different fixed value for each month, or perhaps periodic functions such as sine and cosine. Least squares point estimates of the parameters in these models are obtained by using standard regression techniques, assuming that the error term E represents random fluctuations that are independent and identically distributed. Variables are retained if their corresponding regression coefficients indicate significance (p>0.05). This is a "global" model in the sense that it assumes the trend model and seasonal factors are the same throughout time, and all data points contribute equally to estimating the fixed parameters.

#### Smoothing techniques

The basic exponential smoothing method without a trend or seasonal components uses simple updating equations to decompose a series into a stochastic, smoothly varying level component ( $L_t$ ) and a random component. It is generalised in the Holt-Winters method to handle a time-series with both trend and seasonal components (Chatfield, 1978). The updating equations employ user-specified smoothing parameters  $\alpha$ ,  $\beta$ ,  $\gamma$  as shown below for the multiplicative case (there are analogous formulae for the additive case).

For the level component:

$$L_{t} = \alpha \left(\frac{x_{t}}{S_{t-s}}\right) + (1-\alpha)(L_{t-1} + T_{t-1})$$
 11-3

For the trend component

$$T_{t} = \beta(L_{t} - L_{t-1}) + (1 - \beta)T_{t-1}$$
11-4

For the seasonal component

$$S_{t} = \gamma \frac{Y_{t}}{L_{t}} S_{t-s} + (1 - \gamma)S_{t-s}$$
 11-5

Then the forecasting equation for h periods into the future is:

$$F(h) = (L_t + hT_t) S_{(t-s+h)}$$
 11-6

These techniques give a "local" model in which the components are constantly changing over time, and data points are given more weight in the estimation when they are close to the time of estimation. The Holt-Winters method is a simple method to implement and is widely used in practice (Chatfield, 2004).

### 11.2.3 ARIMA modelling

Moving average

In a time-series with the MA process, the model describes each observation as a function of the previous random errors. The observed series is represented as a weighted moving average of the unobserved error series up to a specified lag q (Armstrong, 2001).

$$Y_t = \theta_1 \varepsilon_{t-1} + \dots + \theta_q \varepsilon_{t-q}$$
 11-7

Where  $\varepsilon_t$  represents the random error at time t and  $\theta_i$  are constants.

Autoregressive (AR)

A simple way to model dependence on past observations is to use ideas from regression. The AR model expresses the current time-series observation as a linear function of its past observations, plus a random error. It can also be regarded as a Markov process (Chatfield, 2004).

$$Y_{t} = \emptyset_{1}Y_{t-1} + \emptyset_{2}Y_{t-2} + \dots + \emptyset_{p}Y_{t-p} + \varepsilon_{t}$$

$$\{\varepsilon_{t}\} \sim iN(0, \sigma_{\varepsilon}^{2})$$

$$11-8$$

ARIMA models

ARMA and ARIMA models were described by Box and Jenkins in 1976 as giving a very flexible framework for the analysis of time-series. The advantages of these models are well documented in the literature (Benschop et al., 2008, Reichert et al., 2004). The ARMA model can be expressed as follows:

 $Y_{t} = \emptyset_{1}Y_{t-1} + \dots + \emptyset_{p}Y_{t-p} + \varepsilon_{t} + \theta_{1}\varepsilon_{t-1} + \dots + \theta_{q}\varepsilon_{t-q}$ <sup>11-9</sup>

Where  $\phi_1 \dots, \phi_p$  and  $\theta_1 \dots, \theta_q$  are constants and  $\varepsilon_t$  represents the random error at time t.

This model combines the features of the MA and AR models. However, a practical drawback of the above ARMA model is that it is valid only for stationary time-series with a constant mean. To extend the ARMA model, it is necessary to consider how non-stationary sources of variation such as trend and seasonal variation can be removed. To transform to a stationary series, the observations usually first need to be filtered (removing of trend and seasonality) by subtracting consecutive observations from each other or subtracting observations at seasonal lags. The resulting combination of filtering to achieve stationarity and ARMA modelling of the stationary series is known as an integrated ARMA model or ARIMA as the stationary model that is fitted to the differenced data has to be summed or integrated to provide a model for the original non-stationary data (Chatfield, 2004).

Thus, each ARIMA process has three parts: (AR), integrated (I) and (MA) parts. The models were written as ARIMA(p,d,q) where p describes the AR part, d the integrated part and q the MA part. Series with seasonal components also need equivalent terms at seasonal lags (S), leading to the full seasonal ARIMA model written as ARIMA(p,d,q)× (P,D,Q)<sub>s</sub>. A multiplicative form of the ARIMA can be obtained by making a logarithmic transformation of the data, fitting an ARIMA to the transformed data, and then back-transforming the predictions.

### ARIMA intervention models

These models can be used when an exceptional, sudden, external event affects the time-series at a particular point in time, for examples strikes, disasters, policy changes, etc. (Bonham and Gangnes, 1996). The NZ campylobacteriosis series experienced an intervention in 2007 when an attempt was made to reduce the rate of campylobacteriosis. ARIMA intervention models try to

accommodate such intervention by the use of an intervention variable which is included in the basic ARIMA model. This variable may be a step function for representing a permanent change or an impulse function to model a temporary effect. Historically, an improvement in forecasting has been noticed by several authors after applying an ARIMA intervention model (Goh and Law, 2002, Min et al., 2010, Rashed et al., 2017). In some cases, estimating the effect of the intervention is the main goal of the analysis.

#### 11.2.4 Model evaluation

The candidate models were fitted using the training data from 1998 to 2008, and quantitatively evaluated based on their predictive ability on the 2009 holdout data using mean square error (MSE) and mean absolute percentage error (MAPE). MAPE measures the average size of the prediction error in relative terms, whereas the MSE penalises the worst predictions more heavily.

# 11.3 Results and Discussion

Inspection of the NZ campylobacteriosis series (Figure 11.1) confirmed that the peak of the annual incidence data was observed during the hot summer months for all the years incorporated in this study, which is in agreement with most international data (Louis et al., 2005, Tam et al., 2006, Fleury et al., 2006, Hartnack et al., 2009). This may be due to human behavioural activity in summer such as picnics, barbeques, other outdoor activities, or possibly the presence of flies (Jepsen et al., 2009, Naumova et al., 2000). This indicates that seasonal models are required.

A significant decline in campylobacteriosis notifications was noticed in 2007 (Figure 11.1) as expected, possibly due to the improved consumers' safety practice at home and reduced level of contamination of chicken carcasses brought to consumers' kitchens. This suggests, as mentioned above, using an intervention variable at this point in the series. The analysis suggested that the best ARIMA was an ARIMA  $(2,0,0) \times (0,1,1)_{12}$ . When this was fitted with an intervention for 2007, using the R software, the intervention constant was found to be significantly negative, indicating evidence of a reduction in the underlying mean number of cases at this time. For the implementation of the Holt-Winters method, the smoothing parameters were left at their default values of 0.2, as this study wanted to investigate the performance of a simple "off-the-shelf" method.

The Holt-Winters and ARIMA with intervention models were found to be the best models for predicting campylobacteriosis in NZ. Both additive and multiplicative versions were considered.

The results for the best models identified for each technique for the training data and the test data are summarised in Table 11.1 with the monthly forecasts of campylobacteriosis incidence by all techniques tested for 2009 given in Table 11.2. The MSE and MAPE for ARIMA and Holt-Winters

method were better than the regression-based decomposition method for both the training data and the test data.



Figure 11-1 Campylobacteriosis notifications in New Zealand from 1998 to 2010 with Holt Winter and ARIMA intervention analysis and forecast.
	Training (1997-2008)			Test (2009)	
Method	$\mathbb{R}^2$	MSE	MAPE	MSE	MAPE
Decomposition (multiplicative)	0.3773	44.84	26.60	142.93	88.34
Decomposition (additive)	0.3778	44.81	26.63	138.65	88.77
Holt-Winters (multiplicative)	0.6228	27.16	18.13	8.99	14.71
Holt-Winters (additive)	0.6604	24.46	18.88	16.24	29.57
Arima	0.7724	16.39	14.11	35.17	40.16
Arima with intervention	0.6959	21.90	17.21	7.69	17.86
Arima (mult.)	0.7875	15.30	12.77	38.65	45.77
Arima (mult.) with intervention	0.7981	14.54	12.09	16.35	30.69

Table 11-1 Time-series statistical model comparisons for campylobacteriosis forecast in New Zealand.

MSE = Mean square error, MAPE = mean absolute percent error

Table 11-2 Time-series model comparisons for campylobacteriosis rate (per 100000) forecasts in NZ for 2009.

Month	TRUE	DCa	DCm	HWa	HWm	ARa	ARm	AR_Ia	AR_Im
jan	18.46	33.40	34.94	19.77	14.60	24.54	23.24	23.24	22.63
feb	13.91	29.13	29.03	13.21	12.02	18.82	19.58	16.98	18.52
mar	11.78	23.58	24.11	9.03	10.14	15.94	17.20	14.67	16.63
apr	9.82	17.06	17.02	2.43	7.51	9.89	12.67	8.17	12.11
may	9.85	19.55	17.25	5.53	8.71	13.49	14.64	10.58	13.24
jun	9.19	19.40	19.49	4.38	8.14	13.15	14.05	9.84	12.63
jul	9.62	21.24	21.14	5.98	9.55	15.45	16.20	11.75	14.29
aug	12.68	23.18	23.18	9.46	11.76	19.25	19.39	15.09	16.78
sep	13.18	24.64	24.39	10.33	12.13	20.83	20.86	16.30	17.85
oct	14.34	27.77	28.41	11.76	12.94	23.00	23.08	18.20	19.60
nov	22.41	32.89	32.73	16.34	15.25	27.97	27.71	22.70	23.09
dec	20.38	32.72	33.70	16.70	15.51	28.94	28.92	23.99	24.50

DCa = Decomposition (additive), etc.

Table 11-3 l	Forecast	performance	of the	number	of	cases	for the	e most	successful	models	for th	ne
year 2010.												

Month	TRUE	HWm	AR_Ia	AR_Im
jan	839	841.8	982.7	912.4
feb	705	686.1	703.2	744.6
mar	571	585.4	610.8	670.5
apr	510	447.2	341.6	494.7
may	449	509.9	432.1	538.5
jun	452	482.9	400.2	514.2
jul	442	564.1	479.6	579.4
aug	628	715.1	623.1	687.7
TOTAI	4596	4832.6	4573.3	5142.0

The basic ARIMA and ARIMA with intervention easily out-performed all other methods in the training data, but the basic ARIMA performed poorly on the test data, presumably because it failed to accommodate the intervention in 2007 and subsequently over-estimate of future values. The Holt-Winters method, being local in nature, quickly adapts to the new regime post-intervention and gives good predictions, comparable with the ARIMA intervention models. A multiplicative model is preferable for the Holt-Winters method, but the choice is less clear for the ARIMA intervention model.

To investigate further, the performance of the Holt-Winters multiplicative and ARIMA intervention methods was also tested for the currently available data of the year 2010. The results are summarised in Table 11.3 and depicted in Figure 11.1 It can be seen that prediction by the additive ARIMA intervention method was slightly better than the prediction of Holt-Winters multiplicative method for the total number of cases over the eight data points currently available, whereas the Holt-Winters forecasts track the monthly figures slightly better. The Holt-Winters method has the additional advantage that it is a simple method to use (Chatfield, 2004), whereas the ARIMA intervention model requires extra statistical expertise to conduct it.

From the above, it is confirmed that classical time-series techniques such as ARIMA with intervention and Holt-Winters can provide a good prediction performance for campylobacteriosis risk. Regression-based decomposition might perform well in a stable situation, but such a global approach is likely to fail when there are structural changes in the series. The NZ campylobacteriosis data pattern, in particular, the sudden drop in incidence following an intervention, may have contributed to the weak performance of this decomposition method.

The general strength of the Holt-Winters forecast method is that it managed to predict the monthly campylobacteriosis cases in NZ accurately. Despite an apparent structural change in the series, and that it can be used easily with a minimum of statistical expertise with the currently available software (Rolfhamre and Ekdahl, 2006). The ARIMA intervention model, despite its much greater complexity and sophistication, did not lead to markedly better forecasts, except for the total number of cases in 2010. Its main advantage is that it can estimate and test the significance of any structural changes expected from a planned intervention. Given the development of appropriate software and the variety of forecast methods available and reviewed (Box et al., 2008, Brockwell and Davis, 2002), evaluation of such interventions can be conducted promptly. These time-series techniques can be useful to health professionals and policymakers in preparing for a predicted epidemic (Myers et al., 2000). The results reported by this study could be useful to the NZ Health and Safety Authority's efforts in addressing the problem of the campylobacteriosis epidemic. NZ has been requested by official international organisations such as Codex Committee on Food Hygiene (CCFH) Codex for NZ to lead the world in developing standards to combat campylobacteriosis. This is possible as a recognition of the initial strategy for targeting the campylobacteriosis epidemic in NZ. However, the number of reported cases in New Zealand is still higher than in other developed countries. Thus, the selected industrial interventions and the consumers' education programme need to be re-evaluated and reviewed to control the disease and achieve a comparable rate with other developed countries.

#### 11.4 Conclusions

An accurate monthly prediction for campylobacteriosis cases based only on previous surveillance data in NZ was achieved using a simple method. The provision of valuable information for public health professionals and policymakers was also be achieved. Evaluation of the selected implemented interventions to combat campylobacteriosis was possible based on the results described above. The study could provide an early alert for the possible repeat increase of campylobacteriosis cases in NZ. The classical time-series analysis could provide an accurate prediction for campylobacteriosis in NZ with minor adjustments.

The Holt-Winters technique was slightly more successful in prediction than the ARIMA with intervention techniques for a general monthly prediction. ARIMA with intervention outperformed Holt-Winters on the total yearly prediction. The summer seasonal peak pattern for NZ data is comparable with international data. The drastic decline in NZ campylobacteriosis reported cases when the study was conducted in 2007, and 2008 hindered the straightforward application of time-series analysis.

# 12 Evaluation of food handling practice among New Zealanders and citizens of other developed countries as the main risk factor for campylobacteriosis rate

#### 12.1 Introduction

The previous studies revealed that the pathogen characteristics including heat resistance or oxygen tolerance, were not unusual and that poultry industry practices were neither more unsafe nor did they significantly contribute to a higher level of chicken contamination. Thus, the most likely remaining hypothesis is, do New Zealanders have poorer home hygiene practices during food preparation than the people of other developed countries? If so, this would cause cross-contamination during food preparation in the kitchen. Internationally, cross-contamination is considered to be the main factor in campylobacteriosis transmission (Kapperud et al., 2003, Mylius et al., 2007). Thus, it is necessary to test whether the high rate of campylobacteriosis in NZ has a significant link to poorer practices by New Zealanders in their home kitchens. Cross-contaminated product during food preparation) has been a growing food safety concern for the last 20 years (Cogan et al., 2002, Gorman et al., 2002, Luber et al., 2006). *Campylobacter* spp. are heat sensitive and cannot multiply or grow below 30°C. Therefore, cross-contamination that occurs during food preparation becomes the most critical food handling risk factor that enables intestinal tract infection of campylobacteriosis in humans (Nauta et al., 2008).

Quantitative risk assessment studies consider the development of campylobacteriosis to be due to direct exposure to *Campylobacter* spp. in animal foods and water. Many of these risk assessment studies consider cross-contamination during food preparation as the major or the only pathway for campylobacteriosis (Brynestad et al., 2008, Lindqvist and Lindblad, 2008, Mylius et al., 2007, Nauta et al., 2007). Reported studies indicate that many cases of foodborne illnesses are related to the contamination of food eaten at home. Van Asselt et al. (2008) and Redmond and Griffith, (2003) reported that up to 87% and 10 - 50% respectively of the cases were related to infections from food usually eaten at home. A lower percentage reported by Redmond and Griffith (2003) is possibly due to the practices in different countries and/or due to the under-reporting of home foodborne illnesses. It is now believed that improper food handling practices contribute to at least 40 - 60% of cases of foodborne illnesses (Cogan et al., 2002). A NZ analysis of gastroenteritis outbreaks for the period between 1998 to 2000 revealed that 39.3% of cases were acquired at home.

(Gilbert et al., 2007). Another NZ study (UMR Research, 2007) reported in the same year showed that about 30% of their participants reported foodborne illnesses in 2007. In the years 2003 and 2005, about 20% of the participants had experienced foodborne illnesses (UMR Research, 2007). However, only 6% of those participants indicated that their illness was caused by food cooked at home (UMR Research, 2007).

Several review papers have discussed food safety practices in homes throughout the world (Patil et al., 2005, Redmond and Griffith, 2003). These studies addressed consumer beliefs about hazards, their knowledge about food safety and their self-reported practices. There are also observational studies on consumer behaviour in both laboratory kitchens and in real environmental conditions. (kitchen). In these studies, consumers were asked to prepare and cook a contaminated meal which was either purposely or naturally contaminated (Evans and Redmond, 2018) Few of these observational studies linked specific consumer behaviour to the microbial contamination of the prepared meal (Redmond et al., 2004). Models for cross-contamination were designed and validated by quantitative observational studies (van Asselt et al., 2009).

The aim of this study was to (i) investigate the reason for the high number of cases of campylobacteriosis in NZ, (ii) provide insights into the relationship between consumer knowledge and food handling practices, and (iii) to discuss the discrepancy between predicted consumer behaviour and observed behaviour. Moreover, the study investigated the best hygiene measures that can be used to provide public health professionals, risk managers and regulatory authorities with the scientific support for future health promotion plans that will discourage consumers from risky, unsafe food preparation practices. Laws and regulations cannot enforce controls and measures to prevent unsafe practises at home, unlike as for the commercial food processors and retailers. The most logical way to change risky consumer practices is by educating consumers.

#### 12.2 Methods

An extensive search of Scopus, ISI web of knowledge, and Google Scholar databases was conducted for publications concerning consumer food handling practices or self-reported practices, consumers' knowledge or perception about food safety and consumers' observed practices. Valuable data was also gained through attendance at international conferences (NZIFST 2008, IAFP 2008, CHRO 2009, and IAFP 2010), personal communication with the science group at the NZ Food Safety Authority (the funding organization of another *Campylobacter* related project, reference number (N0174/06 in 2006) and with *Campylobacter* lead scientists from the ESR. The available data provided an understanding of current consumer behaviour and the perception that can be used to plan effective education strategies. The search focused on consumer-based studies

that used the following methods.

#### 12.2.1 Survey of consumers' knowledge

Personal interviews and self-completed questionnaires were used to collect quantitative and qualitative data. This method provides baseline data about consumers' perception of food safety. Surveys of consumer's knowledge are also used to evaluate the efficacy of health promotion initiatives or before a new health promotion can be formalized or issued (Redmond and Griffith, 2003). The amount and accuracy of consumers' knowledge do not always predict their behaviour towards food safety. However, correct knowledge provides consumers with informed choices about their practices or actions.

#### 12.2.2 Survey of consumers' food handling practice

The consumers' food handling practice survey uses the same technique as the consumers' knowledge survey for collecting data (e.g. interviews and questionnaire). This approach is used mainly to investigate how consumers handle food at home and also to explore why consumers use such handling procedures. The disadvantage of this method is that self-reported practice surveys may or may not reflect actual consumer practice. Usually, consumers report the perceived correct practice rather than their actual practice (Redmond and Griffith, 2003).

#### 12.2.3 Observational study

The direct observational study is considered to be more reliable and accurate for studying consumer behaviour. The observational study can be conducted in either a controlled environment (laboratory) or an uncontrolled environment (home kitchen). Advantages and disadvantages of observational studies are discussed in detail by Redmond and Griffith (2003), and Evans and Redmond (2018). A video camera was used to observe consumer behaviour, or an observer watched the consumer preparing meals and then documented the steps used in the preparation of food. Observational studies have linked cross-contamination practices with the actual microbial (*Campylobacter*) contamination of prepared food in domestic kitchens and the contamination level of the kitchen environment.

#### 12.2.4 Bacterial transfer

A series of microbial studies targeting bacterial transfer from raw food or contaminated sources to cutting boards, hands, or surfaces in kitchens, were also initiated a long time in advance of the fully quantitative, actual kitchen observational studies or interdisciplinary studies (Scott and Bloomfield, 1990, Coates et al., 1987, Deboer and Hahne, 1990).

The simplest and most common equation to calculate the bacterial transfer rate, which is defined as the percentage of cells transferred from the donor surface to the recipient surface (Chen et al., 2001) is expressed as follows:

Transfer rate = 100 x (cfu recipient / cfu donor)

cfu - Colony-forming units of bacteria counted or measured.

### 12.3 Results and discussion

A few reports have applied a different approach to analyse or combine the data obtained from literature or questionnaires, such as the Meta-Analysis (Patil et al. (2005). Some surveys used the Delphi techniques to rank consumer behaviour (Hillers et al., 2003), while other surveys applied a descriptive analysis (Byrd-Bredbenner et al., 2007a, Kwon et al., 2008, Sanlier, 2009, Nesbitt et al., 2009).

The main purpose of this study was to inform regulatory authorities regarding consumers' knowledge and their critical or risky food preparation practices at home that may be contributing to the high rate of campylobacteriosis cases in NZ. This information is essential so that a successful health communication strategy can be designed to provide an effective message to consumers to change unsafe food preparation practices if any, in the future. Therefore, the main findings and conclusions of only the reviewed papers are classified and summarised below:

#### 12.3.1 Survey of consumers' food safety knowledge

The purpose of such surveys is to determine what consumers knew about food safety. It was reported that 75% of U.S. consumers (study participants) understood the concept of cross-contamination In the U.S., 55% of the consumers participating in the study correctly answered questions about cross-contamination (Byrd-Bredbenner et al., 2013), and in Australia, 38-49% (Jay et al., 1999a). In another U.S. study, 79% of consumer participants reported that they avoid cross-contamination from raw meat juices (Meer and Misner, 2000) and in Canada, the majority of the participants were aware of separating raw food and cooked food during preparation or storage (Nesbitt et al., 2014).

However, an European study indicated that 75% of consumers were not informed /aware of potential risks and hence stored poultry and raw meat on the upper shelves of a refrigerator. U.K. studies reported that 64-100% of consumers knew about the use of different utensils for raw and cooked or ready-to-eat food, to prevent raw food from contaminating cooked or ready-to-eat food (Griffith et al., 2001, McCarthy et al., 2007).

In NZ, only 10-18% of the consumers (study participants) agreed that it is better to store cooked food above raw food in the refrigerator (Hodges, 1993, Kerslake, 1995). 38% of New Zealanders were not aware of the need to use separate or clean utensils for the preparation of raw and cooked food (Hodges, 1993). Moreover, it was reported that 28% of New Zealanders cut raw and cooked food with the same knife. The knife was not washed and was only wiped with a "clean" cloth (Kerslake, 1995). A study in NZ (Gilbert et al., 2007) revealed that 89% of the consumers believed that the consumption of chicken might cause foodborne illnesses. 63% of NZ consumers were concerned about *Campylobacter*, and 77% about *Salmonella* (UMR Research, 2007, Lake et al., 2007). Gilbert et al. (2007) indicated that 90% of the consumers had associated *Salmonella* with chicken, and only 70% associated *Campylobacter* with chicken. However, the same study reported that 20% of the consumers were not concerned that raw chicken carried bacteria.

As reported, 95% of consumers in the U.K. (Lader, 1999) and 86% of consumers in the U.S. (Altekruse et al., 1996) believed that washing hands before preparing food is very important. Another U.K study revealed that 100% of consumers identified when and how it was necessary to wash hands (Griffith et al., 2001). Similarly, in the U.S. 79% knew the six occasions (before, during and after preparing meals, before eating, after using the restrooms, before and after caring for children, after touching an animal or sick person, and when the hands are dirty) for washing hands (Redmond and Griffith, 2003).

#### 12.3.2 Survey of consumers' practice

The first survey of consumer practices was conducted in 1975 (Redmond and Griffith, 2003). Hillers et al. (2003) listed 40 consumer practices that could cause 13 foodborne illnesses. Four practices, in particular, were shown to prevent campylobacteriosis, and these are listed below:

- Use a thermometer to assure that the food is cooked to the recommended temperature (75-80 °C).
- Wash food preparation surfaces, knives and cutting boards and sinks with hot water and soap after contact with raw poultry.
- > Wash hands with warm soapy water before and after handling poultry.
- Drink only pasteurised milk.

Most surveys indicate that consumers do not use, or seldom use, thermometers to check if the food has reached the adequate heating temperature (Nesbitt et al., 2009). A U.S. survey indicated 76% of surveyed consumers did not use a thermometer. However, in another U.S. survey, it was reported that 12% use a thermometer for meat, and only 6% used a thermometer to check the

doneness of the cooked food. In another survey, the number of consumers using a thermometer was 3% (Takeuchi et al., 2006). In Canada, 13.7% of consumers use a thermometer to check whether meat is cooked (Nesbitt et al., 2009). There is no NZ data for this practice. The U.K and European studies indicate that 81% of surveyed consumers tended to cook poultry longer than recommended, and 85-92% 'ensured that the food was piping hot'. It has been reported that 28% of American consumers consider the internal pink colour of a burger as sufficiently cooked (Redmond and Griffith, 2003). In NZ, 40% of those surveyed indicated that roasting or baking chicken was the preferred method for cooking chicken and 37.5% of them preferred chicken to be well-done, whereas 35.2% preferred very well-done and only 27.3% preferred medium for roast chicken (Gilbert et al., 2007). Another NZ study revealed that 90% of New Zealanders always check that chicken has been cooked right before they eat it. New Zealander's methods of cooking and preferences for chicken negate the hypothesis that undercooking is considered as a factor for acquiring campylobacteriosis in NZ. Moreover, the heat sensitivity of the most implicated strains are reported in previous chapters and by (Al-Sakkaf and Jones, 2012) support the hypothesis that cross- contamination is the main risk factor responsible for campylobacteriosis in NZ.

With regard to the frequency of washing food preparation surfaces, 80-93% of surveyed consumers in the U.K and the U.S. reported that they always wash the chopping board and utensils after cutting raw chicken. 56-90% of consumers in the U.K., Netherlands, U.S. and Canada use different utensils or cutting boards or the other side of the board for raw and ready-to-cat food (Nesbitt et al., 2009, Redmond and Griffith, 2003, Nauta et al., 2005). In the Netherlands, only 3% of consumers surveyed used the same side of the cutting board for cutting meat and vegetables, and 97% wash the cutting board or use another board or use the other side of the cutting board after cutting raw food without cleaning it first (Byrd-Bredbenner et al., 2013). Another U.S. study reported that 89% of surveyed consumers always wash the chopping board, put it in a dishwasher and/or use another cutting board (Roseman and Kurzynske, 2006). In Ireland, 78% of surveyed consumers reported that they washed the used cutting board with hot water, detergent and bleach, or use another clean cutting board (Jackson et al., 2007).

In NZ, 76.1% of surveyed, reported cleaning chopping boards at least once daily. Other equipment that was reported to be cleaned once daily is as follows: 54.9% clean sinks/tap, 51.4% kitchen sponge, 44.5% dishcloths, 47.5% dish brushes and 43.8% tea towels. Hand towels were cleaned two to three times per week by 46.2% of the surveyed consumers (Gilbert et al., 2007). This same study estimated that 28-41% of New Zealanders would treat knives and surfaces improperly or allow cross-contamination (re-use of knives and surfaces without washing between preparation of

raw and cooked food). Moreover, 5.3% of surveyed New Zealanders placed cooked meat into a raw meat container during a barbeque. The surveys in the U.K, U.S., Europe and Australia indicated that 72-93% of surveyed consumers (Redmond and Griffith, 2003, Nauta et al., 2005) washed their hands with soap and water before handling food and after handling raw food and poultry. In NZ, only 39% of consumers washed their hands before food preparation, and 22% washed hands between food preparation steps. Only 5% washed their hands after food preparation. Moreover, only 26.4% washed their hands correctly (using hot water, soap, and drying) and only 18.8% of the surveyed consumers dried their hands with the same towel that is used for drying dishes (Gilbert et al., 2007). The NZFSA educational film for under school-age children asks kids to wash their hands before eating, but only to wipe their hands with a cloth towel after eating.

#### 12.3.3 Observational consumers practice

Griffith and Redmond (2001) compared observed consumer practice in the U.K with previous knowledge, behavioural intentions, and self-reported practice with regard to hand washing and cross-contamination. It was concluded that in spite of the participants' knowledge (100% of the participants knew the extreme importance of handwashing), attitude and positive intention, only 85% indicated that they would "very likely wash hands" the next time the food is prepared. However, no participants washed their hands adequately all the time during the observed food preparation session. From the findings of the same study, with regard to preventing cross-contamination, although 100% knew the importance of preventing cross-contamination and 80% had a positive attitude, intention and self-reported practice, only 48% used a different utensil for raw or cooked food and to adequately clean surfaces between the preparation of raw food and cooked or ready-to-eat food during that session.

There are few observational studies that compared consumers' knowledge with actual behaviours, such as one by Jay et al. (1999b). This report indicated that half of the study participants did not use detergent or cleaner to clean kitchen surfaces. It concluded that there is a significant difference between consumer knowledge and their actual behaviour and practice. Anderson et al. (2004) and Redmond and Griffith (2003) reported a similar conclusion that 98% of consumers cross-contaminate ready-to-eat or cooked food with raw meat or raw egg. (Anderson et al., 2004) indicated that only 45% of the participants in a U.S. study attempted to wash their hands before beginning to prepare food, although 87% claimed that they wash their hands at all or most of the time. In the U.S., 5-57% of those surveyed neglected to wash their hands or did not wash their hands correctly, and 13-71% used improper-cross contamination procedures (Kendall et al., 2004, Redmond and Griffith, 2003, DeDonder et al., 2009). Only 45% washed their hands before

preparing a meal, and only 84% of those used soap (Anderson et al., 2004). In Australia, approximately 75% failed to wash their hands, 47% failed to wash their hands after handling raw meat, and 44% of those failed to use soap (Jay et al., 1999b). The same study also reported that approximately 35% failed to wash utensils between preparing raw foods and ready-to-eat food, and 30% failed to clean the preparation surfaces before preparing ready-to-eat food. Similar discrepancies among U.S. adults were reported (Abbot et al., 2009), and among U.S. Latinos (Dharod et al., 2007). There appear to be no studies on food preparation observational studies have been conducted in NZ.

#### 12.3.4 Bacterial transfer

Perez-Rodriguez et al. (2008) reviewed and critically analysed published research on bacterial transfer, and they discussed bacterial recovery methods, environmental and intrinsic factors influencing bacterial transfer, and bacterial transfer models used in microbial risk assessment studies (Kusumaningrum et al., 2004, Perez-Rodriguez et al., 2006, Rosenquest et al., 2003, Schaffner, 2004a). Perez-Rodriguez et al. (2008) stated that moisture, pressure and contact time are the main factors that increase bacterial transfer. L. monocytogenes which was attached to polyethylene and wooden cutting boards and recovered after holding time up to 1 h. Transmissions of L. monocytogenes to cooled cooked chicken meat from both types of cutting boards were relatively higher than hot cooked chicken meat (Goh et al., 2014). The currently available bacterial transfer models need more experimental data. Stochastic models (probability distribution) can be utilised as a transient solution for the apparently random and imprecise nature of the current experimental data. One limited study of bacterial transfer in food preparation (Gilbert et al., 2006) and a consumer home practice survey (Gilbert et al., 2007) were conducted after the highest numbers of campylobacteriosis outbreak in NZ history (426 cases per 100,000 persons) reported in 2006. New Zealanders had been lulled into a false sense of security for many years. Gilbert et al. (2006) summarised several bacterial transfer studies in the literature (see Appendix 3; Gilbert et al. study). These transfer rates reported by Kusumaningrum et al. (2003), Luber et al. (2006) and Montville and Schaffner (2003). Gorman et al. (2002) evaluated cross-contamination resulting from naturally contaminated chicken and reported that four pathogens (Salmonella, Campylobacter, Escherichia coli, Staphylococcus aureus) could easily spread from contaminated raw chicken to hands and kitchen surfaces and contact food. Other similar studies evaluated the ease of how different microorganisms can be transferred to food by different consumer handling scenarios (Barker et al., 2003, Haysom and Sharp, 2005, Redmond et al., 2004). A limited number of studies about bacterial transfer between surfaces (Humphrey et al., 1994, Deboer and Hahne, 1990) have also been

conducted. There is a significant discrepancy in the reported values of transfer rate from a donor to recipient and this confirms that identifying the transfer rate is a massive challenge. For example, Chen et al. (2001) reported a transfer rate of 8.7% of the surrogate bacterial indicator (Enterobacter aerogenes) from chicken to hands and Luber et al. (2006) reported a Campylobacter transfer rate of 2.9 - 3.8%. Montville et al. (2001) indicated a transfer rate of 0.71% (of surrogate bacteria). Montville and Schaffner (2003) reported a similar transfer rate of 0.59% in a later study (surrogate bacteria). A NZ study (Gilbert et al., 2006) reported a Campylobacter transfer rate of 0.4 - 6.2%, and Kusumaningrum et al. (2004) 2.4 %. A study by Chen et al. (2001) indicated that the transfer rate could reach 30% from raw poultry to the cutting board and 50% from the cutting board to lettuce, and up to 100% to hands. The transfer rate of *Campylobacter* at 42.5% was significantly higher than for Salmonella (4.8%) and E. aerogenes (12.6%) from a steel surface to food (Kusumaningrum et al., 2004). This confirms the capacity of *Campylobacter* to spread quickly and easily to and from surfaces and to cause contamination and consequently infection if ingested by consumers, more than by any other pathogen (Redmond et al., 2004). Transfer data studies have facilitated the understanding of the microbial risks associated with food handling practices. Many observational studies were conducted with actual bacterial contamination of prepared meals (van Asselt et al., 2009, de Jong et al., 2008, Nauta et al., 2008, Redmond et al., 2004, Luber et al., 2006). Christensen et al. (2005) developed a model for hygiene practice and consumption patterns, and Mylius et al. (2007) built a mechanistic model for cross-contamination during domestic food preparation. Van Asselt et al. (2008) succeeded in developing a model for transfer and validated the model with actual consumer observational data (van Asselt et al., 2009). Although the accuracy of these models is not high, the model prediction can be used to quantify cross-contamination in the home and can be incorporated in a QMRA model.

It has been stated that there is a lack of data in NZ with regards to studies on consumer's knowledge and handling practices. The few studies (Bloomfield and Neal, 1997, Kerslake, 1995, Gilbert et al., 2007) conducted in NZ were not comprehensive and have increased the ambiguity around food handling practices in NZ. Thus, the data extracted from these limited studies should be treated with caution. It is difficult to compare NZ surveys and international surveys due to differences in sample size, objectives, targeted population groups, methodology and the interpretation of results. The limited amount of consumer-based research, which is of significant importance for the safety of New Zealanders, has declined due to a lack of motivation and/or necessary research funds.

It appears, from the findings of this study, that New Zealanders' knowledge of basic food hygiene is less in comparison to people of other developed countries. For example, in preventing the dissemination of *Campylobacter* in New Zealanders' kitchens during preparation such as handwashing, use of different chopping boards, knives and utensils for cooked or ready to eat and raw food, or proper storage, in the kitchen, New Zealanders have scored very low in their knowledge about food safety or hygiene and reported self-practices.

In spite of the limitations of the self-reported practice surveys such as social desirability bias (reported frequently) with questionnaire techniques and telephone interviews rather than with a face to face interviews (Redmond and Griffith, 2003), these surveys can be useful for the evaluation of health promotion initiatives and/or to provide data about awareness and knowledge.

The most critically researched practices that prevent cross-contamination are hand washing and the use of clean and separate utensils for raw and cooked or ready-to-eat food during food preparation. Hand washing is considered the most important practice for preventing gastroenteritis (Mutters and Warnes, 2019). Hand hygiene has been demonstrated to result in reductions in gastrointestinal disease by 30-50% (Aiello et al., 2008, Curtis and Cairncross, 2003, Ejemot et al., 2008). The actual percentage of New Zealanders that cross-contaminate could be higher than reported in NZ studies, as the respondents mostly claimed that they performed the perceived correct practice. This is due to the Hawthorne effect. One observational study observed New Zealanders in the washrooms after using the toilets (Garbutt et al., 2007). This study was an attempt to determine how New Zealanders wash and dry their hands and for how long. It was reported that 13.3% of New Zealanders did not wash their hands after using the toilet (Garbutt et al.) This is a low observed percentage. The actual percentage could be lower as in this study, and all the participants were aware that someone was standing and observing their handwashing practice after using the toilet. The percentage of people who did not wash their hands after using the toilet may be due to the design of many NZ houses especially the older houses where the washing facilities are separate from the room containing the toilet. With this unique constructional and regulatory fault, New Zealanders have to walk to the bathroom or shower-room to access washing facilities for washing their hands after using the toilet. This unique design of older homes in NZ may have contributed to the increased number of participants who did not wash their hands after using the toilet in Garbutt et al. (2007) study. Generally, the lack of a washing facility in toilets may also contribute to the general habit of washing hands in the other five critical instances of washing hands. Thus, the number of patients with Campylobacter, introduced to homes via purchase of chicken, has possibly increased dramatically in the last 20 years in NZ.

In 2007, the NZ Food Safety chief executive considered the level of hygiene practice of New Zealanders to be the reason for the high rate of NZ campylobacteriosis (McKenzie, 2007). In 2008, during an interview with a NZ Radio Station, a prominent NZ and International *Campylobacter* expert (Nigel French) indicated that if New Zealanders improved their practice of washing their

hands, the rate of campylobacteriosis would decline dramatically and would consequently solve the campylobacteriosis problem in NZ. In 2007, a member of the NZ parliament indicated that New Zealanders were too lazy to wash their hands, preferring to wring their hands more often than they washed them, and warned New Zealanders that if they did not wash their hands, they would become ill (Coddington, 2007). Thus, more research on the behaviour of New Zealanders is needed for NZ researchers to understand and address these issues comprehensively in the future.

However, most of the evidence collected in this study appears to confirm the hypothesis that New Zealanders are not as good in food hygiene compared to people in other developed countries. This may have possibly contributed to NZ having the highest rate of campylobacteriosis amongst developed countries. The other reasons which may contribute to the higher rate in NZ are the prevalence of and contamination level of carcasses produced in NZ, leaky packaging of chicken sold in NZ, and inadequacies in the reporting system.

If the present large number of campylobacteriosis increase further, this would result in an increased cost to the health care system, could affect the NZ tourism industry and also agricultural produce since both champion the New Zealand's "clean green" image when advertising abroad.

#### 12.4 Conclusions

Cross-contamination is considered both internationally and in NZ to be the main factor in *Campylobacter* transmission of. A lack of knowledge concerning food safety during domestic food preparation is prevalent amongst NZ consumers. Most of the evidence collected in this study confirmed to some extent the hypothesis that New Zealanders have poor food hygiene practices and food safety knowledge than those in other developed countries, and this is possibly is the reason for the higher rate of campylobacteriosis in NZ than in other developed countries. The practice of handwashing and the separation of raw and cooked or ready-to-eat food during preparation and storage needs to be addressed in any new health promotion initiative. These practices are also crucial to exploring why New Zealanders may deviate from the best microbiological food safety practices and for improving NZ consumers' home practice to control campylobacteriosis. Further research is required to investigate the extent of these unsafe practices and to review the legislation or codes for building homes in NZ so that the habit of handwashing can be encouraged.

## 13 Consumer poultry handling practice survey

#### 13.1 Introduction

The acknowledgement of the importance of adequate consumer food-handling practices has been discussed in the previous chapter(s), and it is widely recognized as the last line of defence against foodborne illness (Nesbitt et al., 2014, Redmond and Griffith, 2003). The prevention of foodborne illness requires the collaboration of all members of the food continuum, from farm producers to consumers. Therefore, research and consumer education regarding the risk of food safety malpractices is an important element in precluding food-borne diseases. It is rational to expect that a fraction of the human exposure to *Campylobacter* spp., specifically, originates from cross-contamination that occurs in home kitchens during food handling (Brynestad et al., 2008, Hartnett et al., 2001, Lindqvist and Lindblad, 2008, Mylius et al., 2007, Nauta et al., 2007). Therefore, the number of human campylobacteriosis cases can be reduced by reducing the degree of *Campylobacter* spp. cross-contamination by improving hygiene in kitchens (Kusumaningrum et al., 2004, Jones et al., 2016).

Bruhn (2014) reported that personal hygiene is insufficient if 65% of the participants do not wash their hand before the meal preparation, and 40 % not wash their hands after handling raw chicken. (Maughan et al., 2016) reported that about 40% wash their hands after handling the chicken breast. They also reported that only 37% of consumers used a thermometer to check the 'doneness' of the chicken breast. Evans and Redmond (2018) observed the handling practice of raw chicken by consumers who are over 60 years old in a model kitchen and reported that older adults did not obey safe food handling practices with 90% failing to implement adequate hand decontamination immediately after handling raw chicken. Diplock et al (2018) investigated the effectiveness of food handler training programs to improve safe food handling behaviour among high school students. This study concluded that despite a significant increase in the correct behaviour, students continued to use risky practices post-intervention, suggesting that the risk of foodborne disease remained. Several studies (Kosa et al., 2015, Sterniša et al., 2018, Koppel et al., 2016, Koppel et al., 2015, Bearth et al., 2013) have addressed consumer reported handling practices of poultry products. All these studies indicated that consumers are not good at safe food handling leading to cross-contamination, caused by improper refrigerated food storage and thawing, undercooking, and general hygiene practices. Recently, (Katiyo et al., 2019) assessed the practice and the knowledge of South African consumers with respect to the safety risks associated with handling raw chicken meat. This study also confirmed unsafe practices during chicken purchasing, thawing, and handwashing by consumers with 62 % not following safe practices during the handling of raw chicken. (Bearth et al., 2014) researched on consumer risk perception and knowledge related to campylobacteriosis and revealed that lack of food safety knowledge and personal risk perception were the central reasons for food safety violations during poultry preparation. (Donelan et al., 2016) investigated the consumer poultry handling behaviour in the grocery store and in-home storage and found that there was no hand sanitizer or wipes available in 71% of grocery store meat sections of the stores visited. Plastic bags were available in the meat section, 85% of the time, but only 25% of the shoppers used the bag for their raw poultry purchases.

#### 13.2 New Zealand consumer studies

In New Zealand, there is a lack of research and also data related to studies on consumer knowledge and handling practices. The few studies conducted in New Zealand (Hodges, 1993, Kerslake, 1995, Gilbert et al., 2007) have indicated that there is lack of knowledge of food safety with 28-41% of New Zealanders re-using knives and food surfaces without washing between preparation of raw and cooked food. However, these studies are not that comprehensive. Thus, the data extracted from these limited studies should be treated with caution. There is no previous study in New Zealand which targeted consumers self-reported practice of handling poultry purchase, transport, storage, and food preparation to provide an insight into campylobacteriosis risk factors. Due to the large economic and health consequences of campylobacteriosis, which has been estimated to be ~ NZ \$ 50 million (Duncan, 2014) urged introducing intervention programmes to reduce the disease rates. Thus, it is clear that there is a need for a more in-depth investigation of safe food preparation at home. Moreover, the chief executive of the -NZ Food Safety Authority revealed that cross-contamination during chicken preparation at New Zealand home as a possible hypothesis to explain the high reported rate of campylobacteriosis (McKenzie, 2007). Thus, it is necessary to test whether the high rate of campylobacteriosis in New Zealand has a significant link to malpractices by New Zealanders in their home kitchens.

Laws and regulations cannot enforce controls and measures to prevent unsafe practises at home, although such controls can be applied to commercial food processors and retailers. The only possible way to change consumer behaviour is through education. To change behaviour, it is important to make people aware of the consequences of their behaviour. A lack of food safety knowledge and gaps in applying knowledge into practice are major obstacles food handlers have to overcome to reduce food contamination effectively. To educate food handlers, it is important

to conduct formative research to determine basic knowledge, perceptions and information of the target audience, formulate a clear food safety message and use diverse platforms to convey the information.

The objective of this research was to fill the data gap in NZ regarding poultry handling practices by consumers. The study will provide an insight into the relationship between consumer knowledge and poultry handling practices. This study will investigate the best hygiene measures that can be used to provide public health professionals, risk managers and regulatory authorities with scientific support for future health promotion plans that will discourage consumers from unsafe poultry preparation practices. It is aimed to determine what consumer groups have inadequate food safety knowledge and hence most likely to practice unsafe poultry preparation at home. Thus, the study goal was two-fold. 1) Assess which information poultry consumers lack and hence should be included in future food safety communications, and to 2) Identify which groups are at more risk so that future food safety education programme can be targeted towards them. The handling of poultry meat preparation data will also be used as an input for further optimisation of a risk assessment strategy for campylobacteriosis in NZ from consumption of knowledge of campylobacteriosis incidents by giving an updated insight on the assessment of knowledge of people on food safety and their practices on preparing poultry at home.

#### 13.3 Methods

Several research designs are available to gather information on consumers' food preparation practices. The main design, which is used in this study, is described briefly below:

#### 13.3.1 The quantitative design

This method tests objective theories by examining the relationship among variables which are measurable (Creswell, 2018). Quantitative questions will result in data that is easy to convert into objective, numbers-based analysis. Survey research (non-experimental) and experimental research are the main strategies for conducting the research design. Survey research provides a numeric or quantitative description of opinions, attitudes, trends of the population by studying a sample of that population. It uses a questionnaire or structured interviews for data collections. The answers to the questions can be assigned numeric values. These numbered data can be analysed using statistical methods. Quantitative techniques include cross-sectional studies which collect data only on time and longitudinal studies which collect data over an extended period of time. A questionnaire, with closed-ended questions, structured interview methodologies are appropriate

when the issue is known, relatively explicit and straightforward, and can be validated with reliable measurement.

#### 13.3.2 The study approaches

The study will use a quantitative survey design, a strategy that has been commonly used for consumer poultry handling investigations involving large populations (Kosa et al., 2015, Sterniša et al., 2018, Koppel et al., 2016, Koppel et al., 2015, Bearth et al., 2013).

A self-reporting questionnaire on consumer behaviour can be administered on a large sample in a short period of time. Thus, interpretations can be made about the behaviours, opinions and attitudes of a population. Moreover, the self-reported questionnaire is a more economical choice than observational study approach. However, a self- reporting questionnaire can show a disparity between observed and reported behaviour (Evans and Redmond, 2018, Redmond and Griffith, 2003). A self-completed survey can provide information on how consumers may apply their knowledge during food handling (Cody and Hogue, 2003). However, specific consideration needs to be given to the structure of the questions to overcome the problem of disparity between the consumer reported practice and actual practice. For example, questions need to be asked the participants on what their practical behaviour is in the kitchen rather than statements that deal with safe food handling. Many of the previously reported consumer food handling studies had used the Likert scale option statements in their questionnaire, such as strongly agree, agree, undecided, disagree, strongly disagree. Moreover, the participant in this study was not directly confronted with the notion of general hygiene, cross-contamination, storage and temperature control but indirectly through questions about their actions and deeds at the last occasion they purchased raw poultry or prepared raw poultry, what utensils were used, what was the status of the utensils they used during preparation of raw poultry or what they did immediately after handling or cooking raw poultry.

#### 13.3.3 Questionnaire development

The biggest challenge to obtain accurate and reliable information about consumer practices and behaviour concerning food safety from studies using self–completed questionnaires, is the design of the questions to guarantee aimed and accurate responses. To do so, it is essential to use language that is understandable to the type of respondents whom you are targeting.

A structured questionnaire is designed based on suitable and relevant questions from previously validated questionnaires applied in similar studies (Byrd-Bredbenner et al., 2007b, Mazengia et al., 2015, Bearth et al., 2014, Kosa et al., 2015, Sampers et al., 2012). The respondents need to be requested to be sincere and encouraged to report their actual behaviour. The innovative method

of a street-intercept survey of main public places such supermarkets (Pak and Save, New World, Countdown) and major public libraries in main district and suburbs of Canterbury region were combined with the distribution of the survey via email (containing an embedded electronic link to the survey) to the Lincoln University academic and non-academic staff. The questionnaire was administered as an online survey using Qualtrics (Qualtrics, Utah). When a print version was used in the survey at supermarkets and public libraries, the respondents were requested to return it on completion on the spot. In addition, the participants who wanted additional time were a postagepaid pre-addressed envelope or to use the electronic link and complete the questionnaire at their convenience.

#### 13.3.4 The Questionnaire

The questionnaire comprised of 31 multiple-choice questions designed to collect information on self-reported practice during the last time they purchased and prepared raw poultry at home. The first four questions in the first section consisted of respondents' food purchasing habit, such as how/where/when they purchased poultry products during their shopping. This was followed by ten questions in the second section, about the respondents' food preparation practices. These questions were designed to address the sequence of events that occurred the last time poultry food was prepared and cooked at home. This was to encourage respondents to report their actual practice rather than their perceived practice and to help minimize biases associated with selfreporting and social desirability. For all questions, respondents selected an answer from a list of five to six answers to avoid signalling respondents to the "correct" response. The questions collected information on respondents' compliance to the recommended food safety practices of the WHO's five keys for food safety manual (Fontannaz-Aujoulat et al., 2019), U.S. Department of Agriculture (USDA) guidelines for handling, storing, and preparing raw poultry at home or following the 3 Cs (clean, cook and chill) food safety tips recommended by New Zealand Food Safety Authority (NZFSA) (Anonymous, 2019). Specifically, there were four questions related to personal food hygiene practices, five questions on prevention of cross-contamination practices, and another five questions related to food temperature control and proper food storage practices. The third section consisted of seven questions related to the respondents' basic food safety awareness, such as awareness of clean, separate, cook, chill cover rules, or the knowledge of whether NZ chicken carries bacteria etc. Each question answered correctly would award the respondent one point and zero for incorrect answer. The maximum score was 21.

The fourth section consisted of 10 questions to obtain information about how frequently poultry was cooked, their source of food safety information, demographic information including gender,

age, education level, occupation and household income etc.

Consent was obtained from the respondents, and they were informed of the research topic and the anonymity of data. The survey was pilot tested with 20 participants with different education level and background for time estimation, question comprehension, and clarity. Face validity and content validity was established by NZ and international experts who agreed that the questionnaire was a valid measure of the 'poultry handling practices and food safety awareness' concept following modification and refinement of the questionnaire. Data collected from the pilot testing was not included in the final data set. A copy of the questionnaire is in appendix G.

#### 13.3.5 Target participants

The participants of the study were people over the age of 18 who lived in Christchurch and its suburbs which has a high incidence of campylobacteriosis notification rate in NZ with restrictions on that they can understand the English questionnaire and handle and cook poultry at home.

#### 13.3.6 Data collection

The participants were selected at random and were approached both in-person with a print version of the survey or an electronic link to the survey which was sent out for participation and completion of the survey. The print version was to be returned on the spot after it was completed, or the respondents were given a postage-paid pre-addressed envelope who preferred because of time constraints. Participants who preferred to fill the electronic version through a tablet/laptop at the site and sometimes or later at their convenience at home were given a link to the survey. The survey was conducted between November 2019 and January 2020.

#### 13.3.7 Data analysis

Data collected via Qualtrics was downloaded into SPSS software for analysis. The survey data were analysed using descriptive statistics (e.g. frequencies,) percentages of correct and incorrect answers for each question. Additionally, each sub-section mean scores of the survey and standard error were calculated. Several normality tests of the data were performed for the data. The z values test of data skewness and kurtosis, Shapiro-Wilk test, Kolmogorov–Smirnov test, histograms, normal Q-Q plot and box plots were the tests performed to analyse the data normality. One-way ANOVA test was used to compare the mean scores between the respondents with a different type of socio-demographics. For all analyses, a p-value < 0.05 was considered statistically significant. A correlation between different sections of the questionnaire was also assessed by the Pearson test.

#### 13.4 Results and discussion

The questionnaire answers were recorded as 1 for a correct answer and 0 for an incorrect answer so that the data can be analysed quantitively as binary data. The reliability test for dichotomous data set Kuder Richardson 20 was used for this study data set. (Vaucher et al., 2012). This calculations used in this test was similar to that used in the Cronbach alfa test, and the mean value for the 21 questions was 0.707. This value is acceptable to confirm the internal consistency, that is, how closely related the tested questions were. All data normality tests confirmed that the data was normally distributed. Therefore, a parametric analysis was considered as the most appropriate method for analysing the data (Sullivan, 2016). Specifically, one way ANOVA was used to determine the impact of the respondents' detail and demographic characteristics on the response scores. For all analyses, a p-value < 0.05 was considered as statistically significant. The number of valid responses was 301, and the mean score was 9.8306 with a SD value of 3.49779 and a SE of 0.20161. The maximum score was 19, and the minimum 2. Since the respondents were randomly selected, some demographic percentages of groups participation were very low due to the small number of the respondents in those groups. So, the data obtained from those groups may not truly reflect the poultry handling practices of these very underrepresented groups in the study sample. Therefore, their results are represented only in the tables for future research and excluded from the detailed analysis and conclusions of the study. Outputs of SPSS analysis are in appendix G.

# 13.4.1 Association between respondents' socio-demographic characteristics and mean score

All the respondent details, demographic characteristics, descriptive analysis, and mean score for each group are summarised in Table 13.1. Approximately 40% of the respondents had the postgraduate qualification, and another about 40% had a university degree or students at a university. The majority of respondents were NZ European (69%). More than 80% of the respondents cooked poultry at home 1 - 3 times per week. Approximately 30% of the respondents had experienced food poisoning 1 - 4 times during the last year. Previous research has shown that demographic factors influence domestic food safety behaviour (Fischer and Frewer, 2008, Al-Sakkaf, 2015). Table 13-1 Respondents' details and demographic characteristics

Respondents' Characteristics	Category	Respond ents (n.)	Percenta ge	Mean	SE
Gender *	Male w/o children	53	17.60	8.76	0.46
	Female w/o children	105	34.90	10.15	0.35
	Male with children	23	7.60	8.61	0.73
	Female with children	117	38.90	10.20	0.31
	Prefer not to answer	3	1	12.67	1.20
Age	<20 years	3	1	7.33	0.89
	20-29 years	42	14	8.69 <sup>a</sup>	0.44
	30-39 years	67	22.30	9.50 <sup>ca</sup>	0.43
	40-49 years	60	19.90	10.20 сь	0.45
	50-59 years	55	18.30	10.20 <sup>cb</sup>	0.46
	60 or older	74	24.60	10.30 cb	0.44
Which of the following	NZ Maori	19	5.90	9.79	0.83
categories describe your ethnic identity? (Please select	NZ European	208	65	9.96	0.24
all that apply)	Other European inc Australian	35	10.90	10	0.58
	Pasifika	6	1.90	8.5	1.65
	Asian	32	10	9.47	0.69
	Latin American	4	1.30	7.75	1.44
	North American	9	2.80	9	1.27
	Middle Eastern	5	1.60	9.2	1.24
	African	2	0.60	8	2
How many occasions in the	1-2 times	69	22.90	9.52	0.42
in your family experienced food poisoning symptoms such as diarrhoea,	3-4 times	16	5.30	8.50	0.84

nausea/vomiting, sudden	5-6 times	3	1	7.33	0.67
aches, lack of energy, dry mouth and tongue?	No experience in past 12 months	87	28.90	10.25	0.39
	No experience in past two years	126	41.90	9.94	0.30
Based on your practices and	Partner	28	9.30	8.07	0.70
your behaviour, who or what has most influenced your personal hygiene habits? *	School/church/leisur e centre/educational campaigns	11	3.70	9.54	0.82
	Doctor, counsellor or health clinic	2	0.70	11.50	1.50
	Parents/grandparents , other relatives or friends	67	22.30	8.79	0.39
	TV, newspaper, radio, media in general	32	10.60	8.84	0.51
	Work colleagues	5	1.70	11.40	0.60
	My knowledge	156	51.80	10.74	0.28
How often do you cook	Six to seven times a	1	0.30	9	0.00
poultry at nome:	week	13	4.30	10.00	1.26
	Four to five times a week	112	37.20	10.08	0.34
	Twice to three times a week	110	36.50	9.75	0.35
	Once a week or in the weekend	65	21.60	9.52	0.34
	Once to twice times per month				
In general, how did you learn to cook? (Please select all	By myself (from internet; videos,	144	25	9.39	0.29
that apply)	article, recipes)	101	17.50	10.12	0.34
	From cookery books/TV	82	14.20	9.52	0.41
	From partner/friends	201	34.90	10.26	0.24

	From other relatives (parents, grandparents)	48	8.30	10.44	0.55
	From a training course (School, church, community centre, etc.)				
What is your highest	No formal schooling	2	0.70	6	0.00
education level?	Kura Kaupapa/Primary school (including intermediate)	2	0.70	9	1
	Secondary school (high school)	53	17.60	9.55	0.52
	University student or completed university, Wānanga, polytechnic or another tertiary	130	43.20	10.32	0.31
	Postgraduate or higher qualification	114	37.90	9.49	0.31
Which of these categories best describe your	Employed — full time (+ 30 hours week)	145	48.20	9.78	0.31
occupational status?"	Employed – part-time (15-30 hours week)	49	16.30	10.61	0.43
	Retired	51	16.90	10.57	0.51
	Unemployed	10	3.30	7.80	1.11
	Housewife/husband /home duties	14	4.70	9.71	0.85
	Student	32	10.60	8.38	0.46
Which one of the following categories best describes the total yearly income of	Lower than \$20,000/yr.	29	9.60	8.38 <sup>a</sup>	0.46
everyone in your household	\$20,001 - 40,000/yr.	40	13.30	9.93 b	0.54
from an sources before tax!	\$40,001 - 60,000/yr.	55	18.30	9.91 в	0.48
	\$60,001 - 80,000/yr.	35	11.60	9.31 b	0.63
	\$80,001 - 100,000/yr.	40	13.30	9.78 <sup>b</sup>	0.59

\$100,001 or more	102	33.90	10.42 в	0.33	
-------------------	-----	-------	---------	------	--

\* Denotes significant difference in the mean scores within all groups, p < 0.05 by one-way ANOVA

test.

Means with different superscripts within columns, for each score category, were significantly different based on the bootstrap method.

#### Gender

There were significant differences in the mean scores observed between the respondents' gender groups (p = 0.017). The mean score of the female with children (10.20) was marginally higher than the mean score of females without children at home (10.15). Those who lived in a family prepared food more safely than those living in a single-person household or the presence of children in the house increased the food hazards concerns (Anonymous, 2012, Ruby et al., 2019). About 25% of the female respondents were > 50 years old, and it was assumed that their children might have grown up and left their home. Therefore, in general, female food safety practices at home and their awareness about food safety were better than males without children at home or with children at home with values of 8.75 and 8.61, respectively. Females tend to judge health risks as having a higher potential of danger than men because females are socialised to be mothers and nurturers (Dosman et al., 2001, Ruby et al., 2019). This finding is in agreement with other studies (Fein et al., 2011, Moreb et al., 2017). New Zealand surveillance data for the Canterbury region revealed that males were more at risk than females. The notification rates for the years 2016 and 2017 were 177 and 152 per 100,000 males respectively, and 141 and 119 females respectively.

#### Age

There were no significant differences between the mean score of the different age groups of the respondents. However, the mean scores of the younger groups, between 20 to 29 or years 30 to 39 years old were 8.69 and 9.50 respectively, lower than the mean score of the 40-year-old groups (10.25). This was expected as the older people were more concerned about food safety practices and hazards. The older age groups were more aware of the food safety issues and also better experienced at safe handling of poultry products compared to the younger groups. This study finding is in agreement with other studies (Katiyo et al., 2019, Kosa et al., 2019). Older people cooked more safely than younger people (Kennedy et al., 2005b). Older people probably learned cooking in school as part of the educational curriculum in the past (Fischer and Frewer, 2008).

Young consumers have less food safety knowledge, and their practices need urgent improvement (Sanlier, 2009). It was reported that age is one of the most important factors in determining public understanding of food safety practices (Mahon et al., 2006). In NZ, the most campylobacteriosis affected population groups were adults between 20 and 39 years old and children < five years (Figure 2.1) (EpiSurv, 2019).

#### Ethnicity

The mean score of European New Zealanders is 9.96. This mark is below 50% of the total score marks, which is 21 marks. The European New Zealanders represented the majority of the study respondents, with 69% of all respondents. The European New Zealanders represents 66% of the population. The is the majority of the New Zealand population. This indicates that poultry handling practices and general food safety awareness of the European New Zealanders were below the average level. This finding is in agreement with the result of previous chapters in this study, (Al-Sakkaf, 2012) and a previously conducted survey in NZ (Gilbert et al., 2007). It is interesting to note that the mean score of other Europeans and Australians (10.00) was higher than the NZ European mean score. This finding is in agreement with the findings in the previous chapters and the previous study (Al-Sakkaf, 2012, McKenzie, 2007). The other Europeans and Australians respondents' of this study proportion is 11.6%. This is slightly higher than the actual representation of the other European and Australian population in NZ which is only 6.8%. However, the other Europeans and Australians proportion in Canterbury region is 7.6%.

Similarly, the European New Zealander respondents' proportion in this study was a slightly higher than the percentage New Zealand population (65%) but less than the proportion in the Canterbury region (76.4%). However, The Maori respondents' proportion (6.3%) was below the national proportion population (16%) but closer to their Canterbury region proportion value (9.4%). The Asian respondents' proportion was (10.6%) close to the percentage in the Canterbury region (11.1%). The 2018 NZ surveillance data showed that New Zealanders of European origin (4,977) were the most affected by campylobacteriosis then the Maori (577) and the Asians (367). This is probably due to the overall higher representation of New Zealand with Europeans origin in NZ. Another reason for the higher rate of reported campylobacteriosis cases among Europeans maybe because they access the public health system for food poisoning than other ethnicities.

#### History of food poisoning

There was no significant difference in the mean score of the number of occasions the respondents' experienced symptoms of food poisoning in the past year or the last two years. Only 1 % (3 respondents) of the respondents reported that they had experienced food poisoning 5 - 6 times in

the previous year and > 5% (16 respondents) reported that they had experienced food poisoning 3 - 4 times in the previous year. However, the NZFSA data reports about 200,000 food poisoning cases in New Zealanders each year (Anonymous, 2019). Over-reporting of "good" and socially desirable behaviour is a well-known phenomenon in self-reported food safety surveys (Dharod et al., 2007, Redmond and Griffith, 2003, Wills et al., 2015). Evidentially, up to 80% of food poisoning cases are acquired at home (van Asselt et al., 2009).

#### Hygiene habits

There were significant differences in the mean scores observed between the respondents' answers about who or what had most influenced their personal hygiene habits (p. 0.000). More than 50% of respondents reported that their knowledge had the most effect on food hygiene habits, and the highest mean score in the group was 10.74. This is a relatively high score for this group and could be interpreted as those respondents of this group mostly sought information about food hygiene and food safety. Another explanation might be that repeated food safety incidences in recent decades in NZ and internationally had raised the respondents' concern of food hygiene and food safety. Either way, the consumer interest in food hygiene and food safety signifies a prospect for public health professionals to target the consumers with new food safety messages or campaigns about risk prevention at the domestic level. The lower mean score groups than the 'My knowledge' group were the respondents who indicated that media (TV, newspapers, radio) and their parents/grandparents or other relatives or friends influenced their food hygiene habits which achieved (8.84) and (8.79) respectively. The lowest mean score group was the group who selected that their partner influenced their food hygiene habits with a score of 8.07. As far as it is concerned, the level of the score and data show that those who identified themselves as their knowledge has influenced their food hygiene habits with the highest mean score within all groups, their score was at the average or the mean score level (10.5). This is a crucial finding, as this target group and the other groups might be unaware of the good food hygiene practices or appropriate food safety knowledge. However, the group with the group of media would benefit from food safety messages or awareness communication campaign used the media channel. For example, messages targeting consumers should include information and education and aim at increasing consumer awareness of good hygiene practices and food safety knowledge, and these must be translated into behavioural changes (Maia et al., 2019). An idea of what might influence the selection of food safety information sources by consumers would be crucial to design a food safety message and also to enhance information provision by a public health professional and health promoters (Tiozzo et al., 2019). The findings of this study will assist in determining the channels which are likely to be the most effective in educating the consumers.

#### Frequency of poultry preparation

People who prepare food least frequently are the most dangerous cooks (Byrd-Bredbenner et al., 2013). Conversely, the people who prepare chicken more often at home would be better at cooking food safely. The results of this study indicated that there was no significant difference in the mean score of all groups. However, the score of those who reported that they prepare poultry more frequently, in general, was marginally better than the score of the groups who prepared poultry less frequently across all the main groups included in the analysis. It is noteworthy to reiterate that the highest score of the respondents in this group with 10.08 was still below the average score (10.5).

#### Learning to cook

It is known that learning to cook also involves learning to prepare food safely with the proper handling, preparation and storage of food. For many people, cooking can be described as a habitual behaviour because it is a frequently repeated process (Fischer and De Vries, 2008). Many cooks, parents and individuals may exhibit bad food-handling habits. Thus, people who learn to cook from those with bad habits will also exhibit similar behaviour. It is sometimes difficult to change such behaviours. The results of this study revealed there was no significant difference in the mean score of each of the above-mentioned groups. However, the highest mean score was achieved by the group who learned cooking through a training course (school, church, community centre). This group was under-represented in this study, just 8.3% of the total number of respondents. The majority of the respondents (35%) learned to cook from parents and grandparents. The home has always been the main place to learn about cooking and food safety mostly by watching others. However, it was apparent from this study that the knowledge of parents and grandparents were not adequate as the mean score was below the average score (10.5).

Similarly, the group who learned cooking by following cookery books/TV shows did not achieve the average mean score either. It appears that TV cooking shows or cookery books rarely stress on food safety practices. The lowest mean score was for those who learned cooking from partner/friends or by themselves from the internet (videos, article, recipes). It is important to teach food safety principles, in-school cooking classes since the behaviour of individuals is mostly influenced by teachers during school years. It has been reported that the basic food safety principles learned before the age of 16 years have an impact on individuals' food safety behaviour (Lange et al., 2018).

#### Education level

The majority of respondents (81%) in this survey had a university degree or was studying at a university. Such a high proportion was expected since the survey was distributed to university staff

and academics (117 respondents from a university out of 301). Moreover, high-level education may have been a reason for wanting to participate in a survey that could contribute to the improvement of consumer food safety in NZ. In general, there were no significant differences between the mean score of all the respondent groups and their education. Surprisingly, the mean scores of the groups with a tertiary qualification or postgraduate qualification were below the average score.

Interestingly, the mean score (9.49) of the group with post-graduate qualifications was lower than the mean score of the group with a university degree or studying at university (10.31). Several studies have reported that individuals with a higher level of education were less worried about food hazards or practised less food safety measures at cooking (Al-Sakkaf, 2015, Katiyo et al., 2019, Fischer and Frewer, 2008). The reason for the lower mean score of the higher education level group was probably because they failed to use their knowledge into practice due to psychological factors such as optimistic bias and habit (Al-Sakkaf, 2015). Another study indicated the higher educated group might habitually cook less because someone else prepared the food for them in restaurants, takeaways, delicatessens or departmental stores etc. and therefore they lack the knowledge of food safety during food preparation (Fischer and De Vries, 2008). In contrast, several studies have reported that individuals with a high level of education were more concerned about food hazards or safe poultry handling practices(Nesbitt et al., 2009, Sterniša et al., 2018, Moreb et al., 2017).

#### Occupation

There is a significant difference in the mean scores of the groups of the respondents based on their occupation status. The mean score of the part-time working group and the retired group were similar 10.61 and 10.57, respectively. These are the only groups which gained the average mark threshold (10.5). The full-time working group score was 9.78, which is below the mean score mark threshold. Previous studies found that work status played a significant role and added that time investment or current lifestyle prevents consumers from following best practice (Brennan et al., 2007, Gkana and Nychas, 2018). Generally, increased working hours or the increased number of women working reduced significantly the time spent at home and in food preparation. The student groups were with the lowest mean score in the group, with a mean score of 8.38. This result is in agreement with other studies which have confirmed that students have low perceptions of health risks, lack the knowledge in food safety and safe food preparation practices (Courtney et al., 2016, Her et al., 2019, Lazou et al., 2012).

#### Income

There is a significant difference observed between the mean score of the lowest income group (<

NZ \$ 20,000) and the higher income groups. This finding is in agreement with the reports of (Gong et al., 2016, Kwon et al., 2008) who reported income as an important factor in an individual's food safety knowledge and proper poultry handling practice. However, surveillance data has confirmed that incidence of campylobacteriosis is higher among the high-income groups than those in earning a low-income (Nichols et al., 2012, Spencer et al., 2012, Taylor et al., 2012, Newman et al., 2015). Income is not reported in NZ surveillance data for the notified cases of campylobacteriosis, and there are only limited studies which have addressed the spatial distribution based on social deprivation area notification. (Spencer et al., 2012) explained that the lower notification rate from deprived urban areas in adults and in children > 6 years as could be explained as only <6 years children would not be charged a fee for visiting a general medical practitioner (GP) which is regarded as the main route of notification. Campylobacteriosis incidence in deprived urban areas for the children <6 years old was higher than in those from the higher-income areas (Newman et al., 2015, Nichols et al., 2012). However, other studies (Nesbitt et al., 2009, Tonsor et al., 2009) have shown that individuals with higher incomes are less worried about food safety, engage in more unsafe practices or have lower risk perception or attitudes than those with a lower income. Proper food-handling skills may be obtained through practice and experience with handling food. For example, if individuals with higher income levels did not prepare meals as regularly as individuals with a lower income level, it is possible that more risky behaviours would be observed among the higher-income individuals (Nesbitt et al., 2014) A few studies found no association between income and the consumers' food safety knowledge and safe preparation practices (Moreb et al., 2017, Pearl et al., 2009)

#### 13.4.2 Consumer food hygiene practices

The mean score of food hygiene practice for respondents is summarised in Table 13.2. It is noted that the respondents scored the lowest value of the mean score in this section (0.25) than any other section of the questionnaire. The majority of the respondents have only been able to answer only one question correctly out of the four food hygiene questions. This result confirmed previous study findings (Al-Sakkaf, 2012, Gilbert et al., 2007)of poor hygiene level of New Zealanders in general. Surprisingly, only < 17% of the respondents selected the cleanness of the premises in their decision about where to shop for their food. Despite that, the majority of the respondents (80%) earned > NZ\$ 40,000 / annum and a similar percentage (80%) had tertiary education level or higher. Only 13% mentioned that they washed and dried their hands correctly. However, Gilbert et al. (2007) study indicated that about 26.4% washed their hands and dried them properly. However, the format of the questions in this study and Gilbert et al. study (2007) was different. For example, the

washing and the drying period for the Gilbert et al. study were only 10 seconds. The question posed in this survey was 20 seconds or more for washing and drying. In both studies, the percentage of the respondents who knew how to wash and dry their hands properly was low compared with other developed countries (Al-Sakkaf, 2012).

Reported practices of personal food hygiene	Correct answer	Respondents (n.)	Percentage (%)
Which of the following factors are most important to you when deciding where to shop for food?	The cleanliness of premises	51	16.90
For each hand washing occasion, how long do you spend washing your hands?	More than 20 seconds	41	13.60
If you wash your hands, how do you usually dry your hands after hand washing in the kitchen?	With a disposable paper kitchen towel	39	13
How often do you clean (using soap or cleaning product) your kitchen sink and bench-top?	After every meal	167	55.50
Mean score		24.75	
SE		1.36	

Table 13-2 Score on reported practices of personal food hygiene

#### 13.4.3 Consumer cross-contamination prevention practices

The mean score of cross-contamination prevention practices is summarised in Table 13.3. The mean score (0.55) marginally improved in this section than in the previous hygiene section as the majority of the respondents were able to select the correct answer for more than two questions but still less than three questions out of a total of five questions. It is only 25% of the respondents who mentioned that they separated raw poultry from other food items they purchased at the supermarket in their carry-on bags and their refrigerators. There was no similar question in Gilbert et al. (2007) study to this question. However, 35% of the respondents reported that they follow safe practice after handling raw poultry to prevent cross-contamination. However, Gilbert et al. (2007) reported that about 60 % of their respondents handled the knife used for poultry and cleaned the surfaces in a way to prevent cross-contamination. In this study, 78% of the respondents correctly answered the specific question about their handling of the cutting board and knife. The content and the structure of the questions in both surveys were not similar. This makes the comparison between the respondents' answers in the two surveys difficult to compare. It is evident

that despite the improvement in the respondents' answer to some of the questions of this section, the overall score of this section was only marginally above the average score. Cross-contamination is considered to be the main factor in the transmission of campylobacteriosis (Mylius et al., 2007).

Table 13-3 Score on repo	orted practices in	cross contamination	prevention

Reported practices in cross contamination prevention	Correct answer	Respondents (n.)	Percentage (%)
Was the poultry bagged separately from other items?	Yes, in the trolley, carry-on bag and freezer/chiller	75	24.90
For each raw poultry handling occasion, which of the following do you usually do immediately after handling poultry?	I use hand sanitiser and water or soap with hot water OR change my gloves	106	35.20
There are many times that raw poultry may be handled or came into direct contact with utensils during preparing or grilling. Please indicate how the raw poultry was handled/came into contact with utensils?	In contact with a previously unused fork/ unused spoon/unused utensil OR handled while wearing previously unworn/gloves OR handled using clean hands.	137	45.50
After cutting or preparing the raw poultry, what was the next thing you did with the cutting board/utensil and knife?	Wash the cutting board/knife with soap and hot water before reuse OR use a different cutting board and knife for preparing the other food	235	78.10
Do you ever place cooked poultry on the same plate/surface where raw poultry meat has been?	No, not at all OR only after washing the plate/surface with hot water and	278	92.40
Mean	uctergent.	55.22	
SE		1.30	

#### 13.4.4 Consumer food temperature control and proper storage practices

The mean score of food temperature control and proper storage practices and the descriptive analysis are summarised in Table 13.4. The mean score (0.49) in this section with five questions is slightly lower than the consumer mean score of the cross-contamination practices section. Only about 5% of the respondents purchased poultry products after purchasing other non-perishable items. The majority followed the flow of the supermarkets. Only a few of the supermarkets in NZ have the appropriate flow design to help consumers to choose perishable food items in the last section. Regarding the question of how long it takes the respondents to reach home after purchasing the poultry products, in this study for 88 % it took < 40 mins which were somewhat similar percentage (88.9%) which reported by Gilbert et al. (2007) study. In terms of thawing the frozen poultry products, 34.2% of the respondents of this study followed the best practice. The respondents of Gilbert et al. study (2007) reported of 25.8% in the refrigerator for up to 12 hours and 4.1% in the sink with cold water. It is not possible to thaw a whole chicken within 12 hours. Therefore, the respondents' of both studies confirmed that they do not thaw safely and thoroughly before cooking. Proper thawing of frozen raw poultry products can minimize the growth of bacterial pathogens and reduce the risk of cross-contamination (Mazengia et al., 2015). Only 47% of the respondents knew when the poultry was properly cooked and safe to eat. 27.3% of respondents of Gilbert et al. (2007) study reported that they preferred medium cooked, roasted chicken and 22.1% of the respondents preferred medium cooked chicken liver. The consumption of undercooked chicken or chicken liver is considered a significant risk factor for campylobacteriosis (Butzler, 2004, Wensley and Coole, 2013).

Reported practices in food temperature control and food storage	Correct answer	Respondents (n.)	Percentage (%)
During your last grocery shopping occasion when you purchased raw poultry, at what stage did you select raw poultry?	After, I had chosen all other non- perishable items	16	5.30
Usually, how long do you leave the raw poultry at room temperature before storing it in a refrigerator or freezer at home (including the time you take to travel from the shop to your home)?	0 minutes– 40 minutes	265	88
When thawing frozen raw poultry for cooking, how do you usually do it?	Thaw it in the bottom shelf of the refrigerator OR in/under running cold water in the sink	103	34.20
How do you check to see if a poultry product you have cooked is safe to eat?	I observe when the exterior colour is golden brown, and the interior juices run clear OR measure the internal temperature of the poultry meat.	142	47.20
Do you cover your food during storage?	Yes, all the time	217	72.10
Mean		49.37	
SE		1.21	

Table 13-4 Score on reported practices in food temperature control and food storage

#### 13.4.5 Consumer general food safety awareness

The mean score of general food safety awareness and descriptive analysis are summarised in Table 13.5. The majority of the respondents were able to select the correct answer for > two questions but <three out of seven questions. It is important to note that the mean score for this section is 0.52, which is at the average food safety awareness level. There were 67% of the respondents who indicated that they think about the basic food safety rules such as clean, separate, cook, chill and cover when they prepare the food at home. However, only 24% of the respondents indicated that

they always follow these rules. It is a common phenomenon that only a low percentage of respondents who always follow the basic food safety rules and that some respondents' knowledge or awareness' of proper chicken handling was not fully translated into practice (Mazengia et al., 2015, Katiyo et al., 2019). Surprisingly, only 50% of the respondents knew that chicken carries *Campylobacter* despite the intensive coverage in the media that NZ has the highest campylobacteriosis notification rate in the developed world and the association of chicken consumption to this highest rate of notification in New Zeeland for more than a decade. This finding confirmed that the information channels governmental institutions and health communication practitioners used to deliver risk/benefit on food safety issues targeting the NZ population should be revised and it is time to rethink about new channels and type of messages especially via social media so that it reaches nearly all of the New Zealand population.

The average level of food safety awareness in this section confirmed the findings of the previous chapters and the finding of another study (Al-Sakkaf, 2012). The respondent scores in each poultry handling practice section provided decisive evidence about the respondents lack food safety awareness as there were statistically significant correlations between all the questionnaire sections analysed by Pearson test at p<0.01 (Table 13.6). Another NZ study (Allan et al., 2018) partially tested the food safety awareness of consumers, and the primary objective of that study was the introduction of comprehensive, high-quality, chicken safety labelling with all chicken products in NZ as an intervention which may lead to changes in consumer behaviour and reduce the incidence of campylobacteriosis in NZ. This study indicated that there are several gaps in consumer knowledge of proper handling of poultry products. It also confirmed the need to introduce a new chicken label containing all the information about safe handling of poultry products.

wareness
ľ

General food safety awareness	Correct answer	Respondents (n.)	Percentage (%)
What temperature should refrigerators operate at for optimum safe food storage?	Between 4 °C and 7 °C	157	52.20
How do you decide whether or not to consume food stored in the fridge?	By the best before and use-by date	145	48.20
When preparing a meal, do you think about the basic food safety rules such as clean, separate, cook, chill and cover?	Yes, I am aware of all of them or most of them	202	67.10
How likely are you to follow these rules clean, separate, cook, chill, cover?	Always	74	24.60
Does raw poultry carry bacteria (bugs) that can make you sick?	Yes, I know that it carries <i>Campylobacter</i> bug	152	50.50
Cross-contamination can be defined as the transfer of harmful bugs to food from other foods, hands and other food contact surfaces. Please indicate your awareness and self-reported practice of this concept below	I have heard about the concept of cross- contamination, and I always take the necessary steps to prevent it	200	66.40
How important do you consider it to follow/implement all good hygiene practices all of the time to protect yourself and your family from foodborne illness?	I consider it as extremely important	157	52.20
Mean		51.59	
SE		1.39	
#### Table 13-6 Correlation between questionnaire sections by Pearson test

	Food Safety Awareness	Temperatu Control an Proper stor	re d rage Hygiene	Cross Contamination Prevention
Food safety awareness	1	0.373** 0.000	0.378** 0.000	0.417** 0.000
Temperature control and Proper storage	0.373** 0.000	1	0.329** 0.000	0.321** 0.000
Hygiene	0.378** 0.000	0.329** 0.000	1	0.237** 0.000
Cross-contamination Prevention	0.417** 0.000	0.321** 0.000	0.237** 0.000	1

\*\*. Correlation is significant at the 0.01 level (2-tailed).

### 13.5 Conclusions

The survey results identified many areas of low adherence to currently recommended food safety practices, based on self-reported data, personal food hygiene practices, prevention of crosscontamination not only at home during poultry preparation, proper storage and temperature control such as determining 'doneness', and proper thawing of raw poultry, shopping and transportation. Additionally, there are several gaps in consumer basic food safety awareness and information about risks and the pathogens related to poultry products. It is noteworthy that about 80 % of the respondents were with a tertiary qualification or postgraduate qualification but still their food safety practices, and food safety awareness was below the average score. It is important to target the less qualified New Zealanders in a future survey as they were an under-represented group in this study sample. It is important to find better ways to convince the less qualified and low-income groups to participate in future research surveys. However, from the score of the higher qualified New Zealander score which can be considered as another decisive evidence that the less qualified New Zealanders food safety practices and food safety awareness could be worse than in those with higher qualifications. Therefore, the evidence collected in this study confirmed the hypothesis that New Zealanders are less educated in food hygiene practices and food safety knowledge than those in other developed countries, and this is possibly a reason for the higher rate of campylobacteriosis in NZ. More research is necessary to recognise and understand the reasons for poor consumer practices in order to develop a more effective food safety message and transmit

that through new information channels including social media so that it reaches a larger audience. This could lead to a reduction in the number of campylobacteriosis cases in NZ.

# 14 General discussion

#### 14.1 Background

The main objective of this thesis was to understand why NZ has a higher rate of reported cases of campylobacteriosis than other developed countries. Thus, it was important to investigate the reasons for the unexplained campylobacteriosis increase, which causes significant economical and health consequences in NZ. The problem is caused by a bacterium, which naturally occurs in the gastrointestinal tract of chickens. Because it is present in the faeces of production-animals, the bacterium can be transferred from the intestine to the fresh meat during slaughtering (Keener et al., 2004). Thus, the investigation of this study targeted the pathogen behaviour and followed the survival of the pathogen under the environmental conditions prevalent in poultry processing plants. In addition to the comprehensive investigation of campylobacteriosis, this study identified the causes for the unusual high endemic rate in NZ and also describes the pathogen fate at the consumer handling phase which contributes to the high incidence (12 -100%) of food poisoning cases occurring at homes (Redmond and Griffith, 2009, van Asselt et al., 2008). Approximately 500 New Zealanders contract food poisoning each day, of which about 40% of the cases believed to be caused by unsafe food handling at home (Anonymous, 2016). Recently, the NZFSA confirmed that about 200,000 New Zealanders contract food poisoning each year (Anonymous, 2019). Moreover, the prevention and control methods at the poultry processing and consumer levels are highlighted in the context of this study. The quantitative microbial risk assessment was conducted using the Bayesian method to determine and evaluate new interventions. Finally, the examination of the time-series models to predict campylobacteriosis risk was also tested.

#### 14.2 The pathogen

#### 14.2.1 Heat resistance

The thermal inactivation kinetic parameters (D and z) for the tested NZ strains of *C. jejuni* isolated from both humans and poultry were obtained by a broth experiment and did not indicate that the most implicated strains in human food poisoning cases were more heat resistant than the international strains recorded in published data overseas. Fitting the survival curves of the tested NZ strains in broth and in food matrix to different models following one-step regression analysis supported the fact that NZ tested strains were heat sensitive. Moreover, NZ strains were, to some

extent, more heat-sensitive than most of the international *C. jejuni* strains. The *D* values for all NZ isolates at 60 °C (1.3 - 4.2 s) as shown in Table 4.1 were slightly lower than the minimum D value of the published international isolates, which ranged from 7 - 30 s (Li et al., 2002, Ray, 2014, Juffs and Deeth, 2007). Similarly, the z values calculated for all isolates and listed in Table 4 -1 and 4 - 2 as ranging from 4.0 ° to 5.2 °C were within the range of the published z values of 2.8 - 5.8 °C (Jackowska et al., 2008, Li et al., 2002, Sorqvist, 2003, Juffs and Deeth, 2007). The results obtained for the tested NZ strains using chicken skin under isothermal conditions are in agreement with most of the published international data. The D and z values obtained by a skin experiment at 60  $^{\circ}$ C (Table 5.2) were 2 to 5 times higher than the values obtained from the broth experiment; 11 to 18 s and 8 to 11 °C, respectively. It is evident that the presence of a food matrix altered the kinetic parameters for Campylobacter heat resistance (Yang et al., 2001). However, a Wageningen University group (Bergsma et al., 2007, de Jong et al., 2012) reported that the D values were not significantly increased. Sampers et al. (2010) with a similar experimental set-up as Bergesma's study and using frying chicken fillet found that 4.5 log cfu/g of C. jejuni inoculated into chicken burgers were reduced to below the detectable level (<10 cfu/g) after 4 minutes of frying to an internal temperature of 57.5 °C, and that 2.5 log cfu/g of the naturally contaminated chicken burger was reduced to below the detectable level after 4 minutes of frying to an internal temperature of 52.1 °C. Although the objective of Samper et al. (2010) was not to determine the kinetic parameters (D and z) in their relatively complex experimental design, the results obtained were useful to determine the minimum time and the internal cooking temperature required to eliminate the contamination. This is in agreement with the conclusions of this study regarding the heat sensitivity of C. jejuni in general. It is assumed that the results reported by the Wageningen University were artefactual in the temperature measurement of the chicken fillet and in the enumeration and identification method used by them. Moreover, the data of de Jong's was presented only at a conference in 2008 and is yet to be published in a peer review journal (Personal communication in 2010 with the coauthor Dr.van Asselt). It was published in 2012 in an open access journal (International Journal of Microbiology with an impact factor in 2019/2020 of 1) by the Hindawai publisher. The heat sensitivity of *Campylobacter* remains a characteristic of this pathogen in NZ and internationally. Moreover, a study by Gunsen (2008) found that baking a chicken drumstick for three or five min at a core temperature of 80 °C or 70 °C killed all Campylobacter cells, which is not in agreement with the findings of the Wageningen University group. Moreover, C. jejuni was not detected following simulated home pan-frying of artificial inoculation of 4 log cfu/g C. jejuni into steaks, fillets, hamburgers and meat strips (Lahou et al., 2015). These meat samples were from a range of animal species, pig, cattle, chicken, sheep, turkey, horse, crocodile and kangaroo. Only after a further

enrichment step, *C. jejuni* was detected in few samples of pork hamburger, horse steak and crocodile steak. Additionally, the *D* and z results obtained under dynamic conditions for NZ strains are broadly in agreement with the published international isothermal data and with isothermal data of this study obtained for the same isolates. None of these indicate that the NZ *Campylobacter* strains are more heat resistant than other strains. The parameters obtained under dynamic conditions are more robust than the isothermal parameters and can be obtained with less experimental effort. The *D* and the zvalues generated under dynamic conditions for *C. jejuni* in this study is the first reported values in the scientific literature. These have not been reported before, probably because of the mathematical complexity of the calculations involved and the instrumentation required to generate such data.

This study results confirm that the tested NZ *C. jejuni* strains do not have unusual heat resistance, and the high rate of campylobacteriosis in NZ is not associated with the emergence of more heat resistant strains in the country. This finding has a significant impact for the poultry industry, regulators, consumers and researchers. There is now ample scientific evidence that shows the importance to maintain the standards for heat treatment practices at the food processing plants, food services and in homes. Any relaxation to the heat treatment protocols can cause unpredictable loss to the poultry industry. It is a challenge for the regulatory bodies to repeatedly convey the message of the importance of cooking temperatures to all consumers, given the reluctance of NZ consumers to use thermometers to verify the cooking temperatures at homes (Gilbert et al., 2007).

#### 14.2.2 Air tolerance

The investigation of NZ strains tested in this study using three methods revealed that the air tolerance of NZ strains was not different to the international strains (Chen et al., 2001, Garenaux et al., 2005, Oh et al., 2017) and the previously tested NZ strains (Chynoweth et al., 1998). The results of this study did not support the hypothesis that the survival of *C. jejuni* was not influenced by the storage atmosphere aerobically or microaerobically, as the *C. jejuni* survival was shorter in both the agar and broth experiments under aerobic conditions than in the microaerobic conditions at all temperatures tested, except in a broth experiment at 10 °C where after the fifth day, the survival rate was similar under both aerobic and microaerobic conditions. This finding confirms the fundamental and best-known hypothesis that *Campylobacter* is a microaerophilic microorganism that requires a specific gas composition for its growth and longer survival.

Biological aspects and molecular mechanisms responsible for microaerobic growth or better survival from a physiological and genetic perspective are not fully understood. It is assumed that due to the inhibition of enzymes at high  $O_2$  concentrations (> maximum level 10 %), the *C.jejuni* 

cells are vulnerable to reactive oxygen species (ROS) and free radicals resulting in changes to molecular components of cells, their function in addition to the production of highly toxic components (Kaakoush et al., 2009). The *C.jejuni* poultry isolates survived longer than the human isolates, especially at the incubation temperatures of 10 and 20 °C. However, there was no significant difference between the survival of the poultry and human isolates when the differences in the initial concentrations of both isolates are accounted for at 4 °C incubation.

This study revealed that the survival of *C. jejuni* under oxidative stress is more influenced by the temperature rather than the incubation atmosphere. A low temperature of 4 °C had less effect on *C. jejuni* and was able to survive more than four weeks in both agar and the broth. Garenaux et al. (2008) assumed this to be due to a less active metabolism, a decrease in catalytic activity, or the  $O_2$  was less toxic to cells at 4 °C.

Despite the fact that this study was not conducted with chicken meat, the combination of three methods to address the oxidative stress on *C. jejuni* increased the validity of the results reported by this study and saved time, resources and cost.

Practically, most international data on C. *jejuni* survival has been for short periods (1 - 2 days)(Solow et al., 2003) or 1-2 weeks (Bhaduri and Cottrell, 2004, Davis and Conner, 2007, Garenaux et al., 2008, Oh et al., 2017). This study results are in agreement with the results obtained in NZ using chicken mince and chicken nuggets (Chynoweth et al., 1998). Internationally, the C. jejuni survival data obtained from 4 biological materials; water (incubation for four weeks), human urine, (five weeks), human faeces and cow milk (three weeks) were similar in spite of the differences between the four biological materials (Blaser et al., 1980). Similar results have been generated by (Garenaux et al., 2008) using Colombia agar plates and by (Bhaduri and Cottrell, 2004) using chicken mince or chicken skin at 4 °C. The implications of the spoilage of chicken meat or chicken skin during sample storage above 4°C hinders the use of a food matrix in long survival studies. Moreover, the irradiation of skin or chicken meat eliminates the microflora (pseudomonads, micrococci, Staphylococci) which are found naturally on poultry, and also inhibits the growth of Campylobacter (Mai, 2003). The conclusion derived from irradiation studies of chicken meat must be treated with caution. Thus, the results of this study reject the hypothesis that the high rate of campylobacteriosis in NZ may be due to the emergence of unusual new strains with more O<sub>2</sub> tolerance or unique survival ability under a range of storage or handling temperatures (4, 10, 20, 25 °C).

Given that the *C. jejuni* infection dose that causes illness is about 500 cells (Robinson, 1981) and the length of time that *Campylobacter* can survive as revealed by this study for NZ strains is a

significant finding from a health perspective as the shelf life of a fresh chicken is eight days at 4  $^{\circ}$ C whereas *Campylobacter* can survive for > 4 weeks at that temperature. Hence it is important for the poultry processing plants to apply the necessary intervention to assure that chicken carcasses and poultry products are released to the retail market with as low a contamination level as practically possible, and theoretically, the counts should not exceed 500 cells per a chicken portion serving.

## 14.3 Poultry Processing Plant

It is important to investigate initially the poultry processing practices which influence the prevalence of *Campylobacter* in the end product, as the prevalence of *Campylobacter* in chicken products in NZ is among the highest of the countries in the world. Thus, the focus should be on the processing practices, which may lead to an increase in the prevalence of *Campylobacter* contamination in chicken carcasses and poultry products. An overview of the current intervention strategies used in poultry plants to reduce contamination and their implications for consumers is discussed below, along with suggestions for alternative more effective interventions.

#### 14.3.1 Marination or tenderisation by needle injection technology:

Traditionally, it has been proven that needle injection technology introduces pathogenic bacteria from the carcass surface into the interior muscle tissue (Gill and McGinnis, 2005). Moreover, since the marination solution is usually recirculated, there is also a risk of spreading contamination to products previously free of pathogens (Ray et al., 2010). In addition, pathogens introduced into the internal tissue may be better able to survive cooking processes and cause illness (Gill et al., 2008).

Generally, marination is mainly used to improve palatability, yield and shelf life by inhibiting spoilage bacteria in treated meat. It is considered that the marination currently used in most poultry processing plants does not inhibit the survival of *Campylobacter*, whereas a new marinade, produced from a mixture of organic acids and different food ingredients that can reduce *Campylobacter*'s contamination level by one log (Ray, (2010). It appears that there is no evidence that the latter has been used commercially in the poultry industry. Others have used organic acids as a *Campylobacter* reduction hurdle (2015).

This study investigated two main marinades used by the largest NZ poultry producer, and it was apparent that the needle injection process could pose a high risk to consumers if marinade "A " (with5% salt) (used for products such as drumsticks) before freezing the products contaminated with *C. jejuni* up to 3.5 log cfu/g. However, after freezing, when the drumsticks were sampled from the retail market, the prevalence of contaminated samples was reduced to 20% (an 80% reduction), and also the contamination was only on the surface with 450 to 900 cfu/portion (= 2.6 to 2.9 log).

The reduction in *C.jejuni* prevalence was possibly due to the freezing process which can cause a reduction of 1.3 - 2.2 log (Boysen and Rosenquist, 2009, El-Shibiny et al., 2009, Georgsson et al., 2006, Hofshagen and Kruse, 2005).

The results of poultry products injected with marinade " B " showed only 30% contamination with *Campylobacter*, and the contamination level was about 450 cfu/portion and only the surface of the drumsticks when tested at the processing plant. The sampling of the injected drumsticks at the retail which sold fresh products revealed that the contamination level of *Campylobacter* was reduced to 20 %, but the sampling of whole chicken carcases injected with marinade or solution " B " and analysed for whole surface contamination, showed 70% prevalence with about 3.9 log cfu/carcass. The Bayesian analysis is convenient to implement when the microbial data contains results which are below the detection limit as it presents the best way to mine all the data available to generate acceptable and unbiased results. However, it was noted that the mean of external or internal samples by Bayesian analysis was similar to the mean obtained by the classical mean calculation by Excel. Moreover, the use of Bayesian analysis was hindered by its complexity and the lack of its general applicability as it is a more useful tool for generating data for the quantitative risk assessment. In spite of the free software available for Bayesian analysis, the method is not that used in the field of food microbiology data and should be encouraged to be used only when the microbiological data are below the detection limit.

The detection of *Campylobacter* in the poultry processing plant and retail chicken samples injected with marinade "B" is questionable. It is because the marinade "B" has shown inhibition capability against *Campylobacter* in a liquid medium, as observed in the initial experiments conducted in a flask at the laboratory (section 8.3.1), as the salt content of marinade "B" exceeded the maximum growth tolerable salt content (>2%) for *Campylobacter*) (Sampers et al., 2010). This may be due to the buffering capacity of the chicken meat which permits *Campylobacter* to survive for a longer time than its survival period in a liquid medium (Perko-Makela et al., 2000), or by the increased bacterial resistance by attachment to poultry skin (Zhang et al., 2013).

The information provided from this study indicated that the contamination level of the fresh products sampled at the market varies and can exceed (at 3.9 log cfu/carcass) which is above the limit of *Campylobacter* contamination level set by the NZFSA (3.78 log cfu/carcass). Thus, it is apparent that *Campylobacter in* un-marinated fresh products exceeds the NZFSA limit significantly. However, the current contamination level is far better than the level reported by a study conducted in 2008, which found up to 6.7 log cfu/carcass in fresh, non-marinated products (Chrystal et al., 2008). A more comprehensive survey covering the entire chicken processing chain in NZ would be useful. This study (see section 8.3.2) also indicated that a 'reduced salt' marinade was not as

effective in reducing the *Campylobacter* contamination level compared to a 'high salt' marinade. However, all products injected with the low salt content solution were sold frozen, which, as reported in section 8.3.3 reduced the initial contamination *Campylobacter* level.

Cumulatively, the processing practices tested in this study using two marinades did not indicate that the marination process could be the reason for increased contamination level of chicken or chicken products available in the NZ retail market. The results of this study are the first study that investigated the *Campylobacter* contamination via needle injection practices in poultry processing plants, as most of the previous studies were focussed on beef or pork processing plants and *E. Coli* (Gill et al., 2005a, Gill and McGinnis, 2005, Gill et al., 2005b, Gill et al., 2009, Gill et al., 2008, Chancey et al., 2013). In addition, all previous marinated poultry studies investigated only the surface contamination of injected or marinated product or the prevalence of contamination (Perko-Makela et al., 2000, Sampers et al., 2008), or explored the best method for detection of *Campylobacter* in marinated products (Katzav et al., 2008) or investigated the most effective marinade against *Campylobacter*; seasoning dry mixture and wine (Ray et al., 2010). So, this study could be classed as pioneering comprehensive research that investigated the effect of marinated practices at poultry processing plants in NZ to gauge the contribution of such practices to *Campylobacter* contamination of chicken meat sold in NZ.

#### 14.3.2 Evaluation of *Campylobacter* mitigation strategies

Chemical intervention (organic acids, chlorine-based and phosphate-based) of broiler carcasses has been used in the US for many years (Oyarzabal, 2005, Lu et al., 2019) and in some other countries, to decontaminate poultry products but has not been that successful (Boysen and Rosenquist, 2009). In NZ poultry processing operations, acidified sodium chlorite (ASC) has been used mainly as a major intervention bacterial reduction step along with the chlorinated washes at many stations in the processing line and also chlorinated spin chillier dipping for about 70 min. Such chemical treatments reduced the bacterial contamination level in chicken especially after NZFSA set a performance target (3.78 log cfu/carcass) in April 2008 with the interim aim of a 90 % contamination reduction level (~1 log cfu/carcass) from the levels observed in 2007 without any restrictions on the type of equipment or materials that could be used to achieve the mandatory target. However, due to adverse health effects caused by chemical to consumers due to the formation of toxic compounds, it was reported that the use of chemicals should not replace the good hygienic practice at all poultry processing plants (Hugas and Tsigarida, 2008, EFSA, 2011b, Schraer and Edgington, 2019). Trihalomethanes (THM), a by-product of chlorine cannot be detected in poultry tissues when exposed to chlorine at 50 ppm or less using either the AOAC or EPA analytical methods. However, the NZ chicken was prior to 2010 were exposed to a higher chlorine concentration than the safe-limit.

A new regulation on the strict use of chemicals in EU (EC) No. 853/2004 was implemented in 2006 which permitted the use of chemicals but only after the European Food Safety Authority (EFSA) has been provided with a risk assessment by the manufacturer or by the poultry processing plant detailing the possible negative impact of the suggested chemical to be used in the poultry processing plant (Hugas and Tsigarida, 2008, EFSA, 2011b, Schraer and Edgington, 2019). To date, no such chemicals have actually been authorised by the EC because of insufficient documentation (Boysen and Rosenquist, 2009, Schraer and Edgington, 2019).

A published survey of 26 EU countries in 2008 indicated that the prevalence of *Campylobacter* was very high in chicken ranging from 5 - 100% (EFSA, 2010). New Zealand government agencies, poultry industry, farmers and growers face a huge challenge with the European policy on the use of chemicals. The consumer groups in NZ have over the years initiated campaigns against the use of chemicals in poultry processing.

To replace the use of chemicals, many other safer methods have been tried. Hot water immersion method reported in section 9.3 provided evidence as a promising, simple, cost-effective alternative which can be used by the poultry industry if a ban on the use of chemicals is implemented in NZ in the future. This study results emphasise the need to conduct commercial trials to better evaluate the feasibility of the hot water immersion intervention to reduce the naturally contaminated chicken in poultry processing environments. In addition, the sensory quality of the treated carcasses needs to be evaluated by poultry industry experts to gauge the best time-temperature combination with the least adverse sensory effect.

Moreover, this study suggested the replacement of the current tap water spray washers with hot water spray washer, which can serve as an additional hurdle to reduce *Campylobacter* contamination level in chicken. Also, in this study, the importance of dry air chilling was highlighted (in section 9.3) as it reduced 0.3 to 1.4 log of *Campylobacter* contamination (Allen et al., 2007, Boysen and Rosenquist, 2009, Huezo et al., 2007, Lu et al., 2019, Rosenquest et al., 2007). Hence, dry air chilling should be considered as an alternative chilling method to the current spin chillier method which has the potential to cross-contaminate (Sánchez et al., 2002, Whyte et al., 2002, Lu et al., 2019) as a comprehensive solution for poultry processors. It is important to encourage research on the physical interventions which would be more acceptable to consumers and also practically applicable to the poultry industry.

## 14.4 The consumer

It is believed that consumers play a major role in preventing campylobacteriosis. Definitely, the broad and complex consumer-aspect cannot be covered in a chapter or two in a thesis but has been addressed in a thesis internationally (Redmond and Griffith, 2003). The main objectives were to investigate the causes of the high rate of campylobacteriosis and then to advise about the possible best remediation and intervention methods that could reduce the high rate in NZ.

#### 14.4.1 The consumer kitchen

Many consumers are unaware that home practices may be responsible for food-borne disease incidence. Investigations on the microbial level and type of microbes harboured in the domestic kitchen ((Davis and Conner, 2007, Kennedy et al., 2005b, Evans and Redmond, 2019), domestic environment (Beumer et al., 2002, Kagan et al., 2002, Ojima et al., 2002) have been conducted. Other studies have concentrated on the kitchen after food preparation (Gorman et al., 2002, Evans and Redmond, 2019). All these studies have reported that most surfaces in the kitchen were heavily contaminated with pathogenic and non-pathogenic bacteria.

It was surprising that a few studies reported the kitchen to be more heavily contaminated than the toilet or bathroom (Ojima et al., 2002, Todd et al., 2009). The most heavily contaminated locations or items in the kitchen were dishcloths, cleaning cloths, sponges, the sink environment, towels and drain areas (Beumer et al., 2002, Doyle et al., 2000, Evans and Redmond, 2019). *Campylobacter,* in particular, has been isolated from most of those places and from chopping boards, work surfaces, floors, refrigerator and door, waste and pedal bin (Redmond and Griffith, 2009). Generally, most bacterial contamination was detected in wet and moist areas of the kitchen with only a few bacteria on dry surfaces (Aiello et al., 2008). The kitchens of NZ consumers will be similar. There is no data in NZ regarding consumer kitchens, and more research is required.

#### 14.4.2 The consumer's knowledge and practices

There is a lack of data and studies in New Zealand with respect to consumer knowledge and handling practices. There are a few studies (Gilbert et al., 2007, Bloomfield and Neal, 1997) but not comprehensive. So, the data from these limited studies should be treated with caution. It is difficult to compare NZ surveys with international surveys because of the differences in sample size, objectives, targeted population groups, methodology and results interpretation. The survey conducted by this study identified that many people did not adhere to currently recommended food safety practices. This conclusion was based on self-reported data based on personal food hygiene practices, cross-contamination prevention at home during poultry preparation, adherence

to proper storage and temperature control including determining 'doneness', proper thawing of poultry, and shopping and transportation. Additionally, there were several gaps in consumer basic food safety awareness and information about risks and the pathogens that can be present in poultry products. It is noteworthy that in spite of about 80 % of the respondents were with a tertiary qualification or postgraduate qualification, their food safety practices, and food safety awareness was below average. In future surveys, it is important to target the less qualified New Zealanders as they were underrepresented groups in this study sample. It is also important to find better ways to convince the less qualified and low-income groups to participate in future research surveys. The evidence collected in this study survey confirmed the hypothesis that New Zealanders are less knowledgeable in food hygiene practices and food safety than people in other developed countries, and this could possibly be the reason for the higher rate of campylobacteriosis in NZ than in other developed countries. More research is required to recognise and understand the reasons for poor consumer practices in order to help develop a more effective food safety message and promote it through new information channels such as through social network sites to improve their practices in the domestic environment. This could, in turn, lead to a reduction in the number of campylobacteriosis cases. In 2007, the New Zealand Food Safety Authority chief executive considered that the level of hygiene practice of New Zealanders could be the reason for the high rate of campylobacteriosis in the country (McKenzie, 2007). In 2008, a prominent NZ and International Campylobacter expert, Professor Nigel French indicated during an interview with a NZ Radio that if New Zealanders could improve their practice of washing their hands, the campylobacteriosis rate would drop dramatically In 2007 a member of the NZ parliament indicated that many New Zealanders are too lazy to wash their hands, just wringing their hands instead of washing them, and warned that if they do not wash their hands, they will become ill (Coddington, 2007).

A large number of campylobacteriosis cases present a high cost on the health care system and could also affect the NZ tourist industry and agricultural produce both of which champion New Zealand's "clean green" image.

#### 14.5 The government

The aim of health professionals, risk managers, policymakers, and regulators is to encourage consumers to practice desirable food safety practices and to discourage improper or unsafe ones. It is not possible to enforce by law consumer handling practices at home. The only possibility is to combat the burden of disease and to optimise consumer risk management by designing an effective communication initiative regarding food safety hazards and proper food handling practices in the

domestic environment (Farias et al., 2019).

A reduction of the disease burden and the incidence of foodborne diseases are possible through improvements in consumer food safety practices (Redmond and Griffith, 2009, Farias et al., 2019). Researchers have identified a gap or a decline in consumers' knowledge which can increase the risk of food poisoning (Kennedy et al., 2005a, Gkana and Nychas, 2018). This indicates that consumer knowledge is correlated with current practices, and this can affect willingness to change from the current practices.

It is clear that the process of communicating the domestic food safety risk is a complex and challenging task. In NZ, it appears that there is a significant deficit in f good hygiene practice and food safety knowledge amongst a majority of the population. The responsibility appears to be for food safety experts and risk communication community to understand the variety of socioeconomic/demographic and psychological factors that prevent the public from changing its behaviour.

Given the cost of campylobacteriosis to NZ economy which is > NZ \$ \$77 Million per year (NZFSA, 2007) and the potential decline in NZ reputation as a safe food-producing country, financial loss in the tourism industry, it may be more cost-effective to utilise social marketing for a future educational or communication strategy aimed at educating New Zealanders to improve their domestic food handling practices. The social marketing approach was used in NZ during a nutritional initiative called the "5 + a day programme" to encourage New Zealanders to eat five or more fresh fruits or vegetables per day. The more expensive social marketing approach has been proven to possibly change consumers' behaviour (Redmond and Griffith, 2006). In order to ensure effective domestic food safety risk communication, a full appreciation of the target audience is required, as each audience brings unique challenges to the development, conveyance and delivery of food safety and safe practice information. The efforts of health professionals are well-coordinated in NZ and are based on a more scientific approach, which can help to educate New Zealanders on food hygiene practices to reduce the high rate of campylobacteriosis cases. The findings of this study have specified which poor hygiene practices need to be improved.

A cost-effective option is to utilise social marketing in future educational and/or communication strategy aimed at educating New Zealanders to improve their domestic food handling practices, which will possibly later help to reduce the number of campylobacteriosis cases in NZ (NZPHO, 2010). NZFSA has conducted a few educational campaigns at shopping malls. However, a complete approach of social marketing as a holistic and long term or permanent strategy, and/or the inclusion and improvement of food safety education at childcares and schools could be

implemented in NZ. In addition to social marketing, publicising foodborne disease outbreaks by the "scare tactics" could also motivate consumers to better understand the severity of food poisoning and persuade them to change their practices and habits and adopt safe food handling practices (Katiyo et al., 2019).

The key findings of this study contributed to exploring the reasons for the high rate of campylobacteriosis in NZ. The results of this study provided some answers to the important and complicated questions regarding the campylobacteriosis crises in NZ. This study results have rejected the hypothesis of a more heat resistance phenotype of *C.jejuni* and its longer survival during storage or handling under the aerobic atmosphere. This would help the scientific community to investigate poultry processing practices to learn about the contribution of the current poultry processors practices to the campylobacteriosis problem and develop interventions at the poultry processing plants. This study investigation at poultry processing plants of one of the critical practices (marination by needle injection) revealed that the current practices could contribute to the increase in *Campylobacter* contamination of carcasses by NZ consumers.

Moreover, the finding from the poultry plant intervention study identified alternate, cost-effective physical interventions with fewer health implications or consequences compared to the current chemical intervention procedures. The use of chemicals in poultry processing plants has been banned in the EU, and consumer groups in NZ have been pressing the NZFSA also ban the use of chemicals in NZ. The physical interventions highlighted in this study provide another solution for a possible chemical ban in New Zealand.

The complete food chain has been modelled by the Bayesian approach, which provides better insight into the food chain and is more informative as it incorporates all the factors that impact the final risk estimation. Thus, it can easily determine the impact of any intervention in the food chain. Therefore, the effect of a new intervention (for example, a consumer education plan or other intervention at poultry plants and farms) is clear for the policymakers, risk managers and health professionals before the implementation of the new intervention, by altering the priors of consumers' hygiene practice for example, and the priors of farms or the poultry plants. The association between the reduction in flock/ poultry prevalence, improvement in hygiene practices and the calculated reduction in risk of human disease was estimated to be 1:1 which is in agreement with international findings (Rosenquest et al., 2007).

A simplified Bayesian model used in this study is able to identify the expected outcomes and provide a fast response and more evidence about the possible main risk drivers in the NZ campylobacteriosis epidemic. The model used the Bayesian methodology but requires refinement

in the future. It is also recommended that another more sophisticated software package be used in the future to facilitate the convergence of the developed model. The procedure followed in this study applied Gellman Robin statistics to determine the model convergence after running three channels (Gelman and Rubin, 1992). However, this was time-consuming as the model convergence occurred after 240, 000 iterations. In addition, to give more confidence and consistency to the model convergence, the model has to complete 700,000 iterations in total before extracting the statistical results (coda and Kernel density plots). There are new software packages which may perform better than WinBUGS, and they conduct a complete sensitivity analysis as an automated routine (Smid et al., 2010). This study showed that the ARIMA intervention models have the lowest forecast error with only 657 cases more than the actual notified cases (see Table 14.1.) The Holt-Winters method, being local in nature, quickly adapts to the new regime post-intervention and gives good predictions which are comparable with the ARIMA intervention models. The Holt-Winters method has an additional advantage in that it is a simple method, whereas the ARIMA intervention model requires extra statistical expertise to conduct it. It is confirmed that classical time-series techniques such as the ARIMA with intervention and the Holt-Winters method can provide a good prediction performance to predict campylobacteriosis risk.

Table 14-1 Comparison of models in predicting campylobacteriosis cases in New Zealand for the year 2010

2010	*TRUE	AR_Ia	HWm	AR_Im	Bayesian	МС
Cases	7333	7990	8442	8666	9250	846451

\*True: actual number of notified cases, HWm: Holt-Winters multiplicative, AR\_Ia: ARIMA intervention additive, AR\_Im: ARIMA intervention multiplicative

Cumulatively, the findings of this PhD study was not limited to identifying solutions to the campylobacteriosis problem in NZ and providing some direction for prevention and control, but they are also extending it to present promising alternative physical interventions for a possible new crisis in the poultry industry if a ban on the use of chemicals is imposed in NZ in the future, as it is the norm in EU countries. The final validation of the proposed method needs to be implemented at the poultry processing plant level.

Additionally, the approaches of investigating the heat resistance and oxygen tolerance of *C.jejuni* in this study were novel and developed new and fairly advanced methodologies in this study investigations.

Finally, the provision of the Bayesian model to assess all the factors which may contribute to the campylobacteriosis risk in NZ, and the time-series models such as the ARIMA intervention model, are considered as significant tools for risk managers, policymakers, and health professionals to enhance food safety in NZ.

It is important to indicate that this study has provided insights into the possible factors for the high rate of campylobacteriosis in NZ compared to other developed countries. However, this study has not explored the factors for the remarkable decline of the campylobacteriosis rate in 2007 and 2008 (see Figure 14.1). This rapid decline is considered one of the largest decline in disease incidence ever recorded for a national epidemic of this type (Baker, 2011) and it remains unclear as to how it was possible for this rapid decline to have been achieved since even the Campylobacter risk management strategy set by the NZFSA in 2007 set only a 50% reduction after five years (Slorach, 2008) but the reduction happened within one year. Moreover, the lack of a reliable and scientific method which can measure the contribution of each intervention to this unexpected decline accurately makes it more questionable for the scientific community in NZ (Nelson and Harris, 2011) and internationally. The source attribution model has not performed well or is not stable, as on one occasion it indicated in a Manawatu survey that poultry attributed to 80% of the campylobacteriosis cases (Mullner et al., 2009a), and another indicated in the same survey that only 67% of cases were attributed to poultry (French, 2008a), and in 2011 a report about the same study showed that >50 % was attributed to poultry (Sears et al., 2011). Furthermore, with changes in the reporting system at the end of 2007 to improve the surveillance system, it was expected that the number of notified cases would have increased. However, questionably, there was a 60% reduction in the number of notified cases in 2008 from 2006, and about a 50% reduction from 2007.

The prohibition of NZ poultry exports to the U.S and E.U for a long time raises more questions about the quality and safety standards implemented in NZ poultry processing plants.



Figure 14-1 Campylobacteriosis Number of cases in New Zealand (EpiSurv, 2019)

Evidentially, the priority for the scientific community, risk managers, policymakers and health professionals is not to celebrate the rapid decline in campylobacteriosis, but to accurately investigate further the proportion of each contributor to the rapid decline and to think about how more can be done to reduce the current rate as NZ still has the highest rate among the developed countries.

Therefore, it is advisable for the NZ government to invest more into prevention and control research of campylobacteriosis. Based on the findings, take appropriate actions and implement interventions that would reduce the disease incidence. New Zealand is a food exporter and also promotes tourism, both of which generate significant revenue. There is a concern that NZ might lose the trust of food importers and tourists if the government does not tackle the food poisoning related to campylobacteriosis. In addition, the government will also save by increased productivity indirectly by reducing the number of New Zealanders who take days off from work due to

associated illnesses. There will also be a saving if the number of hospitalisation which currently cost about 100 millions of NZ \$ a year could be reduced.

#### 14.6 Conclusions and opportunities for future research

#### 14.6.1 Conclusions

The  $D_{60}$  and z values in the broth medium (for the most important strain *C. jejuni* implicated in human cases of campylobacteriosis in NZ) lie within the range from two to five s and 4 to 5.2 °C, respectively. In food, the values ranged from 12 to 18 s and 8 to 11 °C under isothermal conditions, and 13 to 16 s and 8.7 to 10.2 °C under dynamic conditions. These values are broadly in agreement with the published literature and do not indicate that the NZ *Campylobacter* strains are more heat resistant than others. The Weibull model better fitted the survival data than the log-linear model. However, the Weibull shape factor varied significantly between the data sets. When the mean shape factor was considered and the scale factor obtained by re-fitting the model, the goodness of fit was poorer but equivalent to the log-linear model. The scale factor of the Weibull model showed a similar dependency on temperature as the *D* value in the log-linear model, and the time required to achieve a given process extent (e.g. a 7D reduction) was similar for both models. Therefore, the likelihood of a systematic error in using the log-linear model in quantitative risk models appears to be low.

Investigations of the specific *Campylobacter* strains for their ability to survive under other environmental conditions such as atmospheric gas composition at processing plants, handling and storage temperatures in the three methods (liquid media, agar, hydrogen peroxide) revealed that the NZ strains did not differ from most other internationally reported strains in  $O_2$  tolerance or their survival at the investigated temperatures. At 20 or 25°C New Zealand selected strains survived in liquid media (broth) up to only one- week aerobically, but at 4°C, the organisms survived aerobically for > three weeks.

The evaluation of poultry processing practices contributing to the possible increase of *Campylobacter* contamination in NZ chicken and chicken products revealed that the processing practices tested in this study (with two marinades used in the poultry industry) does not support the hypothesis that the marination process is the reason for the significantly high chicken or chicken product contamination in NZ. This study also indicated that a low-salt brine solution was not effective in reducing *C. jejuni* contamination. However, this brine was used for products that would be sold only after freezing, and it reduced the contamination level by at least 2 log of cfu- Higher concentrations of the brine solution did not inhibit the *C. jejuni* survival, and the pathogen could

be detected over the period of product shelf life at the retail outlet up to 10 days. There is no information about other brine solutions or ingredients used in other poultry plants in NZ.

In NZ, government agencies, the poultry industry and the farmers face an immense challenge from the policy on the use of chemicals in poultry production enacted by the EU in 2006. It is most likely the consumers in NZ will also initiate a campaign against the use of chemicals in poultry processing. In this regard, the hot water immersion technique investigated in this study provides a promising, simple, cost-effective alternative which can be used by the poultry industry if a ban on chemicals is implemented in NZ in the future. However, the study emphasised the need for commercial trials to be carried out to evaluate the feasibility of the hot water immersion intervention as a successful intervention to reduce *Campylobacter* in naturally contaminated chicken in a processing environment. In addition, the sensory quality of the treated carcasses also needs to be evaluated by poultry industry experts for the best time-temperature combination with the least adverse sensory effects if this to be universally approved.

Moreover, this study also suggests the replacement of the current tap water spray washers with hot water spray washers, which can serve as an additional aid to reduce the *Campylobacter* contamination level in chicken. The importance of dry air chilling is also highlighted since it reduces the *Campylobacter* level by 0.3 to 1.4 log, as a replacement for the current spine-chiller, which has the potential to cross-contaminate (Sánchez et al., 2002, Whyte et al., 2002). This could be a comprehensive solution with multiple control measures for poultry processors and government agencies.

This study QMRA model was able to identify the consumer hygiene practices, the initial contamination prevalence at the farm and the practices at the processing plant, as the significant factors influencing the final risk estimate. The association between the reduction in flock/ poultry prevalence and improvement in home hygiene practice and the calculated reduction in risk of human disease was estimated to be 1:1, which agrees with international findings. Thus, QMRA model can easily determine the impact of any intervention in the food chain. Therefore, the effect of a new planned intervention such as consumer education on the final risk estimate should be clear for the policymakers, risk managers and health professionals. This can be achieved by altering the priors of the consumer hygiene practices in QMRA model. Similarly, other interventions at poultry plants and farms also be assessed by altering the priors of farms and poultry plants. This study highlighted the importance of the Bayesian model to assess all the factors which may contribute to the campylobacteriosis risk and confirmed that it can provide better conclusions for QMRA than the MC technique because of its interactive link between the data and the parameter

(backward inference).

It is confirmed from this study findings that classical time series techniques such as the ARIMA with intervention and the Holt-Winters method can provide a better prediction performance for the risk of campylobacteriosis than the QMRA by Bayesian approach. Moreover, the general strength of the Holt-Winters forecast method is that it managed to predict accurately the monthly campylobacteriosis cases in New Zealand and it can also be used easily with a minimum of statistical expertise required with the currently available software. The time series models, such as the ARIMA intervention model which has the lowest forecast error and the Holt-Winters method, are considered as significant models for risk managers, policy makers, and health professionals in order to enhance food safety in New Zealand.

This study was aimed to test if the high campylobacteriosis rate in NZ has a significant link with any negligent food handling practices in NZ domestic kitchens. There appears to be a lack of good hygiene practice and cross-contamination prevention at home during poultry preparation, proper storage, temperature control, determination of 'doneness', proper thawing of raw poultry, food shopping and transportation. It is noteworthy that about 80 % of the respondents in the food consumer survey had tertiary qualification or postgraduate qualification but still their food safety practices, and food safety awareness was below the average knowledge.

The lack knowledge concerning food safety during domestic food preparation was prevalent among New Zealand consumers. Most of the evidence collected from this study survey confirmed the hypothesis that New Zealanders are poorer in food hygiene and food safety awareness than the citizens of other developed countries, and this is possibly the major reason for the high rate of campylobacteriosis in NZ compared to the rates in the other developed countries. Other reasons may be the prevalence and contamination level of chicken carcasses in New Zealand. Consumers have a responsibility to handle, store and prepare food in a safe way to avoid the risk of foodborne diseases.

The aim of health professionals and risk managers is to encourage consumers to practice desirable and safe practices and to discourage improper or unsafe ones, by designing an effective communication initiative regarding food safety hazards and proper food handling practices in the domestic environment. It is clear from the literature that the process of communicating domestic food safety risk is a complex and greatly challenging task.

The responsibility is on the experts and risk communication community to understand the variety of socioeconomic/demographic and psychological factors indicated in this study-

The cost of campylobacteriosis to NZ economy is about 100\$ million. Given the potential the loss

of NZ reputation in the world and the financial loss from the tourism industry, this study has recommended using of innovative and effective food safety messages in future educational or communication strategies aimed to educate New Zealanders and improve their domestic food handling practices, which will reduce the number of campylobacteriosis cases that occur in NZ homes.

It is important to state that the efforts of the NZFSA in requesting all poultry processors to collect samples for monitoring the prevalence and contamination level in flocks and carcasses on each day in March 2007 and in February 2008 imposed a performance target limit for the contamination level on chicken carcasses. Later in 2013 imposed the detection limit as well. This forced poultry processors in NZ to increase their efforts to implement interventions throughout the poultry processing lines to further reduce the contamination level in chicken carcasses. Poultry processors achieve the target by using harmful available chemical interventions which may be banned by the authority in the future.

#### 14.6.2 Future research

As the consumer demand increases for more chemical-free products, and with the ban on chemicals used in slaughterhouses in Europe, the poultry industry and government agencies are turning to more consumer acceptable interventions. It is likely that a ban on chemical use will also be enforced in NZ. Thus, further research is required to evaluate the results of this study as a potential, cost-effective, consumer and environment-friendly solution for reducing *Campylobacter* in poultry processing plants in NZ.

It is important to investigate the effect of NZ poultry practices at poultry processing plants in order to gauge the contribution of such practices to *Campylobacter* contamination in NZ chicken meat processing in a more comprehensive survey. Moreover, the model used for the QMRA by the Bayesian approach needs further improvement and investigation.

More research is needed on consumer's food safety knowledge, perception, practice and communication in NZ, as food safety is a complex issue, and requires constant attention, dedication, and persuasive communications. All relevant parties in NZ (the government, researchers, the poultry industry, farmers, growers and consumers) need to collaborate to tackle the complex nature of the campylobacteriosis problem. The success of NZ scientific research in addressing this problem is being watched with interest worldwide. Therefore, it is important to improve disease surveillance and its autonomy and to fund useful research.

Despite the dramatic drop in the campylobacteriosis notification rate in 2007 and 2008, it remains

higher than the rate in other developed countries and some New Zealand *Campylobacter* strains develop antimicrobial drug resistance which threatens the effective treatment of campylobacteriosis (French et al., 2019). The rate 142 cases per 100,000 is seven times higher than the rate in the USA and three times higher than the rate in the Netherlands, France, Ireland and Iceland (ECDC, 2017). Further efforts are required to decrease the incidence rate to at least at a comparable level with other developed countries. Such a reduction could save about NZ \$100 million a year in addition to preserving NZ reputation in the world as a 'clean-green' food exporter which would also indirectly assist NZ as a tourist destination.

# Appendix A

A.1 Code for the programme to estimate *D* and *z* values by one step regression technique performed by R Software (version 2.9.1)

```
#P474
Dref<-42
z<-4.7
K1<-1/Dref
K2 < -10^{(1/z)}
Data<-read.csv("Data_P474.csv")
Start<-read.csv("Start_P474.csv")
#n0<-Start$n0
n0 < -c(7.7, 8.1, 7.9, 8, 8.2, 7.7, 8.9, 7.7, 8.4, 8.1, 7.8)
Startlist<-list(K1=K1,K2=K2,n0.1=n0[1],n0.2=n0[2],n0.3=n0[3],n0.4=n0[4],
n0.5=n0[5], n0.6=n0[6], n0.7=n0[7], n0.8=n0[8], n0.9=n0[9], n0.10=n0[10], n0.11=n0[11])
Fit P474.nls<-nls(n~n0.1*s1+n0.2*s2+n0.3*s3+n0.4*s4+n0.5*s5+n0.6*s6+n0.7*s7+n0.8*s8
+n0.9*s9+n0.10*s10+n0.11*s11 -K1*K2^T.d*t,
trace=TRUE,start=Startlist,data=Data)
summary(Fit_P474.nls)
#P190
Dref<-22
z<-4.3
K1<-1/Dref
K2 < -10^{(1/z)}
Data<-read.csv("Data_P190.csv")
Start<-read.csv("Start_P190.csv")
n0<-Start$n0
\#n0 < -c(7.7, 8.1, 7.9, 8, 8.2, 7.7, 8.9, 7.7, 8.4, 8.1, 7.8)
Startlist<-list(K1=K1,K2=K2,n0.1=n0[1],n0.2=n0[2],n0.3=n0[3],n0.4=n0[4],
n0.5=n0[5], n0.6=n0[6], n0.7=n0[7], n0.8=n0[8])
```

```
Fit_P190.nls<-nls(n~n0.1*s1+n0.2*s2+n0.3*s3+n0.4*s4+n0.5*s5+n0.6*s6+n0.7*s7+n0.8*s8
-K1*K2^T.d*t,trace=TRUE,start=Startlist,data=Data)
```

```
summary(Fit_P190.nls)
#P45
Dref<-30
z<-4
K1<-1/Dref
K2 < -10^{(1/z)}
Data<-read.csv("Data_P45.csv")
Start<-read.csv("Start_P45.csv")
n0<-Start$n0
\#n0 < -c(7.7, 8.1, 7.9, 8, 8.2, 7.7, 8.9, 7.7, 8.4, 8.1, 7.8)
Startlist<-list(K1=K1,K2=K2,n0.1=n0[1],n0.2=n0[2],n0.3=n0[3],n0.4=n0[4],
n0.5=n0[5], n0.6=n0[6])
Fit_P45.nls<-nls(n~n0.1*s1+n0.2*s2+n0.3*s3+n0.4*s4+n0.5*s5+n0.6*s6-K1*K2^T.d*t,
trace=TRUE,start=Startlist,data=Data)
summary(Fit_P45.nls)
#H474
Dref<-38
z<-4.6
K1<-1/Dref
K2 < -10^{(1/z)}
Data<-read.csv("Data_H474.csv")
Start<-read.csv("Start_H474.csv")
```

```
n0<-Start$n0
```

```
\#n0 \le -c(7.7, 8.1, 7.9, 8, 8.2, 7.7, 8.9, 7.7, 8.4, 8.1, 7.8)
```

```
Startlist <-list(K1=K1, K2=K2, n0.1=n0[1], n0.2=n0[2], n0.3=n0[3], n0.4=n0[4], n0.4=n0[4
```

```
n0.5=n0[5],n0.6=n0[6])
```

```
Fit\_H474.nls < -nls(n \sim n0.1 * s1 + n0.2 * s2 + n0.3 * s3 + n0.4 * s4 + n0.5 * s5 + n0.6 * s6 - K1 * K2 ^T.d * t,
```

```
trace=TRUE,start=Startlist,data=Data)
```

```
summary(Fit_H474.nls)
```

```
#H474
Dref<-38
```

```
z<-4.6
K1<-1/Dref
K2<-10^(1/z)
Data<-read.csv("Data_H474.csv")
Start<-read.csv("Start_H474.csv")
n0<-Start$n0
#n0<-c(7.7,8.1,7.9,8,8.2,7.7,8.9,7.7,8.4,8.1,7.8)
Startlist<-list(K1=K1,K2=K2,n0.1=n0[1],n0.2=n0[2],n0.3=n0[3],n0.4=n0[4],
n0.5=n0[5],n0.6=n0[6])
Fit_H474.nls<-nls(n~n0.1*s1+n0.2*s2+n0.3*s3+n0.4*s4+n0.5*s5+n0.6*s6 -K1*K2^T.d*t,
trace=TRUE,start=Startlist,data=Data)
summary(Fit_H474.nls)
#H190</pre>
```

```
Dref<-24
```

z<-5.6

K1<-1/Dref

 $K2 \le -10^{(1/z)}$ 

```
Data<-read.csv("Data_H190.csv")
```

```
Start<-read.csv("Start_H190.csv")
```

```
n0<-Start$n0
```

```
\#n0 < -c(7.7, 8.1, 7.9, 8, 8.2, 7.7, 8.9, 7.7, 8.4, 8.1, 7.8)
```

```
n0.5=n0[5], n0.6=n0[6])
```

```
Fit\_H190.nls < -nls(n \sim n0.1 * s1 + n0.2 * s2 + n0.3 * s3 + n0.4 * s4 + n0.5 * s5 + n0.6 * s6 - K1 * K2 ^T.d * t,
```

```
trace=TRUE,start=Startlist,data=Data)
```

```
summary(Fit_H190.nls)
```

```
#H45
```

Dref<-30 z<-4 K1<-1/Dref K2<-10^(1/z) Data<-read.csv("Data\_H45.csv") Start<-read.csv("Start\_H45.csv")

n0<-Start\$n0

#n0 < -c(7.7, 8.1, 7.9, 8, 8.2, 7.7, 8.9, 7.7, 8.4, 8.1, 7.8)

Startlist<-list(K1=K1,K2=K2,n0.1=n0[1],n0.2=n0[2],n0.3=n0[3],n0.4=n0[4],

n0.5=n0[5], n0.6=n0[6])

 $Fit\_H45.nls < -nls(n \sim n0.1 * s1 + n0.2 * s2 + n0.3 * s3 + n0.4 * s4 + n0.5 * s5 + n0.6 * s6 - K1 * K2 ^T.d * t,$ 

trace=TRUE,start=Startlist,data=Data)

summary(Fit\_H45.nls)

# A.2 Output (Results of Chapter 4)

P474

Parameters:

		Std.		
	Estimate	Error	t value	$Pr(\geq  t )$
K1	0.021676	0.00037	58.54	<2e-16
K2	1.552101	0.017322	89.6	<2e-16
n0.1	7.781425	0.142206	54.72	<2e-16
n0.2	8.803506	0.126776	69.44	<2e-16
n0.3	7.71231	0.100909	76.43	<2e-16
n0.4	7.869977	0.100909	77.99	<2e-16
n0.5	8.224296	0.098875	83.18	<2e-16
n0.6	7.562352	0.125908	60.06	<2e-16
n0.7	9.438821	0.106435	88.68	<2e-16
n0.8	8.471587	0.11869	71.38	<2e-16
n0.9	8.357541	0.106882	78.19	<2e-16
n0.10	7.276851	0.138857	52.41	<2e-16
n0.11	7.760851	0.138857	55.89	<2e-16

Residual standard error: 0.5188 on 276 degrees of freedom

		Std.
	Estimate	Error
Dref	46.13504	0.787949
Z	5.2378	0.132973

H474

Parameters:

		Std.		
	Estimate	Error	t value	$Pr(\geq  t )$
K1	0.024513	0.000634	38.65	<2e-16
K2	1.685991	0.018725	90.04	<2e-16
n0.1	7.710397	0.157679	48.9	<2e-16

n0.2	8.683295	0.145631	59.62	<2e-16
n0.3	7.927546	0.136132	58.23	<2e-16
n0.4	7.608491	0.147141	51.71	<2e-16
n0.5	9.034026	0.155443	58.12	<2e-16
n0.6	8.20164	0.164047	50	<2e-16

Residual standard error: 0.6232 on 159 degrees of freedom

		Std.
	Estimate	Error
Dref	40.79468	1.055439
Z	4.408096	0.093723

P45

Parameters:

		Std.		
	Estimate	Error	t value	$\Pr(\geq  t )$
K1	0.067991	0.001448	46.94	<2e-16
K2	1.772768	0.011702	151.5	<2e-16
n0.1	8.384732	0.094766	88.48	<2e-16
n0.2	7.988861	0.087466	91.34	<2e-16
n0.3	7.114367	0.09764	72.86	<2e-16
n0.4	7.77277	0.079494	97.78	<2e-16
n0.5	7.41625	0.089446	82.91	<2e-16
n0.6	7.827225	0.165366	47.33	<2e-16

# Residual standard error: 0.3484 on 115 degrees of freedom

		Std.
	Estimate	Error
Dref	14.70783	0.313232
Z	4.021686	0.046367

#### H45

Parameters:

		Std.		
	Estimate	Error	t value	$Pr(\geq  t )$
K1	0.063248	0.001819	34.76	<2e-16
K2	1.786784	0.016919	105.61	<2e-16
n0.1	8.292211	0.132925	62.38	<2e-16
n0.2	7.42766	0.124872	59.48	<2e-16
n0.3	7.038507	0.129579	54.32	<2e-16
n0.4	6.663056	0.113924	58.49	<2e-16
n0.5	7.400597	0.117341	63.07	<2e-16
n0.6	7.912529	0.227721	34.75	<2e-16

Residual

standard

error:

0.4932 on

124 degrees

of freedom

# Std.

	Estimate	Error
Dref	15.81078	0.454715
Z	3.96712	0.06472

# P190

Parameters:

## Std.

	Estimate	Error	t value	$Pr(\geq  t )$
K1	0.058477	0.001602	36.49	<2e-16
K2	1.67249	0.020709	80.76	<2e-16
n0.1	7.935879	0.102763	77.22	<2e-16
n0.2	8.209294	0.112955	72.68	<2e-16
n0.3	7.471911	0.150304	49.71	<2e-16

n0.4	8.943868	0.088995	100.5	<2e-16
n0.5	7.282378	0.106337	68.48	<2e-16
n0.6	8.550544	0.100022	85.49	<2e-16
n0.7	7.013655	0.16138	43.46	<2e-16
n0.8	6.526296	0.212263	30.75	<2e-16

Residual

standard

error:

0.4190 on

140 degrees

of freedom

		Std.
	Estimate	Error
Dref	17.10074	0.468481
Z	4.477007	0.107784

# H190

#### Parameters:

		Std.		
	Estimate	Error	t value	$\Pr(\geq  t )$
K1	0.044428	0.000678	65.57	<2e-16
K2	1.683517	0.009732	172.99	<2e-16
n0.1	7.922236	0.062066	127.64	<2e-16
n0.2	8.133696	0.077803	104.54	<2e-16
n0.3	7.841556	0.069883	112.21	<2e-16
n0.4	8.234689	0.064786	127.11	<2e-16
n0.5	7.851038	0.090938	86.33	<2e-16
n0.6	7.379568	0.091492	80.66	<2e-16

Residual

standard

error: 0.2774 on 127 degrees of freedom

Std.

	Estimate	Error
Dref	22.50838	0.343291
Z	4.420526	0.049058

	Measured	Identified							
Time	LOG10(N)	LOG10(N)	Squared difference	Parameters	Parameter S	tandard Error			
0.00	8.04	7.99	0.00	SI (Shoulder length)	79.49	13.61	Mean Su	m of Squared Err	ror 0.0808
0.00	8.04	7.99	0.00	kmax	0.03	0.00	Root Mean Su	m of Squared Err	ror 0.2843
0.00	7.98	7.99	0.00	LOG10(N0)	7.99	0.10		R-Squa	are 0.9783
5.00	8.02	7.99	0.00					R-Square adjust	ed 0.9769
5.00	8.03	7.99	0.00						
5.00	8.03	7.99	0.00	Inactivation model i	dentified				
50.00	7.83	7.87	0.00	N= N0 * exp(-kmax *	t) * ( exp(kma	x * SI))/(1+(exp(l	(max * SI) - 1) *exp	o(-kmax*t)))	
50.00	7.90	7.87	0.00	For identification pu	irposes refor	mulated as			
50.00	8.01	7.87	0.02	Log10(N) = Log10(N0	0) - kmax * t /	Ln(10) + Log10(	Exp(kmax * SI) / (1	+ (Exp(kmax * SI	l) - 1) * Exp(-kmax * t)
100.00	7.31	7.53	0.05	as can be derived fr	om				
100.00	7.32	7.53	0.04	A.H. Geeraerd, C.H.	Herremans ar	nd J.F. Van Impe	2000. Structural m	nodel requirement	ts
100.00	7.25	7.53	0.08	to describe microbial	inactivation d	uring a mild heat	treatment. Interna	tional Journal of	
150.00	6.77	6.96	0.04	Food Microbiology, 5	9(3), 185-209				
150.00	6.79	6.96	0.03						
150.00	6.81	6.96	0.02	10.00					
200.00	6.30	6.28	0.00	10.00					
200.00	6.29	6.28	0.00	8.00					
200.00	6.26	6.28	0.00	<b>F</b>					
250.00	5.81	5.59	0.05	6.00					
250.00	5.83	5.59	0.06						
250.00	5.88	5.59	0.08	<b>5</b> 4.00					
300.00	4.62	4.88	0.07	2 00					
300.00	4.88	4.88	0.00	2.00					
300.00	5.04	4.88	0.02	0.00					
350.00	4.80	4.18	0.39	0.00	100.00	200.00	200.00	400.00	500.00
350.00	4.51	4.18	0.11	0.00	100.00	200.00	300.00	400.00	500.00
350.00	4.63	4.18	0.21			Tir	ne		
400.00	3.87	3.47	0.16						
400.00	3.70	3.47	0.05		Г	Mocourod	Identified		
400.00	3.04	3.47	0.19						
450.00	2.00	2.77	0.59						
450.00	2.48	2.77	0.08						
450.00	2.48	2.77	0.08						
Least Sun	n of Squared	d Error	2.42						

# A.3 An example of the software output from *GlnaFit*.(Chapter 4)

Least Sum of Squared Error

# Appendix B



Figure B1. The aluminium cell used in experiments (Chapters 5 and 6).



Figure B2. The aluminium cells during heating inside the water bath (Chapter 5)



**Figure B3**. Temperature measured on the surface of chicken skin during heating to 60 °C in the programmable heating water bath (Chapter 5).



Figure B4. Temperature measured on the surface of chicken skin during heating to 60 °C in the programmable heating water bath (Chapter 5)



Figure B5. Temperature measured on the surface of chicken skin during heating to 56.5 °C in the programmable heating water bath (Chapter 5).


**Figure B6**. Temperature measured on the surface of chicken skin during heating to 56.5 °C in the programmable heating water bath (Chapter 5).

### Appendix C

Figure C1. Programmable Water Bath used (chapter 6)

A stainless steel electrical 1.8 l kettle (Fast boil 2400-watt concealed element, Breville) was used as a water bath. Eurostar electric stirrer (model ST PB, 300 RPM) was used to assure the temperature uniformity inside the kettle during the heating of the samples. The kettle was connected to a PID controller to input the desired heating temperature and heating time period for each step. (nine steps per pattern are available for program control). The PID has enhanced Communication functions, and it was fitted with a RS 485 interface card to allow communication via a Novus (USBi485) isolated converter to a PC to allow control, display and acquisition of the temperature profile for each run.



# C1. The code for kinetic parameters calculation by MATLAB Optimisation Toolbox (MathWorks Inc. Natick, MA, USA). (Chapter 6)

global Tref K1 K2 Tspline\_8 t\_i\_8 t\_a\_8 tobs\_8 nobs\_8 Tspline\_18 t\_i\_18 t\_a\_18 tobs\_18 nobs\_18;

Tref = 60.75; Tact = 47; K1 = 0.0637; %K1 = 0.077688; K2 = 1.1275; %K2 = 1.1; c0 = 0; %c0 = -6; n0\_8 = 8.8222; n0\_18 = 10.1634;

% 18min expt:

P474\_18min\_in; % load the data t = t-1; % times are measured in seconds starting at 0 b=0:30:max(t); % knotpoints for the spline fit to the temperature data; every 30 seconds Tspline=spline(b,T/spline(b,eye(length(b)),t)); % construct the best spline fit to data Tsp = ppval(Tspline,t);  $t_0_{18} = max(t(Tsp<Tact))+1; %t_0$  is time to reach activation temp  $t_{i_118} = 0:(t_0_{18}-1);$  %t\_i before reaching activation temp  $t_a_{18} = t_0_{18}:max(t);$  %t\_a after reaching activation temp Tspline\_18 = Tspline;  $t_{18} = t;$ 

% 8min expt:

P474\_8min\_in; % load the data

t = t-1; % times are measured in seconds starting at 0

b=0:30:max(t); % knotpoints for the spline fit to the temperature data; every 30 seconds Tspline=spline(b,T/spline(b,eye(length(b)),t)); % construct the best spline fit to data Tsp = ppval(Tspline,t); t\_0\_8 = max(t(Tsp<Tact))+1; %t\_0 is time to reach activation temp t\_i\_8 = 0:(t\_0\_8-1); %t\_i before reaching activation temp t\_a\_8 = t\_0\_8:max(t); %t\_a after reaching activation temp Tspline\_8 = Tspline; t\_8 = t;

x0 = [K1 K2 c0 n0\_8 n0\_18]; sse = ssefun(x0)

[x,resnorm,residual,exitflag,output,lambda,jacobian] = lsqnonlin(@resfun,x0);

x sse = resnorm df = length(tobs\_8)+length(tobs\_18)-5 mse = sse/df Sigma = mse\*inv (jacobian'\*jacobian) SE = sqrt(diag(Sigma))

function d = model(t,y)

global Tref K1 K2 Tspline\_8 t\_i\_8 t\_a\_8 tobs\_8 nobs\_8 Tspline\_18 t\_i\_18 t\_a\_18 tobs\_18 nobs\_18;

% Evaluate the cubic spline fit to the temperature data at time t

T = ppval(Tspline,t); A = -K1\*K2^(T-Tref); d=[A/(1+10^y(2)); A];

return

function  $d = model_{18(t,y)}$ 

global Tref K1 K2 Tspline\_8 t\_i\_8 t\_a\_8 tobs\_8 nobs\_8 Tspline\_18 t\_i\_18 t\_a\_18 tobs\_18 nobs\_18;

% Evaluate the cubic spline fit to the temperature data at time t

T = ppval(Tspline\_18,t); A =  $-K1*K2^{(T-Tref)};$ d=[A/(1+10^y(2)); A];

return

function  $d = model_8(t,y)$ 

global Tref K1 K2 Tspline\_8 t\_i\_8 t\_a\_8 tobs\_8 nobs\_8 Tspline\_18 t\_i\_18 t\_a\_18 tobs\_18 nobs\_18;

% Evaluate the cubic spline fit to the temperature data at time t

 $T = ppval(Tspline_8,t);$   $A = -K1*K2^{(T-Tref)};$  $d=[A/(1+10^{y}(2)); A];$ 

return

%Times and temperatures: t = 1:655; T = [20.5 .....];

%Observed data: tobs\_8 = [360 380 400 420 450 460]; nobs\_8 = [7.837272703 7.745270024 6.515211304 5.161368002 3.844321821 2.640978057];

function res = resfun(x)

global Tref K1 K2 Tspline\_8 t\_i\_8 t\_a\_8 tobs\_8 nobs\_8 Tspline\_18 t\_i\_18 t\_a\_18 tobs\_18

nobs\_18;

K1 = x(1); K2 = x(2); cs = x(3);  $ns_8 = x(4);$  $ns_18 = x(5);$ 

[t\_8,y\_8]=ode45('model\_8',t\_a\_8,[ns\_8,cs]); y\_8 = [repmat([ns\_8 cs],length(t\_i\_8),1); y\_8]; nmod = y\_8(tobs\_8,1)'; res\_8 = (nmod-nobs\_8)';

[t\_18,y\_18]=ode45('model\_18',t\_a\_18,[ns\_18,cs]); y\_18 = [repmat([ns\_18 cs],length(t\_i\_18),1); y\_18]; nmod = y\_18(tobs\_18,1)'; res\_18 = (nmod-nobs\_18)';

res = [res\_8; res\_18]; return

function sse = ssefun(x)

global Tref K1 K2 Tspline\_8 t\_i\_8 t\_a\_8 tobs\_8 nobs\_8 Tspline\_18 t\_i\_18 t\_a\_18 tobs\_18 nobs\_18;

K1 = x(1); K2 = x(2); cs = x(3);  $ns_8 = x(4);$  $ns_18 = x(5);$ 

[t\_8,y\_8]=ode45('model\_8',t\_a\_8,[ns\_8,cs]); y\_8 = [repmat([ns\_8 cs],length(t\_i\_8),1); y\_8]; nmod = y\_8(tobs\_8,1)'; sse\_8 = (nmod-nobs\_8)\*(nmod-nobs\_8)';

[t\_18,y\_18]=ode45('model\_18',t\_a\_18,[ns\_18,cs]); y\_18 = [repmat([ns\_18 cs],length(t\_i\_18),1); y\_18]; nmod = y\_18(tobs\_18,1)'; sse\_18 = (nmod-nobs\_18)\*(nmod-nobs\_18)';

```
sse = sse_8 + sse_18;
return
```

### Appendix D

### D.1 R software code (Chapter 7).

Data<-read.csv("Data.csv") Data\$Temp<-factor(Data\$Temp) Data\$Time<-factor(Data\$Time) Data\$Rep<-paste(Data\$Strain,Data\$Atm,Data\$Temp,Data\$Time,Data\$Replicate) Data\$Rep<-factor(Data\$Rep) Data\$Exp<-paste(Data\$Strain,Data\$Atm,Data\$Temp,Data\$Time) Data\$Exp<-factor(Data\$Exp)

```
write.csv(Data,"Ndata.csv")
```

library(nlme) library(lme4)

```
cuniq<-function(v) length(unique(v))</pre>
```

```
#Temp=4
```

Data4<-Data[Data\$Temp=="4",] Data4\$Time<-factor(Data4\$Time) Data4\$Rep<-factor(Data4\$Rep) Data4\$Exp<-factor(Data4\$Exp) write.csv(Data4,"Data4.csv") Model4.Exp<-lm(LogN~Strain\*Atm\*Time,data=Data4) Model4.Rep<-lm(LogN~Rep,data=Data4) anova(Model4.Exp,Model4.Rep) anova(Model4.Exp) Model4.aov<-aov(LogN~Strain\*Atm\*Time+Error(Rep),data=Data4) summary(Model4.aov)

head(Data4) Emean<-tapply(Data4\$LogN,Data4\$Exp,mean) cPlate<-tapply(Data4\$Plate,Data4\$Exp,cuniq) cRep<-tapply(Data4\$Replicate,Data4\$Exp,cuniq) cbind(Emean,cRep,cPlate) sig2e<-summary(Model4.aov)[[2]][[1]][1,3] sig2r<-summary(Model4.aov)[[1]][[1]][8,3] SE<-sqrt(sig2e/cPlate+sig2r/cRep) Res4<-cbind(Emean,cRep,cPlate,SE)

#### #Temp=10

Data10<-Data[Data\$Temp=="10",] Data10\$Time<-factor(Data10\$Time) Data10\$Rep<-factor(Data10\$Rep) Data10\$Exp<-factor(Data10\$Exp) write.csv(Data10,"Data10.csv") Model10.Exp<-lm(LogN~Strain\*Atm\*Time,data=Data10) Model10.Rep<-lm(LogN~Rep,data=Data10) anova(Model10.Exp,Model10.Rep) anova(Model10.Exp) Model10.aov<-aov(LogN~Strain\*Atm\*Time+Error(Rep),data=Data10) summary(Model10.aov)

#### head(Data10)

Emean<-tapply(Data10\$LogN,Data10\$Exp,mean) cPlate<-tapply(Data10\$Plate,Data10\$Exp,cuniq) cRep<-tapply(Data10\$Replicate,Data10\$Exp,cuniq) sig2e<-summary(Model10.aov)[[2]][[1]][1,3] sig2r<-summary(Model10.aov)[[1]][[1]][8,3] SE<-sqrt(sig2e/cPlate+sig2r/cRep) Res10<-cbind(Emean,cRep,cPlate,SE)

#Temp=20 Data20<-Data[Data\$Temp=="20",] Data20\$Time<-factor(Data20\$Time) Data20\$Rep<-factor(Data20\$Rep) Data20\$Exp<-factor(Data20\$Exp) write.csv(Data20,"Data20.csv") Model20.Exp<-lm(LogN~Strain\*Atm\*Time,data=Data20) Model20.Rep<-lm(LogN~Rep,data=Data20) anova(Model20.Exp,Model20.Rep) anova(Model20.Exp) Model20.aov<-aov(LogN~Strain\*Atm\*Time+Error(Rep),data=Data20) summary(Model20.aov)

#### head(Data20)

Emean<-tapply(Data20\$LogN,Data20\$Exp,mean) cPlate<-tapply(Data20\$Plate,Data20\$Exp,cuniq) cRep<-tapply(Data20\$Replicate,Data20\$Exp,cuniq) sig2e<-summary(Model20.aov)[[2]][[1]][1,3] sig2r<-summary(Model20.aov)[[1]][[1]][8,3] SE<-sqrt(sig2e/cPlate+sig2r/cRep) Res20<-cbind(Emean,cRep,cPlate,SE)

#Using lme:

Data4\$RT<-paste(Data4\$Rep,Data4\$Time) Data4\$RT<-factor(Data4\$RT) Data4.lme<-lme(LogN~Strain\*Atm\*Time,random=~1|RT,data=Data4)

### Appendix E

E.1 CORE MODEL used for risk assessment by the Bayesian approach (Chapter 10). model # starting the model ### ### CHICKEN FARM MODULE .....  $p.f \le exp(lp.f)/(1+exp(lp.f)); #vi$ lp.f ~ dnorm(m.f, tau.f); #.. m.f ~ dnorm(0.0,20.66); #pa s.f ~ dunif(0,0.2); tau.f <-  $1/(s.f^*s.f)$ ; #pa ### BROILER PRODUCTION MODULE .....  $p.b \le exp(lp.b)/(1+exp(lp.b)); #vi$ lp.b ~ dnorm(m.b, tau.b); #.. m.b <- m.f + d.bf; #cv  $d.bf \sim dnorm(0.1,206.6)I(0,); #pa$  $s.b \sim dunif(0,0.2); tau.b <- 1/(s.b*s.b); #pa$ ### HYGIENE MODULE ..... p.h <- p.hc \* p.hh; #vi p.hc ~ dbeta(8,8); #pa p.hh ~ dbeta(166,355); #pa #p.hh ~ dbeta(22,20);# pa ### CONSUMPTION MODULE ..... lambda.cs  $\sim$  dgamma(8,2); #vi ### EXPOSURE MODULE ..... p.ey <- 1 - pow((1-p.e),13); #vi p.e <- 1 - exp(-lambda.e); #cv lambda.e <- p.b \* p.h \* lambda.cs; #cv ### ILLNESS MODULE ..... p.ib <- p.ey \* p.ie; #vi p.it <- p.ib/(1 - (1-p.iq)\*(1-p.ey)); #vi p.ie <- (1-pow(1-p.ne,d)) \* p.in; #cv d <- vd[c.d]; #cv p.ne ~ dbeta(0.024,0.011)I(0.00001,0.999999); #pa p.in <- 0.33; #pa

p.iq ~ dbeta(39,12); #pa  $c.d \sim dcat(c.di[]); #..$ ### CHICKEN FARM MODULE ..... for (i.fs in 1:n.fst) #..  $g.fs[i.fs] \sim dbin(p.fs[i.fs], n.fs[i.fs]); #da$  $p.fs[i.fs] \le exp(lp.fs[i.fs]) / #cv$  $(1 + \exp(\ln fs[i.fs])); #...$  $lp.fs[i.fs] \sim dnorm(m.f,tau.f); #..$ } #.. ### BROILER PRODUCTION MODULE ..... for (i.bs in 1:n.bst) #.. g.bs[i.bs] ~ dbin(p.bs[i.bs],n.bs[i.bs]); #da  $p.bs[i.bs] \le exp(lp.bs[i.bs]) / #cv$  $(1 + \exp(lp.bs[i.bs])); #...$  $lp.bs[i.bs] \sim dnorm(m.b,tau.b); #..$ } #.. ### HYGIENE MODULE ..... ### CONSUMPTION MODULE .....  $n.csb \sim dpois(lambda.csb); #da$ lambda.csb <- n.cs\*lambda.cs;#cv ### EXPOSURE MODULE ..... ### ILLNESS MODULE ..... g.its  $\sim$  dbin(p.its,n.its); #da  $logit(p.its) \le logit(p.it) + err; #cv$ s.s <-1; tau.s <-1/(s.s\*s.s); #pa  $err \sim dnorm(0,tau.s); #..$ list(# starting the doses ..... c.di=c(0.500,0.163,0.222,0.097,0.018), vd=c(1,2,10,100,300)) list(# starting the data incorporated ..... n.fst=50, n.fs=c(188,150,148,195,170,309,211,144,145,158,107,153,150,181,160,204,57,52,62,84,87,40,83,5 4,84,181,160,172,229,141,168,181,176,159,211,163,157,140,176,146,151,67,61,61,70,64,65,60,66,7 5),

n.bst=44,

n.bs = c(75,60,60,72,60,60,60,57,60,72,60,57,57,72,60,72,60,60,60,60,60,72,57,72,60,72,60,54,60,69,42,57,60,72,60,72,60,60,57,72,60,75,60,55,75),

n.cs=21037900,

n.its=4292350,

g.fs=c(133,108,102,128,119,131,142,107,102,117,95,117,125,148,122,

137,20,12,17,34,24,31,50,50,63,66,140,106,133,107,114,121,119,142,154,126,137,132,153,98,83,17 ,15,20,19,16,27,23,34,55),

g.bs=c(34,31,37,31,32,26,35,31,32,45,34,26,34,52,31,41,37,36,

37,46,60,36,43,27,21,30,29,23,29,15,27,24,32,36,39,34,46,39,47,40,55,36,39,32),

n.csb=78634000,

g.its=6926,

E.2 A simplified poultry processing flow diagram.



### Appendix F

F1. The code of R programme used for ARIMA with intervention model used in the timeseries Chapter (11).

```
Inc<-read.csv("Incidence .csv")
head(Inc)
I.ts<-ts(Inc$Inc.10E5, frequency=12, start=c(1997, 1))
plot(I.ts, ylab="Cases per 100,000",type="b")
```

x < -c(rep(0, 122), rep(1, 34))

```
par(mfrow=c(2,1))
acf(I.ts,lag.max=48); pacf(I.ts,lag.max=48)
acf(diff(I.ts,12),lag.max=48); pacf(diff(I.ts,12),lag.max=48)
```

```
I.tr<-ts(Inc$Inc.10E5[1:144], frequency=12, start=c(1997, 1))
x.tr<-x[1:144]
```

```
I.ar<-arima(I.ts, order=c(1,0,0),seasonal=c(0,1,1),xreg=x)
print(I.ar)
par(mfrow=c(2,1))
I.res<-residuals(I.ar)
acf(I.res,lag.max=48); pacf(I.res,lag.max=48)
Box.test(I.res,12)
```

```
I.f<-predict(I.ar,n.ahead=12,newxreg=rep(1,12))
as.numeric(I.f$pred)
```

I.fit<-I.tr-I.res cat(I.fit, sep=" ")

```
I.ar<-arima(I.tr, order=c(1,0,0),seasonal=c(0,1,1),xreg=x.tr)
print(I.ar)
par(mfrow=c(2,1))
I.res<-residuals(I.ar)
```

```
acf(I.res,lag.max=48); pacf(I.res,lag.max=48)
BP<-c(Box.test(I.res,1)$p.val,Box.test(I.res,12)$p.val,Box.test(I.res,24)$p.val)
names(BP)<-c("1","12","24")
BP
```

I.f<-predict(I.ar,n.ahead=12,newxreg=rep(1,12)) as.numeric(I.f\$pred)

```
#Multiplicative - log transform
InI.tr<-ts(log(Inc$Inc.10E5[1:144]), frequency=12, start=c(1997, 1))
I.ar<-arima(InI.tr, order=c(1,0,0),seasonal=c(0,1,1),xreg=x.tr)
print(I.ar)
par(mfrow=c(2,1))
I.res<-residuals(I.ar)
acf(I.res,lag.max=48); pacf(I.res,lag.max=48)
Box.test(I.res,12)</pre>
```

I.fit<-lnI.tr-I.res cat(exp(I.fit), sep=" ")

```
I.f<-predict(I.ar,n.ahead=12,newxreg=rep(1,12))
as.numeric(exp(I.f$pred))
```

```
#Full data
lnI.ts<-ts(log(Inc$Inc.10E5), frequency=12, start=c(1997, 1))
I.ar<-arima(lnI.ts, order=c(1,0,0),seasonal=c(0,1,1),xreg=x)
print(I.ar)
par(mfrow=c(2,1))
I.res<-residuals(I.ar)
acf(I.res,lag.max=48); pacf(I.res,lag.max=48)
Box.test(I.res,12)
```

```
I.f<-predict(I.ar,n.ahead=12,newxreg=rep(1,12))
as.numeric(exp(I.f$pred))
```

### Appendix G

# G1. Research Information Sheet, Ethics Approval and the questionnaire used for the survey study (Chapter 13)

#### Lincoln University

#### [Faculty of Agriculture and Life Sciences. Department of Wine, Food and Molecular Biosciences]

#### **Research Information Sheet**

#### Introduction and invitation

I would like to invite you to participate in a project about poultry preparation practice at home. This will be a part of my PhD research program at Lincoln University. My research has been reviewed and approved by the Lincoln University Human Ethics Committee.

#### What is the aim of the project?

The project aims to investigate the handling practices of consumers during chicken preparation at home. The study would assess which poultry safe handling information poultry consumers lack and hence should be included in food safety campaigns, consumer education programmes and health promotion strategy in the future.

#### What types of participants are being sought?

I am inviting all participants who are 18 and above and cook poultry products at home to take part in the project. The participation in this research is voluntary, and there is no obligation to take part.

#### What will I be asked to do?

Your participation will involve completing a questionnaire, which I estimate will take 5 to 10 minutes to complete. The survey will ask you various questions about the approach you take to preparing the poultry products and about your food safety basic knowledge. You are kindly requested to assist in providing sincere responses to the questions contained in this questionnaire. All questions need to be completed. You may decline to answer any question in the survey. However, any data collated via incomplete surveys will be excluded from the analysis. Submission of the completed survey is deemed consent.

#### What use will be made of my data?

The results of the project will be included in my PhD thesis and submitted for publication in academic journals and officially reported to food safety professionals and health promotion professionals. The survey is anonymous. Your identity would not be recorded, and data will remain private. No one will have access to this information, other than the Human Ethics Committee in the event of an audit. Individual survey data will be stored in an electronic form with secure password protection. Only aggregated data will be presented in any publications.

#### Can I withdraw from the project?

We will not be able to withdraw your answers once submitted as the survey is anonymous.

#### What if I have any questions?

• Contact Information

Mr Ali Al-Sakkaf, Lincoln University Department of Wine, Food and Molecular Biosciences Faculty of Agriculture and Life Sciences RFH Building, P O Box 85084 Lincoln 7647 Christchurch Mobile: 0211030175 Email: <u>Ali.al-sakkaf@lincolnuni.ac.nz</u>

• Supervisor Contact Information Ravi Gooneratne Professor Department of Wine, Food and Molecular Biosciences Faculty of Agriculture and Life Sciences RFH Building, Room 74 P O Box 85084 Lincoln 7647 Christchurch New Zealand Telephone: +64 3 423 0636 Email: Ravi.Gooneratne@lincoln.ac.nz

Research Management Office

T 64 3 423 0817 PO Box 85084, Lincoln University Lincoln 7647, Christchurch New Zealand www.lincoln.ac.nz

#### 23 September 2019

Application No: 2019-53

Title: Survey of NZ consumer poultry handling practices at home to provide an insight into risk factors for campylobacteriosis in New Zealand.

#### Applicant: A Al-Sakkaf

The Lincoln University Human Ethics Committee has reviewed the above noted application. Thank you for your response to the questions which were forwarded to you on the Committee's behalf.

I am satisfied on the Committee's behalf that the issues of concern have been satisfactorily addressed. I am pleased to give final approval to your project.

Please note that this approval is valid for three years from today's date at which time you will need to reapply for renewal.

Once your field work has finished can you please advise the Human Ethics Secretary, Alison Hind, and confirm that you have complied with the terms of the ethical approval.

May I, on behalf of the Committee, wish you success in your research.

Yours sincerely

Con

Grant Tavinor Chair, Human Ethics Committee

**PLEASE NOTE:** The Human Ethics Committee has an audit process in place for applications. Please see 7.3 of the Human Ethics Committee Operating Procedures (ACHE) in the Lincoln University Policies and Procedures Manual for more information.

Please fill in this questionnaire if you are aged 18 or above and cook any raw poultry products at home.

For each of the following questions, please tick <u>one</u> response only.

### Section A: Food purchasing habits

# 1) Which of the following is most important to you when deciding where to shop for food?

	O The diversity of food items for sale.							
	O The cleanliness of the premises.							
	<b>O</b> I know the origin of the food items.							
	O Price.							
	O Close to home.							
	O None of the above.							
2)	During your last grocery shopping occasion when you purchased raw poultry, at what stage did you select raw poultry?							
	O Soon after I entered the shop.							
	OI followed the supermarket/grocery shop layout.							
	O After I had chosen all other non-perishable items.							
	O I have no routine for this.							
	O I do not know.							
3)	During your last grocery shopping occasion when you purchased raw poultry, was the raw poultry bagged separately from other items?							
	O No.							
	O Yes, only at my freezer/chiller after I returned home.							
	O Yes, in my carry-on bags.							
	<b>O</b> Yes, only in the trolley.							
	O Yes, in the trolley, carry-on bag and freezer/chiller.							
4)	Usually, how long do you leave the raw poultry at room temperature before storing it in a refrigerator or freezer at home (including the time you take to travel from the shop to your home)?							

O 0 minutes – 40 minutes.

**O** 41 minutes – One hour.

O More than an hour to one and a half hours.

O More than one hour and a half hours.

**O** I do not know.

### Section B: Preparation of raw poultry in the home

# 5) When thawing frozen raw poultry for cooking, how do you usually do it?

O Thaw it using the microwave.

O Thaw it on the kitchen countertop.

O Thaw it in/under running hot water in the sink.

O Thaw it in the kitchen sink.

O Thaw it in the bottom shelf of the refrigerator OR under running cold water in the sink.

**O** 'Never' thaw raw poultry

# 6) For each raw poultry handling occasion, which of the following do you usually do immediately after handling the poultry?

O I wipe my hands on a disposable towel.

O I wipe my hands on a reusable towel.

O I wash with soap and cold water.

O I use hand sanitiser and water or soap with hot water OR change my gloves.

O I do not clean my hands or change my gloves and continue to handle other utensils/items or foods in the kitchen.

# 7) For each handwashing occasion, how long do you spend washing your hands?

O 10 seconds or less.

O 11 – 20 seconds.

**O** More than 20 seconds.

**O** I do not know.

# 8) If you wash your hands, how do you usually dry your hands after handwashing in the kitchen?

O With the apron or on my clothes I am wearing.

O With an in-use' tea towel.

**O** With a disposable paper kitchen towel.

**O** With a towel used only for drying hands.

**O** I do not dry my hands.

# 9) There are many times that raw poultry may be handled or came into direct contact with utensils during preparation or grilling. Please indicate how raw poultry was handled/came into contact with utensils?

O Handled using bare hands.

O In contact with a previously unused fork/ unused spoon/unused utensil or handled while wearing previously unworn/gloves or handled using clean hands.

O In contact with a previously used fork/ used spoon/used utensil.

O Handled while wearing gloves.

**O** I do not remember.

# 10) After cutting or preparing the raw poultry, what was the next thing you do with the cutting board/utensil and/or knife?

O Continue to use the cutting board/utensil and knife for preparation of other food.

O Rinse the cutting board/utensil and knife with water before reuse or storage.

O Wash the cutting board/knife with soap and hot water before reuse or use a different cutting board and knife for preparing the other food.

O Wipe the cutting board/knife with a dishcloth before reuse or storage.

O Use a different cutting board/ utensil for other food but continue using the same knife.

# 11) Do you ever place cooked poultry on the same plate/surface where raw poultry meat has been?

O Yes, sometimes, especially when I do not have much time.

- **O**Yes, especially when I barbecue poultry.
- **O** Yes, after wiping it with a dishcloth.
- O No, not at all OR only after washing the plate/surface with hot water and detergent.
- **O**Yes, after rinsing it with water.

# 12) How do you check to see if a poultry product you have cooked is safe to eat?

O When it looks cooked.

O After an appropriate cooking time.

- From observing the texture and firmness (by inserting a knife to check uniform meat tenderness).
- OI never check to see if the poultry has been cooked adequately.
- O I observe when the exterior colour is golden brown, and the interior juices run clear OR measure the internal temperature of the poultry meat.

# 13) How often do you clean (using soap/water and cleaning products) your kitchen sink and benchtop?

O After every meal.

**O**Once a day.

Once a week.

O When they look dirty (food debris visible).

OI do not clean the kitchen benchtop.

14) Do you cover your food during storage?

**O**Yes, only if I keep the food for several hours.

**O**No, not at all.

**O**Yes, all the time.

O Yes, but only if I leave it overnight.

**O**Yes, sometimes, especially when I store it in the fridge.

### Section C: Food safety awareness

# 15) What temperature should refrigerators operate at for optimum safe food storage?

**O** I do not know.

O Less than 3°C.

**O** 8°C.

O Between 4°C and 7°C.

O More than 8°C.

# 16) How do you decide whether or not to consume food stored in the fridge?

**O** By the best before and the use-by date.

O I practice – 'first in – first out' techniques for food storage based on my memory.

- **O** I do not know.
- O I think I know when food is spoiled by its appearance and or smell.

O I do not pay attention to the expiry or best before dates.

O I do not pay attention to the use-by dates.

# 17) When preparing a meal, do you think about the basic food safety rules such as clean, separate, cook, chill and cover?

O Yes, I am aware of all them, or most of them.

**O**Yes, I have heard about them, but I do not understand them clearly.

**O**No, I have not heard about them.

O I have heard about a few of them clean, cover and chill.

O I have heard about two of these rules like clean and cover.

# 18) How likely are you to follow these rules clean, separate, cook, chill and cover?

O Always.
O Almost always.
O Sometimes.
O Rarely.
O Never.
19) Does raw poultry carry bacteria (bugs) that can make you sick?
<b>O</b> I do not know.
<b>O</b> Yes, I know that it carries bugs.
<b>O</b> Yes, I know, but I do not know the name of that bug.
<b>O</b> Yes, I know that it carries <i>Campylobacter</i> bug.
<b>O</b> No, I am not concerned about the bugs.
20) Cross-contamination can be defined as the transfer of harmful bugs to food from other foods, hands and other food contact surfaces. Please indicate your awareness and self-reported practice of this concept below:
OI am not familiar with the concept of cross-contamination.
O I have heard about the concept of cross-contamination, but I do not understand it.
O I have heard about the concept of cross-contamination, but I do not always take the necessary steps to prevent it.
O I have heard about the concept cross-contamination, and I always take the necessary steps to prevent it.
O I am not concerned.

21) How important do you consider it hygiene practices all of the time to from foodborne illness?	to follow/implement all good protect yourself and your family						
O I consider it as extremely important.							
O I consider it as important.							
O I consider it not important.							
OI consider it as just important.							
O I am not concerned. Section D: Par	rticipant details						
22) How many occasions in the past ye family experienced food poisoning nausea/vomiting, sudden onset of energy, dry mouth and tongue?	<u>ear</u> have you or anyone in your symptoms such as diarrhoea, fever, chills/muscle aches, lack of						
O 1-2 times.	<b>O</b> No experience in the past 12						
O 3-4 times.	months.						
<b>O</b> 5-6 times.	O No experience in the past 2 years.						
O 7-8 times.	·						
23) Based on your practices and your h influenced your food safety practic	behaviour, <u>who or what has most</u> es during food preparations?						
O Partner.							
O School/church/leisure centre/educati	onal campaigns.						
O Doctor, counsellor or health clinic.							
O Parents/grandparents, other relatives or friends.							
<b>O</b> TV, newspaper, radio, media in general.							
O Colleagues at work.							
<b>O</b> My knowledge.							
24) How often do you cook poultry at home?							
Once to twice per month.	Once a week or in the						

weekend.

244

O 2 to 3 times a week.

 $O_6$  to 7 times a week.

O 4 to 5 times a week.

### 25) In general, how did you learn to cook? (tick all that apply)

O By myself (from internet: videos, recipes, articles).

**O** From cookery books/TV.

O From partner/friends.

- O From other relatives (mum, grandparents).
- O From a training course (school, church, community centre, etc.).

### 26) What is your highest education level?

- **O**No formal schooling.
- O Kura Kaupapa/Primary school (including intermediate).
- O Secondary school (high school).
- O University student or completed university, Wānanga, polytechnic or another tertiary.
- O Postgraduate or higher qualification.

### 27) Which of these categories best describe your occupational status?

- O Employed full time (+ 30 hours week).
- O Employed part-time (15-30 hours week).
- O Unemployed.

O Student.

O Housewife/husband – home duties.

O Retired.

# 28) Which one of the following categories best describes the total yearly income of everyone in your household from all sources before tax?

O Lower than \$20,000/yr.	<b>O</b> \$60,001 – 80,000/yr.
<b>O</b> \$20,001 – 40,000/yr.	<b>O</b> \$80,001 – 100,000/yr.
<b>O</b> \$40,001 – 60,000/yr.	<b>O</b> \$100,001 or more.

29) Which of the following categori Tick all that apply	Which of the following categories describe your ethnic identity? Tick all that apply				
<b>O</b> NZ Maori.	O Latin American.				
<b>O</b> NZ European.	<b>O</b> North American.				
O Other European, including	O Middle Eastern.				
O Pasifika	<b>O</b> African.				
O Asian.	O Other, please specify:				

### 30) What is your gender and which statement best applies?

- O Female without children at home.
- O Female with children at home.
- O Male with children at home.
- O Male without children at home.
- O Prefer not to answer.

### 31) What age category do you fall under?

- **O** Under 20 years.
- **O** 20 29 years.
- **O** 30 39 years.
- **O** 40 49 years.
- **O** 50 59 years
- O 60 years or older.

### Thank you for completing this questionnaire.

# G2. Examples of SPSS analysis results' outputs for the consumer survey(Chapter 13).

	Ν	Me	ean	Std. Deviation	Skewr	ness	Kurtosis	
	Statistic	Statistic	Std. Error	Statistic	Statistic	Std. Error	Statistic	Std. Error
The total score of all sections	301	9.8306	0.20161	3.49779	0.256	0.14	-0.134	0.28
Valid N (listwise)	301							

### **Descriptive Statistics (13.4)**

### Tests of Normality (13.3.7)

	Which of these	Kolmogorov-Smirnov <sup>a</sup>		Shapiro-Wilk		lk	
	categories best						
	describe your						
	occupational status?	Statistic	df	Sig.	Statistic	df	Sig.
Score	Employed – full time	.086	145	.011	.982	145	.059
	(+ 30 hours week)						
	Employed – part-	.156	49	.004	.947	49	.028
	time (15-30 hours						
	week)						
	Retired	.147	51	.008	.964	51	.126
	Unemployed	.212	10	$.200^{*}$	.942	10	.581
	Housewife/husband	.109	14	.200*	.965	14	.804
	- home duties						
	Student	.109	32	$.200^{*}$	.972	32	.559

\*. This is a lower bound of the true significance.

a. Lilliefors Significance Correction



Association between respondents' socio-demographic characteristics and mean score

(13.4.1)

### Score \* How many occasions in the past year have you or anyone in your family experienced food poisoning symptoms such as diarrhoea, nausea/vomiting, sudden onset of fever, chills/muscle aches, lack of energy, dry mouth and tongue?

Score				
How many occasions in the past				
year have you or anyone in your				
family experienced food				
poisoning symptoms such as				
diarrhoea, nausea/vomiting,				
sudden onset of fever,				
chills/muscle aches, lack of				
energy, dry mouth and tongue?	Mean	Std. Error of Mean	Std. Deviation	Ν
1-2 times	9.5217	.41887	3.47939	69
3-4 times	8.5000	.84163	3.36650	16
5-6 times	7.3333	.66667	1.15470	3
No experience in the past 12 months	10.2529	.39280	3.66383	87
No experience in the past two	9.9365	.30337	3.40528	126
years				

Total	9.8306	.20161	3.49779	301

## Score \* Based on your practices and your behaviour, who or what has most influenced your personal hygiene habits?

Score

Based on your practices and your				
behaviour, who or what has most				
influenced your personal hygiene		Std. Error of the		
habits?	Mean	Mean	Std. Deviation	Ν
Partner	8.0714	.69566	3.68107	28
School/church/leisure centre/educational campaigns	9.5455	.82422	2.73363	11
Doctor, counsellor or health clinic	11.5000	1.50000	2.12132	2
Parents/grandparents, other relatives or friends	8.7910	.39188	3.20765	67
TV, newspaper, radio, media in general	8.8438	.51217	2.89727	32
Colleagues at work	11.4000	.60000	1.34164	5
My knowledge	10.7436	.28323	3.53758	156
Total	9.8306	.20161	3.49779	301

#### Score \* How often do you cook poultry at home?

#### Score

How often do you cook poultry at				
home?	Mean	Std. Error of Mean	Std. Deviation	Ν
Six to seven times a week	9.0000			1
Four to five times a week	10.0000	1.25576	4.52769	13
Twice to three times a week	10.0804	.33752	3.57201	112
Once a week or in the weekend	9.7455	.35404	3.71320	110

Once to twice times per month	9.5231	.34328	2.76760	65
Total	9.8306	.20161	3.49779	301

### Score \* What is your highest education level?

Score

What is your highest education level?	Mean	Std. Error of Mean	Std. Deviation	Ν
No formal schooling	6.0000	.00000	.00000	2
Kura Kaupapa/Primary school (including intermediate)	9.0000	1.00000	1.41421	2
Secondary school (high school)	9.5472	.52414	3.81579	53
University student or completed university, Wānanga, polytechnic or other tertiary	10.3154	.31139	3.55038	130
Postgraduate or higher qualification	9.4912	.30603	3.26755	114
Total	9.8306	.20161	3.49779	301

#### Score \* Which of these categories best describe your occupational status?

Score
-------

Which of these categories best				
describe your occupational status?	Mean	Std. Error of Mean	Std. Deviation	Ν
Employed – full time (+ 30 hours week)	9.7793	.30583	3.68267	145
Employed – part-time (15-30 hours week)	10.6122	.43430	3.04012	49
Retired	10.5686	.50971	3.64008	51
Unemployed	7.8000	1.06249	3.35989	10

Housewife/husband – home	9.7143	.84794	3.17269	14
duties				
Student	8.3750	.46174	2.61201	32
Total	9.8306	.20161	3.49779	301

# Score \* Which one of the following categories best describes the total yearly income of everyone in your household from all sources before tax?

Score								
Which one of the following								
categories best describes the total								
yearly income of everyone in your								
household from all sources before								
tax?	Mean	Std. Error of Mean	Std. Deviation	Ν				
Lower than \$20,000/yr.	8.1724	.60999	3.28491	29				
\$20,001 – 40,000/yr.	9.9250	.53993	3.41481	40				
\$40,001 - 60,000/yr.	9.9091	.47869	3.55003	55				
\$60,001 - 80,000/yr.	9.3143	.62681	3.70827	35				
\$80,001 – 100,000/yr.	9.7750	.58888	3.72440	40				
\$100,001 or more	10.4216	.32660	3.29847	102				
Total	9.8306	.20161	3.49779	301				

#### Score \* What is your gender and which statement best applies?

Score

What is your gender and which				
statement best applies?	Mean	Std. Error of Mean	Std. Deviation	Ν
Male without children at home	8.7547	.45709	3.32766	53
Female without children at home	10.1524	.35207	3.60763	105
Prefer not to answer	12.6667	1.20185	2.08167	3
Male with children at home	8.6087	.72977	3.49986	23

Female with children at home	10.1966	.31145	3.36882	117
Total	9.8306	.20161	3.49779	301

### Score \* What age category do you fall under?

Score

What age category do you fall				
under?	Mean	Std. Error of Mean	Std. Deviation	Ν
Under 20 years	7.3333	.88192	1.52753	3
20 – 29 years	8.6905	.44368	2.87536	42
30 – 39 years	9.5075	.43291	3.54355	67
40 – 49 years	10.2000	.45024	3.48751	60
50 – 59 years	10.2000	.45866	3.40152	55
60 or older	10.2973	.43950	3.78074	74
Total	9.8306	.20161	3.49779	301

### **Descriptive Statistics (13.4.2-5)**

	Ν	Mean		Std. Deviation		
	Statistic	Statistic	Std. Error	Statistic	Skewness	Kurtosis
Which of the following	301	.17	.022	.376	1.771	1.145
factors are most important						
to you when deciding						
where to shop for food?						
During your last grocery	301	.05	.013	.225	4.004	14.122
shopping occasion when						
you purchased raw poultry,						
at what stage did you select						
raw poultry?						
Was the poultry bagged	301	.25	.025	.433	1.166	646
separately from other						
items?						

Usually, how long do you	01	.88	.019	.325	-2.356	3.576
leave the raw poultry at						
room temperature before						
storing it in a refrigerator						
or freezer at home						
(including the time you						
take to travel from the						
shop to your home)?						
When thawing frozen raw	301	.34	.027	.475	.669	-1.563
poultry for cooking, how						
do you usually do it?						
For each raw poultry	301	.35	.028	.478	.622	-1.624
handling occasion, which						
of the following do you						
usually do immediately						
after handling the poultry?						
For each hand washing	301	.14	.020	.344	2.132	2.561
occasion, how long do you						
spend washing your hands?						
If you wash your hands,	301	.13	.019	.336	2.217	2.935
how do you usually dry						
your hands after hand						
washing in the kitchen?						
There are many times that	301	.46	.029	.499	.181	-1.980
raw poultry may be						
handled or came into						
direct contact with utensils						
during preparing or						
grilling. Please indicate						
how the raw poultry was						
handled/came into contact						
with utensils?						
After cutting or preparing	301	.78	.024	.414	-1.364	141
the raw poultry wh, at was						
the next thing you did with						
the cutting board/utensil						
and knife?						
Do you ever place cooked	301	.92	.015	.266	-3.205	8.327
poultry on the same						
plate/surface where raw						
poultry meat has been?						

How do you check to see	301	.47	.029	.500	.114	-2.000
have cooked is safe to eat?						
How often do you clean (using soap or cleaning product) your kitchen sink and benchtop?	301	.55	.029	.498	222	-1.964
Do you cover your food during storage?	301	.72	.026	.449	990	-1.027
What temperature should refrigerators operate at for optimum safe food storage?	301	.52	.029	.500	087	-2.006
How do you decide whether or not to consume food stored in the fridge?	301	.48	.029	.500	.074	-2.008
When preparing a meal, do you think about the basic food safety rules such as clean, separate, cook, chill and cover?	301	.67	.027	.471	732	-1.474
How likely are you to follow these rules clean, separate, cook, chill, cover?	301	.25	.025	.431	1.186	596
Does raw poultry carry bacteria (bugs) that can make you sick?	301	.50	.029	.501	020	-2.013
CroCrosscontaminationn is defined as the transfer of harmful bugs to food from other foods, hands and other food contact surfaces. Please indicate your awareness and self- reported action of this concept below.	301	.66	.027	.473	700	-1.520
How important do you	301	.52	.029	.500	087	-2.006
------------------------------	-----	--------	--------	---------	------	--------
consider it to						
follow/implement all good						
hygiene practices all of the						
time to protect yourself						
and your family from						
foodborne illness?						
Score	301	9.8306	.20161	3.49779	.256	134
Valid N (listwise)	301					

## Score (13.4.2-5)

		Frequency	Per cent	Valid Percent	Cumulative Percent
Valid 2.00 3.00 4.00 5.00 6.00 7.00 8.00 9.00 10.00 11.00 12.00 13.00 14.00 15.00 16.00 17.00 18.00 19.00 Total	2.00	3	1.0	1.0	1.0
	5	1.7	1.7	2.7	
	6	2.0	2.0	4.7	
	19	6.3	6.3	11.0	
	6.00	23	7.6	7.6	18.6
	7.00	24	8.0	8.0	26.6
	8.00	31	10.3	10.3	36.9
	31	10.3	10.3	47.2	
	32	10.6	10.6	57.8	
	11.00	29	9.6	9.6	67.4
	12.00	39	13.0	13.0	80.4
	13.00	18	6.0	6.0	86.4
	14.00	14	4.7	4.7	91.0
	15.00	6	2.0	2.0	93.0
	16.00	11	3.7	3.7	96.7
	2	.7	.7	97.3	
	18.00	3	1.0	1.0	98.3
	19.00	5	1.7	1.7	100.0
	301	100.0	100.0		

## **Descriptive Statistics**

	Mean	Std. Deviation	Ν
The score Food Safety Awareness section	3.6113	1.69462	301
The score Temperature Control and Proper Storage section	2.4684	1.05349	301
The score of Hygiene section	.9900	.94687	301
The score of Cross- Contamination Prevention section	2.7608	1.13252	301

## Correlations (13.4.6)

\*\*. Correlation is significant at the 0.01 level (2-tailed).

Correlations					
		The score Food Safety Awareness section	The core temperature Control and Proper Storage section	The score of Hygiene section	The score of Cross Contamination Prevention section
The score of Food Safety Awareness section	Pearson Correlation	1	.373**	.378**	.417**
	Sig. (2- tailed)		0.000	0.000	0.000
	N	301	301	301	301
The score of Temperature Control and Proper Storage section	Pearson Correlation	.373**	1	.329**	.321**
	Sig. (2- tailed)	0.000		0.000	0.000
	N	301	301	301	301
The score Hygiene section	Pearson Correlation	.378**	.329**	1	.237**
	Sig. (2- tailed)	0.000	0.000		0.000
	N	301	301	301	301
The score of Cross Contamination Prevention section	Pearson Correlation	.417**	.321**	.237**	1
	Sig. (2- tailed)	0.000	0.000	0.000	
	N	301	301	301	301

## 15 Bibliography

Abbot, JM, Byrd-Bredbenner, C, Schaffner, D, Bruhn, CM & Blalock, L 2009. Comparison of food safety cognitions and self-reported food-handling behaviors with observed food safety behaviors of young adults. *European Journal of Clinical Nutrition*, 63(4), 572-579.

Adams, MR & Moss, MO 2000. Food microbiology, Cambridge, Royal Society of Chemistry.

- Aiello, AE, Larson, EL & Sedlak, R 2008. Personal health bringing good hygiene home. American Journal of Infection Control, 36(10, Supplement 1), S152-S165.
- Akil, L & Ahmad, HA 2019. Quantitative Risk Assessment Model of Human Salmonellosis Resulting from Consumption of Broiler Chicken. *Diseases*, 7(1), 19.
- Al-Sakkaf, A 2012. Evaluation of food handling practice among New Zealanders and other developed countries as a main risk factor for campylobacteriosis rate. *Food Control* 27(2), 330-337
- Al-Sakkaf, A 2015. Domestic food preparation practices: a review of the reasons for poor home hygiene practices *Health Promotion International*, 30(3), 427-437.
- Al-Sakkaf, A & Jones, G 2012. Thermal inactivation of *Campylobacter jejuni* in broth. *Journal of Food Protection*, 75(6), 1029-1035.
- Al-Sakkaf, A, Jones, G & Mawson, J. 2010. Survival of New Zealand relevant *Campylobacter* strains
   Final Report: N0174/06 Prepared for New Zealand Food Safety Authority (NZFSA).
   Available: <a href="http://www.foodsafety.govt.nz/elibrary/industry/campylobacter-strains/report.pdf">http://www.foodsafety.govt.nz/elibrary/industry/campylobacter-strains</a>
- Albert, I, Grenier, E, Denis, JB & Rousseau, J 2008. Quantitative risk assessment from farm to fork and beyond: a global Bayesian approach concerning food-borne diseases. *Risk Analysis*, 28(2), 557-571.
- Albert, I & Mafart, P 2005. A modified Weibull model for bacterial inactivation. *International Journal* of Food Microbiology, 100(1-3), 197-211.
- Allan, PD, Palmer, C, Chan, F, Lyons, R, Nicholson, O, Rose, M, Hales, S & Baker, MG 2018. Food safety labelling of chicken to prevent campylobacteriosis: Consumer expectations and current practices. *BMC Public Health*, 18(1).
- Allen, VM, Bull, SA, Corry, JEL, Domingue, G, Jørgensen, F, Frost, JA, Whyte, R, Gonzalez, A, Elviss, N & Humphrey, TJ 2007. *Campylobacter* spp. contamination of chicken carcasses during processing in relation to flock colonisation. *International Journal of Food Microbiology*, 113(1), 54-61.
- Allos, BM & Acheson, D 2001. Campylobacter jejuni infections: update on emerging issues and trends.

Clinical Infectious Diseases, 32(8), 1201-1206.

- Altekruse, SF, Street, DA, Fein, SB & Levy, AS 1996. Consumer knowledge of foodborne microbial hazards and food-handling practices. *Journal of Food Protection*, 59, 287-294.
- Alter, T & Scherer, K 2006. Stress response of *Campylobacter* spp. and its role in food processing. *Journal of Veterinary Medicine Series B*, 53(8), 351-357.
- Altizer, S, Dobson, A, Hosseini, P, Hudson, P, Pascual, M & Rohani, P 2006. Seasonality and the dynamics of infectious diseases. *Ecology Letters*, 9(4), 467-484.
- Anderson, JB, Shuster, TA, Hansen, KE, Levy, AS & Volk, A 2004. A Camera's view of consumer food-handling behaviors. *Journal of the American Dietetic Association*, 104(2), 186-191.
- Andritsos, ND, Tzimotoudis, N & Mataragas, M 2020. Estimating the performance of four culture media used for enumeration and detection of Campylobacter species in chicken meat. *LWT*, 118.
- Anonymous. 2012. Consumer Food Behaviours [Online]. Safefood, Cork (Republic of Ireland): Safefood. Available: <u>http://www.safefood.eu/Publications/Research-reports/Food-Behaviours.aspx</u> [Accessed 05th August 2012].
- Anonymous. 2016. Food Health and Safety New Zealand Government, 2017. Available: <u>https://www.govt.nz/browse/consumer-rights-and-complaints/food-safety/food-health-and-safety/</u> [Accessed 11 October ].
- Anonymous. 2017. Infection with Campylobacter jejuni and C. coli. *Manual of diagnostic tests and vaccines for terrestrial animals* [Online]. Available: <u>https://www.oie.int/fileadmin/Home/eng/Health\_standards/tahm/2.09.03\_CAMPYL</u> <u>O.pdf</u> [Accessed 18 May].
- Anonymous. 2019. Tips for food safety. New Zealand Food Safety [Online]. Available: <u>https://www.mpi.govt.nz/food-safety/food-safety-for-consumers/tips-for-food-safety/</u> [Accessed 13/08/2019].
- Armstrong, JS (ed.) 2001. *Principles of forecasting : a handbook for researchers and practitioners,* Boston, MA: Kluwer Academic.
- Avens, JS, Albright, SN, Morton, AS, Prewitt, BE, Kendall, PA & Sofos, JN 2002. Destruction of microorganisms on chicken carcasses by steam and boiling water immersion. *Food Control*, 13(6-7), 445-450.
- Babo Martins, S, Rushton, J & Stärk, KDC 2017. Economics of zoonoses surveillance in a 'One Health' context: An assessment of Campylobacter surveillance in Switzerland. *Epidemiology* and Infection, 145(6), 1148-1158.
- Baker, M. 2011. Campylobacteriosis cases halved following controls on chicken meat [Online]. Press release.

Dunedin: University of Otago. Retrieved 16 May 2011: <u>http://www.otago.ac.nz/news/news/otago018240.html</u>.

- Baker, M, Sears, A, Wilson, N, French, N, Marshall, J, Muellner, P, Campbell, D, van der Logt, P
  & Lake, R 2011. Keep the focus on contaminated poultry to further curtail New Zealand's campylobacteriosis epidemic. *New Zealand Medical Journal*, 124(1338), 135-139.
- Baker, M, Sned, E & Wilson, N 2006a. Is the major increase in notified campylobacteriosis in New Zealand real? *Epidemiology and Infection*, 135(1), 163-170.
- Baker, M, Wilson, N & Edwards, R 2007. Campylobacter infection and chicken: An update on New Zealand's largest 'common source outbreak. New Zealand Medical Journal, 120(1261).
- Baker, M, Wilson, N, Ikram, R, Chambers, S, Shoemack, P & Cook, G 2006b. Regulation of chicken contamination urgently needed to control New Zealand's serious campylobacteriosis epidemic. *New Zealand Medical Journal*, 119(1243).
- Barba, FJ, Koubaa, M, do Prado-Silva, L, Orlien, V & Sant'Ana, AdS 2017. Mild processing applied to the inactivation of the main foodborne bacterial pathogens: A review. *Trends in Food Science & Technology*, 66, 20-35.
- Barker, GC, Talbot, NLC & Peck, MW 2002. Risk assessment for *Clostridium botulinum*: a network approach. *International Biodeterioration & Biodegradation*, 50(3-4), 167-175.
- Barker, J, Naeeni, M & Bloomfield, SF 2003. The effects of cleaning and disinfection in reducing Salmonella contamination in a laboratory model kitchen. Journal of Applied Microbiology, 95(6), 1351-1360.
- Baserisalehi, M, Bahador, N & Kapadnis, B 2006. Campylobacter: An emerging pathogen. *Microbiology*, 1(1), 23-37.
- Bearth, A, Cousin, ME & Siegrist, M 2013. Uninvited guests at the table A consumer intervention for safe poultry preparation. *Journal of Food Safety*, 33(4), 394-404.
- Bearth, A, Cousin, ME & Siegrist, M 2014. Poultry consumers' behaviour, risk perception and knowledge related to campylobacteriosis and domestic food safety. *Food Control*, 44, 166-176.
- Benschop, J, Stevenson, MA, Dahl, J, Morris, RS & French, NP 2008. Temporal and longitudinal analysis of Danish swine salmonellosis control programme data: implications for surveillance. *Epidemiology and Infection*, 136(11), 1511-1520.
- Bergsma, NJ, Fischer, ARH, van Asselt, ED, Zwietering, MH & de Jong, AEI 2007. Consumer food preparation and its implication for survival of *Campylobacter jejuni* on chicken. *British Food Journal*, 109(7), 548-561.
- Berrang, ME & Dickens, JA 2000. Presence and level of Campylobacter spp. on broiler carcasses

throughout the processing plant. Journal of Applied Poultry Research, 9(1), 43-47.

- Berrang, ME, Windham, WR & Meinersmann, RJ 2011. Campylobacter, Salmonella, and Escherichia coli on broiler carcasses subjected to a high pH scald and low pH postpick chlorine dip1. *Poultry Science*, 90(4), 896-900.
- Beumer, R, Bloomfield, SF, Exner, M, Fara, GM, Nath, KJ & Scott, E. 2002. The Infection Potential in the Domestic Setting and the Role of Hygiene Practice in Reducing Infection. *International Scientific Forum on Home Hygiene* [Online]. Available: <u>http://www.ifhhomehygiene.org/best-practice-review/chain-infection-transmission-home-and-everydaylife-settings-and-role-hygiene</u> [Accessed 27/07/2019].
- Bhaduri, S & Cottrell, B 2004. Survival of Cold-Stressed *Campylobacter jejuni* on Ground Chicken and Chicken Skin during Frozen Storage. *Applied and Environmental Microbiology*, 70(12), 7103-7109.
- Biggs, R 2008. *Campylobacter* reductions by poultry processing with chlorine and Sanaova. *Industrial trials*. Auckland: Tegel Foods Ltd.
- Biggs, R 2009. Quality evaluation of sono steam treated chicken Privite Report submitted to Tegel Foods Ltd,. *Plant demonstration for poultry industries* Berlin: Sonosteam.
- Birk, T, Grønlund, AC, Christensen, BB, Knøchel, S, Lohse, K & Rosenquist, H 2010. Effect of organic acids and marination ingredients on the survival of *Campylobacter jejuni* on meat. *Journal of Food Protection*, 73(2), 258-265.
- Black, RE, Myron, ML, Clements, ML, Hughes, TP & Blaser, MJ 1988. Experimental Campylobacter jejuni infection in humans. The Journal of Infectious Diseases, 157(3), 472-479.
- Blankenship, LC & Craven, SE 1982. *Campylobacter jejuni* survival in chicken meat as a function of temperature. *Applied and Environmental Microbiology*, 44(1), 88-92.
- Blaser, MJ, Hardesty, HL, Powers, B & Wang, WL 1980. Survival of *Campylobacter fetus subsp. jejuni* in biological milieus. *Journal of Clinical Microbiology*, 11(4), 309-313.
- Bloomfield, A & Neal, G 1997. Consumer food safety knowledge in Auckland. Auckland, New Zealand: Auckland Healthcare Public Health Protection, .
- Bonham, CS & Gangnes, B 1996. Intervention analysis with cointegrated time series: The case of the Hawaii hotel room tax. *Applied Economics*, 28(10), 1281-1293.
- Bostan, K 2001. Effects of cooking and cold storage on the survival of *Campylobacter jejuni* in meatballs. *Archiv Fur Lebensmittelhygiene*, 52(2), 28-30.
- Box, GEP, Jenkins, GM & Reinsel, GC 2008. *Time series analysis : forecasting and control,* Hoboken, N.J., John Wiley.
- Boxall, N. 2005. The epidemiology of Campylobacter jejuni in commercial broiler flocks in new zealand PhD

thesis. Doctor of philosophy PhD, Massey University.

- Boysen, L & Rosenquist, H 2009. Reduction of thermotolerant *Campylobacter* species on broiler carcasses following physical decontamination at slaughter. *Journal of Food Protection*, 72, 497-502.
- Breand, S, Fardel, G, Flandrois, JP, Rosso, L & Tomassone, R 1998. Model of the influence of time and mild temperature on *Listeria monocytogenes* nonlinear survival curves. *International Journal of Food Microbiology*, 40(3), 185-195.
- Brennan, M, McCarthy, M & Ritson, C 2007. Why do consumers deviate from best microbiological food safety advice? An examination of 'high-risk' consumers on the island of Ireland. *Appetite*, 49(2), 405-418.
- Brockwell, PJ & Davis, RA 2002. Introduction to time series and forecasting, New York, Springer.
- Bromley, J, Jackson, NA, Clymer, OJ, Giacomello, AM & Jensen, FV 2005. The use of Hugin® to develop Bayesian networks as an aid to integrated water resource planning. *Environmental Modelling and Software*, 20(2), 231-242.
- Brooks, SP & Gelman, A 1998. General methods for monitoring convergence of iterative simulations. *Journal of Computational and Graphical Statistics*, 7(4), 434-455.
- Brown, ME 1988. Introduction to thermal analysis : techniques and applications, London; New York, Chapman and Hall.
- Bruhn, CM 2014. Chicken preparation in the home: an observational study. *Food Protection Trends,* 34(5), 318-330.
- Brynestad, S, Braute, L, Luber, P & Bartelt, E 2008. Quantitative microbiological risk assessment of campylobacteriosis cases in the German population due to consumption of chicken prepared in homes. *International Journal of Risk Assessment and Management*, 8(3), 194-213.
- Büchner, C, Thomas, S, Jaros, D & Rohm, H 2012. Fast-responding thermal-death-time tubes for the determination of thermal bacteria inactivation. *Engineering in Life Sciences*, 12(1), 109-112.
- Bumgarner, R 2013. Overview of DNA microarrays: types, applications, and their future. *Current protocols in molecular biology*, Chapter 22, Unit-22.1.
- Burgess, CM, Gianotti, A, Gruzdev, N, Holah, J, Knøchel, S, Lehner, A, Margas, E, Esser, SS, Sela Saldinger, S & Tresse, O 2016. The response of foodborne pathogens to osmotic and desiccation stresses in the food chain. *International Journal of Food Microbiology*, 221, 37-53.
- Busschaert, P, Geeraerd, AH, Uyttendaele, M & Van Impe, JF 2011. Hierarchical Bayesian analysis of censored microbiological contamination data for use in risk assessment and mitigation. *Food Microbiology*, 28(4), 712-719.
- Butzler, J-P 2004. Campylobacter, from obscurity to celebrity. Clinical Microbiology & Infection, 10(10),

868-876.

Butzler, J-P 2014. Campylobacter infection in man and animals, Boca Raton: CRC Press.

- Byrd, JA, Sams, AR, Hargis, BM & Caldwell, DJ 2011. Effect of selected modified atmosphere packaging on Campylobacter survival in raw poultry1. *Poultry Science*, 90(6), 1324-1328.
- Byrd-Bredbenner, C, Berning, J, Martin-Biggers, J & Quick, V 2013. Food safety in home kitchens: a synthesis of the literature. *International journal of environmental research and public health*, 10(9), 4060-4085.
- Byrd-Bredbenner, C, Maurer, J, Wheatley, V, Cottone, E & Clancy, M 2007a. Observed food safety behaviours of young adults. *British Food Journal*, 109(7), 519-530.
- Byrd-Bredbenner, C, Wheatley, V, Schaffner, D, Bruhn, C, Blalock, L & Maurer, J 2007b. Development and implementation of a food safety knowledge instrument. *Journal of Food Science Education*, 6(3), 46-55.
- Cahill, SM 2005. Risk ssessment and campylobacteriosis. In: Smulders, FGM & Colins, JD (eds.) Risk management strategies : monitoring and surveillance. Wageningen: Wageningen Academic Publishers.
- Calistri, P & Giovannini, A 2008. Quantitative risk assessment of human campylobacteriosis related to the consumption of chicken meat in two Italian regions. *International Journal of Food Microbiology*, 128(2), 274-287.
- Campanella, OH & Peleg, M 2001. Theoretical comparison of a new and the traditional method to calculate *Clostridium botulinum* survival during thermal inactivation. *Journal of the Science of Food and Agriculture*, 81(11), 1069-1076.
- Cassin, MH, Lammerding, AM, Todd, ECD, Ross, W & McColl, RS 1998. Quantitative risk assessment for *Escherichia coli* O157:H7 in ground beef hamburgers. *International Journal of Food Microbiology*, 41(1), 21-44.
- Cebrián, G, Condón, S & Mañas, P 2019. Heat resistance, membrane fluidity and sublethal damage in Staphylococcus aureus cells grown at different temperatures. *International Journal of Food Microbiology*, 289, 49-56.
- Cerf, O 1977. Tailing of survival curves of bacterial-spores. *Journal of Applied Bacteriology*, 42(1), 1-19.
- CFSAN 2008. The Bad Bug Book: foodborne pathogenic microorganisms and natural toxins handbook, College Park, Md., US Food and Drug Administration, Center for Food Safety and Applied Nutrition.
- Chan, KF, Tran, HL, Kanenaka, RY & Kathariou, S 2001. Survival of clinical and poultry-derived isolates of *Campylobacter jejuni* at a low temperature (4 degrees C). *Applied and Environmental*

Microbiology, 67(9), 4186-4191.

- Chancey, CC, Brooks, JC, Martin, JN, Echeverry, A, Jackson, SP, Thompson, LD & Brashears, MM 2013. Survivability of Escherichia coli O157:H7 in mechanically tenderized beef steaks subjected to lactic acid application and cooking under simulated industry conditions. *Journal* of Food Protection, 76(10), 1778-1783.
- Chapman, B, Otten, A, Fazil, A, Ernst, N & Smith, BA 2016. A review of quantitative microbial risk assessment and consumer process models for Campylobacter in broiler chickens. *Microbial Risk Analysis*, 2-3, 3-15.
- Chatfield, C 1978. The Holt-Winters Forecasting Procedure. *Journal of the Royal Statistical Society*. Series C (Applied Statistics), 27(3), 264-279.
- Chatfield, C 2004. The analysis of time series : an introduction, Boca Raton, FL, Chapman & Hall/CRC.
- Chen, Y, Jackson, KM, Chea, FP & Schaffner, DW 2001. Quantification and variability analysis of bacterial cross-contamination rates in common food service tasks. *Journal of Food Protection*, 64, 72-80.
- Chick, H 1910. The process of disinfection by chemical agencies and hot water. *Journal of Hygiene*, 10, 237-286.
- Christensen, BB, Rosenquist, H, Sommer, HM, Nielsen, NL, Fagt, S, Andersen, NL & Nørrung, B 2005. A model of hygiene practices and consumption patterns in the consumer phase. *Risk Analysis*, 25(1), 49-60.
- Christopher, FM, Smith, GC & Vanderzant, C 1982. Effect of temperature and pH on the survival of *Campylobacter fetus*. *Journal of Food Protection*, 45(3), 253-259.
- Chrystal, ND, Hargraves, SJ, Boa, AC & Ironside, CJ 2008. Counts of *Campylobacter* spp. and prevalence of *Salmonella* associated with New Zealand broiler carcasses. *Journal of Food Protection*, 71(174), 2526-2532.
- Chung, CA 2004. Simulation modeling handbook : a practical approach, Boca Raton, CRC Press.
- Chung, HJ, Birla, SL & Tang, J 2008. Performance evaluation of aluminum test cell designed for determining the heat resistance of bacterial spores in foods. LWT - Food Science and Technology, 41(8), 1351-1359.
- Chynoweth, RW, Hudson, JA & Thom, K 1998. Aerobic growth and survival of *Campylobacter jejuni* in food and stream water. *Letters in Applied Microbiology*, 27(6), 341-344.
- Cil, GI, Ozdemir, H, Onaran, B, Cengiz, G & Sen, E 2019. Effect of lactic acid and steam treatments on Campylobacter jejuni on chicken skin. *Emirates Journal of Food and Agriculture*, 31(2), 143-147.
- Coates, D, Hutchinson, D & Bolton, F 1987. Survival of thermophilic Campylobacters on fingertips

and their elimination by washing and disinfection. Epidemiology and Infection, 99(2), 265.

Coddington, D. 2007. Poor hygiene is enough to make you sick. New Zealand Herald Sunday [Online]. Available:

https://www.nzherald.co.nz/nz/news/article.cfm?c\_id=1&objectid=10441988 [Accessed 28 May].

- Codex Alimentarius Commission. 1999. Principles and Guidelines for the Conduct of Microbiological Risk Assessment. Joint FAO/WHO food standards programme. [Online]. (CAC/GL-30) Rome, Italy: FAO, WHO. Available: <u>http://www.codexalimentarius.net/download/standards/357/CXG\_030e.pdf</u> [Accessed May 2006].
- Cody, MM & Hogue, MA 2003. Results of the home food safety it is in your hands 2002 survey: Comparisons to the 1999 benchmark survey and healthy people 2010 food safety behaviors objective. *Journal of the American Dietetic Association*, 103(9), 1115-1125.
- Cogan, TA, Slader, J, Bloomfield, SF & Humphrey, TJ 2002. Achieving hygiene in the domestic kitchen: the effectiveness of commonly used cleaning procedures. *Journal of Applied Microbiology*, 92(5), 885-892.
- Cole, MB & Jones, MV 1990. A submerged-coil heating apparatus for investigating thermal inactivation of micro-organisms. *Letters in Applied Microbiology*, 11(5), 233-235.
- Condon, S, Arrizubieta, MJ & Sala, FJ 1993. Microbial heat resistance determinations by the multipoint system with the thermoresistometer TR-SC. Improvement of this methodology. *Journal of Microbiological Methods*, 18(4), 357-366.
- Condon, S, Lopez, P, Oria, R & Sala, FJ 1989. Thermal death determination: design and evaluation of a thermoresistometer. *Journal of Food Science*, 54(2), 451-457.
- Conesa, R, Andreu, S, Fernandez, PS, Esnoz, A & Palop, A 2009. Nonisothermal heat resistance determinations with the thermoresistometer Mastia. *Journal of Applied Microbiology*, 107(2), 506-513.
- Conesa, R, Periago, PM, Esnoz, A, Lopez, A & Palop, A 2003. Prediction of *Bacillus subtilis* spore survival after a combined non-isothermal-isothermal heat treatment. *European Food Research* and Technology, 217(4), 319-324.
- Cools, I, Uyttendaele, M, Caro, C, D'Haese, E, Nelis, HJ & Debevere, J 2003. Survival of *Campylobacter jejuni* strains of different origin in drinking water. *Journal of Applied Microbiology*, 94(5), 886-892.
- Corry, JE, Post, DE, Colin, P & Laisney, MJ 1995. Culture media for the isolation of *Campylobacters*. *International Journal of Food Microbiology*, 26(1), 43-76.

- Corry, JEL, James, SJ, Purnell, G, Barbedo-Pinto, CS, Chochois, Y, Howell, M & James, C 2007. Surface pasteurisation of chicken carcasses using hot water. *Journal of Food Engineering*, 79(3), 913-919.
- Coşansu, S & Ayhan, K 2010. Effects of lactic and acetic acid treatments on *Campylobacter jejuni* inoculated onto chicken leg and breast meat during storage at 4C and -18C. *Journal of food processing and preservation.*, 34(Supplement 1), 98-113.
- Coshall, J 2005. Interventions on UK earnings and expenditures overseas. *Annals of Tourism Research,* 32(3), 592-609.
- Courtney, SM, Majowicz, SE & Dubin, JA 2016. Food safety knowledge of undergraduate students at a Canadian university: results of an online survey. *BMC Public Health*, 16(1), 1-16.
- Covello, VT 2003. Best practices in public health risk and crisis communication. *Journal of Health Communication: International Perspectives,* 8(3 supp 1), 5 - 8.
- Cox, JM & Pavic, A 2009. Advances in enteropathogen control in poultry production. *Journal of Applied Microbiology*, 108(3), 745-755.
- Cox, NA, Richardson, LJ, Bailey, JS, Cosby, DE, Cason, JA, Musgrove, MT & Mead, GC 2005. Bacterial contamination of poultry as a risk to human health, Boca Raton, CRC Press.
- Cressey, P, Nokes, C & Lake, R. 2008. Chlorinated compounds formed during chlorine wash of chicken meat. Available: <u>http://www.nzfsa.govt.nz/science/research-</u> <u>projects/FW0883\_Chlorinated\_compounds\_in\_chicken\_meat.pdf</u> [Accessed 23/04/2009].
- Creswell, JW 2018. Research design : qualitative, quantitative & mixed methods approaches, Los Angeles SAGE. 2018. ©2018.
- Cui, S, Ge, B, Zheng, J & Meng, J 2005. Prevalence and antimicrobial resistance of *Campylobacter* spp. and *Salmonella* serovars in organic chickens from Maryland retail stores. *Applied and Environmental Microbiology*, 71(7), 4108-4111.
- Cunha, LM, Oliveira, FAR & Oliveira, JC 1998. Optimal experimental design for estimating the kinetic parameters of processes described by the Weibull probability distribution function. *Journal of Food Engineering*, 37(2), 175-191.
- Curtis, V & Cairneross, S 2003. Effect of washing hands with soap on diarrhoea risk in the community: a systematic review. *The Lancet Infectious Diseases*, 3(5), 275-281.
- Dasti, JI, Tareen, AM, Lugert, R, Zautner, AE & Groß, U 2010. Campylobacter jejuni: A brief overview on pathogenicity-associated factors and disease-mediating mechanisms. *International Journal of Medical Microbiology*, 300(4), 205-211.
- Davis, BD, Dulbeco, R, Eisen, HN & Ginsberg, HS 1990. Microbiology, Philadelphia, PA, J.B.

Lippincott.

- Davis, MA & Conner, DE 2007. Survival of *Campylobacter jejuni* on poultry skin and meat at varying temperatures. *Poultry Science*, 86(4), 765-767.
- De Cesare, A, Sheldon, BW, Smith, KS & Jaykus, LA 2003. Survival and persistence of *Campylobacter* and *Salmonella* species under various organic loads on food contact surfaces. *Journal of Food Protection*, 66(9), 1587-1594.
- De Greeff, SC, Dekkers, ALM, Teunis, P, Rahamat-Langendoen, JC, Mooi, FR & De Melker, HE 2009. Seasonal patterns in time series of pertussis. *Epidemiology and Infection*, 137(10), 1388-1395.
- de Jong, AEI, van Asselt, ED, Zwietering, MH, Nauta, MJ & de Jonge, R 2012. Extreme heat resistence of *Campylobacter jejuni, Escherichia coli,* and *Salmonella typhimurium* during cooking of chicken breast fillet. *International journal of microbiology,* 2012, 196841-196841.
- de Jong, AEI, Verhoeff-Bakkenes, L, Nauta, MJ & de Jonge, R 2008. Cross-contamination in the kitchen: effect of hygiene measures. *Journal of Applied Microbiology*, 105(2), 615-624.
- De Ledesma, AMR, Riemann, HP & Farver, TB 1996. Short-time treatment with alkali and/or hot water to remove common pathogenic and spoilage bacteria from chicken wing skin. *Journal of Food Protection*, 59, 746-750.
- Deboer, E & Hahne, M 1990. Cross-contamination with *campylobacter-jejuni* and *salmonella-spp* from raw chicken products during food preparation. *Journal of Food Protection*, 53(12), 1067-1068.
- Decordt, S, Vanhoof, K, Hu, J, Maesmans, G, Hendrickx, M & Tobback, P 1992. Thermostability of soluble and immobilized alpha-amylase from *Bacillus licheniformis*. *Biotechnology and Bioengineering*, 40(3), 396-402.
- DeDonder, S, Jacob, CJ, Surgeoner, BV, Chapman, B, Phebus, R & Powell, DA 2009. Self-reported and observed behavior of primary meal preparers and adolescents during preparation of frozen, uncooked, breaded chicken products. *British Food Journal*, 111(9), 915-929.
- Delignette-Muller, ML, Cornu, M, Pouillot, R & Denis, JB 2006. Use of Bayesian modelling in risk assessment: Application to growth of *Listeria monocytogenes* and food flora in cold-smoked salmon. *International Journal of Food Microbiology*, 106(2), 195-208.
- DeLurgio, SA 1998. Forecasting principles and applications, Boston, Irwin/McGraw-Hill.
- Dharod, JM, Perez-Escamilla, R, Paciello, S, Bermudez-Millan, A, Venkitanarayanan, K & Damio, G 2007. Comparison between self-reported and observed food handling behaviors among Latinas. *Journal of Food Protection*, 70(8), 1927-1932.
- Dingle, KE, Colles, FM, Ure, R, Wagenaar, JA, Duim, B, Bolton, FJ, Fox, AJ, Wareing, DRA & Maiden, MCJ 2002. Molecular characterization of Campylobacter jejuni clones: a basis for

epidemiologic investigation. *Emerging infectious diseases*, 8(9), 949-955.

- Dingle, KE, Colles, FM, Wareing, DRA, Ure, R, Fox, AJ, Bolton, FE, Bootsma, HJ, Willems, RJL, Urwin, R & Maiden, MCJ 2001. Multilocus sequence typing system for *Campylobacter jejuni*. *Journal of Clinical Microbiology*, 39(1), 14-23.
- Diplock, KJ, Dubin, JA, Leatherdale, ST, Hammond, D, Jones-Bitton, A & Majowicz, SE 2018. Observation of High School Students' Food Handling Behaviors: Do They Improve following a Food Safety Education Intervention? 81(6), 917-925.
- Dogan, OB, Clarke, J, Mattos, F & Wang, B 2019. A quantitative microbial risk assessment model of Campylobacter in broiler chickens: Evaluating processing interventions. *Food Control*, 100, 97-110.
- Dolan, KD, Yang, L & Trampel, CP 2007. Nonlinear regression technique to estimate kinetic parameters and confidence intervals in unsteady-state conduction-heated foods. *Journal of Food Engineering*, 80(2), 581-593.
- Donelan, AK, Chambers, DH, Chambers, E, Godwin, SL & Cates, SC 2016. Consumer Poultry Handling Behavior in the Grocery Store and In-Home Storage. *Journal of Food Protection*, 79(4), 582-588.
- Donnison, A 2002. Isolation of thermotolerant Campylobacter review & methods for New Zealand laboratories, Prepared for the Ministry of Health under the auspices of the Enteric Zoonotic Disease Research in New Zealand Steering Committee.
- Dosman, DM, Adamowicz, WL & Hrudey, SE 2001. Socioeconomic determinants of health and food safety-related risk perceptions. *Risk Analysis*, 21(2), 307.
- Downes, FP & Ito, K (eds.) 2001. Compendium of methods for the microbiological examination of foods, Washington, D.C.: American Public Health Association.
- Doyle, M, Ruoff, K, Pierson, M, Weinberg, W, Soule, B & Michaels, B 2000. Reducing Transmission of Infectious Agents in the Home Part II: Control Points. *Dairy Food and Environmental Sanitation*, 20(6), 418-425.
- Doyle, ME, Mazzotta, AS, Wang, T, Wiseman, DW & Scott, VN 2001. Heat resistance of *Listeria* monocytogenes. Journal of Food Protection, 64, 410-429.
- Doyle, MP & Roman, DJ 1981. Growth and survival of *Campylobacter fetus* subsp. *jejuni* as a function of temperature and pH. *Journal of Food Protection*, 44(8), 596-601.
- Doyle, TJ, Glynn, MK & Groseclose, SL 2002. Completeness of notifiable infectious disease reporting in the United States: An analytical literature review. *American Journal of Epidemiology*, 155(9), 866-874.
- Duncan, GE 2014. Determining the health benefits of poultry industry compliance measures: The

case of campylobacteriosis regulation in New Zealand. New Zealand Medical Journal, 127(1391), 22-37.

- Duqué, B, Haddad, N, Rossero, A, Membré, J-M & Guillou, S 2019. Influence of cell history on the subsequent inactivation of Campylobacter jejuni during cold storage under modified atmosphere. *Food Microbiology*, 84, 103263.
- Eberhart-Phillips, J, Walker, N, Garrett, N, Bell, D, Sinclair, D, Rainger, W & Bates, M 1997. Campylobacteriosis in New Zealand: results of a case-control study. *Journal of Epidemiology and Community Health*, 51(6), 686-691.
- ECDC. 2017. Reported, confirmed campylobacteriosis cases: number and rate per 100 000 population, EU/EEA, 2010–2014. Available: <u>https://www.ecdc.europa.eu/en/publications-data/table-1-reported-confirmed-</u> <u>campylobacteriosis-cases-number-and-rate-100-000</u> [Accessed 8 January 2019].
- Efimochkina, NR, Pichugina, TV, Stetsenko, VV, Bykova, IB, Markova, YM, Korotkevich, YV, Polyanina, AS & Sheveleva, SA 2017. Optimization of microbiological methods for food control based on the differential-diagnostic media for the isolation and cultivation of bacteria of the genus Campylobacter. *Voprosy Pitaniia*, 86(5), 34-41.
- EFSA. 2010. Analysis of the baseline survey on the prevalence of Campylobacter in broiler batches and of Campylobacter and Salmonella on broiler carcasses in the EU 2008 Part A: *Campylobacter* and *Salmonella* prevalence estimates. 2010. Available: <u>http://www.efsa.europa.eu/fr/scdocs/doc/1503.pdf</u>.
- EFSA 2011a. Panel on Food Contact Materials, Enzymes, Flavourings Processing Aids. Scientific Opinion on the Chemical Safety of Irradiation of Food. *EFSA Journal*, 9(4), 1930.
- EFSA 2011b. Scientific Opinion on Campylobacter in broiler meat production: control options and performance objectives and/or targets at different stages of the food chain. EFSA Journal, 9(4), 2105.
- EFSA 2016. The European Union summary report on trends and sources of zoonoses, zoonotic agents and food-borne outbreaks in 2015. *EFSA Journal*, 14(12), e04634.
- EFSA 2018. The European Union summary report on trends and sources of zoonoses, zoonotic agents and food-borne outbreaks in 2017. *EFSA Journal*, 16(12).
- Egan, J. 2006. Country presentation Ireland. European Union Reference Laboratory for Campylobacter workshop. Held on 23-25 October 2006 [Online]. Available: <u>http://www.sva.se/upload/pdf/CRL/John Egan presentation.pdf</u> [Accessed 11 November].

Ejemot, RI, Ehiri, JE, Meremikwu, MM & Critchley, JA 2008. Hand washing for preventing

diarrhoea. Cochrane Database of Systematic Reviews, (1).

- El-Shibiny, A, Connerton, P & Connerton, I 2009. Survival at refrigeration and freezing temperatures of *Campylobacter coli* and *Campylobacter jejuni* on chicken skin applied as axenic and mixed inoculums. *International Journal of Food Microbiology*, 131(2-3), 197-202.
- Elsgaard, L & Jørgensen, LW 2002. A sandwich-designed temperature-gradient incubator for studies of microbial temperature responses. *Journal of Microbiological Methods*, 49(1), 19-29.
- EpiSurv. 2019. New Zealand notifiable disease surveillance data Available: <u>https://www.esr.cri.nz/our-services/consultancy/public-health/notifiable-diseases-</u> <u>intelligence-dashboard/</u> [Accessed 14 Feb 2020].
- EU 2016. Approving peracetic acid as an existing active substance for use in biocidal products for producttypes 1, 2, 3, 4, 5 and 6. *In:* THE EUROPEAN COMMISSION (ed.)
   *COMMISSION IMPLEMENTING REGULATION (EU) 2016/672* Brussels: Official Journal of the European Union
- Evans, EW & Redmond, EC 2018. Behavioral observation and microbiological analysis of older adult consumers' cross-contamination practices in a model domestic kitchen. *Journal of Food Protection*, 81(4), 569-581.
- Evans, EW & Redmond, EC 2019. Domestic kitchen microbiological contamination and selfreported food hygiene practices of older adult consumers. *Journal of Food Protection*, 82(8), 1326-1335.
- Evers, EG & Bouwknegt, M 2016. Combining QMRA and Epidemiology to Estimate Campylobacteriosis Incidence. *Risk Analysis*, 36(10), 1959-1968.
- Evers, EG & Chardon, JE 2010. A swift Quantitative Microbiological Risk Assessment (sQMRA) tool. *Food Control*, 21(3), 319-330.
- Facciolà, A, Riso, R, Avventuroso, E, Visalli, G, Delia, SA & Laganà, P 2017. Campylobacter: from microbiology to prevention. *Journal of preventive medicine and hygiene*, 58(2), E79-E92.
- Fang, T & Lin, L 1994. Inactivation of Listeria monocytogenes on raw pork treated with modified atmosphere packaging and nisin. *Journal of Food Protection*, 2(3), 189-200.
- Farakos, SMS, Schaffner, DW & Frank, JF 2014. Predicting survival of salmonella in low-water activity foods: An analysis of literature data. *Journal of Food Protection*, 77(9), 1448-1461.
- Farias, ADS, Akutsu, RCCA, Botelho, RBA & Zandonadi, RP 2019. Good practices in home kitchens: Construction and validation of an instrument for household food-borne disease assessment and prevention. *International Journal of Environmental Research and Public Health*, 16(6).
- Fazil, A, Lowman, R, Stern, N & Lammerding, A 1999. Quantitative risk assessment model for

Campylobacter jejuni on chicken. 10th International Workshop on Campylobacter, Helicobacter and related organisms. Baltimore, Maryland, USA.

- FDA/CFSAN. 2000. Kinetics of microbial inactivation for alternative food processing technologies. Available: <u>https://www.fda.gov/files/food/published/Kinetics-of-</u><u>Microbial-Inactivation-for-Alternative-Food-Processing-Technologies.pdf</u> [Accessed 17 October 2007].
- Fein, SB, Lando, AM, Levy, AS, Teisl, MF & Noblet, C 2011. Trends in u.s. consumers' safe handling and consumption of food and their risk perceptions, 1988 through 2010. *Journal* of Food Protection, 74(9), 1513-1523.
- Fernandez, A, Salmeron, C, Fernandez, PS & Martinez, A 1999. Application of a frequency distribution model to describe the thermal inactivation of two strains of *Bacillus cereus*. Trends in Food Science & Technology, 10(4-5), 158-162.
- Fields, JA & Thompson, SA 2008. Campylobacter jejuni CsrA mediates oxidative stress responses, biofilm formation, and host cell invasion. *Journal of Bacteriology*, 190(9), 3411-3416.
- Fischer, A & De Vries, P 2008. Everyday behaviour and everyday risk: An approach to study people's responses to frequently encountered food related health risks. *Health, Risk & Society,* 10(4), 385-397.
- Fischer, A & Frewer, LJ 2008. Food-Safety Practices in the Domestic Kitchen: Demographic, Personality, and Experiential Determinants1. *Journal of Applied Social Psychology*, 38(11), 2859-2884.
- Fleury, M, Charron, DF, Holt, JD, Allen, OB & Maarouf, AR 2006. A time series analysis of the relationship of ambient temperature and common bacterial enteric infections in two Canadian provinces. *International Journal of Biometeorology*, 50(6), 385-391.
- Fontannaz-Aujoulat, F, Frost, M & Schlundt, J 2019. WHO Five Keys to Safer Food communication campaign - Evidence-based simple messages with a global impact. *Food Control*, 101, 53-57.
- Forsythe, SJ 2000. The microbiology of safe food, Oxford ; Malden, MA, Blackwell Science.
- Foster, AM, Ketteringham, LP, Swain, MJ, Kondjoyan, A, Havet, M, Rouaud, O & Evans, JA 2006. Design and development of apparatus to provide repeatable surface temperature-time treatments on inoculated food samples. *Journal of Food Engineering*, 76(1), 7-18.
- Fraser, RW, Williams, NT, Powell, LF & Cook, AJC 2010. Reducing *Campylobacter* and *Salmonella* infection: two studies of the economic cost and attitude to adoption of on-farm biosecurity measures. *Zoonoses and Public Health*, 57(7-8), e109-e115.
- Fravalo, P, Laisney, MJ, Gillard, MO, Salvat, G & Chemaly, M 2009. Campylobacter Transfer from

Naturally Contaminated Chicken Thighs to Cutting Boards Is Inversely Related to Initial Load. *Journal of Food Protection*, 72(9), 1836-1840.

- French, N. 2008a. Enhancing surveillance of potentially foodborne enteric diseases in New Zealand: Human campylobacteriosis in the Manawatu. Available: <u>https://www.mpi.govt.nz/dmsdocument/23128-enhancing-surveillance-of-potentially-foodborne-enteric-diseases-in-new-zealand-human-campylobacteriosis-in-the-manawatu-final-report</u> [Accessed 2nd February 2009].
- French, N. 2008b. Human campylobacteriosis in the Manawatu. Report for NZFSA. Available: <u>http://www.foodsafety.govt.nz/elibrary/industry/enhancing-surveillance-potentially-</u> research-projects-2/Campy\_Attribution\_Manawatu.pdf [Accessed 30/09/2008].
- French, NP, Zhang, J, Carter, GP, Midwinter, AC, Biggs, PJ, Dyet, K, Gilpin, BJ, Ingle, DJ, Mulqueen, K, Rogers, LE, Wilkinson, DA, Greening, SS, Muellner, P, Fayaz, A & Williamson, DA 2019. Genomic Analysis of Fluoroquinolone- and Tetracycline-Resistant Campylobacter jejuni Sequence Type 6964 in Humans and Poultry, New Zealand, 2014-2016. Emerg Infect Dis, 25(12), 2226-2234.
- Frewer, LJ, Fischer, ARH & Kaptan, G 2016. Consumer Perceptions of Risks From Food. Handbook of Hygiene Control in the Food Industry: Second Edition.
- Friedman, CR, Neimann, J, Wegener, HC & Tauxe, RV 2000. Epidemiology of Campylobacter jejuni infections in the United States and other industrialized nations. *Campylobacter*. Washington, USA: ASM International.
- Garaizar, J, Rementeria, A & Porwollik, S 2006. DNA microarray technology: a new tool for the epidemiological typing of bacterial pathogens? FEMS Immunology & Medical Microbiology, 47(2), 178-189.
- Garbutt, C, Simmons, G, Patrick, D & Miller, T 2007. The public hand hygiene practices of New Zealanders: a national survey. *Journal of the New Zealand Medical Association*, 120, 1265.
- Garenaux, A, Jugiau, F, Rama, F, de Jonge, R, Denis, M, Federighi, M & Ritz, M 2008. Survival of *Campylobacter jejuni* strains from different origins under oxidative stress conditions: Effect of temperature. *Current Microbiology*, 56(4), 293-297.
- Garenaux, A, Ritz-Bricaud, M & Federighi, M 2005. *Campylobacter* and food safety: analysis, assessment and risk management. *Bulletin de l'Academie V eterinaire de France*, 158(4), 377-383.
- Garre, A, Huertas, JP, González-Tejedor, GA, Fernández, PS, Egea, JA, Palop, A & Esnoz, A 2018. Mathematical quantification of the induced stress resistance of microbial populations during non-isothermal stresses. *International Journal of Food Microbiology*, 266, 133-141.
- Gaze, JE, Brown, GD, Gaskell, DE & Banks, JG 1989. Heat resistance of Listeria monocytogenes

in homogenates of chicken, beef steak and carrot. Food Microbiology, 6(4), 251-259.

- Geeraerd, AH, Herremans, CH & Van Impe, JF 2000. Structural model requirements to describe microbial inactivation during a mild heat treatment. *International Journal of Food Microbiology*, 59(3), 185-209.
- Geeraerd, AH, Valdramidis, VP & Van Impe, JF 2005. GInaFiT, a freeware tool to assess non-loglinear microbial survivor curves. *International Journal of Food Microbiology*, 102(1), 95-105.
- Gelman, A & Rubin, DB 1992. Inference from iterative simulation using multiple sequences. *Statistical science*, 7(4), 457-472.
- Georgsson, F, Þorkelsson, ÁE, Geirsdóttir, M, Reiersen, J & Stern, NJ 2006. The influence of freezing and duration of storage on *Campylobacter* and indicator bacteria in broiler carcasses. *Food Microbiology*, 23(7), 677-683.
- Gharst, G, Oyarzabal, OA & Hussain, SK 2013. Review of current methodologies to isolate and identify Campylobacter spp. from foods. *Journal of Microbiological Methods*, 95(1), 84-92.
- Gibbens, JC, Pascoe, SJS, Evans, SJ, Davies, RH & Sayers, AR 2001. A trial of biosecurity as a means to control *Campylobacter* infection of broiler chickens. *Preventive Veterinary Medicine*, 48, 85-99.
- Gilbert, SE, Bayne, G, Wong, T, Lake, R & Whyte, R. 2006. Domestic food practices in New Zealand 2005-2006, project report for New Zealand Food Safety Authority. Available: <u>http://www.foodsafety.govt.nz/elibrary/industry/Domestic\_Food-</u> <u>Information\_Home.pdf</u> [Accessed 26 June 2006].
- Gilbert, SE, Whyte, R, Bayne, G, Paulin, SM, Lake, RJ & van der Logt, P 2007. Survey of domestic food handling practices in New Zealand. *International Journal of Food Microbiology*, 117(3), 306-311.
- Gill, CO & Harris, LM 1982. Survival and growth of *Campylobacter fetus* subsp. *jejuni* on meat and in cooked foods. *Applied and Environmental Microbiology*, 44(2), 259-263.
- Gill, CO, McGinnis, CJ, Houde, A, Lamoureux, L & Leblanc, D 2005a. Microbiological conditions of moisture-enhanced pork before and after cooking. *Food Microbiology*, 22(4), 321-327.
- Gill, CO & McGinnis, JC 2005. Factors affecting the microbiological condition of the deep tissues of mechanically tenderized beef. *Journal of Food Protection*, 68(4), 796-800.
- Gill, CO, McGinnis, JC, Rahn, K, Young, D, Lee, N & Barbut, S 2005b. Microbiological condition of beef mechanically tenderized at a packing plant. *Meat Science*, 69(4), 811-816.
- Gill, CO, Moza, LF & Barbut, S 2009. Survival of bacteria in less than thorough cooked, brineinjected steaks. *Food Control*, 20(5), 501-507.
- Gill, CO, Uttaro, B, Badoni, M & Zawadski, S 2008. Distributions of brine and bacteria in beef

primal cuts injected with brine without, or before or after mechanical tenderizing. *Meat Science*, 79(1), 181-187.

- Giovannini, A, Prencipe, V, Conte, A, Marino, L, Petrini, A, Pomilio, F, Rizzi, V & Migliorati, G 2004. Quantitative risk assessment of *Salmonella* spp. infection for the consumer of pork products in an Italian region. *Food Control*, 15(2), 139-144.
- Gkana, EN & Nychas, G-JE 2018. Consumer food safety perceptions and self-reported practices in Greece. *International Journal of Consumer Studies*, 42(1), 27-34.
- Goh, C & Law, R 2002. Modeling and forecasting tourism demand for arrivals with stochastic nonstationary seasonality and intervention. *Tourism Management*, 23(5), 499-510.
- Goh, SG, Leili, A-H, Kuan, CH, Loo, YY, Lye, YL, Chang, WS, Soopna, P, Najwa, MS, Tang, JYH, Yaya, R, Nishibuchi, M, Nakaguchi, Y & Son, R 2014. Transmission of Listeria monocytogenes from raw chicken meat to cooked chicken meat through cutting boards. *Food Control*, 37, 51-55.
- Göksoy, EO, James, C, Corry, JEL & James, SJ 2001. The effect of hot-water immersions on the appearance and microbiological quality of skin-on chicken-breast pieces. *International Journal* of Food Science and Technology, 36(1), 61-69.
- Gölz, G, Kittler, S, Malakauskas, M & Alter, TJCCMR 2018. Survival of Campylobacter in the Food Chain and the Environment. 5(2), 126-134.
- Gong, S, Wang, X, Yang, Y & Bai, L 2016. Knowledge of food safety and handling in households: A survey of food handlers in Mainland China. *Food Control*, 64, 45-53.
- González, M & Hänninen, M-L 2011. Reduction of Campylobacter jejuni counts on chicken meat treated with different seasonings. *Food Control*, 22(11), 1785-1789.
- González, M, Skandamis, PN & Hänninen, M-L 2009. A modified Weibull model for describing the survival of Campylobacter jejuni in minced chicken meat. *International Journal of Food Microbiology*, 136(1), 52-58.
- Gorman, R, Bloomfield, S & Adley, CC 2002. A study of cross-contamination of food-borne pathogens in the domestic kitchen in the Republic of Ireland. *International Journal of Food Microbiology*, 76(1-2), 143-150.
- Greenland, S & Drescher, K 1993. Maximum likelihood estimation of the attributable fraction from logistic models. *Biometrics*, 49(3), 865-872.
- Griffith, CJ, Price, P, Peters, A & Clayton, D 2001. An evaluation of food handlers knowledge, belief and attitudes about food safety and its interpretation using social cognition models.London: Food Standards Agency,.

Grigoriadis, SG, Koidis, PA, Vareltzis, KP & Batzios, CA 1997. Survival of Campylobacter jejuni

inoculated in fresh and frozen beef hamburgers stored under various temperatures and atmospheres. *Journal of Food Protection*, 60, 903-907.

Guerin, MT, Sir, C, Sargeant, JM, Waddell, L, O'Connor, AM, Wills, RW, Bailey, RH & Byrd, JA 2010. The change in prevalence of *Campylobacter* on chicken carcasses during processing: A systematic review. *Poultry Science*, 89(5), 1070-1084.

Guerrero-Legarreta, I 2009. Handbook of poultry science and technology, Hoboken, N.J., Wiley.

- Gunsen, U 2008. Inactivation of *Campylobacter jejuni* inoculated into baked drumsticks. *Archiv Fur Lebensmittelhygiene*, 59(5), 175-179.
- Gunther Iv, NW, Rajkowski, KT & Sommers, C 2015. Survival after Cryogenic Freezing of Campylobacter Species in Ground Turkey Patties Treated with Polyphosphates. 78(2), 419-423.
- Habib, I, Sampers, I, Uyttendaele, M, Berkvens, D & De Zutter, L 2008. Performance characteristics and estimation of measurement uncertainty of three plating procedures for *Campylobacter* enumeration in chicken meat. *Food Microbiology*, 25(1), 65-74.
- Habib, I, Uyttendaele, M & De Zutter, L 2010. Survival of poultry-derived *Campylobacter jejuni* of multilocus sequence type clonal complexes 21 and 45 under freeze, chill, oxidative, acid and heat stresses. *Food Microbiology*, 27(6), 829-834.
- Haddad, N, Burns, CM, Bolla, JM, Prévost, H, Fédérighi, M, Drider, D & Cappelier, JM 2009. Long-term survival of Campylobacter jejuni at low temperatures is dependent on polynucleotide phosphorylase activity. *Applied and environmental microbiology*, 75(23), 7310-7318.
- Hadjicharalambous, C, Grispoldi, L & Goga, BC 2019. Quantitative risk assessment of Listeria monocytogenes in a traditional RTE product. *EFSA Journal*, 17(S2).
- Hanninen, ML, Korkeala, H & Pakkala, P 1984. Effect of various gas atmospheres on the growth and survival of *Campylobacter jejuni* on beef. *Journal of Applied Bacteriology*, 57(1), 89-94.
- Hansen, D & Larsen, B 2007. Reduction of *Campylobacter* on chicken carcasses by sonosteam® treatment. *Proceedings of European Congress of Chemical Engineering (ECCE-6)*. Copenhagen.
- Hansen, J, Holm, L, Frewer, L, Robinson, P & Sandøe, P 2003. Beyond the knowledge deficit: recent research into lay and expert attitudes to food risks. *Appetite*, 41(2), 111-121.
- Hartnack, S, Doherr, MG, Alter, T, Toutounian-Mashad, K & Greiner, M 2009. Campylobacter monitoring in German broiler flocks: An explorative time series analysis. Zoonoses and Public Health, 56(3), 117-128.
- Hartnett, E, Kelly, L, Newell, D, Wooldridge, M & Gettinby, G 2001. A quantitative risk assessment for the occurrence of *Campylobacter* in chickens at the point of slaughter.

Epidemiology and Infection, 127(2), 195-206.

- Harvey, P & Leach, S 1998. Analysis of coccal cell formation by *Campylobacter jejuni* using continuous culture techniques, and the importance of oxidative stress. *Journal of Applied Microbiology*, 85(2), 398-404.
- Havelaar, AH, Evers, EG & Nauta, MJ 2008. Challenges of quantitative microbial risk assessment at EU level. *Trends in Food Science & Technology*, 19(Supplement 1), S26-S33.
- Havelaar, AH, Mangen, M-JJ, De Koeijer, AA, Bogaardt, M-J, Evers, EG, Jacobs-Reitsma, WF, Van Pelt, W, Wagenaar, JA, De Wit, GA, Van Der Zee, H & Nauta, MJ 2007. Effectiveness and efficiency of controlling *Campylobacter* on broiler chicken meat. *Risk Analysis*, 27(4), 831-844.
- Haysom, IW & Sharp, AK 2005. Bacterial contamination of domestic kitchens over a 24-hour period. *British Food Journal*, 107(7), 453-466.
- Hazeleger, WC, Wouters, JA, Rombouts, FM & Abee, T 1998. Physiological activity of *Campylobacter jejuni* far below the minimal growth temperature. *Applied and Environmental Microbiology*, 64(10), 3917-3922.
- Heldman, D & Newsome, R 2003. Kinetic models for microbial survival during processing. *Food Technology*, 57(8), 40.
- Hendricks, RA, Boyle, EAE, Kastner, CL & Fung, DYC 2000. Compilation of intervention methods and conditions, and ingredient limits, for controlling *Campylobacter jejuni* in meat and poultry products. *Journal of Rapid Methods & Automation in Microbiology*, 8(4), 285-305.
- Her, E, Seo, S, Choi, J, Pool, V & Ilic, S 2019. Assessment of food safety at university food courts using surveys, observations, and microbial testing. *Food Control*, 103, 167-174.
- Highmore, CJ, Warner, JC, Rothwell, SD, Wilks, SA & Keevil, CW 2018. Viable-but-Nonculturable Listeria monocytogenes and Salmonella enterica Serovar Thompson Induced by chlorine stress remain infectious. *mBio*, 9(2), e00540-18.
- Hillers, VN, Medeiros, L, Kendall, P, Chen, G & Dimascola, S 2003. Consumer food-handling behaviors associated with prevention of 13 foodborne illnesses. *Journal of Food Protection*, 66(10), 1893-1899.
- Hilmarsson, H, Thormar, H, Thrainsson, J, Gunnarsson, E & Dadadottir, S 2006. Effect of glycerol monocaprate (monocaprin) on broiler chickens: an attempt at reducing intestinal *Campylobacter* infection. *Poultry Science*, 85(4), 588-592.
- Hodges, I 1993. Raw to cooked: Community awareness of safe food handling practices, Wellington, New Zealand, Department of Health Internal Report.
- Hofshagen, M & Kruse, H 2005. Reduction in flock prevalence of Campylobacter spp. in broilers in

Norway after implementation of an action plan. *Journal of Food Protection*, 68(10), 2220-2223. Holdsworth, SD 1998. Thermal processing of packaged foods. *Trans IChemE*, 76(Part C), 63.

- Hovorková, P & Skrivanova, E 2015. Use of Caprylic Acid in Broiler Chickens: Effect on Campylobacter jejuni. *Foodborne Pathogens and Disease*, 12(8), 712-718.
- Hsieh, YH, Simpson, S, Kerdahi, K & Sulaiman, IM 2018. A Comparative Evaluation Study of Growth Conditions for Culturing the Isolates of Campylobacter spp. *Current Microbiology*, 75(1), 71-78.
- Huang, J, Zang, X, Zhai, W, Guan, C, Lei, T & Jiao, X 2018. Campylobacter spp. in chickenslaughtering operations: A risk assessment of human campylobacteriosis in East China. *Food Control*, 86, 249-256.
- Huang, JJ & McBean, EA 2007. Using Bayesian statistics to estimate the coefficients of a twocomponent second-order chlorine bulk decay model for a water distribution system. *Water research*, 41(2), 287-94.
- Huang, LH & Juneja, VK 2001. A new kinetic model for thermal inactivation of microorganisms: Development and validation using Escherichia coli O157 : H7 as a test organism. *Journal of Food Protection*, 64(12), 2078-2082.
- Huezo, R, Northcutt, JK, Smith, DP, Fletcher, DL & Ingram, KD 2007. Effect of dry air or immersion chilling on recovery of bacteria from broiler carcasses. *Journal of Food Protection*, 70(174), 1829-1834.
- Hugas, M & Tsigarida, E 2008. Pros and cons of carcass decontamination: The role of the European Food Safety Authority. *Meat Science*, 78(1-2), 43-52.
- Humphrey, T, O'Brien, S & Madsen, M 2007. Campylobacters as zoonotic pathogens: A food production perspective. *International Journal of Food Microbiology*, 117(3), 237-257.
- Humphrey, TJ & Lanning, DG 1987. Salmonella and Campylobacter contamination of broiler chicken carcasses and scald tank water - the influence of water pH. Journal of Applied Bacteriology, 63(1), 21-25.
- Humphrey, TJ, Martin, KW, Slader, J & Durham, K 2001. *Campylobacter* spp. in the kitchen: Spread and persistence. *Journal of Applied Microbiology Symposium Supplement*, 90(30), 115S-120S.
- Humphrey, TJ, Martin, KW & Whitehead, A 1994. Contamination of hands and work surfaces with *Salmonella enteritidis PT4* during the preparation of egg dishes. *Epidemiology and Infection*, 113(3), 403-409.
- ICMSF 1996. *Microorganisms in Foods 5: Microbiological Specifications of Food Pathogens,* London, Chapman and Hall, for the International Commission on Microbiological Specifications for Foods.

- Ikram, R, Chambers, S, Mitchell, P, Brieseman, M & Ikram, O 1994. A case control study to determine risk factors for *Campylobacter* infection in Christchurch in the summer of 1992-3. New Zealand Medical Journal, 107, 430-432.
- Isohanni, P, Alter, T, Saris, P & Lyhs, U 2010. Wines as possible meat marinade ingredients possess antimicrobial potential against Campylobacter. *Poultry Science*, 89(12), 2704-2710.
- Jackowska, A, Szczawiński, J, Pe, conek, J & Fonberg-Broczek, M 2008. Possibility of *Campylobacter jejuni* inactivation in smoked salmon by high-pressure treatment. *High Pressure Research: An International Journal*, 28(2), 127 - 132.
- Jackson, V, Blair, IS, McDowell, DA, Kennedy, J & Bolton, DJ 2007. The incidence of significant foodborne pathogens in domestic refrigerators. *Food Control*, 18(4), 346-351.
- Jacobs-Reitsma, W, Lyhs, U & Wagenaar, J 2008. *Campylobacter in the food supply,* Washington DC, American Society for Microbiology.
- Jacobs-Reitsma, WF, Jongenburger, I, de Boer, E & Biesta-Peters, EG 2019. Validation by interlaboratory trials of EN ISO 10272 Microbiology of the food chain Horizontal method for detection and enumeration of Campylobacter spp. Part 2: Colony-count technique. *International Journal of Food Microbiology*, 288, 32-38.
- James, C, James, SJ, Hannay, N, Purnell, G, Barbedo-Pinto, C, Yaman, H, Araujo, M, Gonzalez, ML, Calvo, J, Howell, M & Corry, JEL 2007. Decontamination of poultry carcasses using steam or hot water in combination with rapid cooling, chilling or freezing of carcass surfaces. *International Journal of Food Microbiology*, 114(2), 195-203.
- Jasson, V, Uyttendaele, M, Rajkovic, A & Debevere, J 2007. Establishment of procedures provoking sub-lethal injury of *Listeria monocytogenes, Campylobacter jejuni* and *Escherichia coli* 0157 to serve method performance testing. International Journal of Food Microbiology, 118(3), 241-249.
- Jay, LS, Comar, D & Govenlock, LD 1999a. A national australian food safety telephone survey. Journal of Food Protection, 62, 921-928.
- Jay, LS, Comar, D & Govenlock, LD 1999b. A video study of Australian domestic food-handling practices. *Journal of Food Protection*, 62(11), 1285-1296.
- Jeong, J, Kim, H & Seo, KH 2019. Quantitative risk assessment model for salmonellosis in chicken skewers from street food vendors in South Korea. *Journal of Food Protection*, 82(6), 955-962.
- Jepsen, MR, Simonsen, J & Ethelberg, S 2009. Spatio-temporal cluster analysis of the incidence of *Campylobacter* cases and patients with general diarrhea in a Danish county, 1995-2004. *International Journal of Health Geographics*, 8(1).
- Johnson, JYM, McMullen, LM, Hasselback, P, Louie, M & Saunders, LD 2006. Travelers'

knowledge of prevention and treatment of travelers' diarrhea. Journal of Travel Medicine, 13(6), 351-355.

- Jones, AK, Rigby, D, Burton, M, Millman, C, Williams, NJ, Jones, TR, Wigley, P, O'Brien, SJ & Cross, P 2016. Restaurant cooking trends and increased risk for Campylobacter infection. *Emerging Infectious Diseases*, 22(7), 1208-1215.
- Jones, DM, Sutcliffe, EM, Rios, R, Fox, AJ & Curry, A 1993. *Campylobacter jejuni* adapts to aerobic metabolism in the environment. *Journal of Medical Microbiology*, 38(2), 145-150.
- Jones, FS, Orcutt, M & Little, RB 1931. Vibrios (vibrio jejuni, n. sp.) Associated with intestinal disorders of cows and calves, Rockefeller Univ Press.
- Jonsson, M, Heier, B, Norstrom, M & Hofshagen, M 2010. Analysis of simultaneous space-time clusters of *Campylobacter* spp. in humans and in broiler flocks using a multiple dataset approach. *International Journal of Health Geographics*, 9(1), 48.
- Jore, S, Viljugrein, H, Brun, E, Heier, BT, Borck, B, Ethelberg, S, Hakkinen, M, Kuusi, M, Reiersen, J, Hansson, I, Olsson Engvall, E, Løfdahl, M, Wagenaar, JA, van Pelt, W & Hofshagen, M 2010. Trends in *Campylobacter* incidence in broilers and humans in six European countries, 1997-2007. *Preventive Veterinary Medicine*, 93(1), 33-41.
- Joshua, GWP, Guthrie-Irons, C, Karlyshev, AV & Wren, BW 2006. Biofilm formation in *Campylobacter jejuni. Microbiology-SGM*, 152(2), 387-396.
- Juffs, HS & Deeth, H. 2007. Scientific evaluation of pasteurisation for pathogen reduction in milk and milk products. Available: <u>https://www.foodstandards.govt.nz/code/proposals/documents/Scientific%20Evaluati</u> <u>on.pdf</u> [Accessed 16 July 2013].
- Kaakoush, NO, Baar, C, MacKichan, J, Schmidt, P, Fox, EM, Schuster, SC & Mendz, GL 2009. Insights into the molecular basis of the microaerophily of three Campylobacterales: A comparative study. *Antonie van Leeuwenhoek, International Journal of General and Molecular Microbiology*, 96(4), 545-557.
- Kaakoush, NO, Castaño-Rodríguez, N, Mitchell, HM & Man, SM 2015. Global Epidemiology of Campylobacter Infection. *Clinical microbiology reviews*, 28(3), 687-720.
- Kagan, LJ, Aiello, AE & Larson, E 2002. The role of the home environment in the transmission of infectious diseases. *Journal of Community Health*, 27(4), 247-267.
- Kapperud, G, Espeland, G, Wahl, E, Walde, A, Herikstad, H, Gustavsen, S, Tveit, I, Natas, O, Bevanger, L & Digranes, A 2003. Factors associated with increased and decreased risk of *Campylobacter* infection: a prospective case-control study in Norway. *American Journal of Epidemiology*, 158(3), 234-242.

- Karumathil, DP, Upadhyay, A & Venkitanarayanan, K 2016. Chapter 19 Antimicrobial Packaging for Poultry. In: Barros-Velázquez, J (ed.) Antimicrobial Food Packaging. San Diego: Academic Press.
- Katiyo, W, de Kock, HL, Coorey, R & Buys, EM 2019. Assessment of safety risks associated with handling chicken as based on practices and knowledge of a group of South African consumers. *Food Control,* 101, 104-111.
- Katzav, M, Isohanni, P, Lund, M, Hakkinen, M & Lyhs, U 2008. PCR assay for the detection of Campylobacter in marinated and non-marinated poultry products. *Food Microbiology*, 25(7), 908-914.
- Kaur, Ledward, Park & Robson 1998. Factors affecting the heat resistance of Escherichia coli O157 : H7. Letters in Applied Microbiology, 26(4), 325-330.
- Kaur, BP & Rao, PS 2017. Modeling the combined effect of pressure and mild heat on the inactivation kinetics of Escherichia coli, Listeria innocua, and Staphylococcus aureus in black tiger shrimp (Penaeus monodon). *Frontiers in Microbiology*, 8(JUL).
- Keener, KM, Bashor, MP, Curtis, PA, Sheldon, BW & Kathariou, S 2004. Comprehensive review of *Campylobacter* and poultry processing. *Comprehensive Reviews in Food Science and Food Safety*, 3(2), 105-116.
- Keklik, NM, Demirci, A, Puri, VM & Heinemann, PH. Modeling the inactivation of Salmonella typhimurium, Listeria monocytogenes, and Salmonella enteritidis on poultry products exposed to pulsed UV-light. American Society of Agricultural and Biological Engineers Annual International Meeting 2011, ASABE 2011, 2011. 4639-4653.
- Kelly, AF, Park, SF, Bovill, R & Mackey, BM 2001. Survival of *Campylobacter jejuni* during stationary phase: Evidence for the absence of a phenotypic stationary-phase response. *Applied and Environmental Microbiology*, 67(5), 2248-2254.
- Kelly, DJ 2008. Complexity and versatility in the physiology and metabolism of Campylobacter jejuni, Washington, DC, ASM Press.
- Kendall, PA, Elsbernd, A, Sinclair, K, Schroeder, M, Chen, G, Bergmann, V, Hillers, VN & Medeiros, LC 2004. Observation versus self-report: Validation of a consumer food behavior questionnaire. *Journal of Food Protection*, 67, 2578-2586.
- Kennedy, J, Jackson, V, Blair, IS, McDowell, DA, Cowan, C & Bolton, DJ 2005a. Food safety knowledge of consumers and the microbiological and temperature status of their refrigerators. *Journal of Food Protection*, 68, 1421-1430.
- Kennedy, J, Jackson, V, Cowan, C, Blair, I, McDowell, D & Bolton, D 2005b. Consumer food safety knowledge: Segmentation of Irish home food preparers based on food safety

knowledge and practice. British Food Journal, 107(7), 441-452.

- Kerslake, V. 1995. Community awareness of safe food handling practices and food poisoning: Knowledge and experience. Master Master Thesis, Victoria University.
- Klein, Gn 2016. Campylobacter : features, detection, and prevention of foodborne disease, Amsterdam, Netherlands : Academic Press.
- Koidis, P & Doyle, MP 1983. Survival of *Campylobacter-jejuni* in fresh and heated red meat. *Journal of Food Protection*, 46(9), 771-774.
- Koolman, L, Whyte, P, Meade, J, Lyng, J & Bolton, D 2014. Use of chemical treatments applied alone and in combination to reduce Campylobacter on raw poultry. *Food Control*, 46, 299-303.
- Koppel, K, Sosa, M, Gutierrez, NG, Cardinal, P, Godwin, SL, Cates, SC & Chambers, E 2016. Consumer practices for purchase, storage, and preparation of poultry and eggs in selected north and south american countries: A pilot study. *Vitae*, 23(1), 58-64.
- Koppel, K, Timberg, L, Shalimov, R, Vázquez-Araújo, L, Carbonell-Barracchina, AA, Di Donfrancesco, B & Chambers IV, E 2015. Purchase, storage, and preparation of eggs and poultry in selected European countries: A preliminary study. *British Food Journal*, 117(2), 749-765.
- Korzen, S, Sandøe, P & Lassen, J 2011. Don't wash my meat: public perceptions of decontamination in meat production. *British Food Journal*, 113(5), 598-612.
- Kosa, KM, Cates, SC, Bradley, S, Chambers, EI & Godwin, S 2015. Consumer-reported handling of raw poultry products at home: Results from a national survey. *Journal of Food Protection*, 78(1), 180-186.
- Kosa, KM, Cates, SC, Brophy, J, Godwin, S, Chambers, D & Chambers, EI 2019. Older adults and parents of young children have different handling practices for raw poultry. *Journal of Food Protection*, 82(2), 200-206.
- Kou, X, Li, R & Wang, S 2016. Performance of a heating block system designed for studying the heat resistance of bacteria in foods. 2016 ASABE Annual International Meeting. St. Joseph, MI: ASABE.
- Krieg, NR & Hoffman, PS 1986. Microaerophily and Oxygen Toxicity. Annual Review of Microbiology, 40(1), 107-130.
- Kume, T, Furuta, M, Todoriki, S, Uenoyama, N & Kobayashi, Y 2009. Status of food irradiation in the world. *Radiation Physics and Chemistry*, 78(3), 222-226.
- Kusumaningrum, HD, Riboldi, G, Hazeleger, WC & Beumer, RR 2003. Survival of foodborne pathogens on stainless steel surfaces and cross-contamination to foods. *International Journal*

of Food Microbiology, 85(3), 227-236.

- Kusumaningrum, HD, van Asselt, ED, Beumer, RR & Zwietering, MH 2004. A quantitative analysis of cross-contamination of *Salmonella* and *Campylobacter* spp. via domestic kitchen surfaces. *Journal of Food Protection*, 67(9), 1892-1903.
- Kwon, J, Wilson, ANS, Bednar, C & Kennon, L 2008. Food safety knowledge and behaviors of Women, Infant, and Children (WIC) program participants in the United States. *Journal of Food Protection*, 71(8), 1651-1658.
- Lackner, J, Weiss, M, Müller-Graf, C & Greiner, M 2019. The disease burden associated with Campylobacter spp. In Germany, 2014. *PLoS ONE*, 14(5).
- Lader, D 1999. Food Safety in the Home 1998, Office for National Statistics, London.
- Lahou, E, Wang, X, De Boeck, E, Verguldt, E, Geeraerd, A, Devlieghere, F & Uyttendaele, M 2015. Effectiveness of inactivation of foodborne pathogens during simulated home pan frying of steak, hamburger or meat strips. *International Journal of Food Microbiology*, 206, 118-129.
- Lake, R. 2006. Transmission routes for campylobacteriosis in New Zealand. Available: <u>https://www.mpi.govt.nz/dmsdocument/23143-transmission-routes-for-</u> <u>campylobacteriosis-in-new-zealand</u> [Accessed 14 May 2007].
- Lake, R, Adlam, B & Perera, S. 2009. Acute gastrointestinal illness (AGI) Study. Available: <u>http://www.foodsafety.govt.nz/elibrary/industry/acute-illness-study-gastrointestinal-</u> <u>report/</u> [Accessed 1/02/2011].
- Lake, R, Hudson, A, Cressey, P & Bayne, G. 2007. Quantitative risk model: *Campylobacter* spp. in the poultry food chain. Available: <u>https://www.mpi.govt.nz/dmsdocument/25841-</u> <u>quantitative-risk-model-campylobacter-spp-in-the-poultry-food-chain</u> [Accessed 13 April 2009].
- Lake, R, Hudson, A, Cressey, P & Notrje, G. 2003. Risk Profile: *Campylobacter jejuni / coli* in poultry (whole and pieces). Available: <u>https://www.mpi.govt.nz/dmsdocument/5440/direct</u> [Accessed 24 June 2010].
- Lake, RJ, Cressey, PJ, Campbell, DM & Oakley, E 2010. Risk ranking for foodborne microbial hazards in New Zealand: burden of disease estimates. *Risk Analysis*, 30(5), 743-752.
- Lambertz, ST, Nilsson, C, Brådenmark, A, Sylvén, S, Johansson, A, Jansson, LM & Lindblad, M 2012. Prevalence and level of Listeria monocytogenes in ready-to-eat foods in Sweden 2010. *International Journal of Food Microbiology*, 160(1), 24-31.
- Lange, M, Göranzon, H, Fleig, L & Marklinder, I 2018. Adolescents' sources for food safety knowledge and trust. *British Food Journal*, 120(3), 549-562.

- Langiano, E, Ferrara, M, Lanni, L, Viscardi, V, Abbatecola, AM & De Vito, E 2012. Food safety at home: knowledge and practices of consumers. *Journal of Public Health*, 20(1), 47-57.
- Lazou, T, Georgiadis, M, Pentieva, K, McKevitt, A & Iossifidou, E 2012. Food safety knowledge and food-handling practices of Greek university students: A questionnaire-based survey. *Food Control*, 28(2), 400-411.
- Lee, A, Smith, SC & Coloe, PJ 1998. Survival and growth of *Campylobacter jejuni* after artificial inoculation onto chicken skin as a function of temperature and packaging conditions. *Journal of Food Protection*, 61(12), 1609-1614.
- Lewerenz, M 2002. Monte Carlo methods: Overview and basics. NIC Series, 10, 1-24.
- Li, C, Huang, L & Chen, J 2014. Comparative study of thermal inactivation kinetics of Salmonella spp. in peanut butter and peanut butter spread. *Food Control,* 45, 143-149.
- Li, Y, Yang, H & Swem, B 2002. Effect of high-temperature inside-outside spray on survival of *Campylobacter jejuni* attached to prechill chicken carcasses. *Poultry Science*, 81(9), 1371-1377.
- Lindqvist, R & Lindblad, M 2008. Quantitative risk assessment of thermophilic *Campylobacter* spp. and cross-contamination during handling of raw broiler chickens evaluating strategies at the producer level to reduce human campylobacteriosis in Sweden. *International Journal of Food Microbiology*, 121(1), 41-52.
- Lindqvist, R, Sylven, S & Vagsholm, I 2002. Quantitative microbial risk assessment exemplified by Staphylococcus aureus in unripened cheese made from raw milk. International Journal of Food Microbiology, 78(1-2), 155-170.
- Line, JE 2001. Development of a selective differential agar for isolation and enumeration of *Campylobacter* spp. *Journal of Food Protection*, 64(11), 1711-1715.
- Line, JE 2006. Comparison of enrichment vessels for recovery of *Campylobacter* spp. from broiler rinse samples. *Journal of Rapid Methods and Automation in Microbiology*, 14(1), 110-117.
- Lopez, GU, Kitajima, M, Sherchan, SP, Sexton, JD, Sifuentes, LY, Gerba, CP & Reynolds, KA 2015. Impact of disinfectant wipes on the risk of Campylobacter jejuni infection during raw chicken preparation in domestic kitchens. *Journal of Applied Microbiology*, 119(1), 245-252.
- Loretz, M, Stephan, R & Zweifel, C 2010. Antimicrobial activity of decontamination treatments for poultry carcasses: A literature survey. *Food Control,* 21(6), 791-804.
- Lori, S, Buckow, R, Knorr, D, Heinz, V & Lehmacheri, A 2007. Predictive model for inactivation of *Campylobacter* spp. by heat and high hydrostatic pressure. *Journal of Food Protection*, 70(9), 2023-2029.
- Louis, VR, Gillespie, IA, O'Brien, SJ, Russek-Cohen, E, Pearson, AD & Colwell, RR 2005.

Temperature-driven campylobacter seasonality in England and Wales. *Applied and Environmental Microbiology*, 71(1), 85-92.

- Lu, T, Marmion, M, Ferone, M, Wall, P & Scannell, AGM 2019. Processing and retail strategies to minimize Campylobacter contamination in retail chicken. *The Journal of Food Processing and Preservation* 43(12), e14251.
- Luber, P 2009. Cross-contamination versus undercooking of poultry meat or eggs which risks need to be managed first? *International Journal of Food Microbiology*, 134(1-2), 21-28.
- Luber, P, Brynestad, S, Topsch, D, Scherer, K & Bartelt, E 2006. Quantification of *Campylobacter* species cross-contamination during handling of contaminated fresh chicken parts in kitchens. *Applied and Environmental Microbiology*, 72(1), 66-70.
- Lunn, DJ, Thomas, A, Best, N & Spiegelhalter, D 2000. WinBUGS A Bayesian modelling framework: Concepts, structure, and extensibility. *Statistics and Computing*, 10(4), 325-337.
- Lytou, AE, Tzortzinis, K, Skandamis, PN, Nychas, G-JE & Panagou, EZ 2019. Investigating the influence of organic acid marinades, storage temperature and time on the survival/inactivation interface of Salmonella on chicken breast fillets. *International Journal of Food Microbiology*, 299, 47-57.
- Maertens de Noordhout, C, Devleesschauwer, B, Haagsma, JA, Havelaar, AH, Bertrand, S, Vandenberg, O, Quoilin, S, Brandt, PT & Speybroeck, N 2017. Burden of salmonellosis, campylobacteriosis and listeriosis: a time series analysis, Belgium, 2012 to 2020. *Euro surveillance : bulletin Europeen sur les maladies transmissibles = European communicable disease bulletin*, 22(38), 30615.
- Mafart, P, Couvert, O, Gaillard, S & Leguerinel, I 2002. On calculating sterility in thermal preservation methods: application of the Weibull frequency distribution model. *International Journal of Food Microbiology*, 72(1-2), 107-113.
- Mahon, D, Cowan, C & McCarthy, M 2006. The role of attitudes, subjective norm, perceived control and habit in the consumption of ready meals and takeaways in Great Britain. *Food Quality and Preference*, 17(6), 474-481.
- Mai, TL. 2003. Inhibition of Campylobacter jejuni by bacteria isolated from broiler deboning operation. Master Master Thesis, Auburn University.
- Maia, RL, Teixeira, P & Mateus, TL 2019. Risk communication strategies (on listeriosis) for highrisk groups. *Trends in Food Science & Technology*, 84, 68-70.
- Maiden, MCJ 2006. Multilocus Sequence Typing of Bacteria. Annual Review of Microbiology, 60(1), 561-588.
- Maijala, R, Ranta, J, Seuna, E, Pelkonen, S & Johansson, T 2005. A quantitative risk assessment of

the public health impact of the Finnish *Salmonella* control program for broilers. *International Journal of Food Microbiology*, 102(1), 21-35.

- Makridakis, SG, Wheelwright, SC & Hyndman, RJ 1998. Forecasting : methods and applications, New York, Wiley.
- Mangen, M-JJ, Havelaar, AH, Poppe, KP & Wit, GAd 2006. Cost-utility analysis to control *Campylobacter* on chicken meat-dealing with data limitations. *Risk Analysis*, 27(4), 815-830.
- Martinez-Rodriguez, A, Kelly, AF, Park, SF & Mackey, BM 2004. Emergence of variants with altered survival properties in stationary phase cultures of *Campylobacter jejuni*. *International Journal of Food Microbiology*, 90(3), 321-329.
- Mataragas, M, Zwietering, MH, Skandamis, PN & Drosinos, EH 2010. Quantitative microbiological risk assessment as a tool to obtain useful information for risk managers -Specific application to Listeria monocytogenes and ready-to-eat meat products. *International Journal of Food Microbiology*, 141(SUPPL.), S170-S179.
- Mattick, K, Durham, K, Domingue, G, Jørgensen, F, Sen, M, Schaffner, DW & Humphrey, T 2003. The survival of foodborne pathogens during domestic washing-up and subsequent transfer onto washing-up sponges, kitchen surfaces and food. *International Journal of Food Microbiology*, 85(3), 213-226.
- Maughan, C, Chambers, E, Godwin, S, Chambers, D, Cates, S & Koppel, K 2016. Food handling behaviors observed in consumers when cooking poultry and eggs. *Journal of Food Protection*, 79(6), 970-977.
- Mazengia, E, Fisk, C, Liao, G, Huang, H & Meschke, J 2015. Direct observational study of the risk of cross-contamination during raw poultry handling: Practices in private homes. *Food Protection Trends*, 35(1), 8-23.
- McCarthy, M, Brennan, M, Kelly, AL, Ritson, C, de Boer, M & Thompson, N 2007. Who is at risk and what do they know? Segmenting a population on their food safety knowledge. *Food Quality and Preference*, 18(2), 205-217.
- McCarthy, Z, Smith, B, Fazil, A, Wu, J, Ryan, SD & Munther, D 2018. pH dependent C. jejuni thermal inactivation models and application to poultry scalding. *Journal of Food Engineering*, 223, 1-9.
- McIntyre, L. 2008. Domestic food practices in New Zealand: Quantifying The Reduction Of Campylobacter Jejuni On Skin-On Chicken Breasts Commercially Frozen And Stored For Up To 10 Weeks In A Domestic Freezer. Available: http://www.foodsafety.govt.nz/elibrary/industry/domestic-food-practices-researchprojects-

2/FW0799 Campy Commercial Freezing Project final report 26 5 08 web.pdf [Accessed 12 May 2008].

- McIntyre, L. 2009. Quantifying The Reduction Of Campylobacter Jejuni On Skin-On Chicken Breasts Frozen And Stored For Up To 10 Weeks At -12°C. *Domestic Food Practices In New Zealand* [Online]. Available: <u>http://www.nzfsa.govt.nz/science/research-projects/FW0915 Campy -12C Freezing final report 30 April 09.pdf</u> [Accessed 16 April 2009].
- McIntyre, L, Lee, J & Biggs, R 2010. Campylobacteriosis in New Zealand. Part two tackling the problem. . *International Food Hygiene*, 21(6), 21-22.
- McKellar, RC & Lu, X 2004. Modeling microbial responses in foods, Boca Raton, FL, CRC Press.
- McKenzie, A. 2006. NZFSA moves to curb *Campylobacter* rates. *New Zealand Food Safety Authority*, [Online]. Available: <u>http://www.foodsafety.govt.nz/elibrary/industry/Nzfsa Moves-Along\_With.htm</u> [Accessed 30 November 2006].
- McKenzie, A. 2007. Progress in fight against *Campylobacter* bacteria. *Press release* [Online]. Available: <u>https://www.scoop.co.nz/stories/GE0706/S00095.htm</u> [Accessed 27 June 2007].
- Meer, RR & Misner, SL 2000. Food safety knowledge and behavior of expanded food and nutrition education program participants in Arizona. *Journal of Food Protection*, 63, 1725-1731.
- Mégraud, F & Bultel, C. 2003. Appréciation des risques alimentaires liés aux *campylobacters*; Application au couple poulet/*Campylobacter jejuni*. Available: <u>http://www.google.co.nz/search?q=Appr%C3%A9ciation+des+risques+alimentaires+li</u> <u>%C3%A9s+aux+campylobacters%3B+Application+au+couple+poulet%2FCampylobac</u> <u>ter+jejuni&ie=utf-8&co=utf-8&aq=t&rls=org.mozilla:en-GB:official&client=firefox-a</u> [Accessed 14 February 2004].
- Membré, JM & Boué, G 2018. Quantitative microbiological risk assessment in food industry: Theory and practical application. *Food Research International*, 106, 1132-1139.
- Membré, JM, Laroche, M & Magras, C 2011. Assessment of levels of bacterial contamination of large wild game meat in Europe. *Food Microbiology*, 28(5), 1072-1079.
- Meredith, H, Valdramidis, V, Rotabakk, BT, Sivertsvik, M, McDowell, D & Bolton, DJ 2014. Effect of different modified atmospheric packaging (MAP) gaseous combinations on Campylobacter and the shelf-life of chilled poultry fillets. *Food Microbiology*, 44, 196-203.
- Michaud, S, Menard, S & Arbeit, RD 2004. Campylobacteriosis, Eastern Townships, Quebec. *Emerging Infectious Diseases*, 10(10), 1844-1847.
- Mihaljevic, RR, Sikic, M, Klancnik, A, Brumini, G, Mozina, SS & Abram, M 2007. Environmental stress factors affecting survival and virulence of *Campylobacter jejuni*. *Microbial Pathogenesis*,

43(2), 120-125.

- Mikkelä, A, Ranta, J, González, M, Hakkinen, M & Tuominen, P 2016. Campylobacter QMRA: A Bayesian Estimation of Prevalence and Concentration in Retail Foods Under Clustering and Heavy Censoring. *Risk Analysis*, 36(11), 2065-2080.
- Miles, CA & Mackey, BM 1994. A mathematical-analysis of microbial inactivation at linearly rising temperatures - calculation of the temperature rise needed to kill *Listeria monocytogenes* in different foods and methods for dynamic measurements of *D*-value and *z*-value. *Journal of Applied Bacteriology*, 77(1), 14-20.
- Miles, S & Scaife, V 2003. Optimistic bias and food. Nutrition research reviews, 16(01), 3-19.
- Miller, AJ, Bayles, DO & Eblen, BS 2000. Cold shock induction of thermal sensitivity in Listeria monocytogenes. *Applied and Environmental Microbiology*, 66(10), 4345-4350.
- Millstone, E, Lang, T & Marsden, T. 2017. Will the British public accept chlorine-washed turkey for Christmas dinner, after Brexit? FRC Food Brexit Policy Briefing [Online]. Available: <u>https://www.sussex.ac.uk/webteam/gateway/file.php?name=frcbrexitbriefing-</u> <u>chlorinewashedturkeyforchristmas.pdf&site=25</u> [Accessed 22 December 2018].
- Min, JCH, Kung, HH & Liu, HH 2010. Interventions affecting air transport passenger demand in Taiwan. *African Journal of Business Management*, 4(10), 2121-2131.
- Mishra, N & Puri, VM 2013. Modeling the inactivation of listeria monocytogenes by combined high pressure and temperature using weibull model. *Journal of Food Process Engineering*, 36(5), 598-607.
- Montville, R, Chen, Y & Schaffner, DW 2001. Glove barriers to bacterial cross-contamination between hands to food. *Journal of Food Protection*, 64(6), 845-849.
- Montville, R & Schaffner, DW 2003. Inoculum size influences bacterial cross contamination between surfaces. *Applied and Environmental Microbiology*, 69(12), 7188-7193.
- Moore, DS, McCabe, GP & Craig, BA 2009. Introduction to the practice of statistics, New York, W.H. Freeman.
- Moore, JE, Corcoran, D, Dooley, JSG, Fanning, S, Lucey, B, Matsuda, M, McDowell, DA, Mégraud, F, Millar, BC, O'Mahony, R, O'Riordan, L, O'Rourke, M, Rao, JR, Rooney, PJ, Sails, A & Whyte, P 2005. *Campylobacter. Veterinary Research*, 36(3), 351-382.
- Moore, JE & Madden, RH 2000. The effect of thermal stress on Campylobacter coli. *Journal of Applied Microbiology*, 89(5), 892-899.
- Moreb, NA, Priyadarshini, A & Jaiswal, AK 2017. Knowledge of food safety and food handling practices amongst food handlers in the Republic of Ireland. *Food Control*, 80, 341-349.
- Morgan, AI, Goldberg, N, Radewonuk, ER & Scullen, OJ 1996. Surface pasteurization of raw

poultry meat by steam. Lebensmittel-Wissenschaft und-Technologie, 29(5-6), 447-451.

- Morrison, GJ & Fleet, GH 1985. Reduction of *Salmonella* on chicken carcasses by immersion treatments. *Journal of Food Protection*, 48(11), 939-943.
- Možina, SS, Klančnik, A, Kovac, J, Jeršek, B & Bucar, F 2018. Antimicrobial Natural Products Against Campylobacter. *In:* Mérillon, J-M & Riviere, C (eds.) *Natural Antimicrobial Agents*. Cham: Springer International Publishing.
- Mullner, P. 2009. Estimating the contribution of different sources to the burden of human campylobacteriosis and salmonellosis. Doctor of Philosophy (Ph.D.) Doctoral, Massey University.
- Müllner, P, Collins-Emerson, JM, Midwinter, AC, Carter, P, Spencer, SEF, van der Logt, P, Hathaway, S & French, NP 2010. Molecular epidemiology of campylobacter jejuni in a geographically isolated country with a uniquely structured poultry industry. *Applied and Environmental Microbiology*, 76(7), 2145-2154.
- Mullner, P, Jones, G, Noble, A, Spencer, SEF, Hathaway, S & French, NP 2009a. Source attribution of food-borne zoonoses in New Zealand: a modified Hald model. *Risk Analysis,* 29(7), 970-984.
- Mullner, P, Spencer, SE, Wilson, DJ, Jones, G, Noble, AD, Midwinter, AC, Collins-Emerson, JM, Carter, P, Hathaway, S & French, NP 2009b. Assigning the source of human campylobacteriosis in New Zealand: a comparative genetic and epidemiological approach. *Infection, Genetics and Evolution*, 9(6), 1311-9.
- Murphy, C, Carroll, C & Jordan, KN 2003. Identification of a novel stress resistance mechanism in *Campylobacter jejuni*.
- Murphy, C, Carroll, C & Jordan, KN 2005. The effect of different media on the survival and induction of stress responses by *Campylobacter jejuni*. *Journal of Microbiological Methods*, 62(2), 161-166.
- Mutters, R & Warnes, SL 2019. The method used to dry washed hands affects the number and type of transient and residential bacteria remaining on the skin. *Journal of Hospital Infection*, 101(4), 408-413.
- Myers, MF, Rogers, DJ, Cox, J, Flahault, A & Hay, SI 2000. Forecasting disease risk for increased epidemic preparedness in public health. *Advances in Parasitology*, 47, 309-330.
- Mylius, SD, Nauta, MJ & Havelaar, AH 2007. Cross-contamination during food preparation: a mechanistic model applied to chicken-borne *Campylobacter*. *Risk Analysis*, 27(4), 803-813.
- NACMCF 2007. Response to the questions posed by the food safety and Inspection Service regarding consumer guidelines for the safe cooking of poultry products. *Journal of Food Protection*, 70, 251-260.

- Naumova, EN, Chen, JT, Griffiths, JK, Matyas, BT, Estessmargiassi, SA & Morris, RD 2000. Use of passive surveillance data to study temporal and spatial variation in the incidence of giardiasis and cryptosporidiosis. *Public Health Reports*, 115(5), 436-447.
- Nauta, M & Christensen, B 2011. The impact of consumer phase models in microbial risk analysis. *Risk Analysis*, 31(2), 255-265.
- Nauta, M, Fischer, ARH, van Asselt, ED, de Jong, AEI, Frewer, LJ & de Jonge, R 2008. Food safety in the domestic environment: the effect of consumer risk information on human disease risks. *Risk Analysis*, 28(1), 179-192.
- Nauta, M, Hill, A, Rosenquist, H, Brynestad, S, Fetsch, A, van der Logt, P, Fazil, A, Christensen, B, Katsma, E, Borck, B & Havelaar, A 2009. A comparison of risk assessments on *Campylobacter* in broiler meat. *International Journal of Food Microbiology*, 129(2), 107-123.
- Nauta, M, Jacobs-Reitsma, W, Evers, E, van Pelt, W & Havelaar, A. 2005. Risk assessment of *Campylobacter* in the Netherlands via broiler meat and other routes. *CARMA report: Campylobacter* Risk Management and Assessment [Online]. Available: <u>http://www.rivm.nl/bibliotheek/rapporten/250911006.html</u> [Accessed 10 May 2008].
- Nauta, M, Jacobs-Reitsma, WF & Havelaar, AH 2007. A risk assessment model for *Campylobacter* in broiler meat. *Risk Analysis*, 27(4), 845-861.
- Neimann, J, Engberg, J, Molbak, K & Wegener, HC 2003. A case control study of risk factors for sporadic campylobacter infections in Denmark. *Epidemiology and Infection*, 130(03), 353-366.
- Nelson, W & Harris, B 2011. Campylobacteriosis rates show age-related static bimodal and seasonality trends. *The New Zealand Medical Journal*, 124(1337), 33-39.
- Nesbitt, A, Majowicz, S, Finley, R, Marshall, B, Pollari, F, Sargeant, J, Ribble, C, Wilson, J & Sittler, N 2009. High-risk food consumption and food safety practices in a Canadian community. *Journal of Food Protection*, 72(174), 2575-2586.
- Nesbitt, A, Thomas, MK, Marshall, B, Snedeker, K, Meleta, K, Watson, B & Bienefeld, M 2014. Baseline for consumer food safety knowledge and behaviour in Canada. *Food Control*, 38(1), 157-173.
- Neves, MI, Mungai, SN & Nauta, MJ 2018. Can stochastic consumer phase models in QMRA be simplified to a single factor? *Microbial Risk Analysis,* 8, 53-60.
- Newell, DG, Shreeve, JE, Toszeghy, M, Domingue, G, Bull, S, Humphrey, T & Mead, G 2001. Changes in the carriage of *Campylobacter* strains by poultry carcasses during processing in abattoirs. *Applied and Environmental Microbiology*, 67(6), 2636-2640.
- Newman, KL, Leon, JS, Rebolledo, PA & Scallan, E 2015. The impact of socioeconomic status on foodborne illness in high-income countries: A systematic review. *Epidemiology and Infection*,

143(12), 2473-2485.

- Nguyen, HTT, Corry, JEL & Miles, CA 2006. Heat resistance and mechanism of heat Inactivation in thermophilic *Campylobacters*. *Applied and Environmental Microbiology*, 72(1), 908-913.
- Nichols, GL, Richardson, JF, Sheppard, SK, Lane, C & Sarran, C 2012. Campylobacter epidemiology: a descriptive study reviewing 1 million cases in England and Wales between 1989 and 2011. *BMJ Open*, 2(4), e001179.
- Nobre, FF, Monteiro, ABS, Telles, PR & David Williamson, G 2001. Dynamic linear model and SARIMA: A comparison of their forecasting performance in epidemiology. *Statistics in Medicine*, 20(20), 3051-3069.
- Nohra, A, Grinberg, A, Midwinter, AC, Marshall, JC, Collins-Emerson, JM & French, NP 2016. Molecular epidemiology of Campylobacter coli strains isolated from different sources in New Zealand between 2005 and 2014. *Applied and Environmental Microbiology*, 82(14), 4363-4370.
- Nohra, A, Grinberg, A, Midwinter, AC, Marshall, JC, Collins-Emerson, JM & French, NP 2018. Exposure to whole chicken carcasses may present a greater risk of campylobacteriosis compared to exposure to chicken drumsticks. *Zoonoses and Public Health*, 65(7), 822-830.
- Nørrung, B & Buncic, S 2008. Microbial safety of meat in the European Union. *Meat Science*, 78(1-2), 14-24.
- NZFSA. 2007. Joint statement from the New Zealand Food Safety Authority and the Ministry for the Environment. Available: <u>https://www.scoop.co.nz/stories/GE0704/S00041/grant-</u> <u>will-help-fund-campylobacter-research.htm</u> [Accessed 11 April 2007].
- NZFSA. 2009. Caecal testing Review and options assessment Available: <u>https://www.mpi.govt.nz/dmsdocument/27603/direct</u> [Accessed 11 May 2009].
- NZPHO. 2010. Notifiable and other diseases in New Zealand: 2009 Annual surveillance report. Available: http://www.surv.esr.cri.nz/PDF\_surveillance/AnnualRpt/AnnualSurv/2009/2009Annu

<u>alSurvRpt.pdf</u> [Accessed 12 April 2010].

- Oh, E, Andrews, KJ, McMullen, LM & Jeon, B 2019. Tolerance to stress conditions associated with food safety in Campylobacter jejuni strains isolated from retail raw chicken. *Scientific Reports*, 9(1).
- Oh, E, McMullen, L & Jeon, B 2015. Impact of oxidative stress defense on bacterial survival and morphological change in Campylobacter jejuni under aerobic conditions. *Frontiers in Microbiology*, 6(295).
- Oh, E, McMullen, LM, Chui, L & Jeon, B 2017. Differential Survival of Hyper-Aerotolerant
Campylobacter jejuni under Different Gas Conditions. Frontier in Microbiology, 8(954).

- Ojima, M, Toshima, Y, Koya, E, Ara, K, Kawai, S & Ueda, N 2002. Bacterial contamination of Japanese households and related concern about sanitation. *International Journal of Environmental Health Research*, 12(1), 41-52.
- Oosterom, J, Wilde, GJAd, Boer, Ed, Blaauw, LHd & Karman, H 1983. Survival of *Campylobacter jejuni* during poultry processing and pig slaughtering. *Journal of Food Protection*, 46(8), 702-706, 709.
- Oyarzabal, OA 2005. Reduction of *Campylobacter spp.* by commercial antimicrobials applied during the processing of broiler chickens: a review from the united states perspective. *Journal of Food Protection*, 68, 1752-1760.
- Oyarzabal, OA, Oscar, TP, Speegle, L & Nyati, H 2010. Survival of campylobacter jejuni and campylobacter coli on retail broiler meat stored at-20, 4, or 12°C and development of weibull models for survival. *Journal of Food Protection*, 73(8), 1438-1446.
- Pang, H, Lambertini, E, Buchanan, RL, Schaffner, DW & Pradhan, AK 2017. Quantitative microbial risk assessment for Escherichia coli O157:H7 in fresh-cut lettuce. *Journal of Food Protection*, 80(2), 302-311.
- Park, S 2000. Environmental regulatory genes. *In:* Nachamkin, I & Blaser, MJ (eds.) *Campylobacter*.2nd ed. Washington, D.C.: ASM Press.
- Park, S 2005. Campylobacter: stress response and resistance, Cambridge, England, CRC Press, Woodhead.
- Park, SF 2002. The physiology of *Campylobacter* species and its relevance to their role as foodborne pathogens. *International Journal of Food Microbiology*, 74(3), 177-188.
- Parkhill, J, Wren, BW, Mungall, K, Ketley, JM, Churcher, C, Basham, D, Chillingworth, T, Davies,
  RM, Feltwell, T, Holroyd, S, Jagels, K, Karlyshev, AV, Moule, S, Pallen, MJ, Penn, CW,
  Quail, MA, Rajandream, MA, Rutherford, KM, van Vliet, AHM, Whitehead, S & Barrell,
  BG 2000. The genome sequence of the food-borne pathogen *Campylobacter jejuni* reveals
  hypervariable sequences. *Nature*, 403(6770), 665-668.
- Patashnik, M 1953. A simplified procedure for thermal process evaluation. *Food Technology*, 7(1), 1-6.
- Patil, SR, Cates, S & Morales, R 2005. Consumer food safety knowledge, practices, and demographic differences: findings from a Meta-analysis. *Journal of Food Protection*, 68(9), 1884-1894.
- Paulin, S. 2010. Preliminary Study on Campylobacter Recovery from Poultry Carcasses. Available: <u>https://www.mpi.govt.nz/dmsdocument/25904-preliminary-study-on-campylobacter-</u> <u>recovery-from-poultry-carcasses-final-report</u> [Accessed 12 August 2010].

- Pearl, DL, Louie, M, Chui, L, Doré, K, Grimsrud, KM, Martin, SW, Michel, P, Svenson, LW & McEwen, SA 2009. A multi-level approach for investigating socio-economic and agricultural risk factors associated with rates of reported cases of *Escherichia coli* o157 in humans in alberta, canada. *Zoonoses and Public Health*, 56(8), 455-464.
- Peleg, M 2002. A model of survival curves having an 'Activation Shoulder'. *Journal of Food Science*, 67(7), 2438-2443.
- Peleg, M 2003. Calculation of the Non-Isothermal Inactivation Patterns of Microbes Having Sigmoidal Isothermal Semi-Logarithmic Survival Curves. *Critical Reviews in Food Science and Nutrition*, V43(6), 645-658.
- Peleg, M & Cole, MB 1998. Reinterpretation of microbial survival curves. Critical Reviews in Food Science and Nutrition, 38(5), 353-380.
- Peleg, M, Cole, MB, Peleg, M & Cole, MB 2000. Estimating the survival of *Clostridium botulinum* spores during heat treatments. *Journal of Food Protection*, 63(2), 190-5.
- Perez-Rodriguez, F, Todd, ECD, Valero, A, Carrasco, E, Garcia, RM & Zurera, G 2006. Linking quantitative exposure assessment and risk management using the food safety objective concept: An example with *Listeria monocytogenes* in different cross-contamination scenarios. *Journal of Food Protection*, 69(10), 2384-2394.
- Perez-Rodriguez, F, Valero, A, Carrasco, E, Garcia, RM & Zurera, G 2008. Understanding and modelling bacterial transfer to foods: a review. *Trends in Food Science & Technology*, 19(3), 131-144.
- Perko-Makela, P, Koljonen, M, Miettinen, M & Hanninen, ML 2000. Survival of Campylobacter jejuni in marinated and nonmarinated chicken products. *Journal of Food Safety*, 20(4), 209-216.
- Persson, E, Sjöholm, I & Skog, K 2002. Heat and mass transfer in chicken breasts-effect on PhIP formation. *European Food Research and Technology*, 214(6), 455-459.
- Pesciaroli, M, Chardon, JE, Delfgou, EHM, Kuijpers, AFA, Wijnands, LM & Evers, EG 2019. Home style frying of steak and meat products: Survival of Escherichia coli related to dynamic temperature profiles. *International Journal of Food Microbiology*, 300, 53-63.
- Petersen, L, Nielsen, EM & On, SLW 2001. Serotype and genotype diversity and hatchery transmission of *Campylobacter jejuni* in commercial poultry flocks. *Veterinary Microbiology*, 82(2), 141-154.
- Pflug, IJ 1990. *Microbiology and engineering of sterilization processes,* Minneapolis, MN, Environmental Sterilization Laboratory.
- PIANZ. 2011. Campy cases heading downwards. Available: http://www.pianz.org.nz/recent-

posts/campy-cases-heading-downwards. [Accessed 20 May 2011].

Pinheiro, JC 2000. Mixed-Effects Models in S and S-Plus, New York, Springer.

- Poschet, F, Bernaerts, K, Geeraerd, AH, Scheerlinck, N, Nicolai, BM & Van Impe, JF 2004. Sensitivity analysis of microbial growth parameter distributions with respect to data quality and quantity by using Monte Carlo analysis. *Mathematics and Computers in Simulation*, 65(3), 231-243.
- Pouillot, R, Garin, B, Ravaonindrina, N, Diop, K, Ratsitorahina, M, Ramanantsoa, D & Rocourt, J 2012. A risk assessment of campylobacteriosis and salmonellosis linked to chicken meals prepared in households in Dakar, Senegal. *Risk Analysis*, 32(10), 1798-1819.
- Premarathne, JMKJK, New, CY, Ubong, A, Nakaguchi, Y, Nishibuchi, M & Son, R 2017. Risk of escherichia coli o157:H7 infection linked to the consumption of beef. *Food Research*, 1(3), 67-76.
- Prescot, LM, Harley, JP & Klein, DA 1996. Microbiology, Duboque IA WCB.
- Purnell, G, Mattick, K & Humphrey, T 2004. The use of `hot wash' treatments to reduce the number of pathogenic and spoilage bacteria on raw retail poultry. *Journal of Food Engineering*, 62(1), 29-36.
- Ranta, J, Tuominen, P & Maijala, R 2005. Estimation of true Salmonella prevalence jointly in cattle herd and animal populations using Bayesian hierarchical modeling. Risk Analysis, 25(1), 23-37.
- Rashed, Y, Meersman, H, Van De Voorde, E & Vanelslander, T 2017. Short-term forecast of container throughout: An ARIMA-intervention model for the port of Antwerp oa. *Maritime Economics and Logistics*, 19(4), 749-764.
- Ray, AN, Dikeman, ME, Crow, BA, Phebus, RK, Grobbel, JP & Hollis, LC 2010. Microbial translocation of needle-free versus traditional needle injection-enhanced beef strip loins. *Meat Science*, 84(1), 208-211.
- Ray, B 2014. Fundamental food microbiology, Boca Raton, FL : CRC Press.
- Redmond, EC & Griffith, CJ 2003. Consumer food handling in the home: A review of food safety studies. *Journal of Food Protection*, 66(1), 130-161.
- Redmond, EC & Griffith, CJ 2006. A pilot study to evaluate the effectiveness of a social marketingbased consumer food safety initiative using observation. *British Food Journal*, 108(9), 753-770.
- Redmond, EC & Griffith, CJ 2009. The importance of hygiene in the domestic kitchen: Implications for preparation and storage of food and infant formula. *Perspectives in Public Health*, 129(2), 69-76.

- Redmond, EC, Griffith, CJ, Slader, J & Humphrey, T 2004. Microbiological and observational analysis of cross contamination risks during domestic food preparation. *British Food Journal*, 106(8), 581-597.
- Reichert, TA, Simonsen, L, Sharma, A, Pardo, SA, Fedson, DS & Miller, MA 2004. Influenza and the winter increase in mortality in the United States, 1959-1999. *American Journal of Epidemiology*, 160(5), 492-502.
- Ritz, M, Nauta, MJ, Teunis, PFM, van Leusden, F, Federighi, M & Havelaar, AH 2007. Modelling of Campylobacter survival in frozen chicken meat. *Journal of Applied Microbiology*, 103(3), 594-600.
- Robert, CP & Casella, G 2004. Monte Carlo statistical methods, New York, Springer.
- Robinson, DA 1981. Infective dose of Campylobacter jejuni in milk. British Medical Journal, 282(6276), 1584.
- Rolfhamre, P & Ekdahl, K 2006. An evaluation and comparison of three commonly used statistical models for automatic detection of outbreaks in epidemiological data of communicable diseases. *Epidemiology and Infection*, 134(4), 863-871.
- Roseman, M & Kurzynske, J 2006. Food safety perceptions and behaviors of Kentucky consumers. *Journal of Food Protection*, 69(6), 1412-1421.
- Rosenquest, H, Boysen, L & Borck, B. 2007. Interventions to control *Campylobacter* in the broiler production, Report of an International Expert Consultation. Available: <u>https://www.food.dtu.dk//-/media/Institutter/Foedevareinstituttet/Publikationer/Pub-2008/Interventions-to-control-Campylobacter-in-the-broiler-production.ashx</u> [Accessed 27 November 2008].
- Rosenquest, H, Nielsen, NL, Sommer, HM, Norrung, B & Christensen, BB 2003. Quantitative risk assessment of human campylobacteriosis associated with thermophilic *Campylobacter* species in chickens. *International Journal of Food Microbiology*, 83(1), 87-103.
- Ruby, GE, Ungku Zainal Abidin, UF, Lihan, S, Jambari, NN & Radu, S 2019. A cross sectional study on food safety knowledge among adult consumers. *Food Control*, 99, 98-105.
- Rushton, SP, Sanderson, RA, Diggle, PJ, Shirley, MDF, Blain, AP, Lake, I, Maas, JA, Reid, WDK, Hardstaff, J, Williams, N, Jones, NR, Rigby, D, Strachan, NJC, Forbes, KJ, Hunter, PR, Humphrey, TJ & O'Brien, SJ 2019. Climate, human behaviour or environment: individualbased modelling of Campylobacter seasonality and strategies to reduce disease burden. *Journal of Translational Medicine*, 17(1), 34.
- Ryser, ET & Marth, EH 2007. Listeria, listeriosis, and food safety, Boca Raton, Fla.; London, CRC Press.

- Sabirovic, M. 2002. Salmonella brandenburg in sheep meat in New Zealand. Master thesis, Massey University.
- Sagarzazu, N, Cebrián, G, Pagán, R, Condón, S & Mañas, P 2010. Resistance of Campylobacter jejuni to heat and to pulsed electric fields. *Innovative Food Science & Emerging Technologies*, 11(2), 283-289.
- Sampers, I, Berkvens, D, Jacxsens, L, Ciocci, MC, Dumoulin, A & Uyttendaele, M 2012. Survey of Belgian consumption patterns and consumer behaviour of poultry meat to provide insight in risk factors for campylobacteriosis. *Food Control*, 26(2), 293-299.
- Sampers, I, Habib, I, Berkvens, D, Dumoulin, A, De Zutter, L & Uyttendaele, M 2008. Processing practices contributing to *Campylobacter* contamination in Belgian chicken meat preparations. *International Journal of Food Microbiology*, 128(2), 297-303.
- Sampers, I, Habib, I, De Zutter, L, Dumoulin, A & Uyttendaele, M 2010. Survival of *Campylobacter* spp. in poultry meat preparations subjected to freezing, refrigeration, minor salt concentration, and heat treatment. *International Journal of Food Microbiology*, 137(2-3), 147-153.
- Samuel, M, Vugia, D, Shallow, S, Marcus, R, Segler, S, McGivern, T, Kassenborg, H, Reilly, K, Kennedy, M & Angulo, F 2004a. Epidemiology of sporadic *Campylobacter* infection in the United States and declining trend in incidence, FoodNet 1996–1999. *Clinical Infectious Diseases*, 38, S165-S174.
- Samuel, MC, Vugia, DJ, Shallow, S, Marcus, R, Segler, S, McGivern, T, Kassenborg, H, Reilly, K, Kennedy, M, Angulo, F, Tauxe, RV & Emerging Infections Program, F 2004b. Epidemiology of sporadic *Campylobacter* infection in the United States and declining trend in incidence, FoodNet 1996-1999. *Clinical Infectious Diseases*, 38, S165-S174.
- Sánchez, MX, Fluckey, WM, Brashears, MM & McKee, SR 2002. Microbial profile and antibiotic susceptibility of *Campylobacter* spp. and *Salmonella* spp. in broilers processed in air-chilled and immersion-chilled environments. *Journal of Food Protection*, 65(6), 948-956.
- Sandberg, M, Hofshagen, M, ØStensvik, Ø, Skjerve, E & Innocent, G 2005. Survival of Campylobacter on Frozen Broiler Carcasses as a Function of Time. International Association for Food Protection.
- Sanlier, N 2009. The knowledge and practice of food safety by young and adult consumers. *Food Control*, 20(6), 538-542.
- Scanlon, KA, Tiwari, U, Cagney, C, Walsh, D, McDowell, DA & Duffy, G 2015. Modelling the thermal inactivation of five Campylobacteraceae species. *Food Control,* 47, 135-140.
- Schaffner, DW 2004a. Mathematical frameworks for modeling *Listeria* cross-contamination in food. *Journal of Food Science*, 69(6), R155-R159.

- Schaffner, DW 2004b. Mathematical frameworks for modeling *Listeria* cross-contamination in food-processing plants. *Journal of Food Science*, 69(6), R155-R159.
- Schellekens, M, Van Impe, JF, Nicolai, BM, Martens, T & De Baerdemaeker, J 1995. Predictive microbiology in a dynamic environment: a system theory approach. *International Journal of Food Microbiology* 25(3), 227-49.
- Scherer, K, Bartelt, E, Sommerfeld, C & Hildebrandt, G 2006. Quantification of Campylobacter on the surface and in the muscle of chicken legs at retail. *Journal of Food Protection*, 69(4), 757-761.
- Schönberg-Norio, D, Sarna, S, Hänninen, M, Katila, M, Kaukoranta, S & Rautelin, H 2006. Strain and host characteristics of *Campylobacter jejuni* infections in Finland. *Clinical Microbiology and Infection*, 12(8), 754-760.
- Schönberg-Norio, D, Takkinen, J, Hänninen, ML, Katila, ML, Kaukoranta, SS, Mattila, L & Rautelin, H 2004. Swimming and *Campylobacter* infections. *Emerging Infectious Diseases*, 10(8), 1474-1477.
- Schraer, R & Edgington, T. 2019. Chlorinated chicken: How safe is it? *Reality Check* [Online]. Available: <u>https://www.bbc.com/news/uk-47440562</u> [Accessed 5 March 2019].
- Scott, E & Bloomfield, SF 1990. The survival and transfer of microbial contamination via cloths, hands and utensils. *Journal of Applied Microbiology*, 68(3), 271-278.
- Sears, A, Baker, MG, Wilson, N, Marshall, J, Muellner, P, Campbell, DM, Lake, RJ & French, NP 2011. Marked campylobacteriosis decline after interventions aimed at poultry, New Zealand. *Emerging Infectious Diseases*, 17(6), 1007-1015.
- Sebald, M & V'Eron, M 1963. Base DNA content and classification of *Vibrios. Annales de l'Institut Pasteur* 105, 897-910.
- Sharma, H, Mendiratta, SK, Agrawal, RK, Gurunathan, K, Kumar, S & Singh, TP 2017. Use of various essential oils as bio preservatives and their effect on the quality of vacuum packaged fresh chicken sausages under frozen conditions. LWT - Food Science and Technology, 81, 118-127.
- Sheppard, SK, Dallas, JF, Strachan, NJC, MacRae, M, McCarthy, ND, Wilson, DJ, Gormley, FJ, Falush, D, Ogden, ID, Maiden, MCJ & Forbes, KJ 2009. *Campylobacter* genotyping to determine the source of human infection. *Clinical Infectious Diseases*, 48(8), 1072-1078.
- Shrestha, S, Wagle, BR, Upadhyay, A, Arsi, K, Donoghue, DJ & Donoghue, AM 2019. Carvacrol antimicrobial wash treatments reduce Campylobacter jejuni and aerobic bacteria on broiler chicken skin. *Poultry Science*, 98(9), 4073-4083.
- Siekkinen, K-M, Heikkilä, J, Tammiranta, N & Rosengren, H 2012. Measuring the costs of

biosecurity on poultry farms: a case study in broiler production in Finland. *Acta Veterinaria Scandinavica*, 54(1), 12.

- Signorini, ML, Zbrun, MV, Romero-Scharpen, A, Olivero, C, Bongiovanni, F, Soto, LP, Frizzo, LS & Rosmini, MR 2013. Quantitative risk assessment of human campylobacteriosis by consumption of salad cross-contaminated with thermophilic Campylobacter spp. from broiler meat in Argentina. *Preventive Veterinary Medicine*, 109(1), 37-46.
- Singh, P, Lee, HC, Silva, MF, Chin, KB & Kang, I 2017. Trisodium phosphate dip, hot water dip, and combination dip with/without brushing on broiler carcass decontamination. *Food Control*, 77, 199-209.
- Sinhamahapatra, M, Biswas, S, Das, AK & Bhattacharyya, D 2004. Comparative study of different surface decontaminants on chicken quality. *British Poultry Science*, 45(5), 624-630.
- Skarp, CPA, Hänninen, ML & Rautelin, HIK 2016. Campylobacteriosis: the role of poultry meat. *Clinical Microbiology and Infection*, 22(2), 103-109.
- Skirrow, MB, Benjamin, J., Razi, M.H.H., Waterman, S 1982. Isolation, cultivation and identification of *Campylobacter jejuni* and *Campylobacter coli In:* Corry, JEL, Roberts, D. and Skinner, F.A. (ed.) *Isolation and Identification Methods for Food Poisoning Organisms*. London: Academic Press.
- Slorach, S 2008. Food safety risk management in New Zealand: a review of the New Zealand Food Safety Authority's risk management framework and its application. . Wellington New Zealand Food Safety Authority.
- Smid, JH, Verloo, D, Barker, GC & Havelaar, AH 2010. Strengths and weaknesses of Monte Carlo simulation models and Bayesian belief networks in microbial risk assessment. *International Journal of Food Microbiology*, 139(Supplement 1), S57-S63.
- Smith, MG 1992. Destruction of bacteria on fresh meat by hot water. *Epidemiology and Infection*, 109(3), 491-496.
- Smith, MG & Davey, KR 1990. Destruction of *Escherichia-Coli* on sides of beef by a hot water decontamination process. *Food Australia*, 42(4), 195-198.
- Solow, BT, Cloak, OM & Fratamico, PM 2003. Effect of temperature on viability of *Campylobacter jejuni* and *Campylobacter coli* on raw chicken or pork skin. *Journal of Food Protection*, 66, 2023-2031.
- Sorqvist, S 1989. Heat resistance of *Campylobacter* and *Yersinia* strains by three methods. *Journal of Applied Bacteriology*, 67(5), 543-549.
- Sorqvist, S 2003. Heat resistance in liquids of *Enterococcus* spp., *Listeria* spp., *Escherichia coli, Yersinia* enterocolitica, Salmonella spp. and Campylobacter spp. Acta Veterinaria Scandinavica, 44(1/2), 1-

19.

- Spencer, SEF, Marshall, J, Pirie, R, Campbell, D, Baker, MG & French, NP 2012. The spatial and temporal determinants of campylobacteriosis notifications in New Zealand, 2001–2007. *Epidemiology and Infection*, 140(9), 1663-1677.
- Sterniša, M, Smole Možina, S, Levstek, S, Kukec, A, Raspor, P & Jevšnik, M 2018. Food safety knowledge, self-reported practices and attitude of poultry meat handling among Slovenian consumers. *British Food Journal*, 120(6), 1344-1357.
- Stumbo, CR 1973. Thermobacteriology in Food Processing, New York, Academic Press.
- Suzuki, H & Yamamoto, S 2009. *Campylobacter* contamination in retail poultry meats and byproducts in the world: A literature survey. *Journal of Veterinary Medical Science*, 71(3), 255-261.
- Svedhem, A, Kaijser, B & Sjogren, E 1981. The occurrence of campylobacter-jejuni in fresh food and survival under different conditions. *Journal of Hygiene*, 87(3), 421-425.
- Takeuchi, MT, Edlefsen, M, McCurdy, SM & Hillers, VN 2006. Development and validation of stages-of-change questions to assess consumers' readiness to use a food thermometer when cooking small cuts of meat. *Journal of the American Dietetic Association*, 106(2), 262-266.
- Tam, CC, Rodriguez, LC, O'Brien, SJ & Hajat, S 2006. Temperature dependence of reported Campylobacter infection in England, 1989-1999. *Epidemiology and Infection*, 134(1), 119-125.
- Tang, N & Schraft, H 2000. Temperature and mixed-culture effect on Campylobacter jejuni biofilm formation, Dallas, TX.
- Tarlak, F, Ozdemir, M & Melikoglu, M 2020. Predictive modelling for the growth kinetics of Pseudomonas spp. on button mushroom (Agaricus bisporus) under isothermal and nonisothermal conditions. *Food Research International*, 130, 108912.
- Taylor, AW, Coveney, J, Ward, PR, Dal Grande, E, Mamerow, L, Henderson, J & Meyer, SB 2012. The Australian Food and Trust Survey: Demographic indicators associated with food safety and quality concerns. *Food Control*, 25(2), 476-483.
- Teunis, P, Van den Brandhof, W, Nauta, M, Wagenaar, J, Van den Kerkhof, H & Van Pelt, W 2005. A reconsideration of the *Campylobacter* dose-response relation. *Epidemiology and Infection*, 133(4), 583-592.
- Thépault, A, Guyard-Nicodème, M, Rose, V, Quesne, S, Queguiner, M, Houard, E, Mégraud, F, Rivoal, K & Chemaly, M 2018. A representative overview of the genetic diversity and lipooligosaccharide sialylation in Campylobacter jejuni along the broiler production chain in France and its comparison with human isolates. *International Journal of Food Microbiology*, 274, 20-30.
- Tiozzo, B, Pinto, A, Mascarello, G, Mantovani, C & Ravarotto, L 2019. Which food safety

information sources do Italian consumers prefer? Suggestions for the development of effective food risk communication. *Journal of Risk Research*, 22(8), 1062-1077.

- Todd, ECD, Greig, JD, Bartleson, CA & Michaels, BS 2009. Outbreaks where food workers have been implicated in the spread of foodborne disease. Part 6. Transmission and survival of pathogens in the food processing and preparation environment. *Journal of Food Protection*, 72(1), 202-219.
- Toft, N, Innocent, GT, Gettinby, G & Reid, SWJ 2007. Assessing the convergence of Markov Chain Monte Carlo methods: An example from evaluation of diagnostic tests in absence of a gold standard. *Preventive Veterinary Medicine*, 79(2-4), 244-256.
- Tompkins, NM, Avens, JS, Kendall, PA & Salman, MD 2008. Effect of boiling water carcass immersion on aerobic bacteria counts of poultry skin and processed ground poultry meat. *Zoonoses and Public Health*, 55(5), 235-241.
- Tonsor, GT, Schroeder, TC & Pennings, JME 2009. Factors Impacting Food Safety Risk Perceptions. *Journal of Agricultural Economics*, 60(3), 625-644.
- Trachoo, N & Brooks, JD 2005. Attachment and heat resistance of Campylobacter jejuni on Enterococcus faecium biofilm. *Pakistan Journal of Biological Sciences*, 8(4), 599-605.
- Tuominen, P. 2009. Developing risk-based food safety management. PhD thesis, University of Helsinki.
- Tuominen, P, Ranta, J & Maijala, R 2006. *Salmonella* risk in imported fresh beef, beef preparations, and beef products. *Journal of Food Protection*, 69(8), 1814-1822.
- UMR Research. 2007. A quantitative study. Report Prepared for NZFSA [Online]. Available: <u>http://www.foodsafety.govt.nz/elibrary/consumer-survey/2007-survey.pdf</u> [Accessed 26 July 2007].
- Uusitalo, L 2007. Advantages and challenges of Bayesian networks in environmental modelling. *Ecological Modelling*, 203(3-4), 312-318.
- Uusitalo, L, Kuikka, S & Romakkaniemi, A 2005. Estimation of Atlantic salmon smolt carrying capacity of rivers using expert knowledge. *ICES Journal of Marine Science*, 62(4), 708-722.
- Uyttendaele, M, Baert, K, Ghafir, Y, Daube, G, De Zutter, L, Herman, L, Dierick, K, Pierard, D, Dubois, JJ, Horion, B & Debevere, J 2006. Quantitative risk assessment of *Campylobacter* spp. in poultry based meat preparations as one of the factors to support the development of risk-based microbiological criteria in Belgium. *International Journal of Food Microbiology*, 111(2), 149-163.
- Valdramidis, VP, Geeraerd, AH, Bernaerts, K, Devlighere, F, Debevere, J & Van Impe, JF 2004. Accurate Modelling of Non- Loglinear Survival Curves. *Bulletin of the International Dairy Federation*, (392), 97-110.

- Valdramidis, VP, Geeraerd, AH, Bernaerts, K & Van Impe, JFM 2008. Identification of non-linear microbial inactivation kinetics under dynamic conditions. *International Journal of Food Microbiology*, 128(1), 146-152.
- van Asselt, ED, de Jong, AEI, de Jonge, R & Nauta, MJ 2008. Cross-contamination in the kitchen: estimation of transfer rates for cutting boards, hands and knives. *Journal of Applied Microbiology*, 105(5), 1392-1401.
- van Asselt, ED, Fischer, A, de Jong, AEI, Nauta, MJ & de Jonge, R 2009. Cooking practices in the kitchen;observed versus predicted behavior. *Risk Analysis*, 29(4), 533-540.
- van Asselt, ED & Zwietering, MH 2006. A systematic approach to determine global thermal inactivation parameters for various food pathogens. *International Journal of Food Microbiology*, 107(1), 73-82.
- van Boekel, MAJS 1996. Statistical aspects of kinetic modeling for food science problems. *Journal* of Food Science, 61(3), 477-486.
- van Boekel, MAJS 2002. On the use of the Weibull model to describe thermal inactivation of microbial vegetative cells. *International Journal of Food Microbiology*, 74(1-2), 139-159.
- van Boekel, MAJS 2008. Kinetics and Statistics. In: van Boekel, MAJS (ed.) Kinetic Modeling of Reactions In Foods. CRC Press.
- Van Impe, JF, Nicolai, BM, Martens, T, De Baerdemaeker, J & Vandewalle, J 1992. Dynamic mathematical model to predict microbial growth and inactivation during food processing. *Applied and Environmental Microbiology*, 58(9), 2901-2909.
- van Vliet, AHM, Ketley, JM, Park, SF & Penn, CW 2002. The role of iron in Campylobacter gene regulation, metabolism and oxidative stress defense. *Fems Microbiology Reviews*, 26(2), 173-186.
- Varsaki, A, Murphy, C, Barczynska, A, Jordan, K & Carroll, C 2015. The acid adaptive tolerance response in Campylobacter jejuni induces a global response, as suggested by proteomics and microarrays. *Microbial Biotechnology*, 8(6), 974-988.
- Vaucher, P, Bischoff, T, Diserens, EA, Herzig, L, Meystre-Agustoni, G, Panese, F, Favrat, B, Sass,
  C & Bodenmann, P 2012. Detecting and measuring deprivation in primary care:
  Development, reliability and validity of a self-reported questionnaire: The DiPCare-Q. *BMJ Open*, 2(1).
- Vellinga, A & Van Loock, F 2002. The dioxin crisis as experiment to determine poultry-related *Campylobacter enteritis. Emerging Infectious Diseases,* 8(1), 19-22.
- Vigre, H, Barfoed, K, Swart, AN, Simons, RRL, Hill, AA, Snary, EL & Hald, T 2016. Characterization of the Human Risk of Salmonellosis Related to Consumption of Pork

Products in Different E.U. Countries Based on a QMRA. Risk Analysis, 36(3), 531-545.

- Vose, D 2005. Use of microbial risk assessment in decision-making. Dordogne France,: David Vose Consultancy.
- Vose, D 2008. Risk analysis : a quantitative guide, Chichester, England Hoboken, NJ, Wiley.
- Warriss, PD 2000. Meat science an introductory text, Wallingford, UK ; New York, CABI Pub.
- Weisent, J, Seaver, W, Odoi, A & Rohrbach, B 2010. Comparison of three time-series models for predicting campylobacteriosis risk. *Epidemiology and Infection*, 138(6), 898-906.
- Welt, B & Tong, C 1993. Effect of microwave-radiation on thiamin degradation kinetics. *Journal of Microwave Power and Electromagnetic Energy*, 28(4), 187-195.
- Welt, BA, Teixeira, AA, Balaban, MO, Smerage, GH, Hintinlang, DE & Smittle, BJ 1997. Kinetic parameter estimation in conduction heating foods subjected to dynamic thermal treatments. *Journal of Food Science*, 62(3), 529-534.
- Wensley, A & Coole, L 2013. Cohort study of a dual-pathogen point source outbreak associated with the consumption of chicken liver pâté, UK, October 2009. *Journal of Public Health* (United Kingdom), 35(4), 585-589.
- Wesley, RD & Stadelman, WJ 1985. The effect of carbon dioxide packaging on detection of *Campylobacter jejuni* from chicken carcasses. *Poultry Science*, 64(4), 763-4.
- Whiting, RC 1993. Modeling bacterial survival in unfavorable environments. Journal of Industrial Microbiology, 12(3-5), 240-246.
- Whiting, RC, Hogue, A, Schlosser, WD, Ebel, ED, Morales, RA, Baker, A & McDowell, RM 2000. A quantitative process model for *Salmonella enteritidis* in shell eggs. *Journal of Food Science*, 65(5), 864-869.
- Whyte, P, Collins, JD, McGill, K & Monahan, C 2002. Assessment of sodium dichloroisocyanurate in the control of microbiological cross-contamination in broiler carcass immersion chilling systems. *Journal of Food Safety*, 22(1), 55-65.
- Whyte, P, McGill, K & Collins, JD 2003. An assessment of steam pasteurization and hot water immersion treatments for the microbiological decontamination of broiler carcasses. *Food Microbiology*, 20(1), 111-117.
- Whyte, R, Hudson, JA & Graham, C 2006. *Campylobacter* in chicken livers and their destruction by pan frying. *Letters in Applied Microbiology*, 43(6), 591-595.
- Wieczorek, K & Osek, J 2019. Genetic diversity of Campylobacter jejuni isolated from the poultry food chain. *Journal of Veterinary Research (Poland)*, 63(1), 35-40.
- Wilcock, A, Pun, M, Khanona, J & Aung, M 2004. Consumer attitudes, knowledge and behaviour: a review of food safety issues. *Trends in Food Science and Technology*, 15(2), 56-66.

- Wills, WJ, Meah, A, Dickinson, AM & Short, F 2015. 'I don't think I ever had food poisoning'. A practice-based approach to understanding foodborne disease that originates in the home. *Appetite*, 85, 118-125.
- Wingstrand, A, Neimann, J, Nielsen, EM, Gerner-Smidt, P, Wegener, HC & Molbak, K 2006. Fresh chicken as main risk factor for campylobacteriosis, Denmark. *Emerging Infectious Diseases*, 12(2).
- Wong, TL & Wilson, M. 2011. Bacterial concentrations of poultry offal and in mechanically separated meat products at the processing plant. Available: <u>https://www.mpi.govt.nz/dmsdocument/25727-bacterial-concentrations-of-poultry-offal-and-in-mechanically-separated-meat-products-at-the-processing-plant-technical-paper-201159</u> [Accessed 14 December 2011].
- Yamasaki, M, Igimi, S, Katayama, Y, Yamamoto, S & Amano, F 2004. Identification of an oxidative stress-sensitive protein from *Campylobacter jejuni*, homologous to rubredoxin oxidoreductase/rubrerythrin. *FEMS Microbiology Letters*, 235(1), 57-63.
- Yang, H, Li, Y & Johnson, MG 2001. Survival and death of Salmonella typhimurium and Campylobacter jejuni in processing water and on chicken skin during poultry scalding and chilling. Journal of Food Protection, 64(6), 770-776.
- Yang, H, Wang, S, Li, Y & Johnson, MG 2002. Predictive models for the survival/death of *Campylobacter jejuni* and *Salmonella typhimurium* in poultry scalding and chilling. *Journal of Food Science*, 67(5), 1836-1843.
- Yoon, KS, Burnette, CN & Oscar, TP 2004. Development of predictive models for the survival of campylobacter jejuni (atcc 43051) on cooked chicken breast patties and in broth as a function of temperature. *Journal of Food Protection*, 67(1), 64-70.
- Yuen, K-V 2010. Bayesian methods for structural dynamics and civil engineering, Singapore ; Hoboken, NJ, John Wiley & Sons.
- Zaboli, G, Huang, X, Feng, X & Ahn, DU 2019. How can heat stress affect chicken meat quality? - A review. *Poultry Science*, 98(3), 1551-1556.
- Zakarienė, G, Rokaitytė, A, Ramonaitė, S, Novoslavskij, A, Mulkytė, K, Zaborskienė, G & Malakauskas, M 2015. The Antimicrobial Effect of Spice-Based Marinades against Campylobacter jejuni on Contaminated Fresh Broiler Wings. *Journal of Food Science*, 80(3), M627-M634.
- Zhang, L, Singh, P, Lee, HC & Kang, I 2013. Effect of hot water spray on broiler carcasses for reduction of loosely attached, intermediately attached, and tightly attached pathogenic (Salmonella and Campylobacter) and mesophilic aerobic bacteria. *Poultry Science*, 92(3), 804-

810.

- Zhao, T, Doyle, MP & Berg, DE 2000. Fate of *Campylobacter jejuni* in butter. *Journal of Food Protection*, 63(1), 120-122.
- Zhao, T, Ezeike, GOI, Doyle, MP, Hung, YC & Howell, RS 2003. Reduction of Campylobacter jejuni on poultry by low-temperature treatment. *Journal of Food Protection*, 66(4), 652-655.
- Zhou, P, Hussain, S, Liles, M, Arias, C, Backert, S, Kieninger, J & Oyarzabal, O 2011. A simplified and cost-effective enrichment protocol for the isolation of Campylobacter spp. from retail broiler meat without microaerobic incubation. *BMC Microbiology*, 11(1), 175.
- Zwietering, MH & van Gerwen, SJC 2000. Sensitivity analysis in quantitative microbial risk assessment. *International Journal of Food Microbiology*, 58(3), 213-221.